



eMarket Users Guide

Version 6.0 February 2019

Trademarks

"ESCA" and "HABITAT" are registered trademarks of Grid Solutions, a GE and Alstom joint venture. "eterra" is a trademark and/or service mark of E-Terra, LLC, licensed for use by ALSTOM. Other product and company names in these materials may be trademarks or registered trademarks of other companies, and are the property of their respective owners.

Contents

Who Should Use this Document	. x
For More Information	. x
Conventions	. x
Revision Record	. х

1. Introduction	1
1.1 Accessing the eMarket GUI	
1.2 User Roles	
1.3 Agent Users	.2
1.4 Business Rules and Timelines	
1.4.1 Energy Market Timeline	.3
1.4.2 Regulation Market Timeline	
1.4.3 Fuel Price Adjustments Submittal Timeline	6
1.4.4 Rules for Timeline Checks	6
1.5 Effective Dates of Data	.7
1.6 Display Refresh vs Browser Refresh	9
1.7 Browser "Back" Function with eMarket	. 9
1.8 Demand Bid Subaccounts	9

2.1 Common Display Fields and Elements	12
2.2 Navigation	14
2.3 Filters Pane	15
2.4 Messages Pane	16
2.5 Data Displays and Entry	

3. General Tasks	20
3.1 Manage Portfolios	
3.1.1 Create a Portfolio	
3.1.2 View a Portfolio	23
3.1.3 Modify a Portfolio	23
3.1.4 Create a Copy of a Portfolio	
3.1.5 Delete a Portfolio	24
3.2 Upload or Download XML Files	25
3.2.1 Download an XML file	25
3.2.2 Upload an XML Request file	
3.3 Public and Private Messages Displays	

4. Generation Tab	
4.1 Generation Offer Concepts	
4.2 Generation Portfolio Manager Display	
4.3 Unit Default Parameters Display	
4.3.1 Data Validation Rules	

4.3.2 View Specific Unit Information	35
4.3.3 Modify Unit Default Claim 10/30 and Ramp Rate	
4.3.4 Modify Unit Default Status	
4.4 Unit Multiple Ramp Rates Daily Default Display	36
4.4.1 Data Validation Rules	
4.4.2 View Ramp Rates	
4.4.3 Create Ramp Rates	
4.4.4 Modify Ramp Rates	
4.5 Unit Ramp Rate Hourly Updates Display	
4.5.1 Data Validation Rules	
4.5.2 Override Unit's Default Ramp Rate for a Day and Hour	
4.6 Schedule Manager Display	
4.6.1 Create a New Schedule	
4.6.2 Modify a Schedule's Name and/or Description	
4.6.3 Delete a Schedule	
4.7 Schedule Detail Defaults Display	
4.7.1 Data Validation Rules	
4.7.2 View Schedule Detail Defaults	
4.7.3 Modify Schedule Constraint Data	
4.7.4 Modify Schedule Operating Limits	
4.7.5 Enable Schedule for a Market	
4.8 Schedule Offers Daily Default Display	
4.8.1 Data Validation Rules	
4.8.2 Modify Schedule Offers	
4.9 Schedule Offers Hourly Updates Display	
4.9.1 Data Validation Rules	
4.9.2 Override Schedule Offers for One or More Hours of a Market Day	
4.9.3 Remove/Delete Hourly Updates to Schedule Offers	
4.10 Schedule Times Hourly Updates Display	
4.10.1 Data Validation Rules	
4.10.2 Override Startup and Notification Times for a Day and Hour	
4.11 Schedule Selection Display	
4.11.1 Select Schedule for an Operating Day and Market	
4.12 Unit Hourly Updates Display	
4.12.1 Validation rules	
4.12.2 Override MW Operating Limits for a Day and Hour	
4.12.3 Override the Status of a Generating Unit for a Day and Hour	
4.12.4 Self-Schedule a Generating Unit	
4.13 Fuel Price Adjustments Display	
4.13.1 Validation rules	
4.13.2 View Specific Fuel Price Adjustment Information	
4.13.3 Submit a Fuel Price Adjustment Request	
4.13.4 Withdraw a Fuel Price Adjustment Request	
4.13.5 Submit Document Attachments to a Fuel Price Adjustment Request	
4.13.6 Remove Document Attachments of a Fuel Price Adjustment Request	
4.14 Generation by Portfolio Display	

4.14.1 View DA Commitment for each Unit in Portfolio	
4.15 XML Download for Generation Display	
4.15.1 Units XML Report	
4.15.2 Unit Ramp Rates XML Report	91
4.15.3 Unit Schedules XML Report	
4.15.4 Unit Schedule Details XML Report	
4.15.5 Unit Schedule Offers XML Report	
4.15.6 Unit Schedule Selection XML Report	
4.15.7 Unit Hourly Updates XML Report	
4.15.8 Unit Hourly Ramp Rate Updates XML Report	
4.15.9 Schedule Offers Hourly Updates XML Report	
4.15.10 Schedule Times Hourly Updates XML Report	
4.15.11 Fuel Price Adjustments XML Report	
4.15.12 Effective Hourly Fuel Prices of Units with FPAs XML Report	
4.15.13 Generation by Unit XML Report	
5. Regulation Tab	115
5.1 Regulation Offer Concepts	
5.2 Regulation Portfolio Manager Display	
5.3 Regulation Offers Default Display	
5.3.1 Data Validation Rules	
5.3.2 Modify Regulation Offers	
5.4 Regulation Hourly Updates Display	
5.4.1 Data Validation Rules	
5.4.2 Override Regulation Offer Parameters for Day and Hour	
5.5 XML Download for Regulation Display	
5.5.1 Regulation Offers XML Report	
5.5.2 Regulation Hourly Updates XML Report	
6. Demand Tab	
6.1 Demand Portfolio Manager Display	
6.2 Demand Bids Display	
6.2.1 Data Validation Rules	
6.2.2 View Demand Bids	
6.2.3 Submit Fixed Demand Bids	
6.2.4 Submit Price-Sensitive Demand Bids	
6.2.5 Deleting Ranges of Demand Bids	
6.2.6 Duplicating one Day's Demand Bids	
6.3 Increment Offers and Decrement Bids (Inc/Dec) Display	
6.3.1 Data Validation Rules	
6.3.2 Submit/Modify Virtual Bids (Decrement Bids / Increment Offers)	
6.3.3 Deleting Ranges of Virtual Bids	
6.3.4 Duplicating one Day's Virtual Bids	
6.4 Load Response Offer Display	
6.4.1 Data Validation Rules	
6.4.2 View a Load Response Offer	144

6.4.3 Create a new Load Response offer	.144
6.4.4 Modify an existing Load Response offer	.144
6.4.5 Delete a Load Response Offer	.144
6.4.6 Delete a Range of Load Response Offers	. 145
6.5 Demand by Portfolio Display	
6.5.1 View DA Cleared Bids at Pnodes in Portfolio	
6.6 Load Response Results Display	
6.6.1 View Day-Ahead Load Response Cleared Offers	.150
6.7 XML Download for Demand Display	
6.7.1 Bids XML Report	
6.7.2 Demand by Participant XML Report	
6.7.3 Detailed Demand by Participant XML Report	
6.7.4 Load Response Offers XML Report	
6.7.5 Load Response Results XML Report	
•···•• -•••• ··••• ··••• ···•• ····•• ····••	
7 Acces Deleted Demand (ADD) Teh	4 C E
7. Asset-Related Demand (ARD) Tab	
7.1 ARD Offer Concepts	
7.2 ARD Portfolio Manager Display	
7.3 ARD Default Parameters Display	
7.3.1 Data Validation Rules	
7.3.2 View Specific Asset Information:	
7.3.3 Modify Asset Default Claim 10/30 and Ramp Rate	
7.3.4 Modify Asset Default Status	
7.4 ARD Multiple Ramp Rates Daily Default Display	
7.4.1 Data Validation Rules	
7.4.2 View Ramp Rates	
7.4.3 Create Ramp Rates	
7.4.4 Modify Ramp Rates	
7.5 ARD Ramp Rate Hourly Updates Display	
7.5.1 Data Validation Rules	-
7.5.2 Override Unit's Default Ramp Rate for a Day and Hour	
7.6 ARD Schedule Manager Display	
7.6.1 Create a New Schedule	
7.6.2 Modify a Schedule's Name and/or Description	
7.6.3 Delete a schedule	
7.7 ARD Schedule Detail Defaults Display	
7.7.1 Data Validation Rules	. 185
7.7.2 View Schedule Detail Defaults	
7.7.3 Modify Schedule Constraint Data	
7.7.4 Modify Schedule Operating Limits	
7.7.5 Enable Schedule for a Market	
7.8 ARD Schedule Offers Daily Default Display	
7.8.1 Data Validation Rules	
7.8.2 Modify Schedule Offers	
7.9 Schedule Offers Hourly Updates Display	
7.9.1 Data Validation Rules	. 190

7.9.2 Override Schedule Offers for One or More Hours of a Market Day	191
7.10 ARD Schedule Selection Display	193
7.10.1 Select Schedule for an Operating Day and Market	193
7.11 ARD Hourly Updates Display	
7.11.1 Data Validation Rules	
7.11.2 Override MW Operating Limits for a Day and Hour	198
7.11.3 Override the Status of an ARD Asset for a Day and Hour	
7.11.4 Self-Schedule an ARD Asset	
7.12 ARD by Portfolio Display	201
7.12.1 View DA Commitment for each ARD in Portfolio	201
7.13 XML Download for ARD Display	203
7.13.1 Asset-Related Demand XML Report	204
7.13.2 Asset-Related Demand Ramp Rates XML Report	205
7.13.3 Asset-Related Demand Ramp Rate Hourly Updates XML Report	205
7.13.4 Asset-Related Demand Schedules XML Report	
7.13.5 Asset-Related Demand Schedule Details XML Report	208
7.13.6 Asset-Related Demand Schedule Offers XML Report	210
7.13.7 Asset-Related Demand Schedule Offers Hourly Updates XML Report	
7.13.8 Asset-Related Demand Schedule Selection XML Report	
7.13.9 Asset-Related Demand Hourly Updates XML Report	
7.13.10 Asset-Related Demand Results XML Report	215
8. Public Tab	217
	217
8.1 System Summary Display 8.2 Prices Display	
8.1 System Summary Display	218
8.1 System Summary Display 8.2 Prices Display	218 219
8.1 System Summary Display8.2 Prices Display8.3 Binding Constraints Display	218 219 220
 8.1 System Summary Display 8.2 Prices Display 8.3 Binding Constraints Display 8.4 Reactive Interface Limits Display 8.5 Net Tie Schedules Display 8.6 Public XML Download Display 	218 219 220 221 222
 8.1 System Summary Display 8.2 Prices Display 8.3 Binding Constraints Display 8.4 Reactive Interface Limits Display 8.5 Net Tie Schedules Display 8.6 Public XML Download Display 8.6.1 DA System Summary XML Report	218 219 220 221 222 222
 8.1 System Summary Display 8.2 Prices Display 8.3 Binding Constraints Display 8.4 Reactive Interface Limits Display 8.5 Net Tie Schedules Display 8.6 Public XML Download Display 8.6.1 DA System Summary XML Report	218 219 220 221 222 222 223
 8.1 System Summary Display 8.2 Prices Display 8.3 Binding Constraints Display 8.4 Reactive Interface Limits Display 8.5 Net Tie Schedules Display 8.6 Public XML Download Display 8.6.1 DA System Summary XML Report	218 219 220 221 222 222 223 223 224
 8.1 System Summary Display 8.2 Prices Display 8.3 Binding Constraints Display 8.4 Reactive Interface Limits Display 8.5 Net Tie Schedules Display 8.6 Public XML Download Display 8.6.1 DA System Summary XML Report 8.6.2 DA Prices XML Report 8.6.3 DA Binding Constraints XML Report 8.6.4 DA Reactive Limits XML Report 	218 219 220 221 222 222 223 224 224
 8.1 System Summary Display 8.2 Prices Display 8.3 Binding Constraints Display 8.4 Reactive Interface Limits Display 8.5 Net Tie Schedules Display 8.6 Public XML Download Display 8.6.1 DA System Summary XML Report	218 219 220 221 222 222 223 224 224
 8.1 System Summary Display 8.2 Prices Display 8.3 Binding Constraints Display 8.4 Reactive Interface Limits Display 8.5 Net Tie Schedules Display 8.6 Public XML Download Display 8.6.1 DA System Summary XML Report 8.6.2 DA Prices XML Report 8.6.3 DA Binding Constraints XML Report 8.6.4 DA Reactive Limits XML Report 	218 219 220 221 222 222 223 224 224
 8.1 System Summary Display 8.2 Prices Display 8.3 Binding Constraints Display 8.4 Reactive Interface Limits Display 8.5 Net Tie Schedules Display 8.6 Public XML Download Display 8.6.1 DA System Summary XML Report 8.6.2 DA Prices XML Report 8.6.3 DA Binding Constraints XML Report 8.6.4 DA Reactive Limits XML Report 	218 219 220 221 222 222 223 224 224 225
 8.1 System Summary Display 8.2 Prices Display 8.3 Binding Constraints Display 8.4 Reactive Interface Limits Display 8.5 Net Tie Schedules Display 8.6 Public XML Download Display 8.6.1 DA System Summary XML Report	218 219 220 221 222 222 223 224 224 225
 8.1 System Summary Display	218 219 220 221 222 222 223 224 224 225
 8.1 System Summary Display	218 219 220 221 222 222 223 224 225 225 227 230 231
 8.1 System Summary Display	218 219 220 221 222 222 223 224 224 225 225 227 227 230 231 233
 8.1 System Summary Display	218 219 220 221 222 222 223 224 224 225 225 227 227 230 231 233
 8.1 System Summary Display	218 219 220 221 222 222 223 224 225 225 225 227 227 230 231 233 233 234
 8.1 System Summary Display	218 219 220 221 222 222 223 224 224 225 225 227 230 231 233 233 234 234
 8.1 System Summary Display	218 219 220 221 222 222 223 224 224 224 225 227 230 231 233 233 234 234 234
 8.1 System Summary Display	218 219 220 221 222 222 223 224 224 225 225 227 227 230 231 233 233 234 234 234 234 235

	9.5 DRR Schedule Detail Defaults Display	237
	9.5.1 Data Validation Rules	
	9.5.2 View Schedule Detail Defaults	
	9.5.3 Modify Schedule Constraint Data	
	9.5.4 Modify Schedule Operating Limits	
	9.6 DRR Schedule Offers Daily Default Display	
	9.6.1 Data Validation Rules	
	9.6.2 Modify Schedule Offers	
	9.7 DRR Schedule Offers Hourly Updates Display	243
	9.7.1 Data Validation Rules	
	9.7.2 Override Schedule Offers for One or More Hours of a Market Day	247
	9.7.3 Remove/Delete Hourly Updates to Schedule Offers	
	9.8 DRR Schedule Times Hourly Updates Display	250
	9.8.1 Data Validation Rules	
	9.8.2 Override/Remove Startup and Notification Times for One or More Hours of	а
	Market Day	
	9.9 DRR Schedule Selection Display	253
	9.9.1 Select Schedule for an Operating Day and Market	254
	9.10 DRR Hourly Updates Display	
	9.10.1 Validation rules	
	9.10.2 Override MW Operating Limits for a Day and Hour	258
	9.10.3 Override the Status of a DRR for a Day and Hour	259
	9.11 DRR by Portfolio Display	
	9.11.1 View DA Commitment for each DRR in Portfolio	
	9.12 DRR XML Download Display	
	9.12.1 DRRs XML Report	
	9.12.2 DRR Schedules XML Report	
	9.12.3 DRR Schedule Details XML Report	
	9.12.4 DRR Schedule Offers XML Report	
	9.12.5 DRR Schedule Selection XML Report	
	9.12.6 DRR Hourly Updates XML Report	
	9.12.7 Schedule Offers Hourly Updates XML Report	
	9.12.8 Schedule Times Hourly Updates XML Report	
	9.12.9 Reduction by DRR XML Report	271
-		
	0. Continuous Storage Facilities (CSF)	
	10.1 CSF Concepts	272
	10.2 Managing CSFs in eMarket	
	10.3 CSF Generator	
	10.3.1 Unit Daily Default Parameters	
	10.3.1.1 Modify CSF Generator Default Status	
	10.3.1.2 Modify CSF Generator Ramp Rate	
	10.3.2 Schedule Detail Defaults	
	10.3.2.1 Modify Schedule Constraint Data	
	10.3.2.2 Modify Schedule Operating Limits	
	10.3.3 Hourly Updates	2/6

10.3.3.1 Enter Hourly Updates	277
10.3.4 Schedule Offers Default	
10.3.4.1 Update Schedule Offers Default	277
10.3.5 Schedule Offers Hourly Updates	278
10.3.5.1 Update Schedule Offers Default	279
10.4 CSF DARD	279
10.4.1 ARD Daily Default Parameters	
10.4.1.1 Modify CSF DARD Default Status	
10.4.1.2 Modify CSF DARD Ramp Rate	
10.4.2 Schedule Detail Defaults	
10.4.2.1 Modify Schedule Default Data	281
10.4.3 Hourly Updates	
10.4.3.1 Enter Hourly Updates	
10.5 CSF ATRR	
10.5.1.1 Validation rules for CSF ATRR regulation limits	
10.5.2 Regulation Offers Daily Default	
10.5.2.1 Modify CSF ATRR Default Status	
10.5.2.2 Modify CSF ATRR Regulation Range	
10.5.3 Regulation Hourly Updates	
10.5.3.1 Modify CSF ATRR Hourly Status	
10.5.3.2 Modify CSF ATRR Hourly Regulation Range	284

Appendix A. List of Fuel Index Names used by ISO-NE Market	
Monitoring	285

About This Document

This document describes the usage of the eMarket Graphical User Interface (GUI) from the perspective of an end user, such as a public user or Market Participant. It **does not** describe the GUI from the perspective of IT administrators at ISO New England wishing to configure the eMarket application. It does describe each display in the eMarket GUI application along with a set of common tasks users will perform. Some of the tasks change only certain data on a display. Users may certainly change any fields they wish during the steps. The tasks merely provide a quick reference for users who want to do certain things.

Who Should Use this Document

Market Participants wishing to use the eMarket user interface to perform market operations, and public users requesting market information should use this document.

For More Information

Additional information that was used in the creation and development of this document can be found in the following documents.

- eMarket Data Exchange Specification, which is found on the <u>ISO New</u> <u>England website</u> among the documents listed on the Participate > Support > Upload and Download File Format Protocols page.
- ISO-NE Manuals > M-11 Market Operations, which is found on the <u>ISO New</u> <u>England website</u> among the documents listed on the Participate > Rules and Procedures > Manuals page.

Conventions

The following are conventions used throughout this document:

Bolding is used to highlight important screen items.

Revision Record

Version	Date	Comments
1.0	03/23/2012	Original for review
1.1	05/28/2012	Updated with screenshots of production version of eMarket
		and to account for latest changes to displays
1.2	08/31/2012	Corrected errors in the validation rules.
1.3	09/15/2012	Updated screenshots in §4.12, §5.4 and §7.11
2.0	04/25/2013	Updated §1.4 to reflect the changes to the DA timeline.
		Added statement on when the cleared results of the
		Day-Ahead Market are available to be viewed and
		downloaded in §4.15.13, §6.5, §6.7.2, §6.7.3, §7.12,
		§7.13.10.

Version	Date	Comments
3.0	04/15/2014	 Updated to reflect changes for the new Regulation Market and support of Alternative Technology Regulating Resources (ATRRs). Significant changes: Updated the versions of web browsers supported by eMarket in §1.1 Updated §1.4 to describe the Regulation Market timeline for submittals. Updated §3.1 to add Regulation portfolios and concepts of Alternative Technology Regulation Resources (ATRRs) Revised §4.3 Unit Default Parameters Display to remove the regulation limits from the display because the authority to submit these parameters is now governed by the eMarket Regulation roles and this display is governed by the eMarket Generation roles. Created a new major section 5 Regulation Hourly Updates from the Generation Tab section into this section. Note that these two displays are changed significantly. Also moved the two Regulation XML Reports from the Generation Tab section to the new Regulation Tab section.
4.0	09/05/2014	 Updated to rescind the changes specific to the new Regulation Market and support of Alternative Technology Regulating Resources (ATRRs) and to reflect all the changes introduced in support of the Tariff changes for Energy Market Offer Flexibility (EMOF). The significant changes are: Updated the versions of web browsers supported by eMarket in §1.1 Updated §1.4 to describe the changes to the eMarket timeline for submittal of energy and regulation offers and of demand bids, and to introduce the new timeline for Intraday Re-Offer periods for energy and regulation offers. Also introduced the new timelines for submittal of Fuel Price Adjustment requests. Changed the names of displays to emphasize those used to submit daily parameters that carry forward to future market days until they are changed at a future date from those used to submit hourly updates of parameters that are specific to one market day; for example: Unit Manager is now Unit (Daily) Default Parameters; ARD Manager is now ARD (Daily) Default Parameters; (Unit) Ramp Rates is now ARD (Multiple) Ramp Rates Daily Default; ARD Ramp Rates is now ARD (Multiple) Ramp Rates Daily Default;

		 schedule 98 was for the unit's alternate fuel; schedule 97 was the Reference schedule for the unit's primary fuel and schedule 96 was the Reference schedule for the unit's alternate fuel. With Energy Market Offer Flexibility, schedules 95 through 98 are now Reference schedules for a unit's schedule: For single-fuel units, schedule 98 is the only Reference schedule; For dual-fuel units, schedule 98 is the Reference schedule for the unit's fuel #1 (likely its least cost fuel) and schedule 97 is the Reference schedule for the unit's fuel #2 (or alternate fuel); For three-fuel units, schedule 98 is the Reference
		 schedule for the unit's fuel #1 (likely its least cost fuel), schedule 97 is the Reference schedule for the unit's fuel #2 (main alternate fuel), and schedule 96 is the Reference schedule for the unit's fuel #3 (likely a seasonal alternate to either its primary or alternate fuel); Schedule 95 is reserved for the eventuality that a unit may have a seasonal alternate for both its least cost and its alternative fuels. Changed the validation rules for submittal of daily default parameters: when submitting updates to the daily default offer parameters, values must be submitted for <i>all</i> parameters. For example, a value for RTHOL must be submitted when submitting an update to a unit's schedule detail default parameters; it is no longer optional. Changed the validation rules for hourly updates: when submitting hourly updates, values for all related parameters must now be submitted when you want to submit an update for an hour. For example, you can no longer submit an hourly update for only the Economic Max of a unit; you must also submit an update for all the other related parameters are the same as the daily default values. That is especially true for hourly updates of financial offers: if you want to submit an update of your financial offers for an hour, you must submit values for all inancial parameters and their associated fuel blend information.
4.1 11/	1/20/2014	Added appendix A listing fuel index names and references to it throughout the document. Removed reference to SSL3.0

Version	Date	Comments
4.2	12/19/2014	 Revised section 1.1 to refer to the Web Browser Support for SMD Applications section of our website for a list of supported browsers and security protocols. Replaced the unit and ARD Schedule Selection display screenshots in §4.11 and §7.10 and added description of the new Schedule ID data column.
5.0	12/15/2014	 Major Revision # to re-instate all elements for the enhancement of the Regulation Market and for alternative technology regulating resources. Changes in this revision include: Updated §1.4.2 to reflect changes to the eMarket deadlines for the Regulation Market. In particular, changes to the close hour of the Regulation Market Intraday Re-Offer Period and to what can be submitted during this period, and finally changes to the number of minutes before the beginning of the hour that intraday re-offers can be submitted. Updated §3.1 to mention ATRRs in portfolios Revised §5.3 to describe the modified Regulation Offers Default Display and the special validation rules for ATRR offers Revised §5.4 to described the modified Regulation Hourly Updates Display and the special validation rules for ATRR offers Updated §5.5 to show the modified XML Regulation download messages. Replaced "Automatic Ramp Rate" with "Automatic Response Rate" throughout to be consistent with the term used in the Market Rule.
5.1	06/25/2015	 Revision to document the changes to the demand displays and XML download reports for Participants who register subaccounts in the ISO-NE Customer and Asset Management System (CAMS) and other editorial changes. Added note in §1.5 Effective Dates of Data about the need to submit data in successive market day order. Added §1.8 Demand Bid Subaccounts to discuss the nature and particularities of subaccounts. Described changes for subaccounts made to the displays and validation discussed in sections 6.2 Demand Bids Display 6.3 Increment Offers and Decrement Bids (Inc/Dec) Display and 6.4 Load Response Offer Display Described changes to the display and the XML reports discussed in §6.7 XML Download for Demand Display

Version	Date	Comments					
5.2	07/22/2015	 General revision to reflect changes to the eMarket MUI in regards to demand bid subaccounts: Modified §1.8 Demand Bid Subaccounts to state that an inactive subaccount will be shown on the drop-down lists if there are demand bids for the subaccount on the selected market day. This is necessary for Participants to be able to delete demand bids that may have been submitted before the subaccount was made inactive (terminated/deleted). Corrected §6.2, §6.3, §□ and §6.7 to clearly state that active subaccounts and only inactive subaccounts for which there are demand bids on the selected market day are listed on the selector drop-down list for subaccount name. Added note to §6.2.5 and §6.3.3 to clearly state that one can only delete demand bids when selecting an inactive subaccount; bids cannot be modified. Added note to §6.2.6 and §6.3.4 to specify that demand bids cannot be duplicated from one subaccount to another. Corrected the specification of when information on the demand bids that were cleared in the Day-Ahead Market is available in the Demand by Participant (§6.7.3) XML reports. 					

Date	Comments
11/28/2016	 General revision to reflect changes to the eMarket MUI in regards to: Eliminated the Use Startup No Load flag on all displays and in all XML reports. Since Participants must necessarily enter Startup and No Load cost with all their bids/offers, they can just as well submit zero values to achieve the same outcome thereby making the flag a redundant bid/offer parameter. Affected displays: 4.8 Schedule Offers Daily Default Display and 4.9 Schedule Offers Hourly Updates Display. Affected XML reports: 4.15.5 Unit Schedule Offers XML Report and 4.15.9 Schedule Offers Hourly Updates of a unit's Cold Notification or Cold Startup times can impact the unit's eligibility for 10-minute and 30-minute off-line reserve designations, the Schedule Times Hourly Updates display (§4.10) was enhanced to show a unit's off-line reserve eligibility under a selected schedule, both the daily default and for those hours where there are hourly updates of the Cold Notification and Cold Startup times. Added capability for Participant owners of dispatchable ARDs (DARD pumps in pump storage facilities) to submit Maximum Daily Consumption and Maximum Daily Starts for their assets and to view their configured Min Down Time and Min Run Time parameters. Affected displays: 7.7 ARD Schedule Detail Defaults Display. Affected XML reports: 7.13.5 Asset-Related Demand Schedule Details XML Report
	 Display. Affected XML reports: 7.13.5 Asset-Related Demand Schedule Details XML Report Participants can now submit fuel price adjustment (FPA)
	11/28/2016

Version	Date	Comments						
5.4	02/16/2018	 Modified description of deadlines in 1.4.1 Energy Market Timeline: Day-Ahead market for two days from now and beyond Re-Offer period open time Intraday physical parameters submittal for DRRs Added chapter 9 that describes the displays and functions of the new Demand Response Resource (DRR) Tab. Corrected description of the content of all XML reports 						
5.5	07/25/2018	Updated section 4.12.4. After EMOF project implementation, all generating units should price their dispatch level and not bid same EcoMin and EcoMax except non-dispatchable units.						
6.0	02/12/2019	 New revision to reflect support for Continuous Storage Facilities (Energy Storage Devices). Changes in this revision include: New chapter 10 Continuous Storage Facilities (CSF) Correction of timeline rule for Real Time (RT) Intraday Physical Parameter Resubmittal (DRRs only) in section 1.4.1. 						

1. Introduction

This chapter contains introductory material and concepts relevant to the eMarket Graphical User Interface (GUI). It describes such topics as high level GUI functionality, accessing the GUI, user roles, business rules and timelines.

1.1 Accessing the eMarket GUI

Please refer to the **Web Browser Support for SMD Applications** page on the **ISO New England website**, which can be found under Participate > Support > Web Browser Support, for information on the browsers and security protocols ISO New England supports.

All users need to register with ISO-NE Customer Support in order to use the eMarket application to view public information or to submit and view private information in the ISO New England Energy Market. This is done by the user's company Security Administrator using the ISO New England Customer and Asset Management System (CAMS) who will create the user accounts and assign each the various rights to access the eMarket application resulting in a digital certificate being issued to the user. Refer to the *Customer & Asset Management System (CAMS) Frequently Asked Questions* page on the <u>ISO New England website</u>, which can be found under Participate > Support > FAQs, for more information.

Once registered, market system users will be provided with the link to access the SMD home page, which will then allow direct access to the eMarket application, one of the SMD software applications. Access to the SMD Application Sandbox is also restricted to registered market users. Please contact Customer Support for the URL's to these pages.

All questions or inquiries about eMarket access should be addressed to the ISO New England's Customer Support Department at (413) 540-5220 or via email at <u>custserv@iso-ne.com</u>.

1.2 User Roles

Roles define the displays and actions that a user can use to update or view data. Distinct roles are defined for each type of market assets, e.g. Asset Related Demand (ARD), Unit (Generation), Demand, Load Response¹, Demand Response Resource (DRR) and Regulation Resources. As the names indicate, a user can only utilize the displays associated with the asset type. For example, a user with an ARD role can use all the ARD displays, but cannot use any Demand, Generation, Regulation, Load Response or DRR displays.

Roles are further divided into two sub-types; "Read-only" and "Read/Write" (also known as "Update"). Read-only roles restrict the user to strictly viewing data on a

¹ Load Response roles will be retired/discontinued following the implementation of the Price-Responsive Demand (PRD) project in June 2018.

display, while Read/Write roles allow the user to view and submit data using a display. For example, a Participant user with a Read Only ARD role can only view the Participant's market data using the ARD displays, while one with a Read/Write ARD role can view and change the Participant's market data using the ARD displays. The Public displays can be viewed by any user with any role. They are all read-only displays.

Refer to the *Customer and Asset Management System Application Group Roles* document, which is found on the <u>ISO New England website</u> among the documents listed on the Participate > Support > User Guides page, for a complete list and description of the eMarket roles.

1.3 Agent Users

Some users can represent or act on behalf of more than one Participant. Such users are referred to as **agents**. Agents need to specify the Participant they are representing when using eMarket. This is done by selecting the Participant in the upper right of the header.



Note: Although the eMarket application provides the capability to define users as agents, that capability to define agent users in CAMS is not available at this time (August 2014).

1.4 Business Rules and Timelines

The associated business rules and timelines can be found in the document *Market Rule 1*, which is found on the <u>ISO New England website</u> among the documents listed on the Participate > Rules and Procedures > Transmission, Markets, and Services Tariff page, and in the **ISO New England Manuals** that are found on the <u>ISO New England website</u> listed on the Participate > Rules and Procedures > Manuals page.

Submission deadlines are market restrictions established by Market Rule defining deadlines by which data must be submitted. If data is submitted after a submission deadline, that data submittal will be rejected.

1.4.1 Energy Market Timeline

The following submission deadlines for energy offers exist in eMarket:

Deadline	Description				
Day-Ahead Market for two days from now and beyond – Open and Close hour	Normally, the market opens at 12:00 Eastern time or as soon as the Day-Ahead market results for tomorrow are approved and posted, whichever is the latest, each day for the market day two days from now and beyond. Energy offer/bid submittals can be made between 12:00 and 23:59 every day, for the day after tomorrow and up to nine days in the future.				
Day-Ahead Market for tomorrow and beyond – Open and Close hour	Normally, the market opens at 00:00 and closes at 10:00 Eastern time each day for the next day's market. Energy offer/bid submittals can be made between midnight (00:00) and 10:00 every day, for up to ten days in the future. If a bid or offer is to be used in the Day-Ahead Market (DAM), it must be submitted by the 10:00 deadline on the day before the operating day. The markets database is unavailable for submittal of any energy offers/bids for tomorrow and beyond between Day- Ahead Market Close and Re-Offer Period Open, which is when the Day-Ahead commitment process is running. Note : The Real Time Intraday Re-Offer Period for today is however open, so the markets database is available for submittal of intraday re-offers of financial parameters for units and Asset Related Demands (ARDs), and for submittal of intraday updates of physical parameters for Demand Response Resources (DRRs).				
Re-Offer Period Open hour	Normally, the Re-Offer Period opens daily at 12:00 Eastern time or as soon as the Day-Ahead market results are approved and posted, whichever is the latest, but no later than 13:30 Eastern time. While the Re-Offer Period is open, energy offers/bids by units, ARDs, and DRRs can be submitted for tomorrow's market day. For those assets that were not committed in the Day-Ahead Market for tomorrow, changes to both financial and non-financial offer/bid parameters for tomorrow can be submitted during this period of time. Note : During the Re-Offer Period, energy offers/bids for two days from now and up to nine days in the future can also be made.				

Deadline	Description				
Re-Offer Period Close hour	Normally, the Re-Offer Period closes daily at 14:00 Eastern time. For a bid or offer to be in effect for the Real Time Market (RTM), it must be submitted by this deadline on the day before the operating day. After this hour, only hourly updates for tomorrow's market can be submitted when the Real Time Intraday Re-Offer period opens. Note : The Real Time Intraday Re-Offer Period for today is however open, so the submittal of intraday re-offers of financial parameters for units and ARDs and the submittal of physical parameters for DRRs is allowed as per intraday re-offer rules.				
Real Time (RT) Intraday Re-Offer Period – Open hour and special rules for Close hour (units and ARDs only)	 The RT Intraday Re-Offer period opens at 18:30 Eastern time the day before the operating day and implicitly closes at 22:30 of the operating day. During this period of time, hourly updates of unit and ARD energy offers for future hours of the operating day can be submitted. Offers for the next hour can be submitted up to 30 minutes before the beginning of the hour. Therefore After 23:30 the day before the operating day can no longer be submitted After 22:30 of the operating day, hourly updates for the first hour of the operating day can no longer be submitted. After 22:30 of the operating day, hourly updates for the operating day can no longer be submitted. During the RT Intraday Re-Offer period, only hourly updates of financial energy offers can be submitted; that is to say changes to incremental energy offers (MW/Price pairs) and under strict rules, changes to generator's Startup and No Load Cost offers, including changes to the associated fuel blend parameters. All other components of a unit or ARD asset offer/bid cannot be changed during this period. 				
Real Time (RT) Intraday Physical Parameter Resubmittal (DRRs only)	The RT Intraday Resubmittal period for DRRs opens at the time of Re-Offer close (subject to Day Ahead Market approval). During this period of time, hourly updates of DRR physical parameters (See §9.10 for the list of parameters) for current and future hours can be submitted.				

There are some special conditions where the above times may change such as extenuating circumstances when the market window must be extended. Market Participants are notified whenever these times are altered; the notifications are posted as special notices on the ISO website. Market Participants can sign up with the Customer Support Department at ISO New England to receive these notices by email.

1.4.2 Regulation Market Timeline

The following submission deadlines for regulation offers exist in eMarket:

Deadline	Description
Regulation Market for two days from now and beyond – Open and Close hour	The market opens at 18:30 and closes at 23:59 Eastern time each day for the market day two days from now and beyond. Regulation offers, daily and hourly updates, can be submitted between 18:30 and 23:59 every day, for the day after tomorrow and up to nine days in the future. <i>Same as the offers/bids for the Energy market.</i>
Regulation Market for tomorrow and beyond – Open and Close hour	The market opens at 00:00 and closes at 14:00 Eastern time each day for the next day's market and beyond. Regulation offers, daily and hourly updates, can be submitted between midnight (00:00) and 14:00 every day, up to ten days in the future.
	Note that 14:00 is the close of the Energy market Re- Offer Period, when a generator's economic limits for tomorrow are final and cannot be changed. These are the limits against which regulation limits are validated.
Regulation Market Intraday Reoffer Period – Open and Special rules for Close hour	 The Intraday Reoffer period opens at 14:00 Eastern time the day before the operating day and implicitly closes are 22:55 of the operating day. During this period, regulation offers hourly updates for thure hours of the operating day can be submitted. Offers for the next hour can be submitted up to 5 minutes before the start of that hour. Therefore After 23:55 the day before the operating day can no longer be submitted After 22:55 of the operating day, hourly updates for the <u>first</u> hour of the operating day can no longer be submitted After 22:55 of the operating day, hourly updates for the <u>first</u> hour of the operating day can no longer be submitted After 22:55 of the operating day, hourly updates for the <u>operating day</u> can no longer be submitted. During the Intraday Reoffer period, <u>hourly updates of only the regulation financial offer parameters</u> (Regulation Service Price, Regulation Inter-Temporal Opportunity Capacity Cost, and Regulation Other Capacity Cost) can be submitted. For Alternative Technology Regulation Resources (ATRR), hourly updates of both the regulation financial and physical (Availability for Regulation, Automatic Response Rate, Regulation High and Low Limits) can be submitted. What this means is that Participants can submit hourly updates of their regulation financial offers at any time for future hours of today and for any hour of tomorrow.

1.4.3 Fuel Price Adjustments Submittal Timeline

The following submission and processing deadlines and rules for Fuel Price Adjustment (FPA) requests exist in eMarket:

- FPA requests can be submitted at any time with a start hour for any future hours of today and any hour of the next seven days. FPA requests for the next hour of today must be submitted at least 30 minutes before the start of the hour.
- FPA requests are processed within one minute after being received except for periods of time where the processing is suspended:
 - Processing of FPA requests for tomorrow is suspended from the closure of the Day-Ahead Market and the opening of the Re-Offer Period for tomorrow while the commitment software is running to clear the tomorrow's Day-Ahead Market.
 - Processing of FPA requests for tomorrow is suspended from the closure of the Re-Offer Period and the opening of the RT Intraday Re-Offer Period for tomorrow while the commitment software is running to establish and finalize the tomorrow's operating plan.
 - Processing of FPA requests for days after tomorrow is suspended until the beginning of the day before the start date of the FPA request.

FPA requests whose processing is suspended remain in "Pending" state until processing resumes.

1.4.4 Rules for Timeline Checks

Every submittal (display or XML upload) to the eMarket application has an embedded timestamp that cannot be edited. This timestamp tells eMarket when the data was submitted. Almost all the submissions have the ability to enter day and sometimes time information; either for the past or the future. Deadline validation is performed for data submittals only; there is no deadline validation performed on data queries, which means that one can view and retrieve data at any time.

The time against which these validations are done will be the time that the eMarket application receives the submission for processing, which is the earliest time that eMarket can recognize the submission.

Consider the following examples (assumes a submission deadline of 10:00):

• You send a submittal for the next day at 09:59.45 AM. eMarket receives the complete submittal at 09:59:48 AM. Assume the validation and database processing required to submit this transaction is significant, such that the transaction does not complete until 10:00:12 AM. The transaction succeeds, because eMarket received the complete request at 09:59:48 AM, before the cut-off window.

- You send a submittal for the next day at 10:01.45 AM. eMarket receives the complete submittal at 10:01:48 AM. eMarket will reject this submission because it was received after the cut-off time.
- You send a submittal for the next day at 09:59.45 AM. Due to unusual network latency, eMarket receives the complete submittal at 10:00:13 AM. eMarket will reject this submission because it was received after the cut-off time, even though it was sent before the cut-off time. eMarket has no way of knowing that this submittal was sent before the cut-off time, and therefore can only rely on the time the submittal was received.

In the eMarket GUI, data entry fields on displays will be disabled and deemphasized based on the above deadlines to prevent the user from trying to submit data when the market is closed. Only some data submittals are allowed during the Re-Offer Periods. The data entry fields for these data values will be enabled and emphasized during the Re-Offer Periods when data submittal is allowed.

1.5 Effective Dates of Data

Much of the data in the eMarket application is time sensitive. As data is entered, you are typically required to enter the date for which the data applies. As a rule, it is not necessary to submit data for each Market day. With displays labeled as "Default", the submitted data is effective starting on the Market day specified and for all subsequent days up to the day the data is changed.

Note: Data must be submitted for successive market days. This means that you should not submit data for a future date, say July 10, and then submit data for an earlier date, say July 5, because the latter data submittal will overwrite the data previously submitted.

Hourly bid data on the other hand only applies to a specific Market day and hour. Specifically, data on displays labeled as "Hourly Updates" applies only to the specific day and hour for which the override is entered:

- Unit Ramp Rate Hourly Updates
- Schedule Offers Hourly Updates
- Schedule Times Hourly Updates
- Unit Hourly Updates
- Fuel Price Adjustment requests
- Regulation Hourly Updates
- ARD Ramp Rate Hourly Updates
- ARD Schedule Offers Hourly Updates
- ARD Hourly Updates
- Demand Bids
- Inc/Dec Bids

- DRR Hourly Updates
- DRR Schedule Offers Hourly Updates
- DRR Schedule Times Hourly Updates

1.6 Display Refresh vs. Browser Refresh

For cyber security reasons, the eMarket GUI does not use cookies to remember the last eMarket display page that the user accessed.

Consequently, if you perform a browser refresh to reload the web page using

- The F5 function key, or
- The Refresh function in Internet Explorer (toolbar button or Refresh on the View drop-down menu)
- The Reload function in Chrome (toolbar button or Reload on the page tab)
- The Reload function in Mozilla (toolbar button or Reload on the page tab)

eMarket will present its initial web page rather than reload the web page with the display you had selected along with your filter selections. To refresh the information shown in a display page, use the **Refresh** button provided on the eMarket display.

1.7 Browser "Back" Function with eMarket

Since the eMarket GUI does not use cookies to remember the last eMarket display page that the user accessed, using the browser's "Back" function to return to the eMarket GUI after navigating away will <u>not</u> return the user to the eMarket web display tab that was on display prior to navigating away. Rather, the user will be returned to the default eMarket initial web display tab.

Note: The response to an XML file upload and to all XML Download requests will be a downloaded response XML file that the user will be prompted to open or save. The web browsers can be configured in various ways to open the downloaded response file, from selecting the application to open the file (Firefox and Chrome) to using the application associated with the .XML file extension (Firefox and Internet Explorer). If the user's browser is the application designated/selected to open the file, it is recommended that its options be set to open the file in a new tab or window. Otherwise the response file will be opened in the same window display tab used to interact with eMarket and the user will not be able to get back to the eMarket web display tab that was on display prior to the action that caused the response XML file to be generated.

1.8 Demand Bid Subaccounts

Participants with registered subaccounts in the ISO-NE Customer and Asset Management System (CAMS) for their demand bids may now use these subaccounts when their users submit demand bids into the Market using the eMarket application.

Note: *The subaccount feature is optional in the eMarket application*. For Participants who do not register subaccounts in CAMS, the functionality of the eMarket GUI and the XML reports obtained using the eMarket GUI are unchanged. See the *eMarket Data Exchange Specification* for the effect of the subaccount feature on the eMarket Web Services.

For Participants with registered subaccounts in CAMS, a default subaccount named "Default" is automatically created in the eMarket application. Participant users can use this subaccount just like their registered subaccounts to submit and view demand and virtual bids, to view their bids that cleared in the Day-Ahead Market and to request XML Download reports. This subaccount is used when a Participant with registered subaccounts does not specify a subaccount when submitting demand or virtual bids using the SubmitDemandBid XML request.

As described in §0 <?xml version="1.0" encoding="UTF-8" standalone="yes"?> <GetRegHourlyResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages"> <RegHourly day="2014-12-15" ID="444"> <HourlyProfile> <HourlyValues time="2014-12-15T00:00:00-04:00"> <ARR>21.5</ARR> <Available>true</Available> <Capability>37.5</Capability> <RegHighLimit>680</RegHighLimit> <RegLowLimit>400</RegLowLimit> <RegCapOther>70</RegCapOther> <RegITOCap>70</RegITOCap> <RegServPrice>70</RegServPrice> </HourlyValues> <HourlyValues time="2014-12-15T01:00:00-04:00"> <ARR>21.5</ARR> <Available>true</Available> <Capability>37.5</Capability> <RegHighLimit>680</RegHighLimit> <RegLowLimit>400</RegLowLimit> <RegCapOther>70</RegCapOther> <RegITOCap>70</RegITOCap> <RegServPrice>70</RegServPrice> </HourlyValues> <HourlyValues time="2014-12-15T02:00:00-04:00"> <Available>true</Available> <Capability>37.5</Capability> <RegHighLimit>680</RegHighLimit> <RegLowLimit>400</RegLowLimit> <RegCapOther>70</RegCapOther> <RegITOCap>70</RegITOCap> <RegServPrice>70</RegServPrice> </HourlyValues> <HourlyValues time="2014-12-15T23:00:00-04:00"> <Available>true</Available> <Capability>37.5</Capability> <RegHighLimit>680</RegHighLimit> <RegLowLimit>400</RegLowLimit> <RegCapOther>70</RegCapOther> <RegITOCap>70</RegITOCap> <RegServPrice>70</RegServPrice> </HourlyValues> </HourlyProfile> <ShortName>UNIT</ShortName> </RegHourly> </GetRegHourlyResponse>

Demand Tab, users of Participants that have registered subaccounts in CAMS for a given Market Day must select a subaccount in order to view, submit or delete demand bids (be they Fixed, Price-Sensitive or Virtual (Inc/Dec) bids) using the eMarket GUI. Similarly, they must select a subaccount to view the demand bids that cleared in the Day-Ahead Market on the Demand By Portfolio display. They can also select a subaccount to filter the demand bid information returned in the XML Download reports; if a subaccount is not selected, demand bid information for all subaccounts is returned in the report. It is important to note that <u>only subaccounts active for the selected Market day and inactive</u> <u>subaccounts for which demand bids exist for the selected Market day will be listed in the drop-down selection lists presented to the users. Inactive subaccounts will appear with an asterisk (*) appended to their name.</u>

Note: Since Participant users can submit demand bids for several days into the future and it may be possible for the Participant to terminate (inactivate or delete) a subaccount after bids for future Market Days have been submitted, it is important to remember that the bids are not deleted when the subaccount is terminated; they remain in place and will be considered when clearing the Day-Ahead Market on those future days. *It is the responsibility of the Participant users to delete the bids that may be in place for future days when a subaccount is terminated.*

2. Interface Basics

This chapter describes the common topics and concepts used in the eMarket GUI. This chapter's sub-sections describe basic actions such as display navigation, screen layouts, filters, and data entry.

2.1 Common Display Fields and Elements



The basic elements are:

A. Header Area

- 1. **Market Status** Indicator displays "Open" or "Closed" depending on whether the market system is open to receive Participant offers and bids.
- 2. Date and Time current wall clock date and time
- 3. Participant Selector For most users, it simply shows which Participant they represent with their digital certificate. For agent users, it shows the Participant that the user is currently representing; the dropdown list shows all the Participants that the user can represent and provides the means to select a Participant on the list that they want to represent. For ISO New England's Customer Service personnel, it provides them the list of all Participants; they can select from the list the Participant on whose behalf they can submit offer/bids.

- 4. **Certificate User** name of the user associated with the digital certificate used to access the eMarket application
- 5. **User** this will normally be the same as the certificate user name and the drop-down list will show only that user's name. This drop-down box is for ISO New England's Customer Service personnel; it provides them the list of users registered under the selected Participant.

B. Navigation Area

- 1. **Navigation Tabs** allow the user to choose which set of displays of eMarket to view. Only tabs for displays that the user is allowed to view based on his role are shown.
- 2. **Navigation Sub Tabs** allow user to choose specific displays within a category.
- 3. **Data Window Title** –displays the title of the currently selected display from the tabs.
- C. **Filters pane** contains the filters for the selected display. See <u>section</u>2.3 for more details.
- D. Display Pane
 - 1. **Display Title** The display title reflects the type of information that is presented in the display, such as Day-Ahead Market (DAM) LMPs, generation schedules, and demand bids. The data display is determined by selections made from the navigation tabs and filters.
 - 2. **Display Contents** presents the data based on the selected navigation tabs and filters.
- E. Action Buttons A display can present any or all of the following buttons:
 - New use this button to add a new row to a grid display and edit the data in the added row. You can use the button several times to add more than one row and then edit the data in the added rows. The data in the new row is shown on a pink background and it is not updated in the database until you select the Update button. The displayed rows of data will be shown with normal background when it is successfully stored in the database.
 - 2. **Delete** use this button to delete one or more rows of data in the database. Only the highlighted rows in the display are deleted from the database. You can select more than one row to delete at the same time.
 - 3. **Update** use this button after you change the data in the display and you want to submit the data to the database.
 - 4. **Refresh** use this button to retrieve the most recent data from the data base, i.e. to refresh the information shown on the display.

- F. **Messages Pane** shows error, warning and confirmation messages for user actions. It can be collapsed or expanded by clicking the double arrow icon in the pane header. See <u>section</u> 2.4 for more details.
- G. **Display Update Progress Bar** A display popup labeled "Please Wait ..." will be presented when a display update is initiated. The popup will automatically close when the update completes.

2.2 Navigation

Users choose which display to view by using the Navigation Tabs. The tabs are presented in two rows. The top row defines the category of displays to view and the sub tabs show displays within that category.

The first row has up to five tabs which may show based on your defined roles. The five tabs are:

- **Public** for use by all registered market system users; it provides access to displays of information from the cleared DA Market.
- Generation provides access to the generating unit displays
- **Demand** provides access to the demand and load response displays
- **ARD** provides access to the ARD displays
- **Regulation** provides access to the regulation displays
- DRR provides access to the Demand Response Resource displays

The sub tabs row changes based on the navigation tab chosen. For example, choosing the Generation tab shows these sub tabs.

Public Generation Demand ARD Regulation DRR Unit Default Portfolio Private Ramp Rate Ramp Rate Hourly Updates Schedule Schedule Schedule Schedule Offers Schedule Times Schedule Hourly Fuel Price Generation Detail Defaults Offers Default Default Hourly Updates Hourly Updates Adjustments By Portfolio

Users can always tell which display is being shown in two ways:

- A. The selected navigation tabs are highlighted, and
- B. The display name is shown in the display pane.



The eMarket displays are not different web pages. You will notice that the URL shown in your browser address bar does not change as you select different eMarket displays. Consequently, you cannot use the browsers "Back" and "Forward" toolbar buttons to go back to a previously shown eMarket display. You must access it using the display tabs.

2.3 Filters Pane

The Filters pane provides a set of filters applicable to the currently shown display. It sits on the left side of the eMarket GUI window as shown below:

ISO new england			Marke	t Status: OPEN(I		Ve		201407221119)	Interfa 22-Aug-2014 1	
100 A				Participa	nt: ISO New England	i Inc		💙 User: S	uperuser	*
Public Generation Demand ARD Regulation										
Private Unit Default Portfolio Ramp Rate Ramp R Messages Parameters Manager Default Hourly		Schedule Schedule Offers Default Hourly Up			urly Fuel Price dates Adjustments	Generation By Portfolio	XML Download			
Schedule Detail Defaults										
Select Date, Unit and Schedule	Schedule Details for Unit - DFLPRIC	E (Price-based) on 24-Aug-2	D14							
Date: 24-Aug-2014	Schedule Available: Ye		1	4ax Daily Starts:				Emergency Min:	55.0	–
Portfolio: AAB	Day Ahead Market: 🔽		Max Dail	/ Energy (MWh):				Economic Min:	55.0	
Unit: UNIT NAME	Real Time Market: 🔽	- «<	Mi	n Downtime (hr):	1.0			Economic Max:	97.0	
Schedule: DFLPRICE			I	fin Runtime (hr):	1.0		RT Hig	h Operating Limit *:	97.0	

The filter selections vary based on the display. Some displays require only a date filter while others may require date, portfolio and asset filters. When you navigate to a display, the appropriate filter selections are shown in this pane.

The Filters pane can be hidden or resized to allow tailoring the eMarket layout to your liking. By default, the Filters pane is expanded as shown above. If you wish to completely hide the Filters pane, simply click on the double arrow icon in the pane header and the pane will collapse into a single bar on the left as shown below. To restore it, simply click on the double arrow icon again.

IS	new en	gland							Marke	: Status: OPE			Ve		20140722111	er Interfa 19) 22-Aug-2014	
										Partici	pant:	ISO New England	Inc		👻 User:	Superuser	
blic	Generation Dema	ind AR	D Regula	ition													_
vate ssages			Ramp Rate Default	Ramp Rate Hourly Updates	Schedule Manager	Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates		Hourly Updates	Fuel Price Adjustments	Generation By Portfolio	XML Download			
cher	dule Detail D) efault	<u>د</u>														
-	due Details for Unit		-	ased) on 24-Aug-2	2014												
		_	_	· ·													
	Schedule A	vailablet	Yes	_			Map	Cally Starts:					Em	ergency Min:	60.0		
	Day Ahead	Market:	र 📔	»»			Max Daily Er	nergy (MWh):					E	Economic Min:	182.0		
							Min D	owntime (hr): 8,0					F	conomic Max:	102.2		
	Real Time	Market:	7				Philip D	ownerine (i=), 8,0									

2.4 Messages Pane

The Messages pane shows error, warning and action confirmation messages. It appears at the bottom of the eMarket GUI window. By default, only its header bar is shown. To view the full Messages pane at any time, click on the double arrow icon in the pane header.

ISO	new engla	nd														n: 2.0.1(20	0140722111	9)	erface 9-2014 16:24:
										Re	adOnly Pa	rticipant:	ISO New En	gland Inc			💙 User:	Superuser	
iblic General	tion Demand	ARD Reg	ulation																
blic Syst ssages Sum	em Prices mary	Binding Constraints	Reactive Interface Limi	Net Tie Schedules	XML Download														
system S	ummary																		
ect Date			~	System Summa	ry for 23-Au	-2014													
Data:				Name															
Date: 23-	-Aug-2014			Day-Ahead Den	nand		01/13	02 / 14	03 / 15	04 / 16	05 / 17	06 / 18	07 / 19	08 / 20	09/21	10 / 22	11/23	12/24	
							11648	11319	11265	11244	11256	11758	12296	13574	14482	15448	16442	16972	
							17283	17344	17447	17520	17567	17328	17018	16850	16819	15591	13746	12242	
				Day-Ahead Loa	d Forecast		01/13	02 / 14	03 / 15	04 / 16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12/24	
							11600	11020	10720	10600	10550	11290	12660	14070	14900	15490	15930	16270	
							16480	16660	16710	16730	16820	16850	16570	16220	16230	15310	13810	12400	
				Day-Ahead Ope	rating Reserve	Ubjective	01 / 13 2404	02 / 14	03 / 15	04 / 16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11 / 23	12 / 24	
							2404	2324 2328	2337 2715	2337 2715	2337 2715	2352 2721	2324 2715	2324 2715	2324	2324 2334	2324	2324	
							2324	2320	2710	2715	2715	2121	2710	2715	2715	2004	2715	2324	
																	ſ	*	
																	L		
																			Refr

Whenever an update to save entered data is unsuccessful, the Messages pane is automatically expanded (if collapsed) and an error message is shown for each error on the display. Similarly, the pane will be expanded if collapsed when warning messages about the submitted data are posted. When the Messages pane is expanded, it can be collapsed again by clicking on the double arrow icon in the pane header.

ISO new engl	and								ket Status					n: 2.0.1(20	014072211	.19)	terface
μ								Rea	adOnly Pa	ticipant:	ISO New En	gland Inc			🍸 Use	r: Superuse	ər
blic Generation Demand	ARD Re	gulation															
blic System Prices ssages Summary	Binding Constraints	Reactive Interface Limi		XML Download													
ystem Summary																	
ect Date		~	System Summary	y for 23-Aug-20	14												
			Name														
Date: 23-Aug-2014			Day-Ahead Dem	and	01 / 13	02 / 14	03 / 15	04 / 16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12/24	
					11648	11319	11265	11244	11256	11758	12296	13574	14482	15446	16442		٦
					17283	17344	17447	17520	17567	17328	17018	16850	16819	15591	13746		k
			Day-Ahead Load	Forecast	01/13	02 / 14	03 / 15	04 / 16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	⇒	\mathbf{N}
					11600	11020	10720	10600	10550	11290	12660	14070	14900	15490	15930		
					16480	16660	16710	16730	16820	16850	16570	16220	16230	15310	13810	12400	
			A		· ///// 04 / 42	02/44	00.746	04/48	08.147	08.740	07.740	00.700	00/04	10 (22	44.400	42.424	Refre
ssages																	
ne Type	Number	Message Text															
Aug-2014 16:12:04 Error	0	The market is ou	urrently closed at ho	ur ending 24-AU	3-2014 01 (Timestamp	: 22-AUG-14	16:12:03).										

2.5 Data Displays and Entry

Enterable and non-enterable fields are distinguishable from each other by background color.

- A. Enterable fields are shown with a white background
- B. Non-enterable fields are shown with the screen background color or a grey background color

ISO new england		Markat Statur: 05515(01) Patrioparts ISO New England St.	eMarket User Interface Version: 2.0.1(00140722119) 22-Aug-2014 16:55:52 Version: Superviser Superviser
Public Generation Demand ARD Regulation			
Private Unit Default Portfolio Ramp Rate Ramp Rat Messages Parameters Manager Default Hourly U	e Schedule Schedule Schedule Schedule Offers Schedule Times Schedule Hourly Puel Price odates Manager Detail Defaults Offers Default Hourly Updates Hourly Updates Selection Updates Adjustments	Seneration XML by Portfolio Download	
Schedule Detail Defaults	· · · · · · · · · · · ·	A	
Select Date, Unit and Schedule «	Schedule Details for Unit - ISOPRCUT (Price-based) on 24-Aug-2014		
Date: 24-Aug-2014	Schedule Available: No Max Dally:	arts: Energer	cy Min: 0.0
Portfolio: Al_Units V	Day Ahead Markat: 🔽 Max Daily Energy (nic Min: 0.0
Unit: UNIT NAME	Real Time Market: 🔽 Min Downtim		ic Max: 0.0
Schedule: ISOPRCUT	Min Runtim	(hr): 0.0 RT High Operating	Limit *: 0.0
	Fast Start (10 min): Yes Hot to Cold Tim	(hr): 0.0	
	Fast Start (30 min): Yes Hot to Inter Tim	(hr): 0.0 Condense Av	alable: 🥅
	Hot Notification Tim	(hr): 0.0 Condense Startup C	ost (\$): 0.00
	D Inter Notification Tim		
	B Cold Notification Tim		Power: 0.0
	Hot Startup Tim		se (hr): 0.0
	Inter Startup Tim		
	Cold Startup Tim	(hr): 0.0	
		• Will ratifie used by dispetch and comm	simeni
			Update Refresh
Messages			8

To update data, the process is similar throughout the eMarket GUI:

- 1. Click to place the cursor in the enterable field that you want to update or use the TAB key to move to the next enterable field.
- 2. Change the data to a new value.
 - When the Enter key is hit or when another data entry field is selected, the background color of the field where data was entered is highlighted in pink to indicate that the new value is unsaved.
 - The application title in the header changes from white to pink, as shown below.

IS	Ö	newen	igland															Market Sta	itus: OPEN(01) Participanti	C.		n: 2.0.1(20140	Useri Superus		
Public Ge	enera	ation Dema	nd ARD	Regulation																					_
Private	Unit	Default Por	rtfolio Ramp nager Defaul	Rate F		ites M	thedule lanager	Schedule Detail Defau		chedule Iffers Default		Schedule Times Hourly Updates			el Price djustments	Generation By Portfolio	XML Download								
Schedu	lule	Detail D	efaults																						
Select Date,	, Unit	and Schedule				Schedule	Details fo	r Unit-ISOPF	RCUT (Pr	rice-based) o	n 24-Aug-2014														
		-Aug-2014			3		Sche	dule Available:	No						Max Daily	Starts:				Emergency	Min: 0.0				
Portfolio:					~		Day /	Ahead Market:						Max 0	Daily Energy	(Mwh):				Economic	Min: 0.0				
		IT NAME			~		Rea	al Time Market:							Min Downta	ne (hr): 1				Economic N	Max: 0.0				
Schedule:	19	OPRCUT			~										Min Runti	me (hr): 1				RT High Operating Lim	nit *: 0.0				
								Start (10 min):								me (hr): 0.0		 							
							Fast	Start (30 min):	Yes							me (hr): 0.0				Condense Availa					
																me (hr): 0.0				Condense Startup Cost					
																me (hr): 0.0				Condense Hourly Cost					
																me (hr): 0.0				Condense Pov					
																me (hr): 0.0				Condense Notification Time ((hr): 0.0				
																me (hr): 0.0 me (hr): 0.0									
														Ca	io scarcup n	ne (nr): 0.0									
																			• Will not	be used by dispetch and commitm	neni				
																							Update) [_ R	Refresh
Messages																									*

- 3. Click the **Update** button to save the data to the database.
 - The "Please Wait..." progress bar is shown in the header.
 - When the update is complete, the "Please Wait..." progress bar is removed, all unsaved highlights are removed and at least one message is added to the Messages pane indicating the completion status.
 - The changed data values are now stored in the database.

Inevitably users will make mistakes during data entry. When this happens, eMarket automatically expands the Messages pane showing one or more error messages and also highlights the field(s) in error. Similarly, when some of the entered data is accepted but violates certain cautionary checks, eMarket will expand the Messages pane to show warning messages.

When the market is closed (see §1.4), eMarket does not allow data submittals. To prevent data entry while the market is closed, eMarket shows you in several ways as shown below:

- A. The Market Status Indicator will show that the market is closed.
- B. The data cells in the display are shown with a grey background indicating that they are non-enterable.
- C. The "Update", "Delete" and the "New" buttons on applicable displays are not shown.

ISC	new englan	nd										A	Market	Status: CLOSED		eMarke ersion: 2.0.1(20: cate User: Super		erface
															ISO New England Inc		Vseri Superuser	*
Public Ger	neration Demand	ARD Regulation	ion															
	Unit Default Portfolio Parameters Manager	Ramp Rate Default	Ramp Rate Hourly Upda	Schedule Manager	Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Fuel Price Updates Adjustments	Generation By Portfolio	XML Download						
Schedu	ile Detail Defa	ults																
Select Date, L	Init and Schedule			ichedule Details fo	r Unit - ISOPRCUT	(Price-based) a	m 24-Aug-2014								B			
Datas	24-Aug-2014																	
	All Units		-		dule Available: N						ily Starts:		1		Emergency Min			
	UNET NAME		~		Ahead Market: 🔽					Max Daily Ener					Economic Min			
	ISOPROUT		~	Rei	al Time Market: 🔽						time (hr): 0.0				Economic Max			
	20011001										time (hr): 0.0				RT High Operating Limit *	0.0		
					Start (10 min): Ye						Time (hr): 0.0					_		
				Past	Start (30 min): Ye	15					Time (hr): 0.0				Condense Available			
										Hot Notification					Condense Startup Cost (\$)			
										Inter Notification					Condense Hourly Cost (\$)			
										Cold Notification					Condense Power			
											Time (hr): 0.0				Condense Notification Time (hr)	0.0		
											Time (hr): 0.0							
										Cold Startup	Time (hr): 0.0							
																\mathbf{c}		
														• Will not	De used by dispect and commit-	L		
																		Refresh

Another item of note is that if you have new data in the display that has not been submitted to the database, it may be lost if you select the Refresh button or attempt to navigate to a different display. eMarket will warn you of this by presenting this dialog:

Editing Data	×
There may be unsay	ved data. Continue?
	OK Cancel

This allows you to click **OK** to continue and lose your changes, or **Cancel** to cancel your previous action thereby giving you the chance to save the entered data.
3. General Tasks

This section discusses the tasks that apply to one or more categories within the application.

3.1 Manage Portfolios

Portfolios provide filters of data for the eMarket user interface. Portfolios are used to group resources and/or location nodes together to benefit you in viewing and managing data, and can be used in each of the ARD, Demand, DRR, Regulation and Generation sections of eMarket. By using portfolios, you can organize resources in a matter suited to your business. You can create any number of portfolios with each portfolio uniquely named by you to describe its usage. Portfolios that you create belong only to you and cannot be shared.

The combination of a user's portfolios determines the information that is reported in the asset-oriented XML Download reports. For example, the "Generation by Unit" XML report contains information about all the unique generating units *listed in all* of your portfolios.

A portfolio contains one or more resources, and a resource can appear in any number of portfolios. Any defined resource (ARD, DRR, Pnode, ATRR² or Unit) can be included in multiple portfolios. There is no requirement that a resource appear in at least one portfolio, so it is possible that a resource will not appear in any portfolio.

Note: any unit, ARD asset, DRR, regulating resource or node that is not associated with a portfolio does not appear in the filters used for choosing a resource. In other words, only assets listed in portfolios can be selected to view/update its market data. That's because the user must first select a portfolio and only assets listed in the selected portfolio will be available for selection in the asset selection combo box drop-down list.

Portfolios are not effective dated. The data in a portfolio exists from when the portfolio is created until the portfolio is deleted.

² Alternative Technology Regulation Resources

The Portfolio Manager display under Generation, Demand, DRR, Regulation or ARD is used for creating, modifying or deleting portfolios. It looks like this:

ISO new england		Market Status: CLOSED		eMarket User Interface (ersion: 2.0.1(201407221119) Icate User: Superuser 22-Aug-2014 17:16:39
P**		Participant:	ISO New England Inc	Vser: Superuser V
Public Generation Demand ARD Regulation				
Private Unit Default Portfolio Ramp Rate Ramp Rate Schedule Messages Parameters Manager Default Heurly Updates Manager	Schedule Schedule Schedule Offers Detail Defaults Offers Default Hourly Updates	Schedule Times Schedule Hourly Hourly Updates Selection Updates	Fuel Price Generation Adjustments By Portfolio	XML Download
	Detail Detaults Offers Detault Pourly Opdates	Hoony opcares Selection Opcares	Majasanieros By Portifolio	
Portfolio Manager A				G
Portfolios	Units Included in Portfolio Mkt_Adm		Units Not Included in P	Portfolio Mkt_Adm
ААА	AES THAMES	<u> </u>	BELLOWS FALLS	<u> </u>
AAB	ALTRESCO		BERKSHIRE POWER	· · · · · · · · · · · · · · · · · · ·
AAC	AMOSKEAG		BERLIN 1 GT	
AAD	ANP-BELLINGHAM 1		BRANFORD 10	
AAE	ANP-BELLINGHAM 2 ANP-BLACKSTONE ENERGY 1		BRAYTON PT 1 BRAYTON PT 4	
All_Units MITIG	ANP-BLACKSTONE ENERGY 1 ANP-BLACKSTONE ENERGY 2	E	BRIDGEPORT ENERGY 1	
Mkt_Adm	ASCUTNEY GT		BRIDGEWATER	
	AYERS ISLAND		BRISTOL REFUSE	
	AZISCOHOS HYDRO		BULL HILL WIND	
	BAR HARBOR DIESELS 1-4	(F)	CABOT	
	BAR MILLS		CANAL 2	
	BELDENS-NEW		CAPE GT 4	
B	BERKSHIRE WIND POWER PROJECT		CDECCA	
	BETHLEHEM	(D)	CLEARY 9/9A CC	(D)
	BOLTON FALLS		DARTMOUTH POWER	
	BONNY EAGLE/W. BUXTON	•	DIGHTON POWER LLC	
New Copy Delete	Filter:			Filter
Messages				8

The basic features of this display are:

- A. **Portfolio List** this area shows the list of the existing portfolios owned by the user ordered by name.
- B. **Portfolio Action Buttons** these buttons are for managing the portfolios themselves:
 - 1. **New** creates a new portfolio
 - 2. **Copy** creates a copy of an existing portfolio with all of the same contents
 - 3. **Delete** removes an existing portfolio
- C. **Items Included List** this middle section of the display shows the items that are contained within the portfolio currently selected in the **Portfolio List**.
- D. **Filters** you can type in characters which limit the items in the list above the filter. This is used to more easily find items.
- E. Add To Portfolio (left arrow) clicking this icon adds the selected item(s) in the **Items Not Included List** on the right to the **Items Included List**
- F. Remove from Portfolio (right arrow) clicking this icon removes the selected items from the Items Included List and moves them to the Items Not Included List. Removing the last item in a portfolio automatically deletes the portfolio because a portfolio cannot be empty.
- G. **Items Not Included List** this right section of the display shows the items that are not currently in the selected portfolio and can be added. Note that

with the Generation, DRR, Regulation and ARD Portfolio Managers, only items that belong to the Participant that the user represents are shown on this list.

The following sections go through the steps for creating, modifying, and deleting portfolios.

Note: Demand portfolios are particular. When you create a Demand portfolio with a name that is the same as a Generation, Regulation and/or ARD portfolio, the generating units and ARDs listed in those portfolios, but not the ATRRs, are automatically listed in the Demand portfolio that you created. For example, if you create a Generation portfolio named "East" and add unit Gen1 to it. Then create a Demand portfolio named "East" and you will find that the Pnode for Gen1 is already listed. That is because generating units, including regulation capable units, and ARDs have Pnodes associated with them. Therefore if you want to create a Demand portfolio that includes the Pnodes of units and ARDs listed in Generation and ARD portfolios, give the same name to all three portfolios.

3.1.1 Create a Portfolio

There is no limit to the number of portfolios that you can create. Each portfolio is user-specific, meaning that only you can manage and use the portfolios that you create.

To create a portfolio:

- 1. In the navigation sub tabs, click the **Portfolio Manager** tab.
 - The appropriate Portfolio Manager display is shown in the display pane.
- 2. Click the **New** action button.
 - The following dialog is shown:



- 3. Enter the name of the new portfolio in the dialog box.
- 4. Click **Ok**.
 - The portfolio now appears in the Portfolio List
 - The very first item in the Items Not Included List is automatically added to the new portfolio, because a portfolio cannot be empty. You will have to remove it from the Items Included List if you don't want it in the portfolio, but do this only after adding the items that you do want listed in the new portfolio. Removing it while it is the only item listed will automatically delete the portfolio that you just created.

- 5. Select the desired item(s) in the **Items Not Included List.** Multiple contiguous items may be selected by holding down the shift key. Random items may be selected by holding down the Ctrl key.
 - The selected items are highlighted
- 6. Click the **Add to Portfolio** button
 - Items are added to the Items Included List for the selected portfolio
 - Items are removed from the Items Not Included List
- 7. (optional) Enter text into the Filter area below the Items Not Included List.
 - The list is reduced to only items that match the typed filter as shown:

19	i nev	wengland	d						Ma	rket Status:				Ve		1(2014072211		rface
											ticipant:	_	ew England Inc	oerane			Superuser	~
Public	Generation	Demand /	ARD Regula	tion														
Private Messages	Unit Default Parameters	Portfolio Manager	Ramp Rate Default	Ramp Rate Hourly Updates	Schedule Manager	Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Updates			eration ortfolio	XML Download			
Portfo	olio Mana	ager																
Portfolios						Units Include	ed in Portfolio Ml	ct_Adm					Units Not Inclu	ded in Po	rtfolio Mkt_	Adm		
AA						AES THAMES					-		DOMINION BRI	DGEPORT	FUEL CELL			
AB						ALTRESCO							FRAMINGHAM J	ET 1				
AAC						AMOSKEAG							MIDDLETOWN 2	2				
AD						ANP-BELLING	HAM 1						MIDDLETOWN 3	3				
AAE						ANP-BELLING	HAM 2						MILFORD POWE	ER				
All_Units						ANP-BLACKS	TONE ENERGY 1						MILLENNIUM					
AITIG						ANP-BLACKS	TONE ENERGY 2											
4kt_Adm						ASCUTNEY G	т											
						AYERS ISLAN	D											
						AZISCOHOS	HYDRO											
						BAR HARBOR	DIESELS 1-4											
						BAR MILLS												
						BELDENS-NEV												
							VIND POWER PRO	JECT										
						BETHLEHEM												
						BOLTON FALL												
						BONNY EAGL	E/W. BUXTON				-							
			New	Copy	Delete		Filter							F	ilter: MI			
lessages														-				

8. Repeat steps 5-7 until all desired items are listed in the portfolio.

3.1.2 View a Portfolio

- 1. In the navigation sub tabs, click the **Portfolio Manager** tab.
 - The appropriate Portfolio Manager display is shown in the display pane.
- 2. In the **Portfolios List** on the left, select the portfolio that you want to view.
 - The contents of the portfolio are displayed in the **Items Included List**
- 3. (optional) Enter text into the **Filter** below the **Items Included List**.
 - The list is updated to show only those items that match the filter

3.1.3 Modify a Portfolio

- 1. View the portfolio that you want to modify (§3.1.2)
- 2. To add items to the portfolio:

- a. Select the desired item(s) in the **Items Not Included List.** Multiple contiguous items may be selected by holding down the shift key. Random items may be selected by holding down the Ctrl key.
 - The selected items are highlighted
- b. Click the **Add to Portfolio** icon.
 - Items are added to the **Items Included List** for the selected portfolio.
 - Items are removed from the Items Not Included List.
- 3. To remove items from the portfolio:
 - a. Select the desired item(s) in the **Items Included List.** Multiple contiguous items may be selected by holding down the shift key. Random items may be selected by holding down the Ctrl key.
 - The selected items are highlighted.
 - b. Click the Remove from Portfolio icon.
 - Items are removed from the Items Included List
 - Items are added to the Items Not Included List
 - If the item removed is the last in the portfolio, the portfolio is automatically deleted.
- 4. Repeat steps 2-3 as desired.

3.1.4 Create a Copy of a Portfolio

- 1. View the portfolio that you want to copy (§3.1.2)
- 2. Click the **Copy** button.
- 3. The following dialog is shown:



- 4. Enter the name of the new portfolio in the dialog.
- 5. Click the **OK** button to copy the portfolio.

3.1.5 Delete a Portfolio

- 1. View the portfolio that you want to delete (§3.1.2)
- 2. Click the **Delete** button.
- 3. The following dialog is shown:

Delete Portfolio		×
Delete the portfolio "M	IBP_Test" ?	
	ОК	Cancel

4. Click the **OK** button to delete the portfolio.

3.2 Upload or Download XML Files

Each Navigation Tab has an XML Download sub tab that looks similar to this one:

ISO new england		Version: 2.0. Market Status: CLOSED Certificate User: S	
P.,		Participant: ISO New England Inc	Vuser: Superuser V
Public Generation Demand ARD Regulation			
Public System Prices Binding Reactive Messages Summary Constraints Interface Lim	Ret Tie XML Schedules Download		
XML Download		В	
Select Date «	XML Download on 24-Aug-2014		
Date: 24-Aug-2014	Day-Ahead System Summary Day-Ahead Prices Day-Ahead Binding Constraints Day-Ahead Reactive Limits Day-Ahead Net Tie Schedules	Upload XML SOAP Request File:	
Messages			8

The fields on this display are:

- A. **Download Report List** list of reports/files available for download.
- B. Upload XML SOAP Request field and controls used to upload XML files
 - File name of the file containing an XML formatted request that you want to submit to the Market system. The name of the file is set by using the Browse button
 - 2. Browse Button click to find and set the name of the file to upload
 - 3. **Submit Button** click to upload the file once you have defined an upload filename

3.2.1 Download an XML file

To download XML reports/files:

1. In the Navigation sub tabs, click the **XML Download** tab.

- The XML Download display is shown in the display pane.
- 2. Select the desired date from the **Date** filter located in the **Filters pane**.
- 3. Click the name of the specific XML report/file to be downloaded. This will bring up a popup giving you the choice to Open or Save the downloaded XML file. Remember that if you select Open, your browser will attempt to open the files based on its .xml filename extension and that the application designated to view .xml files may be your browser. In the event that it's your browser, the file may open in your current window tab or a new tab depending on the settings of your browser. See note in §1.7.

Public Generation Demand ARD Regulation Public Generation Demand ARD Regulation Public System Prices Brinding Reactive Select Date Constraints Differ Generation Differ Generation Select Date Constraints Date: 23-Aug-2014 File Download Date: 23-Aug-2014 File Download Differ Generation Differ Generation Constraints Day-Ahead Reactive Limits Day-Ahead Net Tie Schedules Open Save Constraints Day-Ahead Net Tie Schedules Use is in the Internet can be useful, some file: can potentially Name: public, getpricerresponse.xml Differ Generation Differ Generation State of the Internet can be useful, some file: can potentially Man: public, defineterresponse.xml Differ Generation Open Save Cancel Visite files from the Internet can be useful, some file: can potentially Name: public, defineter tean be useful, some file: can potentially Name: public, defineter tean be useful, some file: can potentially Name: public, defineter tean be useful, some file: can potentially	ISO (new england							eMarket (Version: 2.0.1(2014 Sertificate User: Superus		
Public Messages System Summary Prices Binding Constraints Reactive Limits XML Download XML Download on 23-Aug-2014 Select Date C Carder System Summary Day-Ahead System Summary Day-Ahead Prices Day-Ahead Reactive Limits Day-Ahead Reactive Limits Day-Ahead Reactive Limits Day-Ahead Net Tie Schedules Open Vinite file from the Internet can be useful, some files can potentially herrory councile. If you do not tuent the source on do gen or a do not gen or sove	ļ.						Particip	ant: ISO New England Inc		Vser: Superuser	~
Messages Summary Constraints Interface Limits Download Select Date Constraints XML Download on 23-Aug-2014 Date: 23-Aug-2014 Day-Ahead System Summary Day-Ahead Prices File Download Image: Select Date Date: 23-Aug-2014 Day-Ahead System Summary Day-Ahead Binding Constraints Day-Ahead Net Tie Schedules File Download Image: Select Date View Date: Day-Ahead Reactive Limits Day-Ahead Net Tie Schedules File Download	Public	Generation	Demand	ARD	Regulation						
Select Date: 23-Aug-2014 Date: 23-Aug-2014 Date: 23-Aug-2014 Date: 23-Aug-2014 Date: Day-Ahead System Summary Day-Ahead Prices Day-Ahead Binding Constraints Day-Ahead Reactive Limits Day-Ahead Net Tie Schedules Open Save Cancel While files from the Internet can be useful, some files can potentially herrow concluster. Hyou do not due to one does on of open or			Prices								
Date: 23.Aug-2014 Bay-Ahead System Summary Day-Ahead Prices Day-Ahead Binding Constraints Day-Ahead Binding Constraints Day-Ahead Reactive Limits Day-Ahead Net Tie Schedules Image: Solution of the state of the internet can be useful, some files can potentially Image: While files from the internet can be useful, some files can potentially	XML Down	load									
Day-Ahead System Summary Day-Ahead Prices Day-Ahead Binding Constraints Day-Ahead Binding Constraints Day-Ahead Reactive Limits Day-Ahead Net Tie Schedules Open Save	Select Date		~	XML Download on 23-Aug-	2014						
	Date: <u>23-AL</u>	G -2014		Day-Ahead Prices Day-Ahead Bindin Day-Ahead Reacti	g Constraints ve Limits	Do you wa Parata Do you wa	In to open or save this f Name: public_getprices Type: XML Document From: emktmuiprj-new 	esponse.xml Save Cancel be useful, some files can potentia	Browse		

4. Alternatively, you can right-click the file that you want to download. That will bring you a popup menu and give you more choices as to what you want to do with the downloaded XML file.

3.2.2 Upload an XML Request file

You can create files containing XML formatted requests to submit or get data from the Market system. Refer to the eMarket Data Exchange Specification for information on how to create all the various kinds of eMarket requests that can be created in an XML file. You can upload a file containing an XML formatted eMarket request using any of the XML Download display tab.

To upload an XML request file:

- 1. Click the **XML Download** tab under any of the Navigation sub tabs.
 - The XML Download display is shown in the display pane.
- 2. In the **Upload XML SOAP Request** box on the display click the **Browse** button.
 - The Windows Choose File dialog opens.

ISO	new england	Market S	Status: CLOSED Participant: I	eMarke Version: 2.0.1(Certificate User: Sup ISO New England Inc	
Public	Choose File to Upload	×			
Public Messages	Go → I → HM (J:) ▼ Testing Data ▼ XML ▼ 2014-08-20		d		
XML Do	Organize - Newfolder 83	- 🗆 🕜			
Select Date	SubmitARDRRHourly starting 2014-08-20 for GDF Suez Energy Marteting NA.xml SubmitARDRSheduleBidHourly_1 starting 2014-08-20 for Bear Swamp Power Company LLC.xml SubmitARDScheduleBidHourly_1 starting 2014-08-20 for GDF Suez Energy Marketing NA.xml SubmitRegPhysicalHourly starting 2014-08-20 for Exelon Generation Company_LLC.xml SubmitRegPhysicalHourly starting 2014-08-20 for Exelon Generation Company_LLC.xml SubmitScheduleBidHourly_1 starting 2014-08-20 for Exelon Generation Company_LLC.xml SubmitScheduleBidHourly_1 starting 2014-08-20 for Exelon Generation Company_LLC.xml SubmitScheduleBidHourly_1 starting 2014-08-20 for Black Bear Hydro Partners_LLC.xml SubmitScheduleBidHourly_1 starting 2014-08-20 for Brayton Point Energy_LLC.xml SubmitScheduleBidH	mitScheduleBidl mitScheduleBidl mitScheduleBidl mitScheduleBidl mitScheduleBidl mitScheduleBidl mitScheduleBidl mitScheduleBidl mitScheduleBidl mitScheduleBidl mitScheduleBidl	Upload XN File:	1L SOAP Request Browse	
Messages					۲

- 3. Select the desired file that you want to upload and click **Open**.
 - The filename is shown in the Upload XML SOAP Request File box.
- 4. Click the **Submit** button
- 5. An XML message file is downloaded in response to every upload request. It contains a completion message or the data that you requested. Note that the name of the response file is distinctive when it's an error response. You will be presented with the following popup giving the choice to open, save or discard the file.



3.3 Public and Private Messages Displays

These eMarket displays provide the means for eMarket to communicate messages to all market system users (Public Messages) or to specific groups of users (Private Messages). The messages are assigned priorities by the sender.

ISO n	ew england					Market Status: CLOSED		Version	rket Use 1: 2.0.1(201407221 2er: Superuser	119)	ace
1						_	ISO New England Ir			: Superuser	~
Public	Generation	Demand	ARD	Regulation							
Public Messages	System Summary	Prices	Binding Constraints	Reactive	Net Tie Schedules	XML Download					
Public Mes	sages	u	u	u u		u					
Select Date		« Me	essages for 23-Aug-201	4							
Date: 23-Auc	5-2014		ID Priority	Text							
											Refresh
L											Reiresn
Messages											
									rket Use		ace
ISO n	ew england					Market Status: CLOSED			n: 2.0.1(201407221 er: Superuser		14 17:30:01
						ReadOnly Participant: 1	ISO New England Ir	nc	Y User	n: Superuser	*
Public	Generation	Demand	ARD	Regulation							
Private Unit Messages Para	Default Portfolio ameters Manager	Ramp Rate Default	Ramp Rate Sch Hourly Update Mar	edule Schedule Detail Defaults	Schedule Offers Default	Schedule Offe Hourly Update Hourly Update	Schedule Selection	Hourly Updates	Fuel Price G Adjustments B	ieneration XM y Portfolio Do	IL wnload
Private Mes	ssages										
Select Date		Me	ssages for 23-Aug-201	4							
Date: 23-Aug	-2014		ID Priority	Text							

Note that this messaging capability is currently not used by ISO New England.

Refresh

4. Generation Tab

To access the Generation Tab, you must be a registered user with a Generation Read/Write or Read-Only role. When you have this access, the Generation tab will be shown and selecting it will show all of its sub tabs as shown below.

 Public
 Generation
 Demand
 ARD
 Regulation

 Private
 Unit Default
 Portfolio
 Ramp Rate
 Ramp Rate
 Schedule
 Schedule
 Schedule Offers
 Schedule Offers
 Schedule Times
 Schedule
 Hourly
 Fuel Price
 Generation
 XML

 Messages
 Parameters
 Manager
 Detail Default
 Offers Default
 Hourly Updates
 Selection
 Updates
 Adjustments
 By Portfolio
 Download

4.1 Generation Offer Concepts

When submitting data, it is important to understand the hierarchy of the Generation offer data and the implications of data overrides. Figure 1 below shows the hierarchy of the Generation displays used to submit data and where the same offer data appears in different displays, the data of which display has precedence over (overrides) the others. [**Note**: only the data elements subject to overrides are shown.]

The Unit Default Parameters display is at the root of the hierarchy. The Unit record itself is initially defined by ISO New England with default offer data when the asset is instantiated in the Market system. Participants must update some data values, like the unit's default dispatch availability status and single value ramp rate, using the display. For units capable of providing off-line reserves, the default Claim10 and Claim30 values must be updated on this display.

To participate in the energy market, a Participant must define at least one schedule for the unit using the Schedule Manager display and all of the child data that goes with it. The schedule defines a container for which detail offer data must be submitted.

Managing generating unit schedules is accomplished by performing the following general steps:

- Create or modify a schedule using the Schedule Manager display
- Create or modify schedule-specific default data, which include MW operating limits, Startup and Notification times, MinRun and MinDown times, and market participation (day-ahead or real time) using the Schedule Detail Defaults display.
- Create or modify schedule-offer default data (Startup and NoLoad costs and Incremental Energy Offers [Price/MW pairs], including fuel blend data for multi-fuel generators) using the Schedule Offers Default display.
- Identify the schedule or schedules to be used in the clearing of the Day-Ahead and Real Time (Balancing) Markets using the Schedule Selection display.

For units that are eligible to participate in the Regulation Market, Regulation Offers must be submitted using the displays under the Regulation asset tab.



Figure 1 – Generation Display Hierarchy and Data Precedence

The other displays allow for overriding schedule based default data or a unit's default dispatch availability status data. Offer data in other displays higher in the hierarchy are overridden by the presence of data in these displays:

- The Unit Ramp Rate Default display can be used to define a default ramp rate curve for a unit that overrides the simple slope defined with the Unit Default Parameters display.
- The limits on the Hourly Updates display (Emergency Min, Economic Max, Economic Min, and RT High Operating Limit) override for specified hours of the day the corresponding limit values defined in the Schedule Detail Defaults display for all schedules.
- The dispatch availability status (Unavailable, Economic, Must Run) on the Hourly Updates display overrides for specified hours of the day the default dispatch availability status (Default Status) defined on the Unit Default Parameters display. For units capable of providing off-line reserves, the display can be used to override for specific hours of the day the default Claim10 and Claim30 values defined on the Unit Default Parameters display.
- The Startup and NoLoad costs and the incremental energy offers (Price/MW pairs) along with their associated fuel blend data for multi-fuel units shown on the Schedule Offers Hourly Updates display override for specified hours of the day the default values defined in the Schedule Offers Default display for the corresponding schedule.
- The Startup and Notification times shown on the Schedule Times Hourly Updates display override for specified hours of the day the default values defined in the Schedule Detail Defaults display for the corresponding schedule.

Note: For Generators that are part of a Continuous Storage Facility (CSF), refer to chapter 10 Continuous Storage Facilities (CSF) for additional considerations.

4.2 Generation Portfolio Manager Display

A generation portfolio is a collection of generating resources that is defined by each individual generation user. The user assigns a unique name to each portfolio when it is created. Once a portfolio is created, the user assigns it one or more generating units. A generating unit can be assigned to more than one portfolio.

Data is filtered by generation portfolios on the following displays:

- Unit Default Parameters
- Unit Multiple Ramp Rates Default
- Unit Ramp Rate Hourly Updates
- Schedule Manager
- Schedule Detail Defaults

- Schedule Offers Default
- Schedule Offers Hourly Updates
- Schedule Times Hourly Updates
- Schedule Selection
- Unit Hourly Updates
- Fuel Price Adjustments
- Generation by Portfolio

Note: A unit that is not associated with a portfolio cannot be selected in these displays' tooltray filters.

See §3.1 Manage Portfolios for more information on the purpose of portfolios and details on using this display.

4.3 Unit Default Parameters Display

The Unit Default Parameters display allows you to view unit ownership information and to update a unit's default ramp rate, dispatch status and when applicable, its off-line reserve Claim Capability values. Generator offer data is defined on a unit basis, as well as on schedule basis. The Unit Default Parameters display is for managing the unit's more static offer data.

ISO New England requires that the Participant submit to the ISO, unit specific information about a generating unit. This information is used to create the initial generating unit record in the market system database and must be submitted in writing to ISO-NE and verified by ISO-NE before the Participant submits offers. (For more information on the data required, see **ISO New England Manual for Market Operations Manual 11**, which is found on the <u>ISO New England</u> <u>website</u> among the documents listed on the Participate > Rules and Procedures > Manuals page.

When the unit-specific information is received, the ISO creates the necessary records in the market system database. Some of this information is shown on the Unit Default Parameters display, and users cannot modify the information from here.

IS	new	england										Market S	tatus: OPEN(0	1)	c		.0.1(2014072:		erface
													Participant	t: ISO New	England Inc		¥ (ser: Superuser	*
Public G	eneration D	emand AF	Regulati	ion															
Private Messages	Unit Default Parameters	Portfolio Manager	Ramp Rate Default	Ramp Ra Hourly I			lule Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Updates	Fuel Price Adjustments	Generation By Portfolio	XML Download				
				ribariy		oger perso		oners b didak	riourly opposite	Trouty opened	Percentit	oputer	r rajasana na	0,10,000	Connoda				
Unit D	aily Defau	ult Para	meters																
Select Date	and Unit			«	Unit Details fo	or UNIT NAM	i on 24-4	Aug-2014											
Date	: 24-Aug-2014					6 h 1	Di 0000	16686686					Land Da	the set ID.	8888888888				
Portfolio	All_Units			~		Unit Short Nar									Participant Nam				
Uni	UNIT NAME			~		Unit Long Nar								le Required:		•			
						Type of Ur								n Resource:					
						.,,	OTIK	1700 #1						[103				
					Ramp F	Rate (MW/Mi	0: 5.0						Def	fault Status: (Economic				
						Ramp Rate Ca									O Unavailable				
					Cla	laim 10 Capab	e: Yes								C Must Run				
						Claim 10 (MV	0: 65.0	1											
						Claim 10 Ca	p: 67												
					Cla	laim 30 Capab	e: Yes												
						Claim 30 (MV	0: 67.0	I											
						Claim 30 Ca	p: 67												
																		Update	Refresh
Messages																			۲

The format of the Unit Default Parameters display is shown below:

The fields on this display are:

- Asset ID the ISO-NE generated unique identifier for the asset. This is the preferred reference number used in doing business with the Customer Service Department.
- Unit Short Name the abbreviated name of the generating unit.

- Unit Long Name the full name or description of the generating unit. This label is shown on other generation displays and in the Unit filter of all generation displays
- Type of Unit descriptive name of the type of unit
- Ramp Rate (MW/Min) the number of MW per minute with which the unit can respond to dispatch signals when it is on-line. This value is used in the calculation of spinning reserves unless it is overridden by a default ramp rate curve on the Ramp Rate Default display or an override for an hour defined on the Ramp Rate Hourly Updates display.
- Ramp Rate Cap the unit's demonstrated ramp rate Capability value that was established during the last audit conducted by ISO New England. The offered value will be capped (constrained) to this value if it is defined. If the demonstrated capability value is undefined, it will be shown as "UD".
- **Claim 10 Capable** indicates if the generating unit is registered as capable of producing Ten Minute Non-Spinning Reserve from an offline state.
- **Claim 10 (MW)** the number of MW with which a resource can respond within 10 minutes from an offline state. Used in the calculation of Ten Minute Non Spinning Reserves.
- **Claim 10 Cap** the unit's demonstrated Claim 10 Capability value that was established during the last audit conducted by ISO New England. The offered value will be capped (constrained) to this value if it is defined. If the demonstrated capability value is undefined or the unit is not Claim 10 Capable, it will be shown as "UD".
- **Claim 30 Capable** indicates if the generating unit is registered as capable of producing Thirty Minute Reserve from an offline state
- Claim 30 (MW) the number of MW with which a resource can respond within 30 minutes from an offline state. Used in the calculation of Thirty Minute (non-spinning) Reserves.
- **Claim 30 Cap** the unit's demonstrated Claim 30 Capability value that was established during the last audit conducted by ISO New England. The offered value will be capped (constrained) to this value if it is defined. If the demonstrated capability value is undefined or the unit is not Claim 30 Capable, it will be shown as "UD".
- Lead Participant ID the ISO-NE assigned unique identifier of the Market Participant whose users are responsible for submitting the operating information of the generating unit using the eMarket application.
- Lead Participant Name the full name of the Market Participant whose users are responsible for submitting the operating information of the generating unit using the eMarket application.
- **DA Schedule Required** indicates if the generating unit is classified as a capacity resource and therefore required to submit generation offers in the DA Market. A capacity resource is a generating unit for which the Participant has acquired a Capacity Supply Obligation (CSO) in the Forward Capacity Market (FCM) and is not an intermittent resource.
- **Regulation Resource** indicates if the generating unit is registered as a regulation resource

• Default Status (Economic/Unavailable/Must Run) – the default dispatch availability status of the unit.

4.3.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- $0 \leq \text{Ramp Rate}$
- A warning will be issued if the Ramp Rate exceeds Ramp Rate Cap when the demonstrated capability is defined (not "UD")
- A warning will be returned if the unit is Claim 10 capable and any of the following conditions exist
 - Claim 10 = 0 or NULL
 - Claim 10 > Claim 10 Cap when the demonstrated capability is defined (not "UD") – meaning that the Claim 10 Cap value will be used instead of the offered value
- A warning will be returned if the unit is Claim 30 capable and any of the following conditions exist
 - Claim 30 = 0 or NULL
 - Claim 30 > Claim 30 Cap when the demonstrated capability is defined (not "UD") – meaning that the Claim 30 Cap value will be used instead of the offered value
 - Claim 30 < Claim 10 meaning that the Claim 10 value will be used instead of the offered Claim 30 value.

4.3.2 View Specific Unit Information

- 1. In the Generation Navigation sub tabs, click the **Unit Default Parameters** tab.
 - The Unit Default Parameters display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired unit from the **Unit** filter.
 - The default unit data for the selected date and selected unit is shown in the Unit Default Parameters display.

4.3.3 Modify Unit Default Claim 10/30 and Ramp Rate

A unit's Claim 10/30 values and its default ramp rate data are specified at the unit level. To define this data:

- 1. View the desired unit's information as described in §4.3.2
- 2. Enter all of the following unit-specific information:

- a. Claim 10 (MW) if Claim 10 Capable
- b. Claim 30 (MW) if Claim 30 Capable
- c. Ramp Rate (MW/Min)
- 3. Click the **Update** button to save the data.

4.3.4 Modify Unit Default Status

- 1. View the desired unit's information as described in §4.3.2
- 2. Select one of the following radio buttons:
 - a. Economic
 - b. Unavailable
 - c. Must Run
- 3. Click the **Update** button to save the selection.

4.4 Unit Multiple Ramp Rates Daily Default Display

This display provides for viewing and modifying a unit's default ramp rate curve. A ramp rate defines how quickly a unit can change its generation in MW per min. Depending on the unit's physical characteristics this rate can vary as a function of the unit's MW output. When that is the case, the Participant can use this display to model the unit's ramp rate curve. Ramp rate curves consist of Energy Ramp Rate (MW/Minute) over a range of the unit's MW output. Up to 10 MW ranges can be defined. The default ramp rate curve shown on the display is effective for the date selected and for all future days until it is redefined or removed at a future date. Note that the effective date of the data may be a date prior to the date selected. See §1.5 Effective Dates of Data.

Note: The existence of a ramp rate curve on this display <u>overrides</u> the simple slope specified in the Ramp Rate field on the Unit Default Parameters display.

The format of this display is shown below:

Proble Displayed Displ	IS	new engla	nd								Market (Status: OPEN(C Participan	·		eMarket User Interface Version: 2.0.1(20140722113) Certificate User: Superuser 22:/kug-2014 19:16:07 Versi Superuser Versi Superuser
Personal Parameters Manager Default Manager Offen Default Hourly Updates Selecto Updates Adjustments By Porticilio Douvlad					. 10				(Catata	 C	((.		
Select Date and Unit Cd Date: 24-Aug-2014 Petrolo: All_bras Urit: Unit: Urit: Unit:				Hourly	ate Updates	Manager		Offers Default	Hourly Updates		Updates	Adjustments		Download	
Date: 24:Aug-2014 Potrfol: All_Links Unit: Unit: Unit: Unit:	Unit Mu	ultiple Ramp F	Rates Daily	y Defa	ult										
Det: 24 Jup 2014 0 Portfolio: Al Links 1	Select Date a	and Unit		«	Unit Rar	mp Rates for l	INIT NAME on 2	4-Aug-2014							
Portfole: Al_Links v Unit: ULTI NAME v Unit: ULTI NAME v ULTI NAME v ULTI NAME v ULTI NAME v V	Date:	24-Aug-2014													
	Portfolio:	All_Units													
	Unit:	UNIT NAME		~											
Messages														New	Delete Update Refreah

The fields on this display are:

- **MW** the unit's MW output level up to which the ramp rate applies.
- Energy Ramp Rate the MW/Minute that the unit can change generation while its MW output is less than or equal to the MW value specified and greater than a lesser MW value specified on a separate row or zero if none. The ramp rate specified for the largest MW value will be used if the unit's MW output is greater than this largest MW point defined for the curve.

4.4.1 Data Validation Rules

- Energy Ramp Rate \geq 0; MW > 0
- eMarket will give you a warning if any of the entered rates is greater than the Ramp Rate Cap shown on the Unit Default Parameters display.
- MW/Ramp Rate pairs can be entered in any order. eMarket will sort them and show them in increasing MW value order after the data is submitted and stored in the database.

4.4.2 View Ramp Rates

To view the ramp rate curve segments of a unit:

- 1. In the Generation Navigation sub tabs, click the **Ramp Rate Default** tab.
 - The Unit Multiple Ramp Rates Daily Default display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired unit from the **Unit** filter.

 The display shows the selected unit's default ramp rate curve data for the selected date; if a ramp rate curve is not defined for the selected date, nothing is shown on the display.

4.4.3 Create Ramp Rates

To create segments of a ramp rate curve for a unit:

- 1. View the unit's ramp rate curve segments (§4.4.2).
- 2. Click the **New** button; you can click more than once if you want to create multiple segments.
 - A new row for each click is added in the grid with a pink background to indicate it has not been saved.
- 3. Type in the desired **MW** and **Energy Ramp Rate** values in the appropriate cells.
- 4. Repeat steps 2-3 as desired.
- 5. Click the **Update** button to save the data.

4.4.4 Modify Ramp Rates

To update ramp rate curve segments:

- 1. View the unit's ramp rate curve segments (§4.4.2).
- 2. (optional) To add new segments to the curve, do as described in §4.4.3.
- 3. (optional) To delete existing segments in the curve:
 - a. Click in the row to be deleted. The row will be highlighted.
 - b. Click the **Delete** button. The row is removed.
- 4. To change the parameters of a curve segment, enter the desired **MW** and **Energy Ramp Rate** values in the appropriate cells.
- 5. Repeat steps 2-4 as needed.
- 6. Click the **Update** button to save the data.

Note: to use the Ramp Rate defined on the Unit Default Parameters display, starting on the date selected, all Ramp Rate segments must be deleted from this display.

4.5 Unit Ramp Rate Hourly Updates Display

This display provides for viewing, entering and modifying hourly updates to a unit's default ramp rate characteristics for a given market day. A ramp rate defines how quickly a unit can change its generation in MW per min. Depending on the unit's physical characteristics this rate can vary as a function of the unit's MW output, in other words the unit's ramp rate characteristic is a curve defined as ramp rate for up to 10 ranges of the unit's MW output. With this display, you can define a unit's ramp rate characteristics on an hourly basis for a given market

day to be either a single ramp rate or a ramp rate curve and thereby account for the unit's physical configuration throughout the day.

Note: The existence of a ramp rate definition for an hour on this display overrides both the simple slope specified in the Ramp Rate field on the Unit Default Parameters display and the default ramp rate curve for the day, if one is defined, specified on the Ramp Rate Default display.

The format of this display is shown below. There are two illustrations. The only difference is in the default values shown in the upper section when there is a default ramp rate curve defined for the unit on the selected market day date and when there is none.



Note: Generation Control A/D Regulator.	te Schedule S pdates Manaper C	Schedule Detail Default		thedule Tiers Defaul			Schedule Tir Hourly Upd					Seneration By Portfolio	XML Download	eMarket User Interface Version: 3.0.1204/2012 Senfect kinas: ORM(3) Senfect kinas: ORM(3
Unit Ramp Rate Hourly Updates	Default Values for UN	IT NAME of	. 24-4-	-2014										
Date: 24-Aug-2014		Daily RR 1.D												
Unit: UNIT NAME	Override Hearly Value	es for UNIT	NAME	on 24-Au	g-2014									
	Hour	RR		MV	RR #	MN	RR	M	RR	F MN	RR	# MV	RR	
	V V Copy		1		2							5		Notice that nothing is shown for a
	D1		1	65.0	8.0 2	75.0	7.0	0.08	6.0	4 85.1	0.0	5 90.0	4.0	
			6	95.0	3.0 7		2.0	105.0	1.5					default ramp rate curve if one is not
	02		8	65.0 95.0	0.0 2 3.0 7		7.0		6.0				4.0	
	83		1	65.0	0.0 2	120.0	1.0			4		6		defined for the selected market day
	D4		1	65.0	0.0 2	120.0	1.0			4		6		date.
	05		6		7							10		adter
	05		1	65.0 95.0	0.0 2 3.0 7		7.0	80.0	6.0	4 85.1 P 120.1			4.0	
	05		1	65.0	8.0 2		7.0		0.0				4.0	
			6	95.0	3.0 7		2.0		1.5		1.0			
	07		1	65.0	0.D 2		1.0					5 10	<u> </u>	
	08		1	65.0	8.0 2		1.0			4		6	ii	
			6		7					•		10		
	09		1	65.0	8.0 2		7.0		0.0				4.0	
	10		6	95.0	3.0 7 8.0 2		2.0	105.0 9 90.0	1.5				4.0	
			0	95.0	3.0 7		2.0		1.5				410	
	11		1	65.0	8.0 2		1.0					6		
			6		8.0 2		1.0			4		5		
	12			85.0	8.0 Z	120.0	1.0					5		
	13		1	65.0	8.0 2	75.0	7.0		0.0	4 85.1	5.0	5 90.0	4.0	

The Default Values section is provided as a reference to show what are the default daily values for the chosen day that are being overridden by the hourly update values entered in the section below. The fields in this section of the display are:

- **Daily RR** the unit's default energy ramp rate defined on the Units Default Parameters display for the selected market day date.
- #/MW/RR the unit's default ramp rate curve segments or MW blocks; # is the segment number, MW is the unit's MW output upper bound for the block and RR is the ramp rate value for the range of MW output from the MW value of the previous segment (0 if there is no previous segment) up to the MW value of the segment. The RR of the last segment is the value used for unit MW output values greater than the MW value of the last segment.

The Default Values section can be collapsed using the double up-arrow icon at the extreme right of the section title bar to show more rows in the hourly section as shown below. The collapsed section can be expanded again to see the default values using the double down-arrow icon at the extreme right of the collapsed section title bar.

ISO new england											eMarket User Inte Verion: 20.1031(0727113) Market Staturi OFEN(31) Certificate User Supervise Pundigiant So New England Str. User Suprimer
c Generation Demand ARD Regulation											And the second s
e Unit Default Portfolio Ramp Rate Ram	np Rate Schedule Sched			s Schedule Ti						XM.	
es Parameters Manager Default Hou	arly Updates Manager Detail	Defaults Offers Defaul	t Hourly Update	s Hourly Upd	stes Selecti	on Updah	s Adjus	tments E	By Portfolio	Download	
t Ramp Rate Hourly Updates											¥ 📐
Date and Unit	C Default Values for UNIT N	ME on 24-Aug-2014									
	Override Hourly Values for	UNIT NAME on 24-Au	-2014								
		RR # MW	RR # MI	RR	MI	RR #	MIX	RR	# MV	RR	
	Y Y Copy	1	2			4			6		
UNIC: UNET NAME	✓	0	7			9			10		
	01	1 65.0	8.0 2 3	0.0 7.0	0.08	6.0 4	85.0	6.0	5 90.0	4.0	
	1	8 95.0	3.0 7 1	0.0 2.0	105.0	1.5 9	120.0	1.0	10		
	02	1 05.0		0.0 7.0		6.0 4	85.0	6.0		4.0	
/	03	6 95.0		0.0 2.0		1.5 9	120.0	1.0	5		
	00	1 65.0	8.0 2 1	0.0 1.0		4			10		
	04	1 65.0		¥.0 1.0		4			6		
01		8	7			9			10		
02	05	1 05.0	8.0 2 :	0.0 7.0	0.08	6.0 4	85.0	6.0	5 50.0	4.0	
03		6 95.0		0.0 2.0	100.0	1.5 9	120.0	1.0			
	00	1 65.0		0.0 7.0		6.0 4	85.0	6.0		4.0	
04	07	5 95.0 1 65.0		0.0 2.0		1.5 9	120.0	1.0	10		
05		8	7			9			10		
06	00	1 05.0	8.0 2 1	0.0 1.0		4			6		
07		8	7			9			10		
08	09	1 05.0		8.0 7.0		6.0 4	85.0	6.0		4.0	
09	10	6 95.0 1 65.0		0.0 2.0 5.0 7.0		1.5 9	120.0	1.0			
	10	1 65.0 5 95.0		5.0 7.0 0.0 2.0		6.0 4 1.5 9	85.0 120.0	6.0 1.0		4.0	
10	11	1 65.0		0.0 1.0		1.0 4	120.0		5		
11		8	7						10		
12	12	1 65.0		0.0 1.0		4			5		
10		6	7			9			10		
	13	1 65.0		8.0 7.0		6.0 4		6.0		4.0	
	14	6 95.0 1 65.0		0.0 2.0 5.0 7.0		1.5 0 6.0 4	120.0	1.0		4.0	
		6 95.0		0.0 2.0		1.5 9	120.0	5.0		4.0	
	15	1 05.0		0.0 1.0		4	_		5		
					, — · ·				**		

The Override Hourly Values section shows each hour of the day and fields to enter overrides. When the fields are blank, it means that there is no override of the default values for that hour. Hence setting all data entry fields for an hour to blank (NULL) means that the applicable daily default value is to be used for the market hour; i.e. it is equivalent to removing the override for the market hour.

The description of the fields in the Override Hourly Values section of the display are the same as those in the Default Values section, except they define the unit's ramp rate characteristic for the hour-ending shown in the Hour column. *Either a single ramp rate override or a ramp rate curve override can be defined for an hour, not both.*

The top row of the grid provides you with the means to enter or clear overrides for a selected range of hours. You select the "from" and "to" hours using the drop-down list boxes under the Hour column, enter a single ramp rate or a ramp rate curve using the data entry fields on the right and then select the "Copy" button. As illustrated below, the values entered in the top row are copied to the rows for each hour in the range selected and the cells where values are being changed are highlighted. This process can be repeated to enter different ramp rate updates for other hour ranges before selecting the "Update" button to save the values in the database. If you do use the top row, you need to **remember that leaving all data entry field on this row blank means that you want to remove the overrides for all hours in the range selected**. For example, if you want to remove overrides for every hour of the day, simply select "01" as the "From" hour and "24" as the "To" hour in the Hour drop-down selections, select Copy to blank out all data entry fields below and then select Update. Since you did not enter anything in the data entry fields, all overrides are removed.

ISO new england																	eMarket User Interfac Version: 2.0.1(201467221119)
																	Market Status: OPEN(01) Centrificate Useri Superiuser 22-Aug-2014 19 Participanti ISO New England Inc Vuseri Superiuser
Public Generation Demand ARD R	Regulation																
Private Unit Default Portfolio Ramp R Messages Parameters Manager Default			Schedule Detail Default	s Sched Offers			le Offers Updates	Schedule Hourly U					Fuel Price Adjustments		neration Portfolio	XML Download	
Unit Ramp Rate Hourly Up	dates																
Select Date and Unit	66	Default Values for UN	ET NAME or	24-Aug-2	014												
Date: 24-Aug-2014			Daily RR	# 100	e 1	RR ¥	MV	RR		MV	RR	# N	eni Re		MIX	RR	
Portfolio: Al Livits			1.0	8	15.0	8.0 2	116.9	1.0	8			9		5	0		
Unit: UNIT NAME	~																
		Override Hourly Value	es for UNIT	NAME on 2	4-Aug-	2014		D.D.									
		Hour		1 1		(K #	MA	NR	2	MAU	кк	8 M	W RR	6	540	RR	
		W - 14 - 0000		8		7			8					10			
		D1		1 0	5.0	10 2	76.0	7.0	3	0.08	6.0	4	85.0	5.0 5	90.0	4.0	
		02			5.0	3.0 7		2.0	•	105.0	1.4		20.0	1,0 10			
		62			5.0 5.0	8.0 2 3.0 7	75.0	7.0		80.0 105.0	6.0 1.5		120.0	1.0 10	90.0	4.0	-
		63			5.0	8.0 2	120.0	1.0	3			4		6			
		D4		6	5.0	8.0 Z	120.0	1.0	8	_		9		10			11
				6		7	140.0	- 1.0	8			•		10			Hour RR
		05			5.0	8.0 2			0	0.08	6.0			5.0 5		4.0	
		08			5.0	3.0 7 8.0 2	100.0	2.0	8	105.0 80.0	1.5			1.0 10 5.0 6		4.0	07 💙 12 💙 Copy 1.5
					5.0	3.0 7	100.0		8	105.0	1.5			1.0 10			07 12 0007 1.0
		67		1 0	5.0	8.0 2		1.0	3	_		4	_	5			
		08			5.0	8.0 2		1.0	3			4		6			
		09		8		7			8			•		10			Before selecting "Copy", example
		Da			5.0 5.0	8.0 2 3.0 7	75.0	7.0		80.0 105.0	6.0 1.5			5.0 5 1.0 10		4.0	
		10			5.0	8.0 2		7.0	3	0.08	0.0		85.0	5.0 5	90.0	4.0	where you want to set the unit's
					5.0	3.0 7 8.0 2			8	105.0	1.5	9 1	20.0	1.0 10			
				6		7	120.0	- 19	8			-		10			ramp rate characteristics to a
		12		1 8	5.0	8.0 2	120.0	1.0	3			4		6			single ramp rate of 1 5 for hours
		13		5	5.0	8.0 2	75.0	7.0	8	80.0		9	85.0	10 5.0 5		4.0	single ramp rate of 1.5 for hours
		L			* 0	0.0 T	100.0			105.0							onding 07 through 12
																	Re

Private Unit Default Po	nd ARD Regulati	on Ramp Rate Hourly Upd		e Scheduli Debal De		Schedule Offers Defer		dule Offers ly Updales		Scheduly Selection				Generation By Portfolio	XML Download	a		eMarket Use Verilion: 20.120144797333 Market Status: OFEN(3) Participant: 30.18ee/EndedSic StatesEndedSic	9) 22-Aug-2014 19:31
Unit Ramp Rate H	Hourly Update	s																	
Select Date and Unit Date: 24-Aug-2014	iouny opuare		Default Values I	Daily Rf		- Aug 2014 MN 05.0	RR 8.0	# MNV 2 115	RR 1.	 ww.	RR #	hter	RR	# 140V	RR				
Portfolio: Al_Units Unit: UNIT NAME		*	Override Hearly					7			9			10					
			01 02 03 04 05 06 05 06 07 09 09		R J 115 115 115 11 115 11 115 115	0000 950 950 950 950 950 950 950 950 950		7 100.7 2 76.7 7 100.0 2 120.7 7 2 2 120.7 7 2 2 76.7 7 100.7 2 76.7 7 100.7 2 76.7 7 100.7 2 76.7 7 100.7 2 76.7 7 100.7 2 76.7 7 100.7 7 100.7	D 2.0 D 7.0 D 2.0 D 1.0 D 1.0 D 1.0 D 7.0 D 2.0	667 00.0 105.0 00.0	RR # 6.0 4 1.6 9 6.0 4 1.6 9 6.0 4 1.5 9 6.0 4 1.6 9 6.0 4 1.6 9 6.0 4 1.6 9 6.0 4 1.6 9 6.0 4 6.0 4 6.0 9 6.0 9 6.0 9 6.0 9 6.0 9 6.0 9 6.0 9	85.0 120.0 130.0	6.0 1.0 5.0	10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 40			After selecting "Copy", the value entered on the top row are copi the rows for each of the selecte hours. The fields where values a changed are highlighted to show where data entries will be made the "Update" button is selected	ied to d are v e when
			11 12 13		1.5 1 1.5 1 6 1		8.0			80.0	4 9 6.0 4	85.0	6.0		0 4.0		L		Update Refrest

P																	Market Status: OPEN(01) Centificate User: Superviser 22-Aup-201: Partisipanti ISO New England Inc V User: Superviser
Generation Demand ARD Re e Unit Default Portfolio Ramp Ra																124	
s Parameters Manager Default	Hourly Up	e Schedule dates Manager	Schedule Detail Defau		ichedule Xffers Defau	t Hour	We Offers y Updates	Hourly	le Times Updøtes	Schedul Selection	e Hour Upda	hes A	Jel Price djustments	By P	eration Portfolio		
Ramp Rate Hourly Upd																	
te and Unit	44	Default Values for U	Daily RR	m 24-A	ag-2014	80		PP			80		00		MV	RR	
Re: 24-Aug-2014			1.6	1	65.0	8.0	2 11	1.0 1.	9 3		-	4		5		hin	
ilo: Al_UNIS	¥			8			7		8			9		10			
NR: UNET NAME	*	Override Hearly Val															
		Hour		1	MV	BB	MA	RR		M	RR J	- MN	RR		MV	RR	
		04 ¥ 06 ¥ Cop		1	60.0	7.0			0	0.00	4.0			.0 5		1.0	
				8	_	_		-	8	_	-	-	_	10	_	_	
		D1) • [65.0	0.8	: 76	.0 7.1	0 0	0.08	6.0		5.0 0	1.0 6	10.0	4.0	
				6	95.0	3.0			8	105.0	1.5			1.0 10			Before selecting "Copy", example
		02		1 8	65.0 95.0	0.0 3.0			0 0	00.0 105.0	0.0 4 1.5 1			1.0 5	90.0	4.0	
		ca		11	65.0	0.0			0	100.0				6			where you want to set the unit's
				8					8		\$			10			
		De		1	65.0	0.0		.0 1.1	8					6			ramp rate characteristics to a
		05		1	65.0	0.0		.0 7.1	0 0	0.08	6.0		5.0 6	1.0 5	90.0	4.0	different ramp rate curve for
				6	95.0	3.0			8	105.0	1.5			1.0 10			unierent ramp rate curve for
		08		1	65.0 95.0	0.8				80.0 105.0	6.0 4 1.5 S			1.0 5 1.0 10	90.0	4.0	
		87	1.4		40.0	3.0		5 21	3	105.0	1.0		1.0 1	5		_	
				6					8					10			
		DB	1.4	1			2		3					6 10			
		09	14	0					3		_			5			
				6					0					10			
		10	1.4	1					3					5			
		11		6					8					10		_	
			1.4	6					8	\rightarrow				10		_	
		12	1.4	1			2		3		•			5			
		13		6	65.0	0.8	75	0 71	0	80.0	0.0		5.0 4	10	90.0	4.0	
		13			00.0	0.0				105.0					40.0	41	
																	Update



4.5.1 Data Validation Rules

- Ramp Rate \geq 0; MW > 0
- eMarket will give you a warning if any of the entered rates is greater than the Ramp Rate Cap shown on the Unit Default Parameters display.
- MW/Ramp Rate pairs can be entered in any order. eMarket will sort them and show them in increasing MW value order after the data is submitted and stored in the database.
- Either a single ramp rate or a ramp rate curve can be defined for an hour, not both.

4.5.2 Override Unit's Default Ramp Rate for a Day and Hour

As discussed in §4.3 Unit Default Parameters Display, you define on this display a unit's daily default ramp rate. Similarly, as discussed in §4.4 Unit Multiple Ramp Rates Daily Default Display, you can define on this display a daily default ramp rate curve for the unit that supersedes the daily default. Both the daily default single ramp rate and the daily default ramp rate curve for a selected market day are shown in the Default Values section of the display. If you want to profile the unit's ramp rate characteristics on an hourly basis,:

- 1. In the Generation Navigation sub tabs, click the **Ramp Rate Hourly Updates** tab.
 - The Unit Ramp Rate Hourly Updates display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired unit from the **Unit** filter.
 - The display reflects the hourly overrides for the selected unit.
- 3. To update the overrides for individual hours:
 - a. Enter the override value for a single ramp rate or for a ramp curve in the hours that you wish. Remember that if you are switching from a ramp rate curve override to a single ramp rate value, you need to blank each segment of the ramp rate curve in addition to entering the single ramp rate value.
 - b. Click the **Update** button to save the data.
- 4. To update the overrides for a range of hours
 - a. Specify the range of hours by selecting "From" and "To" hour-ending values in the **Hour** drop-down list on the top row.
 - b. Enter the override value for either a single ramp rate or a ramp rate curve in the top row.
 - c. Select the **Copy** button to copy the entered values to the rows for each hour in the selected range.
 - d. Repeat steps a through c to enter different values in other range of hours if necessary.
 - e. Click the **Update** button to save the data for the selected hours.

4.6 Schedule Manager Display

Schedule Manager enables users to view, create, and update schedules. The format of this display is shown below:

	new england													Version: 2.0.1(20 Certificate User: Super		-Aug-2014 19:4
												Participant:	ISO New England Inc		💙 User: Super	Jser
blic Generation	Demand ARD Regulat	on														
ate Unit Defau sages Parameter		Ramp Rate Hourly Updat	s Schedule Manager	Schedule S Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Updates	Fuel Price Adjustments	Generation By Portfolio	XML Download				
chedule Ma	anager															
ect Unit		« \$	hedule Manager	for UNIT NAME												
Portfolio: All Unit:		v 8	chedule Name		Description				Schedule Ty	pe						
		•	FLPRICE		DFLPRICE				1 (Price)	~						
Unit: UNIT NA	IAME		NIT_R3		-	8 Ref. Schedule			98 (Cost)	~						
		L	NIT_R2		UNIT_Fuel 2	Ref. Schedule			97 (Cost)	*						
			NIT_R1		-	Ref. Schedule			98 (Cost)	~						
			NIT_M		UNIT_Mitiga	tion Schedule			(Cost) 99	~						

The fields on this display are:

- **Schedule Name** a short name for the schedule, defined by the Participant. This label appears in the Schedule filter on the Generation displays.
- **Description** a brief description of the schedule, defined by the Participant. This label appears in the Generation by Portfolio display, which shows the DAM results for units by generation portfolio.
- Schedule Type identifies the number and type of the schedule, as follows:
 - **0 94** price-based schedules
 - **95 99** cost-based schedules, reserved for use by ISO New England.

By convention, schedules 95 through 98 are **Reference schedules**, none of which will be flagged as available. They show the reference values determined by the ISO-NE Market Monitor for each of the fuels that the unit is configured to use. Schedule 99 is the **Mitigation schedule**, and it is the only one that will be flagged as available. It shows the generator parameters that will be used by the Market clearing software to dispatch the generator and to calculate LMPs at the generator node **should** *the generator be mitigated*.

- Schedule 95, with the name XXXX_R4 where XXXX is the generator short name, is the reference schedule for the generator's fourth fuel;
- Schedule 96, with the name XXXX_R3 where XXXX is the generator short name, is the reference schedule for the generator's third fuel;
- Schedule 97, with the name XXXX_R2 where XXXX is the generator short name, is the reference schedule for the generator's second fuel.
- Schedule 98, with the name XXXX_R1 where XXXX is the generator short name, is the reference schedule for the generator's first fuel; for single-fuel units, this will be the only reference schedule.
- Schedule 99, with the name XXXX_M where XXXX is the generator short name, is the mitigation schedule for the generator.

4.6.1 Create a New Schedule

Users can define more than one price-based schedule for a generating unit. To assist in creating multiple schedules for unit, the Schedule Manager display allows you to assign a schedule name and a description to a specific unit schedule. To create a new schedule:

- 1. In the Generation Navigation sub tabs, click the **Schedule Manager** tab.
 - The Schedule Manager display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired portfolio from the **Portfolio** filter.
 - b. Select the desired unit from the **Unit** filter.
 - Existing schedules for the selected unit are listed.
- 3. Click the **New** button.
- 4. Type the following information in the new row:
 - a. Schedule Name
 - b. Description
- 5. Select an unused type of schedule from the **Schedule Type** drop-down menu.
- 6. Click the **Update** button to save the data.

4.6.2 Modify a Schedule's Name and/or Description

After a schedule is created and saved to the database, you cannot modify the Schedule Type field. Only the schedule's name and description can be changed.

To modify an existing schedule's parameters:

- 1. In the Generation Navigation sub tabs, click the **Schedule Manager** tab.
 - The Schedule Manager display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired portfolio from the **Portfolio** filter.
 - b. Select the desired unit from the **Unit** filter.
 - The existing schedules for the selected unit are listed.
- 3. Modify either or both of the following in any of the listed schedules:
 - a. Schedule Name
 - b. Description
- 4. Click the **Update** button to save the data.

4.6.3 Delete a Schedule

- 1. In the Generation Navigation sub tabs, click the **Schedule Manager** tab.
 - The Schedule Manager display is shown in the display pane.

- 2. In the **Filters pane**:
 - a. Select the desired portfolio from the **Portfolio** filter.
 - b. Select the desired unit from the **Unit** filter.
 - The existing schedules for the selected unit are listed.
- 3. Select a cell in the row of the schedule that you want to delete.
- 4. Click the **Delete** button. If the schedule is not flagged available in any of the market days for the period that market information is preserved in the market database, it is deleted; otherwise you will get an error message.

4.7 Schedule Detail Defaults Display

The Schedule Detail Defaults display is used to view and manage a generating unit's default market offer parameters. The format of the display is shown below:

Constraint Constraint </th <th>Marce Marce District <thdistrict< th=""> Distrin Distrin</thdistrict<></th> <th></th> <th>eMark Version: 2.0.1(2</th> <th>014072211</th> <th>19)</th> <th></th>	Marce Marce District District <thdistrict< th=""> Distrin Distrin</thdistrict<>															eMark Version: 2.0.1(2	014072211	19)	
Beseration Reseration Reseration <th>E Generality Rundy Rundy Rundy Rundy Strady Rundy S</th> <th>150</th> <th>new englar</th> <th>Id</th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th>Market Sta</th> <th></th> <th></th> <th></th> <th></th> <th>2014 19:5</th>	E Generality Rundy Rundy Rundy Rundy Strady Rundy S	150	new englar	Id										Market Sta					2014 19:5
b Description Description <thdescription< th=""> <thdescription< th=""> <thdes< th=""><th>b b</th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th>Participant:</th><th>ISO New England Inc</th><th>🎽 User</th><th>Superuser</th><th></th></thdes<></thdescription<></thdescription<>	b b														Participant:	ISO New England Inc	🎽 User	Superuser	
gen Parander Manger Opfail Manger Opfail Manger Opfail Manger Opfail Manger Manger Manger Opfail Manger Manger Opfail Manger Mang	get Parender Manger Opfall Hondry (bolds Name Opfall Hondry (bulds Name Opfall Opfall Hondry (bulds Name Opfall Opfall Opfall Name Name Opfall Name Nam Name Name Name Name	Generatio	on Demand	ARD Regula	tion														
heduel Detail Defaults	Adduzionali Defaulta Image: Staduali Image: Stad																		
Data Schedule Avaluation Max: Data Max: Data Max: Data Schedule Avaluation	Date: 44.42-2014 Image: Condense Available: Yes Associal Status, Time (hr); 10 Ensemption (hr);				riourly ope	narage		One's Delada	Thousy opuses	nourly optimes	SHECKON	Opdates		. Dy Fordolo	Cominad				
Date: 24.4ug-2014 Image: Schedule Available: Yes Max. Dafy Starts: Emergency Min: 55.0 Economic Min: 55.0 Image: Schedule Available: Yes Min. Downthine (hr): 1.0 Economic Min: 55.0 Image: Schedule Available: Yes Min. Downthine (hr): 1.0 Economic Min: 15.0 Image: Schedule Available: Yes Min. Downthine (hr): 1.0 Economic Min: 15.0 Image: Schedule Available: Yes Image: Schedule Available: Yes Min. Downthine (hr): 1.0 Economic Min: 15.0 Image: Schedule Available: Yes Image: Schedule Available: Yes Image: Schedule Available: Yes Yes Min. Downthine (hr): 1.0 Economic Min: 15.0 Image: Schedule Available: Yes Image: Schedule Available:	Date: 24.4ug-2014 Image: Schedule Available: Yes Max. Daty Scats: Energon(YMh): Economic Mos 50.0 Unit: Unit: Unit: Image: Schedule Available: Image: Schedule Available: Image: Schedule Available: Image: Schedule Available: Economic Mos 50.0 Unit: Unit: Image: Schedule Available: Image: Sch			ults															
Milleris Mill	MLDels w </td <td>t Date, Unit an</td> <td>nd Schedule</td> <td></td> <td>*</td> <td>Schedule Deta</td> <td>ils for UNIT NAME - DI</td> <td>FLPRICE (Price-b</td> <td>ased) on 24-Aug-2</td> <td>014</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	t Date, Unit an	nd Schedule		*	Schedule Deta	ils for UNIT NAME - DI	FLPRICE (Price-b	ased) on 24-Aug-2	014									
orfidio: ALLeAS V URL WAYE V House V House V Fast Start (10 mir): No Fast Start (10 mir): Yes V S Start (10 mir): Yes V S Start (10 mir): Ves V S Start (10 mir): V	ordiols: ALLeRS V V Unit: VITAME V headle: VITAME V FRETTION VITAME V Read Time Market:	Date: 24-A	ug-2014				Schedule Available: y					Max Daily	Starts:			Emergency Min	45.0		
hedde reinondrate (r); 10 10 100 100 Fast Start (10 min); 16 100 RT High Operating Line 7: 120 Fast Start (10 min); 16 10 100 Condenes Available: Fast Start (10 min); 16 100 Condenes Available: 00 Hot Natricators There (r); 10 20 Condenes Available: 00 Hot Natricators There (r); 10 02 Condenes Available: 00 Cold Matricators There (r); 10 20 Condenes Available: 00 Hot Natricators There (r); 10 20 Condenes Available: 00 Cold Matricators There (r); 10 20 Condenes Natricators There (r); 10 00 Hot Starting There (r); 10 20 Condenes Natricators There (r); 10 00 Line Starting There (r); 10 20 Condenes Natricators There (r); 10 00	hedde Intel line line line line line line line li	ortfolio: All_U	Inits		~						Max	Daily Energy	(MWh):			Economic Min	55.0		
Impact (Impact	Impact de la contract de la contra	Unit: UNIT	NAME		~														_
Past Start (30 min): yes 12.0 Condense Available: Image: Condense Available: Hot Natification Time (hr): 0.2 Condense Barbur, Cost (3): 0.00 Linter Hatrication Time (hr): 0.2 Condense Havailable: Condense Havailable: Cold Markinsk Time (hr): 0.2 Condense Havailable: Condense Havailable: Linter Hatrication Time (hr): 0.2 Condense Havailable: Condense Havailable: Linter Hatrication Time (hr): 0.2 Condense Havailable: Condense Hatrication Time (hr): 0.0 Linter Startup Time (hr): 0.29 Condense Hatrication Time (hr): 0.0 Cold Startup Time (hr): 0.29 Condense Hatrication Time (hr): 0.0	Fast Start (30 mir); Yes Hot to birter Time (hr); 12.0 Hot to birter Time (hr); 12.0 Hot to birter Time (hr); 0.2 Condense Startup Cost (J); 0.00 Linter Hothlication Time (hr); 0.2 Condense Hourd Cost (J); 0.00 Hot XBartup Time (hr); 0.2 Condense Hourd Cost (J); 0.00 Inter Startup Time (hr); 0.2 Condense Hourd Cost (J); 0.0	thedule: DFLP	RICE		~														
Hot Natification Time (hr): 0.2 Condense Startup Cost (s): 0.0 Inter Natification Time (hr): 0.2 Condense Hourly Cost (s): 0.0 Cold Matrification Time (hr): 0.2 Condense Hourly Cost (s): 0.0 Hot Startup Time (hr): 0.2 Condense Matrification Time (hr): 0.0 Inter Startup Time (hr): 0.29 Condense Matrification Time (hr): 0.0 Cold Startup Time (hr): 0.29 Condense Matrification Time (hr): 0.0	Hot Mudikation Time (hr): 0.2 Condernes Starbup Codt (s): 0.00 Inter NetWickton Time (hr): 0.2 Condernes Houthy Codt (s): 0.00 Cdd Mudikation Time (hr): 0.2 Condernes Houthy Codt (s): 0.00 Hot Starbup Time (hr): 0.2 Condernes Houthy Codt (s): 0.00 Inter Starbup Time (hr): 0.29 Condernes Houthy Codt (s): 0.00 Condernes Nathup Time (hr): 0.29 Condernes Houthy Codt (s): 0.0						Fast Start (10 min): N	0			H	lot to Cold Ti	me (hr): 24.	D					
Inter Natification Time (hr): 0.2 Condense Hourh' Cost (b): 0.0 Cold Natification Time (hr): 0.2 Condense Hourh' Cost (b): 0.0 Hot Startup Time (hr): 0.2 Condense Hourh' Cost (b): 0.0 Inter Startup Time (hr): 0.2 Condense Hourh' Cost (b): 0.0 Cold Startup Time (hr): 0.2 Condense Hourh' Cost (b): 0.0	Inter Modification Time (hr): 0.2 Condense Hourly Codd (k): 0.0 Codd Hot/Fickton Time (hr): 0.2 Condense Hourly Codd (k): 0.0 Hot Structop Time (hr): 0.2 Condense Hourly Codd (k): 0.0 Inter Structop Time (hr): 0.2 Condense Hourly Codd (k): 0.0 Condense Hourly Codd (k): 0.0 Condense Hourly Codd (k): 0.0 Inter Structop Time (hr): 0.2 Condense Hourly Codd (k): 0.0						Fast Start (30 min): y	55			н	ot to Inter Ti	me (hr): 12.	0		Condense Available			
Cold Matrikation Time (hr); 0.2 Condense Power: 0.0 Hot: Startup Time (hr); 0.29 Condense Nublikation Time (hr); 0.0 Inter Startup Time (hr); 0.29	Cold Intuitication Time (hr): 0.0 0.0 Hot Startup Time (hr): 0.0 0.0 Inter Startup Time (hr): 0.29 Condense Notification Time (hr): 0.0 Cold Startup Time (hr): 0.29 Condense Notification Time (hr): 0.0										Hot N	lotification Ti	me (hr): 0.2			Condense Startup Cost (\$)	0.00		
Hot Startup Time (hr): 0.29 Condense Hot/Risation Time (hr): 0.0 Inter Startup Time (hr): 0.29 Cold Startup Time (hr): 0.29	Hot Startup Time (hr): 0.29 Conderse Notification Time (hr): 0.0 Inter Startup Time (hr): 0.29 Cold Startup Time (hr): 0.29										Inter N	lotification Ti	me (hr): 0.2			Condense Hourly Cost (\$)	0.00		
Inter Startup Time (Hr): 0.29 Cold Startup Time (Hr): 0.29	Inter Startup Time (hr): 0.29 Cold Startup Time (hr): 0.29										Cold N	lotification Ti	me (hr): 0.2			Condense Power	0.0		
Cold Startup Time (hr): 0.29	Cold Startup Time (hr): 0.29										н	ot Startup Ti	me (hr): 0.2	9		Condense Notification Time (hr)	0.0		
											Int	er Startup Ti	me (hr): 0.2	9					
* Wil million and by displicit and summinum i	• With maj dar and by dignalish and community (Co	old Startup Ti	me (hr): 0.2	9					
																• Will not be used by dispatch and commitment			
																		Update	Refre

The fields on this display are:

- Schedule Available indicates whether the schedule is to be used. This is managed on the Schedule Selection display.
- Day Ahead Market this check box indicates whether the schedule is enabled for Day Ahead Market use.
- **Real Time Market** this check box indicates whether the schedule is enabled for Real Time Market use.
- **Fast Start (10 min)** if the unit is Claim 10 capable³, this indicates whether the entered parameters qualify the unit for 10-min fast startup operation and therefore make it eligible for off-line non-spinning reserves.
- **Fast Start (30 min)** if the unit is Claim 30 capable³, this indicates whether the entered parameters qualify the unit for 30-min fast startup operation and therefore make it eligible for off-line non-spinning reserves.

³ The indication of whether the unit is Claim 10 or Claim 30 Capable can be seen on the Unit Default Parameters display (§4.3)

- **Max Daily Starts** the maximum number of times that the unit can be started in one operating day under **normal** operating conditions. A blank field means the value is unlimited.
- **Max Daily Energy** the maximum amount of energy that the unit can produce in one operating day. A blank field means the value is unlimited.
- **Min Downtime** the minimum number of hours that the generator owner requires between the time the generator is released for shutdown by the ISO and the time the generator is scheduled to be released for dispatch to the ISO.
- **Min Runtime** the minimum number of hours that the unit must remain online after it has been scheduled to reach its Economic Minimum Limit before it can be released for shutdown from its Economic Minimum Limit.
- Hot to Cold Time the time (in hours) for the unit to transition from hot to cold status.
- Hot to Inter Time the time (in hours) for the unit to transition from hot to intermediate status.
- Hot Notification Time the notification time (in hours) needed for the unit to synchronize to the system from a hot temperature state; the time is measured from the time the startup dispatch instruction is received from the ISO.
- Inter Notification Time the notification time (in hours) needed for the unit to synchronize to the system from an intermediate temperature state; the time is measured from the time the startup dispatch instruction is received from the ISO.
- **Cold Notification Time** the notification time (in hours) needed for the unit to synchronize to the system from a cold temperature state; the time is measured from the time the startup dispatch instruction is received from the ISO.
- Hot Startup Time the time (in hours) for the unit to startup from hot temperature state after the notification time elapses.
- Inter Startup Time the time (in hours) for the unit to startup from intermediate temperature state after the notification time elapses.
- **Cold Startup Time** the time (in hours) for the unit to startup from cold temperature state after the notification time elapses.
- Emergency Min the lowest level of energy in MW the unit can produce and maintain a stable level of operation. The Participant will allow operation of the unit at this level during a Minimum Generation Emergency.
- Economic Min the lowest MW level at which the unit can be economically dispatched.

- Economic Max the highest MW level at which the unit can be economically dispatched.
- **RT High Operating Limit** the maximum MW output that the unit can achieve in response to an ISO New England request for energy.
- **Condense Available** indication of whether the unit is available to condense.
- **Condense Startup Cost** the price for starting up the unit in synchronous condensing mode.
- **Condense Hourly Cost** the hourly cost of running the unit in synchronous condensing mode.
- **Condense Power** the unit's MW consumption while running in synchronous condensing mode.
- **Condense Notification Time** the notification time (in hours) needed for the unit to engage in synchronous condensing mode.

4.7.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- Data is entered for each of the following values, i.e. non-blank values appear in the data entry cells:
 - Min Downtime and Min Runtime
 - Hot to Cold and Hot to Inter times
 - All Startup and Notification times
 - All limits (Emergency Min, Economic Min, Economic Max and RT High Operating)

Only Max Daily Starts, Max Daily Energy and the Condense parameter values can be blank (unspecified).

- Emergency Min > 0 if Economic Max > 0
 Economic Min > 0 if Economic Max > 0
 i.e. the Emergency and Economic Minimum limits must be greater than zero if the Economic Maximum limit is greater than zero
- Economic Max ≥ CSO if CSO > 0 Warning

 for a resource with a Capacity Supply Obligation, its Economic Maximum
 Limit *should* be at least as high as its Capacity Supply Obligation. A warning
 will be returned if the rule is violated; it's only a warning because the resource
 may not be fully available.
- Emergency Min ≤ Economic Min ≤ Economic Max ≤ RT High Operating Limit This is the *limit nesting rule*. The data submitted is rejected if this rule is violated.

If the schedule is flagged as available at the time the values are submitted and the unit is a regulating resource, the limit nesting rule is extended to the regulation limits (i.e. Economic Min \leq Regulation Low < Regulation High \leq Economic Max) where the regulation limits are the more constraining of the limits among the daily default (those entered on the Regulation Offers Daily Default display) and the hourly updates (those entered on the Regulation Hourly Updates display) for the selected market day. A warning message is returned when the limit nesting rule with regulation limits is violated.

- Hot To Cold Time \geq Hot To Inter Time \geq 0
- Cold Startup Time \geq Inter Startup Time \geq Hot Startup Time \geq 0
- Cold Notification Time \geq Inter Notification Time \geq Hot Notification Time \geq 0
- Min Downtime ≥ 0
- $24 \ge Min Runtime \ge 0$
- Max Daily Energy must be blank to specify an unlimited capacity or a value greater than zero if you want to define a limit on the unit's energy capability and be treated as a Limited Energy Generation (LEG) resource. If a non-blank value is entered, then Max Daily Energy should be greater than or equal to Economic Min x Min Runtime; a warning message will be reported if it is not.
- When a unit self schedules for a total number of hours greater than its Min Runtime, the Max Daily Energy (if defined) must be greater than or equal to Economic Min x the number of hours that the unit self schedules or a warning will be returned.
- For Claim 10 capable Units, a warning will be returned when the sum of cold notification and cold startup time exceeds the Claim 10 criteria, i.e. ≥ 10min because time values make the unit ineligible for off-line reserve capability.
- For Claim 30 capable Units, a warning will be returned when the sum of cold notification and cold startup time exceeds the Claim 30 criteria, i.e. ≥ 30min because time values make the unit ineligible for off-line reserve capability.
- For Claim 10 capable Units and Claim 30 capable Units, a warning will be returned when Min Runtime > 1 or Min Downtime > 1 because time values greater than 1 hour make the unit ineligible for off-line reserve capability.
- An error message will be returned if any of the following offer parameter values exceed their threshold value established by the ISO-NE Market Monitor⁴:
 - Cold Notification Time, Cold Startup Time
 - Inter Notification Time, Inter Startup Time
 - Hot Notification Time, Hot Startup Time
 - Hot To Cold Time, Hot To Inter Time
 - Min Downtime, Min Runtime
- An error message will be returned if the sum of each of the following offer parameter values is greater than the sum of their respective reference values by more than a threshold value established by the ISO-NE Market Monitor⁵. The reference values used for this validation are those found in the unit's

⁴ Refer to Market Rule 1, Appendix A, Section III.A.6.1.

⁵ Refer to Market Rule 1, Appendix A, Section III.A.6.1. Currently defined as 6 hours

cost-based Reference schedule 98; they are fuel independent, hence the use of schedule 98 because there is such a schedule for every unit.

- Hot Notification Time, Hot Startup Time, Min Downtime, Min Runtime
- Inter Notification Time, Inter Startup Time, Min Downtime, Min Runtime
- Cold Notification Time, Cold Startup Time, Min Downtime, Min Runtime
- A warning message will be returned if any of the following offer parameter values exceed their threshold value established by the ISO-NE Market Monitor⁶: Economic Max, Economic Min, Emergency Min, Max Daily Energy, and Max Daily Starts.

4.7.2 View Schedule Detail Defaults

To view the default values of a generator schedule:

- 1. In the Generation Navigation sub tabs, click the **Schedule Detail Defaults** tab.
 - The Schedule Detail Defaults display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired unit from the **Unit** filter.
 - d. Select the desired schedule from the **Schedule** filter.
 - The display shows the selected schedule's default detail information.

4.7.3 Modify Schedule Constraint Data

Constraint data includes minimum downtime, minimum runtime, maximum daily starts and maximum daily energy. To modify schedule constraints:

- 1. View the unit's schedule details (§4.7.2)
- 2. Enter the desired changes for any or all of the following offer parameters:
 - a. Minimum Downtime (hours)
 - b. Minimum Runtime (hours)
 - c. Maximum Daily Starts
 - d. Maximum Daily Energy (MWh)
- 3. Click the **Update** button to save the data.

4.7.4 Modify Schedule Operating Limits

- 1. View the unit's schedule details (§4.7.2)
- 2. Enter the desired changes for any or all of the following operating limits:

⁶ Refer to Market Rule 1, Appendix A, Section III.A.6.1.

- a. Economic Max (MW)
- b. Economic Min (MW)
- c. Emergency Min (MW)
- d. RT High Operating Limit (MW)
- 3. Click the **Update** button to save the data.

4.7.5 Enable Schedule for a Market

Another schedule parameter is market participation. The Participant selects the schedules that are enabled for the Day-Ahead Market (DAM) and/or the Real-Time market (RTM). The following steps enable a schedule for the specified market(s):

- 1. View the unit's schedule details (§4.7.2)
- 2. To enable/disable a schedule for use in the DAM, click the **Day Ahead Market** checkbox.
- 3. To enable/disable a schedule for use in the RTM, click the **Real Time Market** checkbox.
- 4. Click the **Update** button to save the data.

Note: If the schedule is flagged as available, you will not be able to change the schedule's applicable market flags. That is because one price-based schedule must be flagged as available for each of the markets at all times. To change a schedule's market eligibility, you must first make one or more other schedules available (one for each market).

4.8 Schedule Offers Daily Default Display

The purpose of this display is to view and update the unit's daily default Startup and No Load costs values, its incremental energy offers, its Use Bid Slope flag and if the unit is a multi-fuel unit, the fuel blend associated with each financial offer parameter. Incremental energy offers consist of price/MW pairs, also referred to as schedule offer segments. Schedule offer segments specify a MW amount and the price per MWH for generation from the previous segment's MW amount up to this segment's MW amount. Up to 10 segments can be defined for each schedule.

Offers in MW are cumulative. For example,

- To offer the first block of 15 MW of generation at \$10.00, you enter an offer segment of 15 MW @ \$10.00
- To offer the next block of 20 MW of generation at \$30.00, you enter an offer segment of 35 MW @ \$30.00

And so on. You can enter the segment offers in any order of MW values; eMarket will organize the segments and display them in increasing MW value order. The price value of each segment must however be greater than the price of the previous segment after the segments are sorted in increasing MW value order. In other words, the offer segments must be monotonically increasing in MW and price value.

The slope from zero to the first MW point (which can be zero) is equal to zero; that is to say that the price/MWH is fixed and equal to the price of the first segment. The price along the segment between MW points is a calculated slope or block, depending on the status of the Use Offer Slope checkbox.

The last MW point should be greater than or equal to the Economic Maximum Limit. If it is not, the curve is extended with a slope equal to zero.



This is all illustrated in Figure 2.

Units are eligible to change segment offer Price/MW pairs and Startup and No Load cost offers after the close of the DA market during the daily Re-Offer period and during the RT Intraday Re-Offer period.

The format of the display is different for multi-fuel units and single-fuel units as shown below; for multi-fuel units, the <u>fuel index</u> names of each fuel that the unit is configured to use are shown along with data entry cells to specify the fuel blend percentages associated with each of the financial offer parameters. **See Appendix A for list of Fuel Index Names registered with Market Monitoring.**

ISC	new	england													tus: OPEN(01 Participant:	ISO New England In	Version Certificate U	: 2.0.1(201407 ser: Superuser	221119)	Interfac 3-Aug-2014 09 ruser	
_			RD Regula			X	8			Ŷ											
		Portfolio Manager	Ramp Rate Default	Ramp Rate Hourly Up	a odates		Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Updates	Fuel Price Adjustments	Generation By Portfolio	XML Download						
hedu	le Offer	s Daily	Default																		
t Date, U	Jnit and Sched	lule		«	Schedu	ile Offer for UN	NIT NAME - DF	LPRICE (Price-bas	ed) on 24-Aug-2	014											
Date:	24-Aug-2014						Fuel Selection:	No2													
Portfolio:	All_Units			~				1902													
Unit:	UNIT NAME			~		Hot Sta	rtup Cost (\$):	849	5.00												
chedule:	DFLPRICE			~	In	itermediate Sta		849													
							rtup Cost (\$): Load Cost (\$):	840													
						Incre	mental Energy Ot	fers													
						Us	se Offer Slope:														
							M99	Prie													
							55.0 60.0	22													
							65.0	22													
							70.0	23	1.83												
							75.0	23													
							80.0	24													
							90.0		1.52												

Figure 3 – Schedule Offers Default Display of a single-fuel unit
IS	nev	w englar	ıd											Market Sta	tus: OPEN(01 Participant:	 V. Certific	eMark ersion: 2.0.1(cate User: Sup	201407221 peruser		ug-2014 0	
Public Ge	eneration	Demand	ARD Regi	ulation																	-
	Unit Default Parameters	Portfolio Manager	Ramp Rate Default		np Rate urly Updates	Schedule Manager	Schedule Detail Defaults	Schedule Offers Default	Schedule Offer Hourly Updates			Hourly Updates	Fuel Price Adjustments	Generation By Portfolio	XML Download						
Sched	ule Offe	ers Daily	/ Defaul	lt																	
elect Date,	Unit and Sch	edule			« Sche	dule Offer fo	r UNIT NAME - DI	FLPRICE (Price-ba	red) on 24-Aug	2014											
Date	: 24-Aug-20	14			•					Fuels											
	All_Units				×				ма	natgas2 %	No2	*									
	UNIT NAM	F			~	Hot	:Startup Cost (\$):	0.47	5.00	% 100		* 0									
	DFLPRICE						Startup Cost (\$):		5.00	100		0									
						Cold	Startup Cost (\$):	846	5.00	100		0									
							No Load Cost (\$):	528	2.00	100		0									
							Use Offer Slope:	E Pr	0e	%		%									
							65.0		9.78	100		0									
							60.0		5.40	100		0									
							65.0		1.02	100		0									
							110.0					-									
												_									
												_									
																			Hedete		efresh
																			Update	K	
essages																					

Figure 4 – Schedule Offers Default Display of a multi-fuel unit

The fields on this display are:

- Hot Startup Cost the startup cost of the unit from a hot temperature (state).
- Intermediate Startup Cost the startup cost of the unit from an intermediate temperature (state).
- **Cold Startup Cost** the startup cost of the unit from a cold temperature (state).
- No Load Cost the inherent cost of running the unit.
- Use Offer Slope this check box indicates whether to use the offer slope for the incremental energy offers within the market clearing and dispatch software as illustrated in Figure 2.
- **MW** the net MW unit output associated with this segment. The last MW point should be greater than or equal to the Economic Maximum Limit.
- **Price** the \$/MWH price offered for this segment.

4.8.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- A non-blank value is entered for each Startup and No Load cost and associated fuel blend percentage if applicable.
- At least one non-blank Price/MW pair value is entered along with fuel blend percentage if applicable.

- Cold Startup Cost \geq Inter Startup Cost \geq Hot Startup Cost \geq 0
- No Load Cost ≥ 0
- An error message will be returned if any of the following offer parameter values exceed their threshold value established by the ISO-NE Market Monitor⁷: Cold/Intermediate/Hot Startup Cost, No Load Cost; for multi-fuel units, the thresholds are dynamically calculated using the unit's reference levels for each fuel and the fuel blend information submitted with the cost offers.
- Fuel blend percentages for a financial offer must add up to 100; only whole numbers between 0 and 100 are allowed for a fuel blend percentage value.
- MW quantity must be strictly positive or zero; i.e. $MW \ge 0$
- Price of Incremental Energy Offers must be less than or equal to the Energy Offer Cap price and greater than or equal to the Energy Offer Floor price; i.e. Energy Offer Floor price ≤ Price ≤ Energy Offer Cap price⁸.
- Offers must be monotonically increasing. The price value for each segment must be greater than the one of the previous segment after the segments are ordered in increasing MW value⁹.

4.8.2 Modify Schedule Offers

- 1. In the Generation Navigation sub tabs, click the **Schedule Offers Default** tab.
 - The Schedule Offers Daily Default display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired unit from the **Unit** filter.
 - d. Select the desired schedule from the **Schedule** filter.
 - The display reflects the selected schedule's offer data for the selected date.
- 3. Enter or modify the values on the display. To delete an incremental energy offer segment, simply blank each value on the row where the segment is shown, including the fuel blend percentages.
- 4. Click the **Update** button to save the data.

⁷ Refer to Market Rule 1, Appendix A, Section III.A.5.5.

⁸ Energy Offer Floor price and Energy Offer Cap price are both defined in Market Rule 1, section I.2.2.

⁹ The same price for successive segments is allowed; however it doesn't really make sense when considering Participant offers because the two segments are equivalent to a single segment with a MW value equal to that of the second segment. It is allowed only to permit mitigation reference price values for successive offer segments to be equal.

4.9 Schedule Offers Hourly Updates Display

This display provides for viewing, entering and modifying hourly updates to a unit schedule's default offers for Startup and No Load cost, incremental energy price/MW pairs, and Use Bid Slope parameters for a given market day, including the fuel blend associated with each financial offer parameter for multi-fuel units. Refer to §4.8 Schedule Offers Daily Default Display for a description of these schedule offer parameters and how incremental energy offers must be structured. *This is the only Generation display enabled for submittal of hourly updates during the RT Intraday Re-Offer period*.

The format of the display for a single-fuel unit is shown below. For a multi-fuel unit, the columns showing the *fuel index* of the fuels that the unit is configured to use with data entry cells to enter the fuel blend percentages are shown just like with the Schedule Offers Daily Default display. As you can see, there are multiple display tabs on this display:

- The "DEF" tab shows the default offers for the selected market day. These were entered and are shown on the Schedule Offers Default display. They cannot be changed using this display.
- There is one tab for each hour of the market day. Each tab is labeled with the hour-ending value of the market hour. There is an extra tab labeled "02*" for the extra hour on the Fall Daylight Saving Time transition day. On the Spring Daylight Saving Time transition day, the tab labeled "03" will not be shown. When the selected market day date is today, the tab labels for the hours that have passed are de-emphasized to indicate that the values for those hours cannot be changed.

				eMarket User Interface
ISO new england				cate User: Superuser 23-Aug-2014 09:53:31
por a second			Participant: ISO New England Inc	Viser: Superuser V
Public Generation Demand ARD Regulation				
Private Unit Default Portfolio Ramp Rate Ramp Ra Messages Parameters Manager Default Hourly L				
Schedule Offers Hourly Updates				
Select Date, Unit and Schedule	Schedule Offer for UNIT NAME - ISOPRICE (Price-based) on 2			
Date: 23-Aug-2014	DEF 01 02 03 04 05 06 07 08 0	10 11 12 13 14 15 16 17 18 19 20	21 22 23 24	
Portfolio: Al_Units	Fuel Selection: MA_natgas2			
Unit: UNIT NAME			Legend	
Schedule: ISOPRICE	Hot Startup Cost (\$): 9100.00		No Override: 🔳	
	Intermediate Startup Cost (\$): 10497.00	\sim	False:	
	Cold Startup Cost (\$): 18087.00		True:	
	No Load Cost (\$): 2008.73			
Copy From: Default		Displa	y tab labels are de-en	nphasized for
Copy To: V Through: V	Incremental Energy Offers	those	hours that are in the	past when the
	Use Offer Slope: 💟			
Сору	MW Price		ed market day date is	
	167.0 -90.00	"Delet	te" button is also hido	len when one
	162.0 49.47	of the	co dicalov toba is colo	stad That's
	187.0 50.15		se display tabs is sele	
	177.0 51.55 187.0 53.07	becau	se the hourly updates	s values for
	187.0 53.07		hours cannot be char	
	207.0 56.63			igeu oi
	217.0 68.83	remov	/ed.	
	227.0 61.43			
	237.0 64.86			
				Update Refresh
Messages				A A A A A A A A A A A A A A A A A A A A A A A

As illustrated below, whenever there are values entered for an hour that are not yet saved, the display tab label is highlighted in red. Selecting the hour's display tab will show the modified data entry fields highlighted in pink. Furthermore, if there are updates for an hour (whether saved in the database or entered on the screen) that can be changed for the selected market day, a "**Delete**" button is provided on the display. Selecting the button sets all the data entry fields to blank and the checkboxes to "No Override" state without saving the entered data. You must select the "**Update**" button to actually delete the hourly updates. That way, you can delete the updates to multiple hours with one "Update" selection.



4.9.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks for each market hour where data has been entered to ensure that the entered data is valid:

- A non-blank value is entered for each Startup and No Load cost and associated fuel blend percentage if applicable except when you are deleting all overrides for the selected market hour.
- At least one non-blank Price/MW pair value is entered along with fuel blend percentage if applicable except when you are deleting all overrides for the selected market hour.
- A selection has been entered for the Use Offer Slope checkbox; it cannot be left in the "No Override" state except when you are deleting all overrides for the selected market hour.
- Cold Startup Cost \geq Intermediate Startup Cost \geq Hot Startup Cost \geq 0
- No Load Cost ≥ 0

- An error message will be returned if any of the following offer parameter values exceed their threshold value established by the ISO-NE Market Monitor¹⁰: Cold/Intermediate/Hot Startup Cost, No Load Cost; for multi-fuel units, the thresholds are dynamically calculated using the unit's reference levels and the fuel blend information submitted with the cost offers.
- Fuel blend percentages for a financial offer must add up to 100; only whole numbers between 0 and 100 are allowed for a fuel blend percentage value.
- MW quantity must be strictly positive or zero; i.e. $MW \ge 0$
- Price of Incremental Energy Offers must be less than or equal to the Energy Offer Cap price and greater than or equal to the Energy Offer Floor price; i.e. Energy Offer Floor price ≤ Price ≤ Energy Offer Cap price¹¹.
- Incremental Energy Offers must be monotonically increasing. The price value for each segment must be greater than or equal to the one of the previous segment after the segments are ordered in increasing MW value¹².
- The market hour for which updates are being submitted must be a future hour. In other words, the selected market day date must be tomorrow or a later date. During the RT Intraday Re-Offer period when the selected market day date is today, if the market hour is the next hour from now, the request must be made 30 minutes before the start of the hour.

4.9.2 Override Schedule Offers for One or More Hours of a Market Day

- 1. In the Generation Navigation sub tabs, click the **Schedule Offers Hourly Updates** tab.
 - The "DEF" Schedule Offers Hourly Updates display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired unit from the **Unit** filter.
 - d. Select the desired schedule from the **Schedule** filter.

¹⁰ Refer to Market Rule 1, Appendix A, Section III.A.5.5. Threshold values are different for supply offer values submitted while the DA market and the Re-Offer periods are open versus those in effect during the RT Intraday Re-Offer period. As outlined in the Market Rule, you cannot increase your offer of Startup and No Load costs during the RT Intraday Re-Offer period without either a corresponding change to your fuel blend or submitting a Fuel Price Adjustment request. The allowable change in the Startup and No Load supply offers during the RT Intraday Re-Offer period is referenced in Market Rule 1, Appendix A, Section III.A.6.2.

¹¹ Energy Offer Floor price and Energy Offer Cap price are both defined in Market Rule 1, section I.2.2.

¹² The same price for successive segments is allowed; however it doesn't really make sense when considering Participant offers because the two segments are equivalent to a single segment with a MW value equal to that of the second segment. It is allowed only to permit mitigation reference price values for successive offer segments to be equal.

- The "DEF" display tab reflects the selected schedule's default offer data for the selected date.
- 3. To update the overrides of individual hours:
 - a. Select the display tab of the hour for which you want to enter new hourly overrides of the schedule offers or modify any existing hourly overrides.
 - b. Enter the override value for each of the schedule offers or modify the existing values.
 - c. Repeat steps a and b for each hour that you want to enter new overrides or modify the existing ones.
 - d. Click the **Update** button to save the data.
- 4. To update the overrides for a range of hours; this involves using the "Copy From:" tool in the tooltray. It is designed to minimize the amount of data entry that you are required to perform, especially when there are blocks of hours where the override values are to be the same.
 - a. First create a template of the data that you want to enter for multiple hours. When entering new overrides, you can minimize your data entry by first copying the schedule's default values to the first hour of the range that you want to override and then editing the values for that hour. When modifying hourly updates that are already in place, you pick an hour in the range of hours that you want change and modify the values for this hour.
 - b. Then copy this hour's values to the various range (one or more) of hours where you want this set of overrides to apply.
 - c. You can repeat this process (steps a and b) for other ranges of hours where you want to enter different sets of overrides.
 - d. When you are done entering/modifying the hourly overrides to the schedule's offer, click the **Update** button to save the data.

4.9.3 Remove/Delete Hourly Updates to Schedule Offers

- 1. In the Generation Navigation sub tabs, click the **Schedule Offers Hourly Updates** tab.
 - The "DEF" Schedule Offers Hourly Updates display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired unit from the **Unit** filter.
 - d. Select the desired schedule from the **Schedule** filter.
 - The "DEF" display tab reflects the selected schedule's default offer data for the selected date.
 - A horizontal bar appears above the hour-ending value of the display tabs where there are overrides/updates in place.

- 3. To remove/delete the overrides of individual hours:
 - a. Select the display tab of the hour for which you want to remove the hourly overrides. If you are allowed to remove the overrides for the selected hour, a "Delete" button will be shown in the lower display toolbar.
 - b. Click on the "Delete" button. All of the data entry fields on the display are set to blank and the two checkboxes are set to "No Override" state.
 - c. Repeat steps a and b for each hour that you want to remove the overrides.
 - d. Click the **Update** button to permanently delete the override data in the database.
- 4. To remove/delete the overrides for a range of hours; this involves using the "Copy From:" tool in the tooltray.
 - a. Pick the display tab of the first hour in a range of hours where you want to delete the overrides; click the "Delete" button to set all data entry fields to blank and the checkboxes to "No Override" state. You don't need to save ("Update") this data entry.
 - b. Copy the selected hour's values to the various range (one or more) of hours where you want to remove the overrides. *Note*: if there is an hour with no overrides in the selected market day, you can skip step a and copy this hour to the various range (one or more) of hours where you want to remove the overrides.
 - c. Repeat this process (steps a and b) for other ranges of hours where you want to also remove the overrides.
 - d. When you are done, click the **Update** button to permanently delete the override data in the database.

4.10 Schedule Times Hourly Updates Display

This display provides for viewing, entering and modifying hourly updates to a unit schedule's default Notification and Startup times. Refer to §4.7 Schedule Detail Defaults Display for a description of these schedule offer parameters. To show the impact of these parameters on the unit's eligibility for off-line reserve designations, it also shows

- The unit's default Minimum Run Time and Minimum Down Time that were entered on the Schedule Detail Defaults display (§4.7).
- The unit's eligibility to be treated as a 10-minute and 30-minute Fast Start resource; i.e. its eligibility to receive 10-minute and 30-minute off-line (Non-Spin) reserve designations.

To be treated as a 10-minute resource, the unit must be designated as Claim10 capable. Similarly to be treated as a 30-minute resource, the unit must be designated as Claim 30 capable. These two capability flags are asset registration parameters and they are shown on the Unit Default Parameters display (§4.3). In addition each eligibility is a function of the unit's Cold Notification Time + Cold Startup Time along with its Minimum Run Time and Minimum Down Time. That is why the information is shown both at the schedule default level and for each hour of the operating day where the hourly time values override the default values and can be different for each hour of the operating day. Thus a unit can be eligible based on the schedule's default offer parameters but not eligible for a particular hour of the operating day because of the hourly overrides of either Cold Notification or Startup offer parameters.

																eMarke		er Interi	face
S	nev nev	v englan	d												c.	rtificate User: Supe	eruser	23-Aug-201	4 10:07
														Participant: ISO N	lew England Inc		Y User	Superuser	
dic 1	Generation	Demand	ARD Regul	ation															
ate sages	Unit Default Parameters	Portfolio Manager	Ramp Rate Default	Ramp Rate Hourly Up		Schedule Detail Defaults	Schedule Offers Defaul	Schedule Offers Hourly Updates			Hourly Updates	Fuel Price Adjustments		XML Download					
	11	-	11			11	Cites Delada	Thomy optimes	I Houry opuates	Descourt	opoints	Aufumiens	by Fordoid	Comions					
			Notifica		ne Hourly U	pdates													
t Date	e, Unit and Sch	edule		*															
Dat	e: 24-Aug-20	14			Default Values fo	UNIT NAME - D		e-based) on 24-Au	-2014										
Portfol	io: All Units			~			Notifical	ion Times Inter	Cold H	Start	ap Times	Cold	Min Run Time	Min Down Time	Other Fast Start (10 min)	Fast Start (30 min)			
	It: UNIT NAME	F		~			0.08	0.08		0.08	0.08	0.08			No	Yes			
	e: DFLPRICE	L		~															
criedu	DELEBRICE			•	Override Hourly	Values for UNIT	AME - DFLPR	CE (Price-based)	n 24-Aug-2014										
					Hour		Notificati	on Times		Startup	Times		c	ither					
							Hot	Inter	Cold Ho	et i i i i i i i i i i i i i i i i i i i	Inter	Cold	Fast Start (10 min)	Fast Start (30 min)					
					• •	Copy													
					01		0.08	0.08	0.08	0.08	0.08	0.08	Yes						
					02		0.08	0.08		0.08	0.08	0.08	Yes						
					03		0.08	0.08		0.08	0.08	0.08	Yes						
					04		0.08	0.08		0.08	0.08	0.08	Ves						
					05		0.08	0.08		1.08	0.08	0.06	No						
					05					0.08	0.08	0.08	No						
					05		0.08	0.08	0.5										
							0.08	0.08		1.08	0.08	0.08	No	No					
					07				0.6		0.08	0.08	No						
					D7 D8 D9 10		0.08 0.08 0.08	0.08 0.08 0.08	0.5 0	1.08 1.08 1.08	0.08 0.08	0.08	No No	No					
					07 08 09 10 11		0.08 0.08 0.08 0.08	0.08 0.08 0.08 0.08 0.08	0.6 0 0.5 0 0.5 0	0.08 0.08 0.08	0.08 0.08 0.08	0.08 0.08 0.08	No No	No No					
					07 08 09 10 11 12		0.08 0.08 0.08 0.08 0.08 0.08	0.08 0.08 0.08 0.08 0.08	0.6 0 0.5 0 0.5 0 0.6 0 0.6 0	1.08 1.08 1.08 1.08	0.08 0.08 0.08 0.08	0.08 0.08 0.08 0.08	No No No	No No No					
					07 08 09 10 11 12 13		0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08	0.08 0.08 0.08 0.08 0.08 0.08	0.6 0 0.5 0 0.5 0 0.5 0 0.5 0 0.5 0 0.5 0 0.5 0 0.5 0	1.08 1.08 1.08 1.08 1.08 1.08	0.08 0.08 0.08 0.08 0.08	0.08 0.08 0.08 0.08 0.08	No No No No	No No No No No					
					07 08 09 10 11 12		0.08 0.08 0.08 0.08 0.08 0.08	0.08 0.08 0.08 0.08 0.08	0.6 0 0.5 0 0	1.08 1.08 1.08 1.08	0.08 0.08 0.08 0.08	0.08 0.08 0.08 0.08	No No No	No No No No No No					
					07 08 09 10 11 12 13 13 14		0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08	0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08	0.5 00 0.5 00 0.5 00 0.5 00 0.5 00 0.5 00 0.5 00 0.5 00 0.5 00 0.5 00	1.08 1.08 1.08 1.08 1.08 1.08 1.08	80.0 80.0 80.0 80.0 80.0 80.0	0.08 0.08 0.08 0.08 0.08 0.08	Na No Na Na Na	No No No No No Yes					
					07 08 09 10 11 12 13 13 14 15		0.08 0.09 0.08 0.08 0.08 0.08 0.08 0.08	0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08	0.5 00 0.5 0000000000	1.08 1.08 1.08 1.08 1.08 1.08 1.08 1.08	0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08	0.08 0.08 0.08 0.08 0.08 0.08 0.08	No No No No No No No No No	No No No No No No No Yes Yes					
					07 08 09 10 11 12 13 14 15 16 15 16 17 18		0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08	0.08 0.09 0.08 0.08 0.08 0.08 0.08 0.08	0.8 0 0.5 0 0.1 0 0.0 0 0.0 0 0.0 0 0.0 0 0.0 0 0.0 0 0.0 0 0.0 0 0.0 0 0	0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08	80.0 80.0 80.0 80.0 80.0 80.0 80.0 9.0 0.0 0.0 0.0 0.0 0.0 0.0	0.08 0.08 0.08 0.08 0.06 0.06 0.08 0.08	Na Na Na Na Na Na Na Na Na Na Na Na Na	No No No No No Yes Yes Yes					
					07 09 09 10 11 12 13 14 15 16 17 17 18 18		0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.00 0.0 0.	0.08 0.09 0.08 0.08 0.08 0.08 0.08 0.08	0.6 0 0.5 0 0.1 0 0.0 0 0.0 0 0.0 0 0.0 0 0.0 0 0.0 0 0.0 0 0.0 0 0.0 0 0	0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.00 0.00 0.00 0.00 0.00	80.0 80.0 80.0 80.0 80.0 80.0 80.0 9.0 0.0 0.0 0.0 0.0 0.0 0.0	0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08	Na Na Na Na Na Na Na Na Na Na Na Na Na N	No No No No No No No No No Yes Yes Yes Yes					
					07 09 10 11 12 13 14 15 16 17 16 17 19 20		0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.00 0.0 0.	0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08	0.8 00 0.5 0000000000	0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.00	80.0 80.0 80.0 80.0 80.0 80.0 80.0 80.0	80.0 80.0 80.0 80.0 80.0 80.0 80.0 80.0	Na Na Na Na Na Na Na Na Na Na	No No No No No Yes Yes Yes Yes					
					07 09 09 10 11 12 13 14 15 16 17 17 18 18		0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.00 0.0 0.	0.08 0.09 0.08 0.08 0.08 0.08 0.08 0.08	0.6 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.00 0.00 0.00 0.00 0.00	80.0 80.0 80.0 80.0 80.0 80.0 80.0 9.0 0.0 0.0 0.0 0.0 0.0 0.0	0.08 0.08 0.08 0.08 0.08 0.08 0.08 0.08	Na Na Na Na Na Na Na Na Na Na Na Na Na N	No No No No No Yes Yes Yes Yes					

The format of the display is shown below.

The Default Values section is provided as a reference to show what are the default schedule offers for the chosen day that are being overridden by the hourly update values entered in the section below. Only the Notification Times and Startup Times in the Hourly Values section are enterable. The other information shown in this section is derived from the entered values as explained above and is for reference only.

4.10.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- Data is entered for each of the parameter values for an hour, i.e. non-blank values appear in all the data entry cells, if any one of the cells contains a nonblank value. In other words, when you want to override one time value of an hour, you must enter values for all the other time parameters even if they are the same as the default values. When all data entry cells for an hour are set to blank, it means that the override for the hour is to be removed/deleted.
- Cold Startup Time ≥ Inter Startup Time ≥ Hot Startup Time ≥ 0
- Cold Notification Time ≥ Inter Notification Time ≥ Hot Notification Time ≥ 0
- An error message will be returned if any of the following offer parameter values for an hour exceed their threshold value established by the ISO-NE Market Monitor¹³:
 - Cold Notification Time, Cold Startup Time
 - Inter Notification Time, Inter Startup Time
 - Hot Notification Time, Hot Startup Time
- An error message will be returned if the sum of each of the following offer parameter values is greater than the sum of their respective reference values by more than a threshold value established by the ISO-NE Market Monitor ¹⁴. The reference values used for this validation are those found in the unit's cost-based Reference schedule 98. The daily default offer values of Min Downtime and Min Runtime as defined in the schedule's Detail defaults and the corresponding reference values are used in the calculations.
 - Hot Notification Time, Hot Startup Time, Min Downtime, Min Runtime
 - Inter Notification Time, Inter Startup Time, Min Downtime, Min Runtime
 - Cold Notification Time, Cold Startup Time, Min Downtime, Min Runtime

4.10.2 Override Startup and Notification Times for a Day and Hour

As discussed in §4.7 Schedule Detail Defaults Display, you define on this display a unit's daily default Startup and Notification time values under the selected

¹³ Refer to Market Rule 1, Appendix A, Section III.A.6.1. The reference level values from which the threshold values are derived are found in the unit's Reference schedule 98 (the one with the name xxxx_R1 where xxxx is the unit's short name).

¹⁴ Refer to Market Rule 1, Appendix A, Section III.A.6.1. Currently defined as 6 hours (August 2014)

schedule. These values are shown in the Defaults section of the display; they are the values used for those hours where an override is not defined. If you want to profile the unit's times on an hourly basis (to reflect the unit's actual manned hours of the day for example), you override the daily default schedule values using this display as follows:

- 1. In the Generation Navigation sub tabs, click the **Schedule Times Hourly Updates** tab.
 - The Schedule Startup and Notification Time Hourly Updates display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired unit from the **Unit** filter.
 - d. Select the desired schedule from the **Schedule** filter.
 - The display reflects the hourly overrides for the selected unit schedule.
- 3. To update the overrides for individual hours:
 - a. Enter override values for each time parameter in the hours that you wish. Enter blanks if you want to remove an override. Remember that you must enter values for all time parameters.
 - b. Click the **Update** button to save the data.
- 4. To update the overrides for a range of hours using the first row in the grid:
 - a. Enter override values for each time parameter in the top row. Enter blanks if you want to remove overrides.
 - b. Specify the range of hours to override by selecting the "From" (left dropdown selector) and "To" (right drop-down selector) hours in the top row



- c. Click the **Copy** button; this will copy the data that you entered in the top row to each row for the hours in the range that you selected.
- d. Repeat steps a through c for other range of hours as desired.

e. Click the **Update** button to save the data for the selected hours.

4.11 Schedule Selection Display

This display is used to make schedules available for a given day and market. The format of the display is shown below:

ISC	new e	england							Market	: Status: OP	EN(01)			rsion: 2.0.7(2 te User: Sup	2014121116		014 18:22:
	1									Partic	ipant:	ISO New England	Inc		Y User	Superuser	
blic Ge	eneration De	mand ARD	Regulat	ion													
			mp Rate fault	Ramp Rat Hourly U			Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Updates	Fuel Price Adjustments	Generation By Portfolio	XML Download			
chedu	ule Select	ion															
ect Date a	and Portfolio			«	Schedules fo	Portfolio AAA on 2	1-Dec-2014										
Date	21-Dec-2014				Unit		Schedule I	D Type	Schedul	e		Day Ahead	Real Time	Available			
					UNIT NAME		36598	Cost	UNIT_R1			x	x				
Portfolio:	AAA			•	UNIT NAME		36597	Cost	UNIT_R2	2		x	x				
					UNIT NAME		36599	Cost	UNIT_M			х	x	v			
					UNIT NAME		36501	Price	DFLPRIC	E		x	х	V			

The fields on this display are:

- **Unit** the name of the unit. All units listed in the selected portfolio along with each of their schedules are shown on this display.
- Schedule ID the unique identifier of the schedule; it is the concatenation of the unique unit identifier and of the schedule type number.
- **Type** the type of schedule: Price or Cost
- **Schedule** the name of the schedule as it was entered on Schedule Manager display
- **Day Ahead** an X-mark indicates that the schedule is enabled to be used in the clearing of the Day Ahead market if the schedule is flagged as available.
- **Real Time** an X-mark indicates that the schedule is enabled to be used in the clearing of the Real Time market if the schedule is flagged as available.
- **Available** a checkmark indicates that the schedule is to be use in the clearing of the selected markets on the selected operating day.

4.11.1 Select Schedule for an Operating Day and Market

The Unit Schedule Selection operation is for modifying the availability of a schedule for a specific day.

Note: One and only one price-based schedule can be available for a market on any given day. Therefore to make a schedule that is enabled for both Day Ahead and Real Time markets unavailable, you must at the same time make another schedule that is

enabled for both markets available or two other schedules, one enabled for the Day Ahead market and another enabled for the Real Time market, available.

When a schedule for a unit that is a regulating resource is made available, irrespective of whether it is a schedule designated for use in the DA market or the RT market or both, the regulation limit nesting rule is checked; i.e. Economic Min \leq Regulation Low < Regulation High \leq Economic Max where the economic and regulation limits are the more constraining of the limits among the daily default (those entered on the Schedule Detail Defaults display and on the Regulation Offers Daily Default display) and the hourly updates (those entered on the Unit Hourly Updates and on the Regulation Hourly Updates and on the Regulation Hourly Updates display) for the selected market day. A warning message is returned when this rule is violated.

To change the availability of schedules:

- 1. In the Generation Navigation sub tabs, click the **Schedule Selection** tab.
 - The Schedule Selection display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - The schedules for all units in the chosen portfolio are displayed.
- 3. Toggle the **Available** checkboxes as desired.
- 4. Click the **Update** button to save your modifications.

Note: To change the market to which a schedule applies you must use the Schedule Detail Defaults display.

Note: You cannot change the availability of the cost-based schedules. The availability of these is managed by ISO New England.

4.12 Unit Hourly Updates Display

This display is used to view and modify the hourly overrides of a unit's MW limits, default dispatch availability status and when it is applicable, its off-line reserve capabilities for a specified market day. Changes made to the unit data using the Unit Hourly Updates display supersede all schedule limits (Schedule Detail Defaults display) and default unit parameters (Unit Default Parameters Display).

E	new england									Market Statu	IS: OPEN(01)			2.0.1(201407221 er: Superuser	23-Aug-2	2014 10:3
											Participant: I	50 New England :	Inc	• Use	superuser	
ic Ge	meration Demand ARD Regulation															
	Unit Default Portfolio Ramp Rate Ramp	Rate Schedule	Schedule	Schedule	Schedule Offer	s Schedule Tin	es Schedule	Hourly F	Fuel Price	Generation	KML					
ages	Parameters Manager Default Houri	y Updates Manager	Detail Defaults	Offers Default	Hourly Update	s Hourly Upda	tes Selection	Updates /	Adjustments	By Portfolio	Download				-	
nit Ho	ourly Updates															
		«													\sim	
Date a	ind Onit															
Date:	24-Aug-2014	Default Values for U														
ortfolio:	All Units		Emerg Min 45.0	Econ Min 55.0	Econ Max 120.0	RTHOL Type 125.0 Price	Day Ahead	Real Time	Available	 Capat 	Claim 10	Claim30 Yes	Default Status:	C Unavailable		
			55.0	55.0	97.0	97.0 Cost	x	x	v V	Defa		60		Economic		
Unit:	UNIT NAME ~	UNIT_R1	55.0	55.0	97.0	97.0 Cost	×	x		c	ap 18.5	61.1				
		1												C Must Run		
		Override Hourly Va	dues for UNIT NA	ME on 24-Aug	2014					-						
		Hour	Emerg Min	Econ Min	Econ Max	RTHOL	Claim10	Claim30	Unavailable	Economic	Must Run					
		v v Co	ру						0	0	C					
		01	50	0 55.	0 117.5	117.5	16.5	59.5		۲	C					
		02	49				17.6	60.5		ě	Õ					
		03	48	D 63.	0 119.5	119.5	18.5	61.6		۲	õ					
		04	49	.0 54.	0 118.5	118.5	19.5	62.5	0	۲	Ċ					
		05	49	D 55.	0 117.5	117.5	20.5	63.5	5 O	۲	C					
		06	61	J 68.	0 116.5	116.5	19.5	62.6	0	۲	C					
		07	50	.0 55.	0 117.5	117.5	18.5	61.5		۲	0					
		08	61	.0 56.	0 116.5	116.5	17.5	60.5		۲	C					
		09	62	D 67.	0 117.5	117.5	16.5	69.6		۲	0					
		10	53	J) 57.	8 118.5	118.5	17.5	60.5		۲	C					
		11	52				18.5	61.5		۲	0					
		12	61				19.5	62.5		۲	C					
		13	50				20.5	63.5		۲	Ç					
		14	49				19.5	62.5		۲	C					
		15	48				18.5	61.5		۲	0					
		16	49				17.5	60.6		۲	0					
		17	50				10.5	69.5			0					
		18	40				17.5	60.5			0					
		20	48				18.5	61.0		۲	C					
		20	4/				19.5	62.5			0					
		22	40				19.5	62.5			0					
		20	40				18.5	61.5		۲	0					
							17.5	60.5			Ő					
		24	51													

The format of this display is shown below:

The Default Values section is provided as a reference to show what are the limit values in the various schedules and the default dispatch availability status and off-line reserve claim capability for the chosen day that are being overridden by the hourly update values entered in the section below.

- Schedule The name of the schedule
- Emerg Min The emergency minimum limit specified in the schedule detail
- Econ Min The economic minimum limit specified in the schedule detail
- Econ Max The economic maximum limit specified in the schedule detail
- **RTHOL** The RT high operating limit specified in the schedule detail
- Claim 10 and Claim 30 Capable Indicate whether the unit is registered as capable (Yes/No) of providing 10 and 30 minute non-spinning reserves from an offline state.
- Claim 10 and Claim 30 Default The daily default off-line 10 and 30 minute off-line reserve capability for the market day as submitted using the Unit Default Parameters display.

- Claim 10 and Claim 30 Cap The demonstrated claim capabilities that were established during the last audit conducted by ISO New England. If the demonstrated capability value is undefined, it will be shown as "UD".
- Default Status (Economic/Unavailable/Must Run) The dispatch availability status (Default Status) specified on the Unit Default Parameters display.

The Default Values section can be collapsed using the double up-arrow icon at the extreme right of the section title bar to show more rows in the hourly section as shown below. The collapsed section can be expanded again to see the default values using the double down-arrow icon at the extreme right of the collapsed section title bar.

~	new england										OPEN(01)		Certificate User: S	uperuser	23-Aug-201
											articipant: ISO Ne	w England Inc		Y User:	Superuser
Gener	ation Demand ARD Regulation													_	
		np Rate Schedi urly Updates Manag			Schedule Offers Hourly Updates	Schedule Time Hourly Update			Fuel Price Adjustments	Generation XM By Portfolio Do	L wnload			8	
										-,					\sim
	ly Updates												L		
and t	Unit	*													
e: 24	4-Aug-2014	L ²	s for UNIT NAME on 24-A												
0: Al		Verride Hou	irly Values for UNIT NAM												
it: In	NIT NAME	→ Hour	Emerg Min	Econ Min	Econ Max	RTHOL	Claim10	Claim30	Unavailable	Economic	Must Run				
Ĩ			Сору												
	· ·	01	50.0	55.0 54.0	117.5	117.5	16.5	59.5 60.5			0				
	· · · · · · · · · · · · · · · · · · ·	03	48.0	63.0	110.5	110.5	12.5	61.5			ŏ				
	01	04	49.0	54.0	118.5	118.5	19.5	62.5		۲	ŏ				
	02	05	49.0	55.0	117.5	117.5	20.5	63.5		ě	õ				
	02	06	51.D	56.D	116.5	116.5	19.5	62.6	0	۲	Ó				
	03	07	50.0	55.0	117.5	117.5	18.5	61.5		۲	0				
		08	51.0		116.5	116.5	17.5	60.5		۲	0				
	04	09	62.0		117.5	117.5	16.5	69.6		۲	0				
	05	10	53.0		118.5	118.5	17.5	60.5		۲	0				
	0.5	12	52.0 51.0		119.5	119.5	18.5	61.5			0				
	06	13	50.0		119.5	119.5	20.5	63.5			ŏ				
		14	49.0	67.8	119.5	119.5	19.5	62.5		۲	ŏ				
	07	15	48.0	67.D	118.5	118.5	18.5	61.6		ě	ŏ				
	08	16	49.0	58.0	118.5	118.5	17.5	60.5		۲	õ				
		17	50.0	55.0	117.5	117.5	16.5	59.5		۲	0				
	09	18	40.0		117.5	117.5	17.6	60.6		۲	0				
	10	19	48.0		116.5	116.5	18.5	61.5		۲	0				
	10	20	47.0		116.5	116.5	19.5	62.5		۲	0				
	11	21 22	46.0	54.D 58.D	117.5	117.5	20.5	63.6		۲	0				
		23	45.0	0.00	118.5	118.5	19.5	62.5			0				
	12	24	51.0		118.5	118.5	17.5	60.6		ě	ŏ				
	10 💌														

The Override Hourly Values section shows each hour of the day and fields to enter overrides. When the fields are blank and the radio buttons unchecked, it means that there is no override of the schedule or default values for that hour.

- **Hour** Hour ending values are shown in the first column for each hour in the day. The row for hour ending 03 is not shown for the Spring Daylight Saving Time transition day. Similarly, the value on the row of the repeating hour ending 02 of the Fall Daylight Saving Time transition day is marked with an asterisk (*).
- Emerg Min Column to enter overrides of the Emergency Minimum limit for any hour of the day.
- Econ Min Column to enter overrides of the Economic Minimum limit for any hour of the day.
- Econ Max Column to enter overrides of the Economic Maximum limit for any hour of the day.

- **RTHOL** Column to enter overrides of the RT High Operating limit for any hour of the day.
- **Claim 10** Column to enter overrides of the Claim 10 capability for any hour of the day if the unit is Claim 10 capable. If the unit is not Claim 10 capable, there will be no data entry cells in this column.
- **Claim 30** Column to enter overrides of the Claim 30 capability for any hour of the day if the unit is Claim 30 capable. If the unit is not Claim 30 capable, there will be no data entry cells in this column.

The following override the Default Status specified on the Unit Default Parameters display:

- **Unavailable** When checked, it indicates that the unit is unavailable for commitment and dispatch for that hour of the operating day.
- **Economic** When checked, it indicates that the unit is available for economic commitment and dispatch for that hour of the operating day.
- **Must Run** When checked, it indicates that the unit is self-scheduling for that hour of the operating day; that is, the unit must be committed at its Economic Minimum limit and can be dispatched up to Economic Maximum limit based on economics.

If a value is entered for an hour in any of these columns, values must also be entered in every other column even if they are copies of the default values. Setting these limit values to blank (NULL) means that the corresponding values defined in the Schedule Detail data of the schedules that are flagged as available for the market day are to be used for that market hour; i.e. it is equivalent to removing the override of the values for that market hour. Setting the off-line reserve claim values to blank (NULL) means that the corresponding values defined in the Unit Default Parameters data for the market day are to be used for that market hour. Similarly clearing all of the Unavailable/Economic/Must Run flags means that the unit's default status defined in the Unit Default Parameters data for the market day is to be used for that market hour.

The top row of the grid provides you with the means to enter or clear overrides for a range of hours. If you do use the top row, you need to **remember that leaving all data entry fields on this row blank means that you want to remove the overrides for all hours in the range selected**. For example, if you want to remove all overrides for every hour of the day, select 01 for the "From" hour (left drop-down selector) and 24 for the "To" hour (right drop-down selector) in the Hour drop-down selection lists, select the "Copy" button and then select "Update". Since you did not enter anything in the limit fields or select any of the dispatch availability override flags, all overrides are removed. You cannot use the top row to selectively update the overrides in one column for a range of hours while retaining the override values in the other columns.

4.12.1 Validation rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- If an Econ Max override value is entered, then Econ Max ≥ CSO¹⁵ if CSO >0 – Warning if not i.e. for a resource with a Capacity Supply Obligation, its Econ Max Limit should be at least as high as its Capacity Supply Obligation. A warning will be returned if the rule is violated; it's only a warning because the resource may not be fully available.
- Emerg Min ≤ Econ Min ≤ Econ Max ≤ RTHOL (RT High Operating Limit). This is the **limit nesting rule**. The data submitted is rejected if this rule is violated.
- For units capable of providing regulation, the limit nesting rule is extended to include the regulation limits, i.e.
 Emerg Min ≤ Econ Min ≤ Reg Low < Reg High ≤ Econ Max ≤ RTHOL where Reg Low and Reg High are the regulation low limit and the regulation high limit for the corresponding hour as defined on the Regulation Offers Default display and overridden on the Regulation Hourly Updates display for the selected market day. A warning message is returned when this limit nesting rule is violated.
- When a unit is self-scheduled for a total number of hours (not necessarily consecutive) greater than its Min Runtime, the Max Daily Energy (if defined in the DA and RT Schedule Detail Defaults of the schedules that are flagged as available) **should** be greater than or equal to Economic Min x the total number of hours that the unit is self-scheduled. A warning will be returned if the rule is violated.
- When a unit is self-scheduled for an hour (through the combination of the unit's default status for the day and the hourly overrides of the status), it should not be declared unavailable until its Min Runtime is elapsed. Therefore if the unit is set to Must Run for any hour, it **should** be set for a consecutive number of hours sufficient to cover its Min Runtime. A warning will be returned if the rule is violated.
- When a unit is declared unavailable for an hour (through the combination of the unit's default status for the day and the hourly overrides of the status), it should not be declared available (Economic or Must Run) until its Min Downtime is elapsed. Therefore if the unit is set to Unavailable for any hour, it should be set for a consecutive number of hours sufficient to cover its Min Downtime. A warning will be returned if the rule is violated.

¹⁵ CSO = unit's Capacity Supply Obligation from the Forward Capacity Market.

- A warning message will be returned if any of the following hourly override values exceed their associated threshold value established by the ISO-NE Market Monitor¹⁶: Economic Max, Economic Min, Emergency Min
- A warning will be returned if the unit is Claim 10 capable, a Claim 10 override is entered and any of the following conditions exist
 - Claim 10 = 0
 - Claim 10 > Claim 10 Cap meaning that the Claim 10 Cap value will be used instead of the offered value
- A warning will be returned if the unit is Claim 30 capable, a Claim 30 override is entered and any of the following conditions exist
 - Claim 30 = 0
 - Claim 30 > Claim 30 Cap meaning that the Claim 30 Cap value will be used instead of the offered value
 - Claim 30 < Claim 10 meaning that the Claim 10 value will be used instead of the offered Claim 30 value.

4.12.2 Override MW Operating Limits for a Day and Hour

As discussed in §4.7 Schedule Detail Defaults Display, you define on this display a unit's MW operating limits under the selected schedule. These limit values are shown in the Defaults section for each schedule that you've created for the unit. If you want to profile the unit's MW operating limits on an hourly basis, irrespective of which schedule is designated as available for a given market, you override the schedule values using the Unit Hourly Updates display as follows:

- 1. In the Generation Navigation sub tabs, click the **Hourly Updates** tab.
 - The Unit Hourly Updates display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired unit from the **Unit** filter.
 - The display reflects the hourly overrides for the selected unit.
- 3. To update the overrides for individual hours:
 - a. Enter override values for all of the limits in the hours that you wish. You must also enter unit dispatch availability status overrides. If the unit is Claim 10 or Claim 30 capable, you must also enter Claim 10 and Claim 30 override values. If you want to remove an override, you must enter blanks in every data entry cells and deselect all status overrides.
 - b. Click the **Update** button to save the data.
- 4. To update the overrides for a range of hours

¹⁶ Refer to Market Rule 1, Appendix A, Section III.A.6.1.

- a. Enter override values for all of the limits in the top row. You must also enter unit dispatch availability status overrides. If the unit is Claim 10 or Claim 30 capable, you must also enter Claim 10 and Claim 30 override values. If you want to remove overrides, leave all the data entry cells on the first row blank and the status overrides deselected.
- b. In the Hour column, select a "From" hour (left drop-down selector) and a "To" hour (right drop-down selector) and then select the **Copy** button. This will copy the data entered in the top row to every row for the range of hours selected.
- c. Repeat steps a and b if you want to enter/remove overrides for other ranges of hours.
- d. Click the **Update** button to save the data for the selected hours.

4.12.3 Override the Status of a Generating Unit for a Day and Hour

As discussed in §4.3 Unit Default Parameters Display you define on this display a unit's default dispatch availability status (Default Status) for an entire market day. This status is shown in the Defaults section. If you want to profile the unit's dispatch availability status on an hourly basis, you override this default status using the Unit Hourly Updates display.

- 1. In the Generation Navigation sub tabs, click the **Hourly Updates** tab.
 - The Unit Hourly Updates display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired unit from the **Unit** filter.
 - The display reflects the hourly overrides for the selected unit.
- 3. To update the overrides for individual hours:
 - a. Select the Unavailable/Economic/Must Run status for the hours that you want to override the default status. You must also enter override values for all the limits even if they are copies of the default values. If the unit is Claim 10 or Claim 30 capable, you must also enter Claim 10 and Claim 30 override values. If you want to remove an override, deselect all the status overrides and enter blanks in all other data entry cells.
 - b. Click the **Update** button to save the data.
- 4. To update the overrides for a range of hours
 - a. Select the status override value in the first row. You must also enter override values for all the limits even if they are copies of the default values. If the unit is Claim 10 or Claim 30 capable, you must also enter Claim 10 and Claim 30 override values. If you want to remove overrides, leave all the data entry cells on the first row blank and the status overrides deselected.

- b. In the Hour column, select a "From" hour (left drop-down selector) and a "To" hour (right drop-down selector) and then select the **Copy** button. This will copy the data entered in the top row to every row for the range of hours selected.
- c. Repeat steps a and b if you want to enter/remove overrides for other ranges of hours.
- d. Click the **Update** button to save the data for the selected hours.

4.12.4 Self-Schedule a Generating Unit

If you want to self-schedule (must run) a generating unit, you must submit a **Must Run** status and indicate whether the unit is block loaded or if it can follow ISO dispatch instructions. To self-commit a unit and allow the ISO to issue dispatch instructions, submit a must run status for the unit, with an Economic Min limit less than the Economic Max limit.

Note: For Non-dispatchable units the Economic Min and Economic Max values *should* be equal.

4.13 Fuel Price Adjustments Display

The purpose of this display is to submit Fuel Price Adjustment (FPA) requests for a fuel that a specific unit is configured to use and for a range of hours of a specific market day. Such requests are submitted when the Participant determines that their cost for the fuel on which their offers are based is higher than the market index price of the fuel and their financial offers are at risk of mitigation if the unit's financial reference levels are not updated to reflect their actual fuel purchase price.

The requested price in an FPA must be greater than the market index price of the fuel¹⁷. FPA requests are automatically either approved at the price requested or capped at the Fuel Price Adjustment Limit (FPAL). The FPAL for each fossil fuel is a value that the ISO-NE Market Monitor determines for each fuel's flow day. When an FPA request is processed, it is <u>approved</u> as submitted if the requested price is equal to or below the FPAL. Otherwise the request is approved but the approved price is <u>capped</u> at the FPAL. The ISO-NE Market Monitor has the ability to accept a request price submitted greater than the FPAL, in essence overriding the FPAL, but will do so only at the request of the Participant and on an FPA basis. It is important to keep in mind that natural gas FPA requests for a given market day span two fuel flow days where the FPAL can be different for each fuel flow day. Therefore the FPA requested price can be approved for one fuel flow day and capped in the other.

During the RT Intraday Re-Offer period of a market day, the ISO-NE Market Monitor can override the default FPAL of a fuel on an hourly basis. When that action is taken, all of the FPAs for the fuel will be re-processed to account for the new hourly FPAL. When an FPAL is raised, an FPA price may transition from a capped price to an approved price equal to the requested price, or from a capped price to a higher capped price. If on the other hand the FPAL is lowered, all previously approved/capped hourly FPA prices remain unchanged; in other words, once an FPA price for an hour is set when the FPA is first processed or raised when it is re-processed, it is never lowered by the system.

FPA requests can be submitted only for future hours of today and all hours of any of the next seven days. FPA requests can be submitted at any time. The FPA price of requests for today and tomorrow will be calculated as the FPAs are received and as FPA Limits are updated, meaning that the effective hourly prices of the FPA requests will be calculated immediately as these events occur. The FPA price of requests submitted for days beyond tomorrow will be calculated shortly after the start of day before the requested market day, i.e. when the requested market day becomes tomorrow. The state of the FPAs submitted for days beyond tomorrow will be FPAs submitted for days beyond tomorrow will be shown as Pending on the FPA display until the FPA price is calculated.

¹⁷ The FPA requested price must be greater than the Fuel Index Price of the fuel by an amount greater than or equal to a threshold established by the ISO-NE Market Monitor as specified in Market Rule 1, Appendix A; III.A.3.4 for specific rules on Fuel Price Adjustments.

FPA prices are used to update the financial reference values in the Reference schedule affected by the FPA request. This updating is normally performed at the same time as the effective hourly prices of an FPA are updated. The immediate updating of the financial reference values in the Reference schedules affected by FPAs for tomorrow is however suspended during the following two time windows:

- During the period of time allocated to the clearing of the DA Market, i.e. from the closing of the DA Market for energy offers (10:00) to the opening of the Re-Offer Period for tomorrow (time when the DAM case is approved but no later than 13:30). The effective hourly price of FPA requests for tomorrow received in this time period are calculated immediately but the Reference schedules are updated when the Re-Offer Period opens.
- During the period of time allocated to the preparation of tomorrow's unit operating plan, i.e. from the closing of the Re-Offer Period (14:00) and the opening of the RT Intraday Re-Offer Period (18:30). The effective hourly price of FPA requests for tomorrow received in this time period are calculated immediately but the Reference schedules are updated when the RT intraday Re-Offer period opens.

The state of the FPAs submitted during these periods will be shown as Pending on the FPA display because the reference levels in the affected Reference schedules are not updated. When the Reference schedules have been updated, the state of the FPAs will be updated to Approved or Capped. Figure 5 below illustrates these processing time windows for one operating day. Note that reference levels for days beyond tomorrow are not calculated. Hence FPA requests for market days beyond tomorrow remain shown as Pending on the FPA display until the requested market day becomes tomorrow and reference values are calculated.



Figure 5 – Fuel Price Adjustment Submittal and Processing for Operating Day

The basic components of an FPA request are by unit, a fossil fuel that the unit is configured to use, a market day, a range of hours in that market day and a requested price. Any number of FPA requests can be submitted for a given market day. If the range of hours specified in one FPA request overlaps the range of hours of a previous FPA request, the last one received supersedes all previous requests, meaning that it is the requested price of the last one received that constitutes the FPA request for the unit's fuel where the FPA requests overlap. In the absence of an FPA request for a market hour, it is the Fuel Index Price that is used to calculate the reference levels in the Reference schedule associated with the unit's fuel. This is illustrated in Figure 6.

The concept of the last FPA request superseding all FPA requests that it overlaps is important when considering the withdrawal of an FPA requests. A Participant can withdraw an FPA request at any time provided that it is not fully passed, i.e. where the end hour of the request is still in the future. Withdrawing an FPA means that for the hours remaining in the range spanned by the request, the Fuel Index Price becomes the fuel price of those hours, even if there are underlying FPA requests that the withdrawn FPA request superseded. This is illustrated in Figure 7.



Figure 6 – Illustration of Overlapping FPA Requests



Figure 7 – Illustration of Withdrawing an Overlapping FPA Request

The eMarket display that is used to view and submit Fuel Price Adjustment requests is shown below. You must select a market day date and one of your generation portfolios. You must then select

- A generating unit listed in your selected portfolio; note that only units configured to use a fossil fuel (coal, oil or gas) will be listed in the drop-down list of units;
- A fuel; note that only the *fuel index* names of each fuel that the unit is registered with IMM to use are listed in the drop-down list.

Once your selection is made, the display will refresh and show information on all FPA requests in place for the selected date, unit and fuel, if there are any.

There are three sections to the display:

- Fuel Price Adjustment Submittal section used to submit an FPA request;
- Submitted Fuel Price Adjustments section that lists all the FPA requests that have been submitted for the selected market day;
- Effective Hourly Fuel Price Adjustments section that shows which FPA overrides the Fuel Index Price for each hour of the selected market day.

SO new england							eMarket User Interfac
						Participanti ISO New Engle	Certificate Useri Superuser 23-Aug-2014 11 ind Inc V Useri Superuser
c Generation Demand ARD Regulation							
rte Unit Default Portfolio Ramp Rate Ram ages Parameters Manager Default Hou	mp Rate Schedule S	Schedule Schedule Schedule O Detail Defaults Offers Default Hourly Upd	fers Schedule Times Sched	le Hourly Fuel Price	Generation 30%		
el Price Adjustments	any optimes in mininger in		es rouri opones seros	in oponis neptitiens	Station Stations		
t Date, Unit and Feel	Const Dates Adjustment	nt Submittal for MA_natgas2 - UNIT NAME e	a 24-8aa 2014				
		V To HE: V Price:	Submit				
		Adjustments for MA_natgas2 - UNIT NAME	on 24-Aug-2014				
	FPA HE	HE Requested Submitted End Price	Submitted By	# of Status Attch	Effective Withdrawn Withdrawn By Price HE	Withdrawn	
Fuel: MA_natgas2	✓ # Start 1 13	21 5.00 23-Aug-2014 10:57	Superuser	2 Capped	4.12		
	2 04	11 3.10 23-Aug-2014 10.55		1 Approved	3.10		
	3 11	14 4.50 23-Aug-2014 11:00	Superuser	0 Withdrawn	11 Superater	23-Aug-2014 11:04	
	4 19	22 4.10 23-Aug-2014 11:00	Superuser	1 Approved	4.10		
	6 21	23 0.50 23-Aug-2014 11:14		0 Capped	5.12		
	0 22	23 7.55 23-Aup-2014 11:17	ISO-NE IMM	0 Approved	7.65		
					View Attachments Withdraw		
		Price Adjustments for MA_natgas2 - UNIT	NAME on 24-Aug-2014				
	Hour FPA	Effective Status Price					
	08 2	3.10 Approved					
	07 2	3.10 Approved					
	08 2	3.10 Approved					
	09 2	3.10 Approved					
	10 2	3.10 Approved					
	11 3	Index Withdrawn					
	12 3	Index Withdrawn Index Withdrawn					
	14 3	Index Withdrawn					
	15 1	4.12 Capped					
	10 1	4.12 Capped					
	17 1	4.12 Capped					
	18 1	4.12 Capped					
	19 4	4.10 Approved					
	20 4	4.10 Approved					
	20 4 21 5	6.12 Capped					
	20 4						

The Fuel Price Adjustment Submittal section is used to submit an FPA request. If the selected market day date is in the past, this section is not shown at all because FPA requests cannot be submitted for days in the past. The data entry fields in this section are:

- From HE An hour-ending drop-down selector that is used to specify the hour-ending value of the first hour of the market day for the period spanned by the request;
- **To HE** An hour-ending drop-down selector that is used to specify the hourending value of the last hour of the market day for the period spanned by the request;
- **Price** The FPA requested price in \$/MMbtu.

A **Submit** button is provided to send the FPA request to the system of validation and processing.

The ISO-NE Market Monitor can lockout¹⁸ a Participant's generation unit from submitting FPA requests for a unit. When a lockout condition exists for the selected unit and market day, a message to that effect is shown in the FPA Submittal section of the display as shown below; the data entry fields and the **Submit** button are disabled. While a Participant is locked out, the Participant must contact the ISO-NE Market Monitor to have the FPA requests entered on their behalf. Note that while a Participant is locked out, the Participant can still withdraw an FPA request and must still administer the FPA document attachments for the FPAs previously submitted or submitted on their behalf by the ISO-NE Market Monitor.

¹⁸ Refer to Market Rule 1, Appendix A; III.A.3.4 for specific rules on Fuel Price Adjustments and Lockout provisions.

ISO new england											rsion: 2.0.1(2014072211 ate User: Superuser	19) 23-Aug-2014 11:
									Participanti	150 New England Inc		Superuser
lic Generation Demand ARD Regulation												
ate Unit Default Portfolio Ramp Rate Ram	no Rate Sche	duk Sch	edule Schedule	Schedule Offers	Schedule Times S	chedule Hourly Fuel Price	Generation 33%					
sages Parameters Manager Default Hou	arly Updates Man	iger Deta	al Defaults Offers Default	Hourly Updates	Hourly Updates S	election Updates Adjustments	By Portfolio Downlos	5				
el Price Adjustments												
t Date, Unit and Fuel	C Fuel Price #	idjustment S	ubmittal for MA_natgas2 - I	NIT NAME on 24	Aug-2014							
Date: 24-Aug-2014	From HE		To HE:	Price:		Unit is locked out from Fuel Price						
	×					Adjustmente submittele	1					
	Submitted F		justments for MA_natgas2									
	FPA	HE F Start Er	IE Requested Subm nd Price	tted	Submitted By	Attch	Effective Will Price	hdrawn Withdrawn By HE	With drawn			
Fuel: MA_natgas2	1	13		2014 10:57	Superuser	2 Capped	4.12					
	2	04	11 3.10 23-A	2014 10:59	Superuser	1 Approved	3.10					
	3	11	14 4.50 23-A	2014 11:00	Superuser	0 Withdrawn		11 Superuser	23-Aug-2014 11:04			
	4	19		2014 11:00	Superuser	1 Approved	4.10					
	6	21			Superuser	0 Capped	6.12					
	0	22	23 7.55 23-A	p2014 11:17	ISO-NE IMM	0 Approved	7.55					
							View Attachme	ds With draw				
	Effective He	ourly Fuel Pr	ice Adjustments for MA_na	pas2 - UNIT NAM	E on 24-Aug-2014							
	Hour	FPA	Effective Status Price									
	01											
	02											
	03											
	04	2	3.10 Approved									
	05	2	3.10 Approved									
	08	2	3.10 Approved									
	07	2	3.10 Approved									
	08	2	3.10 Approved									
	10	2	3.10 Approved 3.10 Approved									
	11	2	Index Withdrawn									
	12	3	Index Withdrawn									
	13	3	Index Withdrawn									
	14	э	Index Withdrawn									
	15	1	4.12 Capped									
	10	1	4.12 Capped									
	17	1	4.12 Capped									
	18	1	4.12 Capped									

The Submitted Fuel Price Adjustments section lists all the FPA requests that have been submitted for the selected unit, fuel and market day. They are listed in the chronological order that they were submitted. The display frame of this section can be collapsed/expanded using the double vertical arrows at the extreme right of the section title bar. The display fields in this section are:

- **FPA #** The ordinal number of the FPA request. This number is used in the section below to uniquely identify the FPA request that affects each hour of the market day.
- **HE Start** The hour ending first hour of the market day spanned by the FPA request;
- **HE End** The hour ending last hour of the market day spanned by the FPA request;
- Requested Price The FPA request price in \$/MMbtu;
- Submitted The date and time that the FPA request was submitted.
- **Submitted By** The name of the Participant user who submitted the FPA request. If the FPA request was submitted by the ISO-NE Market Monitor, the name will be "ISO-NE IMM".
- **# of Attch** The number of documents that have been uploaded and attached to the FPA request.
- Status The status of the FPA request. There are four possibilities:
 - Pending This status indicates that the request has been received and accepted but the reference prices in the associated unit Reference schedule have not yet been updated to account for the approved FPA requested fuel price. This state will persist longer than a minute only during those periods of time where the processing of the Reference schedules stemming from FPA requests is suspended.

 Approved – The FPA requested price has been approved as submitted and is the effective fuel index price used to calculate the reference levels in the unit's associated Reference schedule. The effective price shown will be the requested price unless the published Fuel Index Price (FIP) is greater than the FPA requested price, a condition that can arise as explained below. In this latter case, "Index" will be shown as the effective price.

Note that the requested price for an FPA submitted on behalf and at the request of a Participant by the ISO-NE Market Monitor is always approved, even if the requested price is greater than the Fuel Price Adjustment Limit for any hour in the range spanned by the FPA request.

- **Capped** The FPA requested price is greater than the FPA Limit for at least one of the hours spanned by the FPA request. The effective price shown will be equal to the lowest FPA Limit for the hours spanned by the request.
- *Withdrawn* The FPA request was withdrawn. Further information on the withdrawal is shown to the right.
- WithdrawnPending A user has submitted a request to withdraw the FPA. This status indicates that the request has been received and accepted but the reference prices in the associated unit Reference schedule have not yet been updated to account for the withdrawn FPA. This state will persist longer than a minute only during those periods of time where the processing of the Reference schedules stemming from FPA requests is suspended.
- Effective Price This is the approved/capped price for the FPA. This price is
 - Blank as long as the FPA request is in Pending state.
 - The requested price if the FPA is Approved; note that the FPA will be set to Approved if the requested price is less than or equal to the Fuel Price Adjustment Limit for each hour spanned by the FPA (since the Fuel Price Adjustment Limit can be different for each hour spanned by the FPA), or if the FPA requested price was accepted as submitted by the ISO-NE Market Monitor thereby overriding the Fuel Price Adjustment Limits, of if the FPA was submitted by the ISO-NE Market Monitor on behalf and at the request of the Participant.
 - **The lowest Fuel Price Adjustment Limit** at which the FPA was Capped when it was last processed; note that the FPA will be set to Capped if the requested price is greater than the Fuel Price Adjustment Limit for any hour spanned by the FPA (since the Fuel Price Adjustment Limit can be different for each hour spanned by the FPA). This price value can change (increase) when the Fuel Price Adjustment Limit is raised. If the Fuel Price Adjustment Limit is raised above the FPA requested price for all hours spanned by the FPA, the FPA status will change to Approved and the effective price will be set equal to the requested price. The effective price will never be reduced by the system as a result of a decrease in the Fuel Price Adjustment Limit.

- The approved/capped price of the FPA at the time it was *withdrawn*. The effective price of an FPA will never change after the FPA is withdrawn.
- "*Index*" if the FPA requested price is less than the Fuel Index Price for some hours spanned by the FPA since the Fuel Index Price can be different for each hour spanned by the FPA. This can only happen when the Fuel Index Prices are updated after the FPA was submitted. For example, just before the Re-Offer Period closes, the Fuel Index Prices of gas are updated with the published market index prices. These values can be higher than those used to clear the DA market and for which a Participant may have submitted an FPA request with a price above the FIP at the close of the DAM but which is now less than the updated FIP for the next gas day at the close of the Re-Offer Period.
- Withdrawn HE The hour-ending value of the effective end hour of a withdrawn FPA; blank otherwise. All hours of the original FPA request between Withdrawn HE and HE End are effectively cancelled, as well as all hours in the same range of all previously submitted FPA requests. The withdrawn HE value is equal to the HE value of the FPA start hour or of the withdrawal date/time (shown on the right) + 60 minutes (i.e. the HE of the hour following the HE time that the FPA was withdrawn), whichever is the later hour.
- Withdrawn by The name of the Participant user who withdrew the FPA request; blank otherwise.
- Withdrawn The date and time that the FPA request was withdrawn; blank otherwise.

Two <u>normally disabled</u> buttons appear at the bottom of this section. They are <u>enabled when one of the rows in the grid is selected</u>. They provide the following functionality:

- Withdraw This button brings up a popup requesting the user to confirm that the selected FPA request is to be withdrawn. When the user confirms the action, a request is sent to the database to withdraw the FPA. When the request is acknowledged, the FPA status is changed to WithdrawnPending. When the affected Reference schedule has been updated, the FPA status changes to Withdrawn and the Withdrawn HE, Withdrawn by, and Withdrawn date/time fields on the selected FPA are filled in. If the selected FPA is already Withdrawn or if the FPA HE End is equal to the HE value of the current hour (i.e. there are no hours spanned by the FPA beyond the current hour), the button will be disabled.
- View Attachments This button brings up a pop-up (shown below) on which all of the electronic document files attached to the FPA are listed as shown below.

achn	nents to Fuel Price Adjustment for MA_natgas2 - UNIT NAM	E on 24-Aug-2014		2
#	Document File Name	Submitted	Submitted By	
1	Energy market changes December MC2.pptx	23-Aug-2014 11:32	Superuser	
2	2012_014 Energy Market Offer Flexibility Project MDDN red	23-Aug-2014 11:33	Superuser	
	Add Remove	Open	Close	

The displays fields on the pop-up display are:

- **#** The ordinal number of the document. Documents are numbered in the order that they are uploaded.
- Document File Name The filename of the document that was uploaded.
- **Submitted** The date and time that the document was uploaded.
- **Submitted By** The name of the Participant user who uploaded the document.

The buttons at the bottom of the pop-up display allow the user to:

- **Add** – Upload documents and attach them to the FPA request. This button brings up a file selection popup that the user must use to select the document file to upload. *If the selected market day is more than 5 business days in the past, a warning message stating that the attachment is late appears in the popup.*

Only "safe" files can be uploaded. The following types of files are considered "safe":

- Microsoft Word documents (.doc, .docx) but not macro-enabled documents (.docm)
- Microsoft Excel documents (.xls, .xlsx) but not macro-enabled documents (.xlsm)
- Microsoft Power Point documents (.ppt, .pptx) but not macro-enabled documents (.pptm)
- Adobe Acrobat document files (.pdf)
- JPEG and JPG images, GIF images
- Rich-Text-Format documents (.rtf) ex: copies of email messages
- Simple text documents (.txt)
- Remove Select a previously uploaded document and remove it from the list of documents attached to the FPA request
- **Open** Select a previously uploaded document, download it and either save it on the user's computer or open it in the application associated with the document type.
- **Close** Close the pop-up window.

The Effective Hourly Fuel Price Adjustments section shows which FPA request price overrides the Fuel Index Price for each hour of the selected market day and is used to calculate the reference levels for that hour in the associated Reference schedule. The display frame of this section is vertically adjustable (by selecting the upper border and dragging it up or down). The row for a given market hour will be empty if there are no FPA requests that span the hour. Otherwise the following information is shown in each column of the grid:

- **Hour** The hour-ending value of market hour. The extra hour of the Fall Daylight Saving Time transition day will be shown as "02*". On the transition day of the Spring Daylight Saving Time, hour-ending 03 will not be shown.
- **FPA #** The ordinal number of the FPA request listed in the section above that establishes the effective fuel price for the hour. If the FPA is actually withdrawn for the hour and there is no FPA subsequently submitted to cover the hour, the FPA ordinal is shown, the FPA status is shown as withdrawn and the effective fuel price is shown as "Index", meaning the published Fuel Index Price.
- Effective Price The effective price of the FPA request for the hour. This price may be different than the effective price for the FPA shown above, in particular if the FPA is capped or withdrawn. Since the Fuel Price Adjustment Limit can be different for each hour spanned by the FPA, the FPA price can be different from one hour to the next. It can be "Index" even though the FPA is not withdrawn as explained above.
- **Status** (of the referenced FPA request) This is the FPA status for the hour: Approved / Capped / Withdrawn. Since the Fuel Price Adjustment Limit can be different for each hour spanned by the FPA, the FPA status can be different from one hour to the next.

4.13.1 Validation rules

When you select the Submit button, eMarket makes the following checks to ensure that the entered FPA request is valid:

- The selected market day date must be today or a date up to seven days in the future.
- The start hour of the FPA request, "From HE", must be greater than or equal to the hour-ending value of the current hour plus 30 minutes when the selected market day is today. In other words, an FPA request with an hour-ending value of the next hour of the current day as the start must be submitted at least 30 minutes before the start of that hour.
- The end hour of the FPA request, "To HE", must be greater than or equal to the hour-ending value of the start hour.
- The FPA requested price must be less than \$1000.

 The FPA requested price must be greater than the most recent published Fuel Index Price of the fuel plus a threshold established by the ISO-NE Market Monitor¹⁹ for all hours spanned by the FPA request.

4.13.2 View Specific Fuel Price Adjustment Information

- 1. In the Generation Navigation sub tabs, click the Fuel Price Adjustments tab.
 - The Fuel Price Adjustments display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired unit from the **Unit** filter; note that only units configured to use a fossil fuel (coal, oil or gas) will be listed in the drop-down list of units.
 - d. Select the desired fuel from the **Fuel** filter; note that only the <u>fuel index</u> names of each fuel that the unit is registered with IMM to use are listed in the drop-down list.
 - The display is refreshed and shows information on all FPA requests that have been submitted for the selected date, unit and fuel, if there are any

4.13.3 Submit a Fuel Price Adjustment Request

- 1. View the desired unit's FPA information as described in §4.13.2
- 2. Enter all of the following information in the FPA Submittal section:
 - a. From HE Select a value between 01 and 24; on the Fall Daylight Saving Time transition day, the extra hour will be listed as 02*; on the Spring Daylight Saving Time transition day, hour-ending 03 will not be listed.
 - b. To HE Select a value between 01 and 24; on the Fall Daylight Saving Time transition day, the extra hour will be listed as 02*; on the Spring Daylight Saving Time transition day, hour-ending 03 will not be listed.
 - c. Price Enter a value for the requested price
- 3. Click the **Submit** button to save the data.

Your FPA request will be listed in the section below.

4.13.4 Withdraw a Fuel Price Adjustment Request

- 1. View the desired unit's FPA information as described in §4.13.2
- 2. In the Submitted FPA section of the display, select the FPA request that you want to withdraw.
- 3. Click the **Withdraw** button to withdraw the FPA; the button will be enabled only if the FPA can still be withdrawn.

¹⁹ Refer to Market Rule 1, Appendix A, Section III.A.3.4

The FPA status changes to WithdrawnPending and some time later, it changes again to Withdrawn.

4.13.5 Submit Document Attachments to a Fuel Price Adjustment Request

- 1. View the desired unit's FPA information as described in §4.13.2
- 2. In the Submitted FPA section of the display, select the FPA request to which you want to attach documents.
- Click the View Attachments button. A pop-up display will be presented. Select the Add button and then browse to find the file that you want to upload and attach to the FPA. Select the OK button to upload the file.

The selected file is listed on the pop-up display. You can repeat this operation to upload additional documents. Simply click the *Close* button when you are done because the eMarket user interface cannot otherwise be used while the popup is open.

4.13.6 Remove Document Attachments of a Fuel Price Adjustment Request

- 1. View the desired unit's FPA information as described in §4.13.2
- 2. In the Submitted FPA section of the display, select the FPA request from which you want to remove document attachments.
- 3. Click the **View Attachments** button. A pop-up display will be presented listing the documents currently attached to the FPA request. Select the document that you want to remove and then select the **Remove** button.

The selected document is no longer listed on the pop-up display. You can repeat this operation to remove additional documents. Simply click the *Close* button when you are done because the eMarket user interface cannot otherwise be used while the popup is open.

4.14 Generation by Portfolio Display

You can view the cleared results of the Day-Ahead Market for each generating unit listed in a selected portfolio using the Generation by Portfolio display that is accessible on one of the Generation sub tabs. The day-ahead generation information includes the name of the schedule used when committing the unit. That will be the schedule that was designated to be available for the DA market on the selected day. If the unit was mitigated, this will be the cost-based mitigation schedule 99. The other information is the MW amount that the unit is committed to provide along with the Locational Marginal Price (LMP) at the generator node for each hour of the market day. If the unit was not committed, "(No Schedule Committed)" will be shown instead of the schedule name. Information for a 24-hour period is shown on two lines: the top line details the hours ending 01 through 12; the second line details hours ending 13 through 24. The last row of the display is labeled **Total Portfolio MW** and shows the total MW amount for each hour committed in the Day-Ahead market by all the units in your portfolio. Information on the display is shown only after the Day-Ahead market has been cleared.

IC	new	england																<i>larke</i> :: 2.0.1(201		r Inter	face
5	i new	engianu												OSED		Cert	tificate U	lser: Superu	ser	23-Aug-20	14 14:47
												Read	Only Particip	ant: IS	O New England	Inc			💙 User:	Superuser	
blic 🚺	Generation C	Demand A	RD Re	egulation																	
vate	Unit Default	Portfolio	Ramp Ra	ate Ran	np Rate	Schedule	Schedu	de So	thedule	Schedule Of	fers Sche	dule Times	Schedule	Hourly	Fuel Price	Genera	tion	XML			
ssages	Parameters	Manager	Default	Hou	urly Upda	es Manager	Detail D	efaults O	ffers Default	Hourly Upda	ites Hou	ly Updates	Selection	Updates	Adjustments	By Por	tfolio	Download			
ana	ration By	Portfoli	`																		
	•		,																		
ect Date	and Portfolio					nit By Portfoli	o for ZZZ or	n 24-Aug-20	014												
Dat	e: 24-Aug-201-	4				lame															
Portfoli	0: ZZZ					JNIT1 NAME Unit 1 Schedul	88)	01 / 13 155@	02 / 14	03 / 15 155@	04716 155@	05 / 17 155@	06 / 18 155@	07 / 19 155@	08/20 226@	09/21 225@	10 / 22 224@		12/24 221@		
)	\$19.91		\$18.06	\$16.90	\$18.86	\$19.72	\$22.35	\$22.46	\$22.48	\$25.32		\$31.06		
								221@	221@	220@	220@	220@	221@	222@	224@	226@	226@		155@		
								\$31.68	\$36.24	\$38.62	\$40.95	\$41.00	\$39.94	\$35.43	\$33.47	\$31.44	\$23.74	\$22.48	\$21.33		
						JNIT2 NAME		01/13	02 / 14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	2 11/23	12 / 24		
					(Unit 2 Schedul	88)	155@	155@	155@	155@	155@	155@	155@	226@	225@	224@	-	221@		
								\$19.91		\$18.06	\$16.90	\$18.86	\$19.72	\$22.35	\$22.46	\$22.46	\$25.32		\$31.06		
								221@	221@	220@	220@	220@	221@	222@	224@	226@	226@	-	155@		
						JNIT3 NAME		\$31.68		\$38.62	\$40.95	\$41.00	\$39.94	\$35.43	\$33.47	\$31.44	\$23.74		\$21.33		
						JNITS NAME Unit 3 Schedul		01 / 13 150@		03 / 15 150@	04 / 16 150 @	05 / 17 150@	06 / 18 150@	07 / 19 150@	08720 150@	09721 150@	10 / 22 190@		12/24 230@	-	
						onit o ochedan		\$19.96		\$18.10	\$16.93	\$18.90	\$19.77	\$22.42	\$22.53	\$22.55	\$25.46		\$31.27		
								237@	237@	236@	235@	236@	237@	238@	240@	240@	154.20		150@		
								\$31.89	\$36.47	\$38.88	\$41.50	\$41.25	\$40.18	\$35.65	\$33.68	\$31.63	\$23.87		\$21.39		
						JNIT4 NAME	_	01/13	02 / 14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	2 11/23	12 / 24		
					(Unit 4 Schedul	88)	150@	150@	150@	150@	150@	150@	150@	238@	236@	235@	233@	232@		
								\$19.96	\$18.12	\$18.10	\$16.93	\$18.90	\$19.77	\$22.42	\$22.53	\$22.55	\$25.46	\$28.27	\$31.27		
Tot	tals sho	wn or						231@	231@	230@	229@	230@	231@	232@	235@	237@	237@		150@		
-		-		~				\$31.89		\$38.88	\$41.50	\$41.25	\$40.18	\$35.65	\$33.68	\$31.63	\$23.87		\$21.39		
	t rows o	Ji the				lotal Portfolio N	NV.	01 / 13	02 / 14 610	03 / 15 610	04 / 16 610	05 / 17	06 / 18 610	07 / 19	08 / 20 840	09/21	10 / 22 873	2 11/23	12/24		
dis	play							910	910	906	904	610 906	910	610 914	923	836 929	8/3		904 610		
								910	910	900	804	800	910	014	92.0	47.4	040.2	0.42	010		

The format of this display is shown below:

The following generation information is displayed for each generating unit included in the selected generation portfolio:

- **Unit/Schedule** the name of the generating unit and of the schedule on which it was committed. If the unit was not committed, "No Schedule Committed" is shown instead of a schedule name.
- **MW** @ **Price** the MW amount that unit is committed to provide and the \$/MWH DA LMP at the generator Pnode for each hour of the day.

4.14.1 View DA Commitment for each Unit in Portfolio

To view the commitment in the Day-Ahead market of the units listed in your portfolios and the Day-Ahead LMPs at their generator Phodes:

- 1. In the Generation Navigation sub tabs, click the **Generation by Portfolio** tab
 - The Generation by Portfolio display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - The display reflects the selected market day's day-ahead commitment and LMPs for each of the units listed in the selected portfolio. If the Day-Ahead market has not yet been cleared for the selected day, the following text is shown in the display title bar: "The day ahead market results are not available ..."

4.15 XML Download for Generation Display

The XML Download display is accessible through the Generation Navigation sub tab. This display enables you to download pre-defined Generation specific XML report files of data that you maintain in the ISO-NE Market System, including results from the cleared Day-Ahead market. It also provides you the means to upload data in XML formatted data files into the ISO-NE Market System to be used to clear the Day-Ahead and Real Time markets. The format of the XML files can be found in the **eMarket Data Exchange Specification**, which is found on the <u>ISO New England website</u> among the documents listed on the Participate > Support > Upload and Download File Format Protocols page.

ISO new england		Market Status: CLOSED Participanti ISO New England Inc	eMarket User Interface Version: 2.0.1(201407221119) Certificate User: Superuser 23-Aug-2014 11:42:48 Viser: Superuser V
Public Generation Demand ARD Regulation			
Private Unit Default Portfolio Ramp Rate Ramp R Messages Parameters Manager Default Hourly I		ty Fuel Price Generation XML Adjustments By Portfolio Download	
XML Download			
Select Date	XML Download on 23-Aug-2014		
Date: 23-Aug-2014	Units	Upload XML SOAP Request	
	Unit Ramp Rate	File: Browse	
	Schedules	Lines Lines	
	Schedule Details	Submit	
	Schedule Offers Schedule Selection		
	Unit Hourly Updates		
	Unit Hourly Ramp Rate Updates		
	Schedule Offers Hourly Updates		
	Schedule Times Hourly Updates		
	Fuel Price Adjustments		
	Effective Hourly Fuel Prices of Units with FPAs		
	Generation by Unit		
Messages			(*)

The following XML reports of generation data are available for downloading:

- Units
- Unit Ramp Rates
- Unit Schedules
- Unit Schedule Details
- Unit Schedule Offers
- Unit Schedule Selection
- Unit Hourly Updates
- Unit Hourly Ramp Rate Updates
- Schedule Offers Hourly Updates
- Schedule Times Hourly Updates
- Fuel Price Adjustments
- Effective Hourly Fuel Prices of Units with FPAs
- Generation by Unit

All but the Generation by Unit returns an XML formatted report for the selected date and for every unit that belongs to you as Lead Participant. The Generation by Unit report returns information for all the unique units listed in all of the user's portfolios.

See §3.2 for details on uploading and downloading files. The following sections provide samples of the downloaded reports.

4.15.1 Units XML Report

This report contains the information that is shown on the Unit Default Parameters display for every unit that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetUnitResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <Unit day="2014-08-24" ID="888">
    <CapacityResource>true</CapacityResource>
    <Claim10Settings claim10Capable="true">
      <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
      <Claim10Cap>0</Claim10Cap>
    </Claim10Settings>
    <Claim30Settings claim30Capable="true">
      <Claim30>0</Claim30>
      <Claim30Cap>0</Claim30Cap>
    </Claim30Settings>
    <DefaultStatus>Economic</DefaultStatus>
    <LeadParticipantID>88888</LeadParticipantID>
    <LeadParticipantName>Participant R Us</LeadParticipantName>
    <NodeID>888</NodeID>
    <NodeName>UN.NODE 13.8NODE</NodeName>
    <RampRateCap>2</RampRateCap>
    <RampRateDefault>0.5</RampRateDefault>
    <RegResource>false</RegResource>
    <FuelIndexList>
      <FuelIndex>No2</FuelIndex>
      <FuelIndex>MA natgas2</FuelIndex>
    </FuelIndexList>
    <TypeOfUnit>CC</TypeOfUnit>
    <TypeOfUnitLongName>CT Combined Cycle</TypeOfUnitLongName>
    <UnitLongName>IAMUNIT</UnitLongName>
    <UnitNumber>88</UnitNumber>
    <UnitShortName>UNIT</UnitShortName>
  </Unit>
  <Unit day="2014-08-24" ID="999">
    <CapacityResource>true</CapacityResource>
    <Claim10Settings claim10Capable="true">
      <Claim10>11.3</Claim10>
      <Claim10Cap>8.6</Claim10Cap>
    </Claim10Settings>
    <Claim30Settings claim30Capable="true">
      <Claim30>11.8</Claim30>
      <Claim30Cap>8.6</Claim30Cap>
    </Claim30Settings>
    <DefaultStatus>Economic</DefaultStatus>
    <LeadParticipantID>99999</LeadParticipantID>
    <LeadParticipantName>Participant R Us 2</LeadParticipantName>
    <NodeID>999</NodeID>
    <NodeName>UN.NODENODE99 NODE</NodeName>
    <RampRateCap>1.5</RampRateCap>
    <RampRateDefault>1.5</RampRateDefault>
    <RegResource>false</RegResource>
    <FuelIndexList>
```

```
<FuelIndex>No2</FuelIndex>
</FuelIndexList>
<TypeOfUnit>CT</TypeOfUnit>
<TypeOfUnitLongName>Single Engine Jet</TypeOfUnitLongName>
<UnitLongName>IAMUNIT2</UnitLongName>
<UnitNumber>99</UnitNumber>
<UnitShortName>UNIX</UnitShortName>
</Unit>
</GetUnitResponse>
```

4.15.2 Unit Ramp Rates XML Report

This report contains the information that is shown on the Unit Ramp Rates display for every unit that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetUnitRampRateResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
 <UnitRampRate day="2014-08-24" ID="888">
   <RampRateCap>5</RampRateCap>
   <RampRateCurve>
     <RampRatePoint rate="5" MW="170"/>
   </RampRateCurve>
   <ShortName>UNIT</ShortName>
 </UnitRampRate>
  <UnitRampRate day="2014-08-24" ID="999">
   <RampRateCap>2.5</RampRateCap>
   <RampRateCurve>
     <RampRatePoint rate="1.5" MW="150"/>
     <RampRatePoint rate="2.5" MW="216"/>
     <RampRatePoint rate="1" MW="226"/>
   </RampRateCurve>
   <ShortName>UNIX</ShortName>
 </UnitRampRate>
</GetUnitRampRateResponse>
```

4.15.3 Unit Schedules XML Report

This report contains the information that is shown on the Unit Schedule Manager display for all schedules of every unit that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetScheduleResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <Schedule schedID="88805">
   <SchedLongName>UNIT Price</SchedLongName>
   <SchedShortName>UNITPric</SchedShortName>
   <SchedType>Price</SchedType>
   <SchedTypeID>5</SchedTypeID>
   <UnitID>888</UnitID>
   <UnitShortName>UNIT</UnitShortName>
 </Schedule>
  <Schedule schedID="88896">
   <SchedLongName>UNIT Fuel 3 Ref. Schedule</SchedLongName>
   <SchedShortName>UNIT R3</SchedShortName>
   <SchedType>Cost</SchedType>
   <SchedTypeID>96</SchedTypeID>
   <UnitID>888</UnitID>
```
```
<UnitShortName>UNIT</UnitShortName>
  </Schedule>
  <Schedule schedID="88897">
   <SchedLongName>UNIT Fuel 2 Ref. Schedule</SchedLongName>
   <SchedShortName>UNIT R2</SchedShortName>
   <SchedType>Cost</SchedType>
   <SchedTypeID>97</SchedTypeID>
   <UnitID>888</UnitID>
   <UnitShortName>UNIT</UnitShortName>
  </Schedule>
  <Schedule schedID="88898">
   <SchedLongName>UNIT Fuel 1 Ref. Schedule</SchedLongName>
   <SchedShortName>UNIT_R1</SchedShortName>
   <SchedType>Cost</SchedType>
   <SchedTypeID>98</SchedTypeID>
   <UnitID>888</UnitID>
   <UnitShortName>UNIT</UnitShortName>
  </Schedule>
  <Schedule schedID="88899">
   <SchedLongName>UNIT_Mitigation Schedule</SchedLongName>
   <SchedShortName>UNIT M</SchedShortName>
   <SchedType>Cost</SchedType>
   <SchedTypeID>99</SchedTypeID>
   <UnitID>888</UnitID>
   <UnitShortName>UNIT</UnitShortName>
 </Schedule>
</GetScheduleResponse>
```

4.15.4 Unit Schedule Details XML Report

This report contains the information that is shown on the Unit Schedule Detail display for all schedules of every unit that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetScheduleDetailResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <ScheduleDetail day="2014-08-24" ID="88805">
    <ColdNotifyTime>3.5</ColdNotifyTime>
    <ColdStartupTime>4</ColdStartupTime>
    <CondenseAvailable>false</CondenseAvailable>
    <CondenseHourlyCost>0</CondenseHourlyCost>
    <CondenseNotifyTime>0</CondenseNotifyTime>
    <CondensePower>0</CondensePower>
    <CondenseStartupCost>0</CondenseStartupCost>
    <EcoMax>153</EcoMax>
    <EcoMin>110</EcoMin>
    <EmerMin>110</EmerMin>
    <HotNotifyTime>3</HotNotifyTime>
    <HotStartupTime>1.5</HotStartupTime>
    <HotToColdTime>48</HotToColdTime>
    <HotToInterTime>24</HotToInterTime>
    <InterNotifyTime>3</InterNotifyTime>
    <InterStartupTime>2.5</InterStartupTime>
    <MarketType>Both</MarketType>
    <MaxDailyEnergy xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
    <MaxDailyStarts>1</MaxDailyStarts>
    <MinDownTime>5.5</MinDownTime>
    <MinRunTime>4</MinRunTime>
```

<RTHighOperLimit>153</RTHighOperLimit> <ScheduleAvailable>true</ScheduleAvailable> <ShortName>UNITPric</ShortName> </ScheduleDetail> <ScheduleDetail day="2014-08-24" ID="88896"> <ColdNotifyTime>1.5</ColdNotifyTime> <ColdStartupTime>4</ColdStartupTime> <CondenseAvailable>false</CondenseAvailable> <CondenseHourlyCost xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <CondenseNotifyTime xmIns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <CondensePower xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <CondenseStartupCost xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <EcoMax>153</EcoMax> <EcoMin>110</EcoMin> <EmerMin>110</EmerMin> <HotNotifyTime>1</HotNotifyTime> <HotStartupTime>1.5</HotStartupTime> <HotToColdTime>48</HotToColdTime> <HotToInterTime>24</HotToInterTime> <InterNotifyTime>1</InterNotifyTime> <InterStartupTime>2.5</InterStartupTime> <MarketType>Both</MarketType> <MaxDailyEnergy xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MaxDailyStarts xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MinDownTime>5.5</MinDownTime> <MinRunTime>4</MinRunTime> <RTHighOperLimit>153</RTHighOperLimit> <ScheduleAvailable>false</ScheduleAvailable> <ShortName>UNIT R3</ShortName> </ScheduleDetail> <ScheduleDetail day="2014-08-24" ID="88897"> <ColdNotifyTime>1.5</ColdNotifyTime> <ColdStartupTime>4</ColdStartupTime> <CondenseAvailable>false</CondenseAvailable> <CondenseHourlyCost xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <CondenseNotifyTime xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <CondensePower xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <CondenseStartupCost xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <EcoMax>153</EcoMax> <EcoMin>110</EcoMin> <EmerMin>110</EmerMin> <HotNotifyTime>1</HotNotifyTime> <HotStartupTime>1.5</HotStartupTime> <HotToColdTime>48</HotToColdTime> <HotToInterTime>24</HotToInterTime> <InterNotifyTime>1</InterNotifyTime> <InterStartupTime>2.5</InterStartupTime> <MarketType>Both</MarketType> <MaxDailyEnergy xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MaxDailyStarts xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MinDownTime>5.5</MinDownTime> <MinRunTime>4</MinRunTime> <RTHighOperLimit>153</RTHighOperLimit> <ScheduleAvailable>false</ScheduleAvailable> <ShortName>UNIT_R2</ShortName> </ScheduleDetail> <ScheduleDetail day="2014-08-24" ID="88898">

<ColdNotifyTime>1.5</ColdNotifyTime>

<ColdStartupTime>4</ColdStartupTime> <CondenseAvailable>false</CondenseAvailable> <CondenseHourlyCost xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <CondenseNotifyTime xmIns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <CondensePower xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <CondenseStartupCost xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <EcoMax>153</EcoMax> <EcoMin>110</EcoMin> <EmerMin>110</EmerMin> <HotNotifyTime>1</HotNotifyTime> <HotStartupTime>1.5</HotStartupTime> <HotToColdTime>48</HotToColdTime> <HotToInterTime>24</HotToInterTime> <InterNotifyTime>1</InterNotifyTime> <InterStartupTime>2.5</InterStartupTime> <MarketType>Both</MarketType> <MaxDailyEnergy xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MaxDailyStarts xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MinDownTime>5.5</MinDownTime> <MinRunTime>4</MinRunTime> <RTHighOperLimit>153</RTHighOperLimit> <ScheduleAvailable>false</ScheduleAvailable> <ShortName>UNIT R1</ShortName> </ScheduleDetail> <ScheduleDetail day="2014-08-24" ID="88899"> <ColdNotifyTime>1.5</ColdNotifyTime> <ColdStartupTime>4</ColdStartupTime> <CondenseAvailable>false</CondenseAvailable> <CondenseHourlyCost xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <CondenseNotifyTime xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <CondensePower xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <CondenseStartupCost xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <EcoMax>153</EcoMax> <EcoMin>110</EcoMin> <EmerMin>110</EmerMin> <HotNotifyTime>1</HotNotifyTime> <HotStartupTime>1.5</HotStartupTime> <HotToColdTime>48</HotToColdTime> <HotToInterTime>24</HotToInterTime> <InterNotifyTime>1</InterNotifyTime> <InterStartupTime>2.5</InterStartupTime> <MarketType>Both</MarketType> <MaxDailyEnergy xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MaxDailyStarts>1</MaxDailyStarts> <MinDownTime>5.5</MinDownTime> <MinRunTime>4</MinRunTime> <RTHighOperLimit>153</RTHighOperLimit> <ScheduleAvailable>true</ScheduleAvailable> <ShortName>UNIT M</ShortName> </scheduleDetail>

</GetScheduleDetailResponse>

4.15.5 Unit Schedule Offers XML Report

This report contains the information that is shown on the Unit Schedule Offers display for all schedules of every unit that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetScheduleBidResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <ScheduleBid day="2014-08-24" ID="88801">
    <PriceCurve>
      <PricePoint price="888.99" MW="8.5">
        <FuelIndex>
         <Name>Hydro_Run_of_River</Name>
         <Percentage>100</Percentage>
        </FuelIndex>
      </PricePoint>
    </PriceCurve>
    <UseOfferSlope>false</UseOfferSlope>
    <HotStartupCostAndFuel>
      <Cost>0</Cost>
      <FuelIndex>
        <Name>Hydro_Run_of_River</Name>
       <Percentage>100</Percentage>
      </FuelIndex>
    </HotStartupCostAndFuel>
    <InterStartupCostAndFuel>
      <Cost>0</Cost>
      <FuelIndex>
        <Name>Hydro Run of River</Name>
        <Percentage>100</Percentage>
      </FuelIndex>
    </InterStartupCostAndFuel>
    <ColdStartupCostAndFuel>
      <Cost>0</Cost>
      <FuelIndex>
        <Name>Hydro_Run_of_River</Name>
       <Percentage>100</Percentage>
      </FuelIndex>
    </ColdStartupCostAndFuel>
    <NoLoadCostAndFuel>
      <Cost>0</Cost>
      <FuelIndex>
        <Name>Hydro_Run_of_River</Name>
        <Percentage>100</Percentage>
      </FuelIndex>
    </NoLoadCostAndFuel>
    <ShortName>DFLPRICE</ShortName>
  </ScheduleBid>
<ScheduleBid day="2014-08-24" ID="99905">
    <PriceCurve>
      <PricePoint price="99.69" MW="35">
        <FuelIndex>
         <Name>Coal HS</Name>
         <Percentage>100</Percentage>
        </FuelIndex>
        <FuelIndex>
         <Name>MA_natgas1</Name>
         <Percentage>0</Percentage>
```

</FuelIndex> </PricePoint> <PricePoint price="99.78" MW="57"> <FuelIndex> <Name>Coal_HS</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>MA_natgas1</Name> <Percentage>0</Percentage> </FuelIndex> </PricePoint> </PriceCurve> <UseOfferSlope>false</UseOfferSlope> <HotStartupCostAndFuel> <Cost>12345.16</Cost> <FuelIndex> <Name>Coal HS</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>MA_natgas1</Name> <Percentage>0</Percentage> </FuelIndex> </HotStartupCostAndFuel> <InterStartupCostAndFuel> <Cost>12345.67</Cost> <FuelIndex> <Name>Coal HS</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>MA_natgas1</Name> <Percentage>0</Percentage> </FuelIndex> </InterStartupCostAndFuel> <ColdStartupCostAndFuel> <Cost>11111.11</Cost> <FuelIndex> <Name>Coal_HS</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>MA_natgas1</Name> <Percentage>0</Percentage> </FuelIndex> </ColdStartupCostAndFuel> <NoLoadCostAndFuel> <Cost>1234.56</Cost> <FuelIndex> <Name>Coal_HS</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>MA_natgas1</Name> <Percentage>0</Percentage> </FuelIndex> </NoLoadCostAndFuel>

<ShortName>UNXPric</ShortName> </ScheduleBid </GetScheduleBidResponse>

4.15.6 Unit Schedule Selection XML Report

This report contains the information that is shown on the Unit Schedule Selection display for all schedules of every unit that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetScheduleSelectionResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <ScheduleSelection day="2014-08-24">
    <Schedule ID="88805">
      <Available>true</Available>
      <ShortName>UNITPric</ShortName>
    </Schedule>
    <Schedule ID="88896">
      <Available>false</Available>
      <ShortName>UNIT_R3</ShortName>
    </Schedule>
    <Schedule ID="88897">
      <Available>false</Available>
      <ShortName>UNIT R2</ShortName>
    </Schedule>
    <Schedule ID="88898">
      <Available>false</Available>
      <ShortName>UNIT_R1</ShortName>
    </Schedule>
    <Schedule ID="88899">
      <Available>true</Available>
      <ShortName>UNIT M</ShortName>
    </Schedule>
    <Schedule ID="99905">
      <Available>true</Available>
      <ShortName>UNIXPric</ShortName>
    </Schedule>
    <Schedule ID="99996">
      <Available>false</Available>
      <ShortName>UNIX_R3</ShortName>
    </Schedule>
    <Schedule ID="99997">
      <Available>false</Available>
      <ShortName>UNIX R2</ShortName>
    </Schedule>
    <Schedule ID="99998">
      <Available>false</Available>
      <ShortName>UNIX_R1</ShortName>
    </Schedule>
    <Schedule ID="99999">
      <Available>true</Available>
      <ShortName>UNIX_M</ShortName>
    </Schedule>
  </ScheduleSelection>
</GetScheduleSelectionResponse>
```

4.15.7 Unit Hourly Updates XML Report

This report contains the information that is shown on the Unit Hourly Updates display for every unit that belongs to you as Lead Participant, and for which hourly updates exist. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetUnitHourlyResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <UnitHourly day="2014-08-24" ID="666">
    <HourlyProfile>
      <HourlyValues time="2014-08-24T00:00:00-04:00">
       <EmerMin>9</EmerMin>
        <EcoMax>9</EcoMax>
        <EcoMin>9</EcoMin>
        <RTHighOperLimit>9</RTHighOperLimit>
        <ResourceStatus>MustRun</ResourceStatus>
        <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
       <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
      </HourlyValues>
      <HourlyValues time="2014-08-24T01:00:00-04:00">
       <EmerMin>9</EmerMin>
       <EcoMax>9</EcoMax>
       <EcoMin>9</EcoMin>
       <RTHighOperLimit>9</RTHighOperLimit>
        <ResourceStatus>MustRun</ResourceStatus>
        <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
      </HourlyValues>
      <HourlyValues time="2014-08-24T02:00:00-04:00">
        <EmerMin>9</EmerMin>
        <EcoMax>9</EcoMax>
        <EcoMin>9</EcoMin>
        <RTHighOperLimit>9</RTHighOperLimit>
       <ResourceStatus>MustRun</ResourceStatus>
        <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
      </HourlyValues>
      <HourlyValues time="2014-08-24T22:00:00-04:00">
        <EmerMin>9</EmerMin>
       <EcoMax>9</EcoMax>
        <EcoMin>9</EcoMin>
        <RTHighOperLimit>9</RTHighOperLimit>
        <ResourceStatus>MustRun</ResourceStatus>
        <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
      </HourlyValues>
      <HourlyValues time="2014-08-24T23:00:00-04:00">
       <EmerMin>9</EmerMin>
        <EcoMax>9</EcoMax>
       <EcoMin>9</EcoMin>
       <RTHighOperLimit>9</RTHighOperLimit>
        <ResourceStatus>MustRun</ResourceStatus>
        <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
      </HourlyValues>
    </HourlyProfile>
```

```
<ShortName>UNIT</ShortName>
  </UnitHourly>
  <UnitHourly day="2014-08-24" ID="777">
    <HourlyProfile>
      <HourlyValues time="2014-08-24T00:00:00-04:00">
        <EmerMin xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <EcoMax xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <EcoMin xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <RTHighOperLimit xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <ResourceStatus xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
      </HourlyValues>
      <HourlyValues time="2014-08-24T01:00:00-04:00">
        <EmerMin>11</EmerMin>
        <EcoMax>32</EcoMax>
        <EcoMin>22.6</EcoMin>
        <RTHighOperLimit>34</RTHighOperLimit>
        <ResourceStatus>Economic</ResourceStatus>
        <Claim10>15.2</Claim10>
        <Claim30>25</Claim30>
      </HourlyValues>
      <HourlyValues time="2014-08-24T02:00:00-04:00">
        <EmerMin>11</EmerMin>
        <EcoMax>32</EcoMax>
        <EcoMin>22.6</EcoMin>
        <RTHighOperLimit>34</RTHighOperLimit>
        <ResourceStatus>Economic</ResourceStatus>
        <Claim10>15.2</Claim10>
        <Claim30>25</Claim30>
      </HourlyValues>
      <HourlyValues time="2014-08-24T22:00:00-04:00">
        <EmerMin>11</EmerMin>
        <EcoMax>32</EcoMax>
        <EcoMin>22.6</EcoMin>
        <RTHighOperLimit>34</RTHighOperLimit>
        <ResourceStatus>Economic</ResourceStatus>
        <Claim10>15.2</Claim10>
        <Claim30>25</Claim30>
      </HourlyValues>
      <HourlyValues time="2014-08-24T23:00:00-04:00">
        <EmerMin xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <EcoMax xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <EcoMin xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <RTHighOperLimit xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <ResourceStatus xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
      </HourlyValues>
    </HourlyProfile>
    <ShortName>UNIX</ShortName>
  </UnitHourly>
</GetUnitHourlyResponse>
```

4.15.8 Unit Hourly Ramp Rate Updates XML Report

This report contains the information that is shown on the Unit Ramp Rate Hourly Updates display for every unit that belongs to you as Lead Participant, and for which hourly updates exist. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetUnitRampRateHourlyResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <UnitRampRateHourly day="2014-08-24" ID="666">
    <HourlyProfile>
      <HourlyValues time="2014-08-24T00:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T01:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T02:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T03:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T04:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T05:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T06:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T07:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T08:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T09:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T10:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T11:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T12:00:00-04:00">
       <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T13:00:00-04:00">
       <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T14:00:00-04:00">
        <RampRateCurve>
         <RampRatePoint rate="1.6" MW="50.1"/>
         <RampRatePoint rate="2.6" MW="90.1"/>
         <RampRatePoint rate="3.6" MW="120.1"/>
        </RampRateCurve>
```

```
</HourlyValues>
      <HourlyValues time="2014-08-24T15:00:00-04:00">
        <RampRateCurve>
         <RampRatePoint rate="1.6" MW="50.1"/>
         <RampRatePoint rate="2.6" MW="90.1"/>
         <RampRatePoint rate="3.6" MW="120.1"/>
        </RampRateCurve>
      </HourlyValues>
      <HourlyValues time="2014-08-24T16:00:00-04:00">
       <RampRateCurve>
         <RampRatePoint rate="1.6" MW="50.1"/>
         <RampRatePoint rate="2.6" MW="90.1"/>
         <RampRatePoint rate="3.6" MW="120.1"/>
        </RampRateCurve>
      </HourlyValues>
      <HourlyValues time="2014-08-24T17:00:00-04:00">
        <RampRateCurve>
         <RampRatePoint rate="1.6" MW="50.1"/>
         <RampRatePoint rate="2.6" MW="90.1"/>
         <RampRatePoint rate="3.6" MW="120.1"/>
        </RampRateCurve>
      </HourlyValues>
      <HourlyValues time="2014-08-24T18:00:00-04:00">
        <RampRateCurve>
         <RampRatePoint rate="4.1" MW="118.1"/>
       </RampRateCurve>
      </HourlyValues>
      <HourlyValues time="2014-08-24T19:00:00-04:00">
        <RampRateCurve>
         <RampRatePoint rate="1.6" MW="50.1"/>
         <RampRatePoint rate="2.6" MW="90.1"/>
         <RampRatePoint rate="3.6" MW="120.1"/>
       </RampRateCurve>
      </HourlyValues>
      <HourlyValues time="2014-08-24T20:00:00-04:00">
        <RampRateCurve>
         <RampRatePoint rate="1.6" MW="50.1"/>
         <RampRatePoint rate="2.6" MW="90.1"/>
         <RampRatePoint rate="3.6" MW="120.1"/>
       </RampRateCurve>
      </HourlyValues>
      <HourlyValues time="2014-08-24T21:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T22:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
      <HourlyValues time="2014-08-24T23:00:00-04:00">
        <FixedRampRate>4.9</FixedRampRate>
      </HourlyValues>
    </HourlyProfile>
    <ShortName>UNIT</ShortName>
  </UnitRampRateHourly>
</GetUnitRampRateHourlyResponse>
```

4.15.9 Schedule Offers Hourly Updates XML Report

This report contains the information that is shown on the Unit Schedule Offers Hourly Updates display for every unit that belongs to you as Lead Participant, and for which hourly updates exist. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetScheduleBidHourlyResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <ScheduleBidHourly day="2014-08-24" ID="66697">
    <HourlyProfile>
      <HourlyValues time="2014-08-24T00:00:00-04:00">
        <PriceCurve>
         <PricePoint price="183.71" MW="55">
           <FuelIndex>
              <Name>MA_natgas2</Name>
             <Percentage>0</Percentage>
           </FuelIndex>
           <FuelIndex>
             <Name>No2</Name>
             <Percentage>100</Percentage>
           </FuelIndex>
         </PricePoint>
         <PricePoint price="210.18" MW="60">
           <FuelIndex>
              <Name>MA_natgas2</Name>
              <Percentage>0</Percentage>
           </FuelIndex>
            <FuelIndex>
              <Name>No2</Name>
             <Percentage>100</Percentage>
           </FuelIndex>
         </PricePoint>
        </PriceCurve>
       <UseOfferSlope>true</UseOfferSlope>
        <HotStartupCostAndFuel>
         <Cost> 9999.12 </Cost>
         <FuelIndex>
           <Name>MA_natgas2</Name>
           <Percentage>0</Percentage>
         </FuelIndex>
         <FuelIndex>
           <Name>No2</Name>
           <Percentage>100</Percentage>
         </FuelIndex>
        </HotStartupCostAndFuel>
        <InterStartupCostAndFuel>
         <Cost>14999.12</Cost>
         <FuelIndex>
           <Name>MA_natgas2</Name>
           <Percentage>0</Percentage>
         </FuelIndex>
         <FuelIndex>
           <Name>No2</Name>
           <Percentage>100</Percentage>
         </FuelIndex>
        </InterStartupCostAndFuel>
       <ColdStartupCostAndFuel>
         <Cost>99999.12</Cost>
```

<FuelIndex> <Name>MA_natgas2</Name> <Percentage>0</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>100</Percentage> </FuelIndex> </ColdStartupCostAndFuel> <NoLoadCostAndFuel> <Cost>5047.0</Cost> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>0</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>100</Percentage> </FuelIndex> </NoLoadCostAndFuel> </HourlyValues>

•

<HourlyValues time="2014-08-24T23:00:00-04:00"> <PriceCurve> <PricePoint price="183.71" MW="55"> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>0</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>100</Percentage> </FuelIndex> </PricePoint> <PricePoint price="210.18" MW="60"> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>0</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>100</Percentage> </FuelIndex> </PricePoint> </PriceCurve> <UseOfferSlope>true</UseOfferSlope> <HotStartupCostAndFuel> <Cost>9999.12 </Cost> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>0</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>100</Percentage> </FuelIndex> </HotStartupCostAndFuel>

<InterStartupCostAndFuel> <Cost>14999.12</Cost> <FuelIndex> <Name>MA natgas2</Name> <Percentage>0</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>100</Percentage> </FuelIndex> </InterStartupCostAndFuel> <ColdStartupCostAndFuel> <Cost>99999.12</Cost> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>0</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>100</Percentage> </FuelIndex> </ColdStartupCostAndFuel> <NoLoadCostAndFuel> <Cost>5047.0</Cost> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>0</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>100</Percentage> </FuelIndex> </NoLoadCostAndFuel> </HourlyValues> </HourlyProfile> <ShortName>UNIT_R2</ShortName> </ScheduleBidHourly> <ScheduleBidHourly day="2014-08-24" ID="66698"> <HourlyProfile> <HourlyValues time="2014-08-24T00:00:00-04:00"> <PriceCurve> <PricePoint price="83.71" MW="55"> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>0</Percentage> </FuelIndex> </PricePoint> <PricePoint price="110.18" MW="60"> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name>

<Percentage>0</Percentage> </FuelIndex> </PricePoint> </PriceCurve> <UseOfferSlope>true</UseOfferSlope> <HotStartupCostAndFuel> <Cost>6666.12 </Cost> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>0</Percentage> </FuelIndex> </HotStartupCostAndFuel> <InterStartupCostAndFuel> <Cost>11666.12</Cost> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>0</Percentage> </FuelIndex> </InterStartupCostAndFuel> <ColdStartupCostAndFuel> <Cost>166666.12</Cost> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>0</Percentage> </FuelIndex> </ColdStartupCostAndFuel> <NoLoadCostAndFuel> <Cost>2547.0</Cost> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>0</Percentage> </FuelIndex> </NoLoadCostAndFuel> </HourlyValues>

<HourlyValues time="2014-08-24T23:00:00-04:00"> <PriceCurve> <PricePoint price="83.71" MW="55"> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>100</Percentage>

</FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>0</Percentage> </FuelIndex> </PricePoint> <PricePoint price="110.18" MW="60"> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>0</Percentage> </FuelIndex> </PricePoint> </PriceCurve> <UseOfferSlope>true</UseOfferSlope> <HotStartupCostAndFuel> <Cost>6666.12 </Cost> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>0</Percentage> </FuelIndex> </HotStartupCostAndFuel> <InterStartupCostAndFuel> <Cost>11666.12</Cost> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>0</Percentage> </FuelIndex> </InterStartupCostAndFuel> <ColdStartupCostAndFuel> <Cost>166666.12</Cost> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex> <Name>No2</Name> <Percentage>0</Percentage> </FuelIndex> </ColdStartupCostAndFuel> <NoLoadCostAndFuel> <Cost>2547.0</Cost> <FuelIndex> <Name>MA_natgas2</Name> <Percentage>100</Percentage> </FuelIndex> <FuelIndex>

<Name>No2</Name> <Percentage>0</Percentage> </FuelIndex> </NoLoadCostAndFuel> </HourlyValues> </HourlyVrofile> <ShortName>UNIT_R1</ShortName> </ScheduleBidHourly>

4.15.10 Schedule Times Hourly Updates XML Report

This report contains the information that is shown on the Unit Schedule Times Hourly Updates display for every unit that belongs to you as Lead Participant, and for which hourly updates exist. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetScheduleTimesHourlyResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <ScheduleTimesHourly day="2014-08-24" ID="77797">
    <HourlyProfile>
      <HourlyValues time="2014-08-24T00:00:00-04:00">
        <HotStartupTime>1.5</HotStartupTime>
        <InterStartupTime>2.5</InterStartupTime>
        <ColdStartupTime>4</ColdStartupTime>
        <HotNotifyTime>1</HotNotifyTime>
        <InterNotifyTime>1</InterNotifyTime>
        <ColdNotifyTime>1.5</ColdNotifyTime>
      </HourlyValues>
      <HourlyValues time="2014-08-24T23:00:00-04:00">
        <HotStartupTime>1.5</HotStartupTime>
        <InterStartupTime>2.5</InterStartupTime>
        <ColdStartupTime>4</ColdStartupTime>
        <HotNotifyTime>1</HotNotifyTime>
        <InterNotifyTime>1</InterNotifyTime>
        <ColdNotifyTime>1.5</ColdNotifyTime>
      </HourlyValues>
    </HourlyProfile>
    <ShortName>UNIX_R2</ShortName>
  </ScheduleTimesHourly>
  <ScheduleTimesHourly day="2014-08-24" ID="77798">
    <HourlyProfile>
      <HourlyValues time="2014-08-24T00:00:00-04:00">
        <HotStartupTime>1.5</HotStartupTime>
        <InterStartupTime>2.5</InterStartupTime>
        <ColdStartupTime>4</ColdStartupTime>
        <HotNotifyTime>1</HotNotifyTime>
        <InterNotifyTime>1</InterNotifyTime>
        <ColdNotifyTime>1.5</ColdNotifyTime>
      </HourlyValues>
      <HourlyValues time="2014-08-24T23:00:00-04:00">
        <HotStartupTime>1.5</HotStartupTime>
        <InterStartupTime>2.5</InterStartupTime>
        <ColdStartupTime>4</ColdStartupTime>
        <HotNotifyTime>1</HotNotifyTime>
        <InterNotifyTime>1</InterNotifyTime>
        <ColdNotifyTime>1.5</ColdNotifyTime>
      </HourlyValues>
    </HourlyProfile>
    <ShortName>UNIX R1</ShortName>
  </ScheduleTimesHourly>
</GetScheduleTimesHourlyResponse>
```

4.15.11 Fuel Price Adjustments XML Report

This report contains the information that is shown in the Submitted FPA section of the Fuel Price Adjustments display for every unit that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetFuelPriceAdjustmentUnitResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <FuelPriceAdjustmentUnit day="2014-08-24" ID="666">
    <UnitLongName>UNITLongName</UnitLongName>
    <FuelPriceAdjustmentFuel>
      <FuelIndex>MA natgas2</FuelIndex>
      <FuelPriceAdjustment FPAID="466188495">
       <StartHour>2014-08-24T08:00:00-04:00</StartHour>
        <EndHour>2014-08-24T10:00:00-04:00</EndHour>
        <RequestedPrice>4.5</RequestedPrice>
       <EffectivePrice>4.05</EffectivePrice>
       <EffectivePriceIsIndex>false</EffectivePriceIsIndex>
        <SubmittedBy>GenUser</SubmittedBy>
       <SubmitAcceptedTime>2014-08-23T08:32:13-04:00</SubmitAcceptedTime>
        <WithdrawnBy xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <WithdrawnHour xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <WithdrawnAcceptedTime xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <Status>Capped</Status>
        <Attachments>
          <Attachment>
            <AttachmentID>466188624</AttachmentID>
            <FileName>error message.docx</FileName>
            <UploadedBy>GenUser</UploadedBy>
            <UploadedTime>2014-08-23T08:34:37-04:00</UploadedTime>
          </Attachment>
        </Attachments>
      </FuelPriceAdjustment>
      <FuelPriceAdjustment FPAID="466188795">
        <StartHour>2014-08-24T14:00:00-04:00</StartHour>
        <EndHour>2014-08-24T19:00:00-04:00</EndHour>
        <RequestedPrice>4.11</RequestedPrice>
       <EffectivePrice>4.11</EffectivePrice>
        <EffectivePriceIsIndex>false</EffectivePriceIsIndex>
        <SubmittedBy>GenUser</SubmittedBy>
        <SubmitAcceptedTime>2014-08-23T08:35:35-04:00</SubmitAcceptedTime>
        <WithdrawnBy xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
       <WithdrawnHour xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <WithdrawnAcceptedTime xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
       <Status>Approved</Status>
      </FuelPriceAdjustment>
      <FuelPriceAdjustment FPAID="466203107">
        <StartHour>2014-08-24T20:00:00-04:00</StartHour>
        <EndHour>2014-08-24T22:00:00-04:00</EndHour>
        <RequestedPrice>4.75</RequestedPrice>
       <EffectivePrice>4.75</EffectivePrice>
        <EffectivePriceIsIndex>false</EffectivePriceIsIndex>
        <SubmittedBy>ISO-NE IMM</SubmittedBy>
       <SubmitAcceptedTime>2014-08-23T13:33:47-04:00</SubmitAcceptedTime>
        <WithdrawnBy xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
       <WithdrawnHour xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
       <WithdrawnAcceptedTime xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <Status>Approved</Status>
```

```
</FuelPriceAdjustment>
    </FuelPriceAdjustmentFuel>
  </FuelPriceAdjustmentUnit>
  <FuelPriceAdjustmentUnit day="2014-08-24" ID="999">
    <UnitLongName>UNIT X</UnitLongName>
    <FuelPriceAdjustmentFuel>
      <FuelIndex>No6 070</FuelIndex>
      <FuelPriceAdjustment FPAID="466207278">
        <StartHour>2014-08-24T03:00:00-04:00</StartHour>
        <EndHour>2014-08-24T07:00:00-04:00</EndHour>
        <RequestedPrice>30</RequestedPrice>
        <EffectivePrice>16.64</EffectivePrice>
        <EffectivePriceIsIndex>false</EffectivePriceIsIndex>
        <SubmittedBy>GenUser</SubmittedBy>
        <SubmitAcceptedTime>2014-08-23T14:56:32-04:00</SubmitAcceptedTime>
        <WithdrawnBy xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <WithdrawnHour xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <WithdrawnAcceptedTime xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <Status>Capped</Status>
      </FuelPriceAdjustment>
    </FuelPriceAdjustmentFuel>
  </FuelPriceAdjustmentUnit>
</GetFuelPriceAdjustmentUnitResponse>
```

4.15.12 Effective Hourly Fuel Prices of Units with FPAs XML Report

This report contains the information that is shown in the Effective Fuel Price Adjustments section of the Fuel Price Adjustments display for every unit that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetFuelPriceAdjustmentUnitHourlyResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <FuelPriceAdjustmentUnitHourly day="2014-08-24" ID="999">
    <UnitLongName>UNITXLongName</UnitLongName>
    <FuelPriceAdjustmentFuelHourly>
      <FuelIndex>MA natgas2</FuelIndex>
      <HourlyProfile>
        <HourlyValues time="2014-08-24T08:00:00-04:00">
          <FPAID>466188495</FPAID>
          <EffectivePrice>41.5</EffectivePrice>
          <EffectivePriceIsIndex>false</EffectivePriceIsIndex>
          <Status>Approved</Status>
        </HourlyValues>
        <HourlyValues time="2014-08-24T09:00:00-04:00">
          <FPAID>466188495</FPAID>
          <EffectivePrice>41.5</EffectivePrice>
          <EffectivePriceIsIndex>false</EffectivePriceIsIndex>
          <Status>Approved</Status>
        </HourlyValues>
        <HourlyValues time="2014-08-24T10:00:00-04:00">
          <FPAID>466188495</FPAID>
          <EffectivePrice>41.5</EffectivePrice>
          <EffectivePriceIsIndex>false</EffectivePriceIsIndex>
          <Status>Approved</Status>
        </HourlyValues>
        <HourlyValues time="2014-08-24T11:00:00-04:00">
          <FPAID>466188495</FPAID>
          <EffectivePrice>40.5</EffectivePrice>
```

<EffectivePriceIsIndex>false</EffectivePriceIsIndex> <Status>Capped</Status> </HourlyValues> <HourlyValues time="2014-08-24T12:00:00-04:00"> <FPAID>466188495</FPAID> <EffectivePrice>40.5</EffectivePrice> <EffectivePriceIsIndex>false</EffectivePriceIsIndex> <Status>Capped</Status> </HourlyValues> <HourlyValues time="2014-08-24T13:00:00-04:00"> <FPAID>466188495</FPAID> <EffectivePrice>40.5</EffectivePrice> <EffectivePriceIsIndex>false</EffectivePriceIsIndex> <Status>Capped</Status> </HourlyValues> <HourlyValues time="2014-08-24T14:00:00-04:00"> <FPAID>466188795</FPAID> <EffectivePrice>40.5</EffectivePrice> <EffectivePriceIsIndex>false</EffectivePriceIsIndex> <Status>Capped</Status> </HourlyValues> <HourlyValues time="2014-08-24T15:00:00-04:00"> <FPAID>466188795</FPAID> <EffectivePrice>40.5</EffectivePrice> <EffectivePriceIsIndex>false</EffectivePriceIsIndex> <Status>Capped</Status> </HourlyValues> <HourlyValues time="2014-08-24T16:00:00-04:00"> <FPAID>466188805</FPAID> <EffectivePrice>39.95</EffectivePrice> <EffectivePriceIsIndex>false</EffectivePriceIsIndex> <Status>Approved</Status> </HourlyValues> <HourlyValues time="2014-08-24T17:00:00-04:00"> <FPAID>466188805</FPAID> <EffectivePrice>39.95</EffectivePrice> <EffectivePriceIsIndex>false</EffectivePriceIsIndex> <Status>Approved</Status> </HourlyValues> <HourlyValues time="2014-08-24T18:00:00-04:00"> <FPAID>466188805</FPAID> <EffectivePrice>39.95</EffectivePrice> <EffectivePriceIsIndex>false</EffectivePriceIsIndex> <Status>Approved</Status> </HourlyValues> <HourlyValues time="2014-08-24T19:00:00-04:00"> <FPAID>466188805</FPAID> <EffectivePrice>39.95</EffectivePrice> <EffectivePriceIsIndex>false</EffectivePriceIsIndex> <Status>Approved</Status> </HourlyValues> <HourlyValues time="2014-08-24T20:00:00-04:00"> <FPAID>466188805</FPAID> <EffectivePrice>39.95</EffectivePrice> <EffectivePriceIsIndex>false</EffectivePriceIsIndex> <Status>Approved</Status> </HourlyValues> <HourlyValues time="2014-08-24T21:00:00-04:00">

```
<FPAID>466203107</FPAID>
<EffectivePrice>41.75</EffectivePrice>
<EffectivePriceIsIndex>false</EffectivePriceIsIndex>
<Status>Approved</Status>
</HourlyValues>
<HourlyValues time="2014-08-24T22:00:00-04:00">
<FPAID>466203107</FPAID>
<EffectivePrice>41.75</EffectivePrice>
<EffectivePrice>41.75</EffectivePrice>
<EffectivePriceIsIndex>false</EffectivePriceIsIndex>
<Status>Approved</Status>
</HourlyValues>
</HourlyValues>
</FuelPriceAdjustmentFuelHourly>
</FuelPriceAdjustmentUnitHourly>
```

4.15.13 Generation by Unit XML Report

This report contains the same information that is shown on the Generation by Portfolio display. The report contains information only after the Day-Ahead market has been cleared and the time for the opening of the Re-Offer Period of the selected day has passed. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetGenByParticipantResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <GenByParticipant day="2014-08-24" ID="666">
    <UnitShortName>UNIT</UnitShortName>
    <UnitScheduleID>66699</UnitScheduleID>
    <UnitScheduleShortName>UNIT M</UnitScheduleShortName>
    <HourlyProfile>
      <HourlyValues priceCapped="false" time="2014-08-24T00:00:00-04:00" MW="23" price="29.62"/>
      <HourlyValues priceCapped="false" time="2014-08-24T01:00:00-04:00" MW="23" price="29.53"/>
      <HourlyValues priceCapped="false" time="2014-08-24T02:00:00-04:00" MW="23" price="30.78"/>
      <HourlyValues priceCapped="false" time="2014-08-24T03:00:00-04:00" MW="23" price="35.55"/>
      <HourlyValues priceCapped="false" time="2014-08-24T04:00:00-04:00" MW="23" price="38.69"/>
      <HourlyValues priceCapped="false" time="2014-08-24T05:00:00-04:00" MW="23" price="41.67"/>
      <HourlyValues priceCapped="false" time="2014-08-24T06:00:00-04:00" MW="23" price="42.09"/>
      <HourlyValues priceCapped="false" time="2014-08-24T07:00:00-04:00" MW="23" price="40.75"/>
      <HourlyValues priceCapped="false" time="2014-08-24T08:00:00-04:00" MW="23" price="38.95"/>
      <HourlyValues priceCapped="false" time="2014-08-24T22:00:00-04:00" MW="23" price="36.45"/>
      <HourlyValues priceCapped="false" time="2014-08-24T23:00:00-04:00" MW="23" price="34.44"/>
    </HourlyProfile>
  </GenByParticipant>
  <GenByParticipant day="2014-08-24" ID="666">
    <UnitShortName>UNIT</UnitShortName>
    <UnitScheduleID>66605</UnitScheduleID>
    <UnitScheduleShortName>UNITPric</UnitScheduleShortName>
    <HourlyProfile>
      <HourlyValues priceCapped="false" time="2014-08-24T09:00:00-04:00" MW="13" price="129.62"/>
      <HourlyValues priceCapped="false" time="2014-08-24T10:00:00-04:00" MW="13" price="129.53"/>
      <HourlyValues priceCapped="false" time="2014-08-24T11:00:00-04:00" MW="13" price="130.78"/>
      <HourlyValues priceCapped="false" time="2014-08-24T12:00:00-04:00" MW="13" price="135.55"/>
      <HourlyValues priceCapped="false" time="2014-08-24T13:00:00-04:00" MW="13" price="138.69"/>
      <HourlyValues priceCapped="false" time="2014-08-24T14:00:00-04:00" MW="13" price="141.67"/>
      <HourlyValues priceCapped="false" time="2014-08-24T15:00:00-04:00" MW="13" price="142.09"/>
      <HourlyValues priceCapped="false" time="2014-08-24T16:00:00-04:00" MW="13" price="140.75"/>
      <HourlyValues priceCapped="false" time="2014-08-24T17:00:00-04:00" MW="13" price="138.95"/>
      <HourlyValues priceCapped="false" time="2014-08-24T18:00:00-04:00" MW="13" price="136.45"/>
```

```
<HourlyValues priceCapped="false" time="2014-08-24T19:00:00-04:00" MW="13" price="134.44"/>
<HourlyValues priceCapped="false" time="2014-08-24T20:00:00-04:00" MW="13" price="136.54"/>
<HourlyValues priceCapped="false" time="2014-08-24T21:00:00-04:00" MW="13" price="129.33"/>
</HourlyProfile>
```

</GenByParticipant>

<GenByParticipant day="2014-08-24" ID="77776">

denbyrarticipant day- 2014-0

<UnitShortName>UNIX</UnitShortName> <UnitScheduleID>7777601</UnitScheduleID>

<UnitScheduleShortName>DFLPRICE</UnitScheduleShortName>

<HourlyProfile>

<HourlyValues priceCapped="false" time="2014-08-24T00:00:00-04:00" MW="11" price="225.96"/> <HourlyValues priceCapped="false" time="2014-08-24T01:00:00-04:00" MW="11" price="223.91"/> <HourlyValues priceCapped="false" time="2014-08-24T02:00:00-04:00" MW="11" price="222.45"/> <HourlyValues priceCapped="false" time="2014-08-24T03:00:00-04:00" MW="11" price="221.42"/> <HourlyValues priceCapped="false" time="2014-08-24T04:00:00-04:00" MW="11" price="222.47"/> <HourlyValues priceCapped="false" time="2014-08-24T05:00:00-04:00" MW="11" price="223.82"/> <HourlyValues priceCapped="false" time="2014-08-24T06:00:00-04:00" MW="11" price="228.04"/> <HourlyValues priceCapped="false" time="2014-08-24T07:00:00-04:00" MW="11" price="231.35"/> <HourlyValues priceCapped="false" time="2014-08-24T08:00:00-04:00" MW="11" price="231.44"/> <HourlyValues priceCapped="false" time="2014-08-24T09:00:00-04:00" MW="11" price="232.27"/> <HourlyValues priceCapped="false" time="2014-08-24T10:00:00-04:00" MW="11" price="233.56"/> <HourlyValues priceCapped="false" time="2014-08-24T11:00:00-04:00" MW="11" price="235.06"/> <HourlyValues priceCapped="false" time="2014-08-24T12:00:00-04:00" MW="12" price="239.89"/> <HourlyValues priceCapped="false" time="2014-08-24T13:00:00-04:00" MW="12" price="243.62"/> <HourlyValues priceCapped="false" time="2014-08-24T14:00:00-04:00" MW="17" price="246.19"/> <HourlyValues priceCapped="false" time="2014-08-24T15:00:00-04:00" MW="17" price="246.79"/> <HourlyValues priceCapped="false" time="2014-08-24T16:00:00-04:00" MW="17" price="246.33"/> <HourlyValues priceCapped="false" time="2014-08-24T17:00:00-04:00" MW="12" price="243.79"/> <HourlyValues priceCapped="false" time="2014-08-24T18:00:00-04:00" MW="12" price="240.52"/> <HourlyValues priceCapped="false" time="2014-08-24T19:00:00-04:00" MW="12" price="237.71"/> <HourlyValues priceCapped="false" time="2014-08-24T20:00:00-04:00" MW="12" price="239.22"/> <HourlyValues priceCapped="false" time="2014-08-24T21:00:00-04:00" MW="11" price="232.22"/> <HourlyValues priceCapped="false" time="2014-08-24T22:00:00-04:00" MW="11" price="230.35"/> <HourlyValues priceCapped="false" time="2014-08-24T23:00:00-04:00" MW="11" price="228.97"/> </HourlyProfile>

</GenByParticipant>

</GetGenByParticipantResponse>

5. Regulation Tab

To access the Regulation Tab, you must be a registered user with a Regulation Read/Write or Read-Only role. When you have this access, the Regulation tab will be shown and selecting it will show all of its display sub tabs as shown below.

Public Gen	eration	emand	ARD	Regulation	
Private	Portfolio	Regula	ition	Regulation	XML
Messages	Manager	Offers	Default	Hourly Updates	Download

5.1 Regulation Offer Concepts

Regulation resources are specially equipped generating resources that are able to increase or decrease generation output every four seconds in response to signals it receives from ISO New England in order to control changes in the system load and fluctuations on the system tie-lines. There are two types of regulation resources:

- Generating units that are capable of providing regulation. These are flagged as Regulation Resources in the database. This status is shown on the Generation Unit Default Parameters display.
- Alternative Technology Regulation Resources (ATRRs) These are a new type of assets in the Market database that are capable of providing only regulation capability and therefore are limited to participating in only the Regulation Market.

When submitting data, it is important to understand the hierarchy of the Regulation offer data and the implications of data overrides. Figure 8 below shows the hierarchy of the Regulation displays used to submit data and where the same offer data appears in different displays, the data of which display has precedence over (overrides) the others. [Note: only the data elements subject to overrides are shown.]

The Regulation Offer display is at the root of the hierarchy. Participants must submit a complete set of Regulation offer data using this display when the asset is first instantiated in the database. These offers persist for future Market days until they are changed. The Regulation Hourly Updates displays allow the Participant to profile his physical and financial regulation offers on an hourly basis for a given Market day. These hourly updates offer data supersede the corresponding offer data for the Market day only; they don't extend to future market days.



Figure 8 – Regulation Display Hierarchy and Data Precedence

Note: For ATRRss that are part of a Continuous Storage Facility (CSF), refer to chapter 10 Continuous Storage Facilities (CSF) for additional considerations.

5.2 Regulation Portfolio Manager Display

A regulation portfolio is a collection of regulation resources that are identified by each user. The user assigns a unique name to each portfolio when it is created. There is no limit to the number of regulation portfolios a user can create. Once a portfolio is created the user assigns one or more regulation resource. There is no limit to the number of regulation resources that can be "assigned" to a portfolio. A regulation resource can be assigned to more than one portfolio. It must be noted that only regulation resources can be assigned to regulation portfolios. Therefore where you are looking at the Portfolio Manager display for regulation portfolios, only generating units capable of providing regulation and ATRRs that belong to the Participant that you represent will be listed as Regulation Assets.

Data is filtered by the regulation portfolio on the following displays:

- Regulation Offers
- Regulation Hourly Updates Display

Note: A regulation resource that is not associated with a portfolio cannot be selected in these displays' tooltray filters.

See §3.1 Manage Portfolios for more information on the purpose of portfolios and details on using this display.

5.3 Regulation Offers Default Display

A regulation offer is submitted by a regulation Market Participant to sell regulation services. This display is to view and update the Participant's regulation offers for every regulation resource in the user's selected portfolio.

Regulation offers for a regulation resource consist of regulation high and low limits, a regulation (automatic) ramp rate, offers for regulation capacity (two parts) and service, and the resource's availability to participate in the Regulation market. Regulation offers for a given market day can be updated up to the close of the daily Regulation Market period. See §1.4 for a complete definition of the eMarket timeline for regulation offers.

ISC	nev	w england											Regulat	ion Market			ISO New E		Version: 3	rket U 3.0.0(2014110 r: Superuser ~	061037)	13-Dec-20:	face 14 19:21:18 *
ublic Ge	eneration	Demand ARD	Regulation																				
ivate essages	Portfolio Manager	Regulation Offers Default	Regulation Hourly Updates	XML Download																			
Regula	ation C	Offers Daily	Default																				
elect Date	e and Portfo	olio	~	Regulation Off	fers for Portfolio A	ALL_ATRRS o	on 15-Dec-20	14															
Date	: 15-Dec-2	2014		Resource Short Name	Reg Capability	Reg Cap ITO Cost	Reg Cap Offer Oth	Reg Serv Price	ARR	ARR Cap	Dispatch Method	Reg High	RHL Ceil	Reg Low	RLL Floor	Unavail	Derate Factor	Derate Comment	Service Factor	Service Comment	Exclude	Exclude Comment	
	ALL_ATR		×	A123	4	2.00	3.00	4.00	277.0	999.0	ENC	-1.0	10.0	-9.0	-10.0		1.0	Initial defa	10.0	Initial defa			
Portiolo.	ALL_ATH	(KS	•	APNB	1.5	3.00	3.00	3.00	120.0	999.0	ENC	2.0	10.0	-1.0	-10.0		1.0	Initial defa	10.0	Initial defa			
				BPW1	3.7	3.00	4.00	3.00	999.0	999.0	ENT	4.0	10.0	-3.5	-10.0		1.0	Initial defa	10.0	Initial defa			
				BPW2	5	2.00	1.00	2.00	88.0	999.0	ENT	4.0	10.0	-8.0	-10.0	1	1.0	Initial defa	10.0	Initial defa			
				CAR2	8.5	2.00	2.00	2.00	300.0	999.0		9.0	10.0	-8.0	-10.0		1.0	Initial defa	10.0	Initial defa			
				DSYS	2	1.00	2.00	3.00	88.0		CONT	4.0	10.0	0.0	-14.0			Initial defa		Initial defa			
				RLT1	3	1.00	1.00	1.00	240.0	999.0	CONT	8.0	10.0	2.0	-10.0		1.0	Initial defa	10.0	Initial defa			
																						date	Refresh

The format of this display is shown below:

The fields on this display are:

- **Resource Short Name** The name of the regulation resource.
- Reg Capability The regulation capability: a calculated value equal to the day's derate factor times the minimum of 5 times the regulation ramp rate (ARR) and half the difference between the regulation high and low limits; i.e., Reg Capability = Derate Factor x Min (5 x ARR, [Reg High²⁰ Reg Low²⁰]/2).
- **Reg Cap ITO Cost** The price requested to cover the regulation intertemporal opportunity capacity cost (\$/MWH) of the resource.
- **Reg Cap Offer Other** The price requested to cover all other regulation capacity cost (\$/MWH) of the resource.
- **Reg Serv Price** The price requested to cover the regulation service cost (\$/MWH) of the resource.
- ARR (Automatic Response Rate) The MW/Minute rate at which the resource can change its MW output in response to regulation dispatch instructions.

²⁰ Effectively, it is the minimum of the Reg High and Reg Ceil limits and the maximum of the Reg Low and Reg Floor limits.

- **ARR Cap** The resource's demonstrated ARR Capability value that was established during the last audit conducted by ISO New England. The offer value for ARR will be capped (constrained) to this value within the generation dispatch software if it is defined. If the demonstrated capability value is undefined, it will be shown as blank.
- **Dispatch Method** The dispatch methodology of the regulating resource. The possibilities are:
 - CONT for Continuous; not energy neutral (for traditional thermal units, but can apply to ATRRs)
 - ENC for Energy Neutral Continuous (applicable to ATRRs only)
 - ENT for Energy Neutral Trinary (applicable to ATRRs only)
- **Reg High** The highest level of energy, in MW, that the Participant will allow for the operation of the resource while it is providing regulation (on Automatic Generation Control). This is the Regulation High Limit (RHL).
- **RHL Ceil** The highest level of energy, in MW, recognized by ISO New England for the operation of the resource while it is providing regulation (on Automatic Generation Control). For generators, this is the unit's <u>daily</u> Economic Maximum limit shown on the Schedule Detail Defaults display for the schedule designated for the RT Market on the selected market day. For ATRRs, this is a value set for the resource by the ISO New England.
- **Reg Low** The lowest level of energy, in MW, that the Participant will allow for operation of the resource while it is providing regulation (on Automatic Generation Control). This is the Regulation Low Limit (RLL).
- RLL Floor The lowest level of energy, in MW, recognized by ISO New England for the operation of the resource while it is providing regulation (on Automatic Generation Control). For generators, this is unit's <u>daily</u> Economic Minimum limit shown on the Schedule Detail Defaults display for the schedule designated for the RT Market on the selected market day. For ATRRs, this is a value set for the resource by the ISO New England.
- **Unavail** When checked it indicates that the resource is not available for regulation; entered by the Participant.
- **Derate Factor** This is a value between 0 and 1 that is calculated based on historical performance that provides an estimate of the resource's future capability to respond to an AGC setpoint. It is set by the ISO New England.
- **Derate Comment** A comment entered by the ISO New England the last time the Derate Factor value was changed.
- Service Factor A factor that is calculated based on historical performance that provides an estimate of the resource's regulation service as a multiple of capacity. It is set by the ISO New England.
- Service Comment A comment entered by the ISO New England the last time the Service Factor value was changed.
- **Exclude Flag** When checked it indicates that the resource is excluded from the regulation market by the ISO New England.
- Exclude Comment A comment entered by the ISO New England the last time the Exclude Flag value was changed.

5.3.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered value is valid:

- Automatic Response Rate cannot be less than a minimum allowable response rate that is set by ISO New England.
- A warning will be issued if the Automatic Response rate exceeds the ARR Cap, when one is defined.
- Reg Floor ≤ Reg Low < Reg High ≤ Reg Ceil Note that for ATRRs, the regulation limits can be negative. For generating units, the regulation limits must be strictly positive; the economic limits (Reg Floor = Economic Minimum, Reg Ceil = Economic Maximum) are the more constraining of the limits among the daily default (those entered on the Schedule Detail Defaults display) and the hourly updates (those entered on the Unit Hourly Updates display) for the selected market day; thus the limits used for validation are not necessarily the ones shown on the display in the Reg Floor and Reg Ceil columns. This is the *limit nesting rule*. A warning message is returned when this rule is violated.
- Total regulation capacity offer price (Reg Cap ITO Cost + Reg Cap Offer Other) cannot exceed \$100.00 and must be greater than the Regulation Floor Price determined by the ISO New England.
- Regulation service offer price (Reg Serv Price) cannot exceed \$10.00 and must be greater than the Regulation Floor Price determined by the ISO New England.

5.3.2 Modify Regulation Offers

- 1. In the Regulation Navigation sub tabs, click the **Regulation Offers Default** tab.
 - The Regulation Offers Daily Default display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - The detail of the regulation offers for the resources in the selected portfolio is displayed.
- 3. Enter any or all of the following information for the desired resources:
 - a. Regulation Prices
 - b. Automatic Response Rate
 - c. Available/Unavailable
 - d. Regulation limits
- 4. Click the **Update** button to save the data.

5.4 Regulation Hourly Updates Display

This display is used to view and modify the hourly overrides of a resource's daily (default) regulation offer parameters for a specified market day. Refer to section 1.4 Business Rules and Timelines for the specification of when regulation hourly updates can be submitted into the market and which can be submitted during the Regulation Intraday Re-Offer period.

ISC	new england						Regul	ation Mark	ket Status: Of	PEN(01)			'ersion: 3.0 cate User: !	.0(2014110	61037)	nterfac 3-Dec-2014 19:
	P								Parti	cipant: ISC) New Englan	d Inc		~	User: Sup	eruser
olic Ger	neration Demand ARD	Regulation														
	Portfolio Regulation	Regulation	XML													
	Manager Offers Default	Hourly Updates	Download													
egula	ation Hourly Upo	dates														
ect Date	and Resource	~	Regulation Daily Defa	ult Values for	A123 on 1	5-Dec-2014	-									
Date:	15-Dec-2014		Resource Short Name	Reg Capability	Reg Low	Reg High	ARR	Unavail	Reg Cap ITO Cost	Reg Cap Offer Oth	Reg Serv Price	RLL Floor	RHL Ceil	Active ARR Cap	Dispatch Method	
	ALL ATRRS	~	A123	4	-9.0	-1.0	277.0		2.00	3.00	4.00	-10.0	10.0	999.0	ENC	
esource:	_	~	Regulation Hourly Up	dates for A12	3 on 15-De	c-2014										
	1120		Hour	Reg Capability	Reg Low	Reg High	ARR	Unavail	Reg Cap ITO Cost	Reg Cap Offer Oth	Reg Serv Price		Legend			
			🗙 👻 Copy										No Over	ride: 🔳		
			01	4.0								-	Available	. 🗖		
			02	4.0									Unavailab			
			03	4.0									Ullavallat	Jie. 💌		
			04	4.0												
			06	4.0												
			07	4.0												
			08	4.0												
			09	4.0												
			10	4.0												
			11	4.0												
			12	4.0												
			13	4.0												
			14	4.0												
			18	4.0												
			17	4.0												
			18	4.0												
			19	4.0												
			20	4.0												
			21	4.0												
			22 23	4.0												
			23	4.0								-				
			27	4.0								_				

The format of this display is shown below:

During the Regulation Intraday Re-Offer period, the rows for hours where regulation offers cannot be updated will be grayed out. For generating resources the data entry cells for future hours of regulation financial offers will be the only ones shown (open) on the display, as illustrated below.

ISC	new engla	and						Regul	ation Marl	ket Status: Ol	PEN(23)		Ve		.0(2014110	061037)	nterfa 3-Dec-2014 :	
										Parti	icipant: IS	O New England	d Inc		~	User: Supe	eruser	
dic Ger	eration Demand	ARD	Regulation															
ate	Portfolio Regulat	ion	Regulation	XML														
ssages	Manager Offers		Hourly Updates	Download														
egula	ition Hourly	Upda	ates															
ect Date	and Resource		«	Regulation Daily D	afault Values for		on 13-Dec-	2014										
cer bute				Resource				ARR	Unavail	Reg Cap	Reg Cap	Reg Serv	RLL	RHL	Active	Dispatch		
Date:	13-Dec-2014		•	Short Name	Reg Capability	Reg Low	Reg High			Reg Cap ITO Cost	Reg Cap Offer Oth	Reg Serv Price	Floor	Ceil	ARR Cap	Method		
Portfolio:	ALL_UNITS		~	ALTR	4.5	149.0	159.0	3.0		0.00	0.7	5 0.00	150.0	160.0		CONT		
esource:	ALTRESCO		~	Regulation Hourly	Updates for ALT	RESCO on 1	3-Dec-2014	1										
				Hour	Reg	Reg	Reg	ARR	Unavail	Reg Cap	Reg Cap	Reg Serv		Legend				
					Capability	Low	High			ITO Cost	Offer Oth	Price		-				
				Y Y Cop										No Over	ride: 🔳			
				01	4.5									Available	: 🔳			
				02	4.5									Unavailat	ble: 🔽			
				03	4.5									Onavailat	. <u>.</u>			
				04	4.5 4.5													
				06	4.5													
				07	4.5													
				08	4.5													
				09	4.5													
				10	4.5													
				11	4.5													
				12	4.5													
				13	4.5													
				14	4.5													
				15	4.5													
				16	4.5													
				17	4.5													
				18	4.5													
				19	4.5													
				20	4.5													
				21	4.5													
				22	4.5													
				23	4.5													
				24	4.5													
																Upd	ate F	Refre

There are two frames in this display:

- Regulation Daily Default Values This frame shows the daily default regulation offer parameters for the selected regulation resource and market day. They are the values shown on the Regulation Offers Default display. You cannot change these values using this display; they are shown for reference only.
- Regulation Hourly Updates This frame shows the hourly overrides of the selected resource's regulation parameters for the selected market day. The fields in this frame are:
 - **Hour** Hour ending values are shown in the first column for each hour in the day. The row for hour ending 03 is not shown for the Spring Daylight Saving Time transition day. Similarly, the value on the row of the repeating hour ending 02 of the Fall Daylight Saving Time transition day is marked with an asterisk (*).
 - **Reg Capability** The MW regulation capability for the hour calculated from the ARR for the hour, the regulation limits for the hour and the derate

factor for the day. It is equal to the day's derate factor times the minimum of 5 times the regulation ramp rate (ARR) and half the difference between the regulation high and low limits; i.e.,

Reg Capability = Derate Factor x Min (5 x ARR, [Reg High²¹ – Reg Low²¹]/2).

In the absence of an ARR and regulation limits for the hour, the daily offer values are used.

- **Reg Low** The regulation low limit override value if it is not blank.
- **Reg High** The regulation high limit override value if it is not blank
- **ARR (Automatic Response Rate)** The regulation response rate override value if it is not blank.
- **Unavail** Checkbox that serves to override the default status for the day. Refer to the legend shown of the display for the definition of the each state of the tri-state checkbox.
- **Reg Cap ITO Cost** The regulation inter-temporal opportunity capacity cost (\$/MWH) override value if it is not blank.
- **Reg Cap Offer Oth** The other regulation capacity cost (\$/MWH) override value if it is not blank.
- **Reg Serv Price** The regulation service cost (\$/MWH) override value if it is not blank.

The top row of the grid provides you with the means to enter or clear overrides for a selected range of hours. You select the range of hours to override by selecting a "From" hour-ending value (left) and a "To" hour-ending value and then selecting the "Copy" button to copy the values entered in the top row to the hourly rows that you specified for the hourly range before you select the "Update" to save the data that you entered on the screen.

Hourly overrides of the physical parameters (Reg Low, Reg High, ARR and Unavail) and the financial parameters (Reg Cap ITO Cost, Reg Cap Offer Oth and Reg Serv Price) are independent. In other words, you can submit overrides of the physical parameters for some hours without also submitting overrides of the financial parameters for these hours; the financial parameter values for these hours would be the daily default values. Vice versa, you can submit overrides of the financial parameters for some hours without also submitting overrides of the physical parameters for these hours; the physical parameter values for these hours would be the daily default values. In either case, you must submit a complete set of override values for an hour. In other words, when you submit an override of the physical parameters for an hour, you must submit override values for all four parameters: Reg Low, Reg High, ARR and Unavail, even if it means entering the daily default value. Similarly, when you submit an override of the financial parameters for an hour, you must submit override values for all three parameters: Reg Cap ITO Cost, Reg Cap Offer Oth and Reg Serv Price, even if it means entering the

²¹ Effectively, it is the minimum of the Reg High and Reg Ceil limits and the maximum of the Reg Low and Reg Floor limits for the hour.

daily default value. This rule also applies to entries in the top row of the grid that you want to copy to a range of hours. Note however that when entering values in the top row, if you leave either set of values blank, the blank fields will be copied to the range of hours selected, meaning that any previously entered hourly overrides of these values in the range are being deleted and returning the values in these hours to their daily default values. You cannot use the top row to selectively update the overrides in one column for a range of hours while retaining the override values in the other columns.

5.4.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- The hourly ARR (Automatic Response Rate) cannot be less than a minimum allowable response rate that is set by ISO New England.
- A warning will be issued if the hourly ARR exceeds the Active ARR Cap, when one is defined. The Active ARR Cap is shown in the Daily Default Values frame.
- On an hourly basis, RLL Floor < Reg Low < Reg High < RHL Ceil
- The RLL Floor and the RHL Ceil are shown in the Daily Values frame. Note that for ATRRs, the regulation limits can be negative. For generating units, the regulation limits must be strictly positive.
- Total regulation capacity offer price (Reg Cap ITO Cost + Reg Cap Offer Other) for an hour cannot exceed \$100.00 and must be greater than the Regulation Floor Price determined by the ISO New England.
- Regulation service offer price (Reg Serv Price) for an hour cannot exceed \$10.00 and must be greater than the Regulation Floor Price determined by the ISO New England

5.4.2 Override Regulation Offer Parameters for Day and Hour

To override a resource's default daily regulation limits, availability to provide regulation and financial regulation offers:

- 1. In the Regulation Navigation sub tabs, click the **Regulation Hourly Updates** tab.
 - The Regulation Hourly Updates display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired unit from the **Resource** filter. Note: only regulation resources listed in the selected portfolio will be listed in the drop-down list.
 - The display reflects the regulation daily default offers and the hourly overrides for the selected unit.

- 3. To update the overrides of physical parameters for individual hours:
 - a. Enter the override value for each of the limits, for the ARR and for the resource availability in the hours that you wish; you must enter a value for each parameter. If you want to remove an override of the physical parameters, you enter a blank for each limit and the ARR and set the Unavail flag to "No Override".
 - b. Click the **Update** button to save the data
- 4. To update the overrides of financial parameters for individual hours:
 - a. Enter the override value for each of financial offers in the hours that you wish; you must enter a value for each of the three parameters. If you want to remove an override, you must enter blanks for each parameter.
 - b. Click the **Update** button to save the data
- 5. To update the overrides for a range of hours:
 - a. Enter the override value for each of the limits, for the ARR and for the resource availability in the top row. Enter the override value for each of the financial regulation offer parameters in the top row. Remember: if you leave blank the fields of either set of parameters, it means that you want to remove the overrides of that set of parameters.
 - b. Select the range of hours to override on the top row. Then select the "Copy" button. This will copy all of the values on the top row (including blanks) to the hourly rows for the range of hours selected.



c. Click the **Update** button to save the data for the selected hours.

5.5 XML Download for Regulation Display

The XML Download display is accessible through the Regulation Navigation sub tab. This display enables you to download pre-defined Regulation specific XML report files of data that you maintain in the ISO-NE Market System. It also provides you the means to upload data in XML formatted data files into the ISO-NE Market System to be used to clear the Day-Ahead and Real Time markets. The format of the XML files can be found in the **eMarket Data Exchange Specification**, which is found on the <u>ISO New England website</u> among the documents listed on the Participate > Support > Upload and Download File Format Protocols page.

ISO new england	Regulation Market Status:		eMarket User Interface Version: 2.0.1(201407221119) Rificate User: Superuser 23-Aug-2014 17:51:08
per-	Pa	ticipant: ISO New England Inc	Vuser: Superuser V
Public Generation Demand ARD Regulation			
Private Portfolio Regulation Regulation XMI Messages Manager Offers Default Hourly Updates Dev	L wnload		
XML Download			
Select Date 🔍	XML Download on 25-Aug-2014		
	Regulation Offers Regulation Hourly Updates	Upload XML SOAP Request File: Submit	Browse
Messages			

The following XML reports of generation data are available for downloading:

- Regulation Offers
- Regulation Hourly Updates

Each returns an XML formatted report for the selected date and for every regulating unit or ATRR that belongs to you as Lead Participant.

See §3.2 for details on uploading and downloading files. The following sections provide samples of the downloaded reports.

5.5.1 Regulation Offers XML Report

This report contains the information that is shown on the Regulation Offers Default display for every regulating unit or ATRR that belongs to you as Lead Participant. The following is a sample of the report.

<Excluded>false</Excluded> <RegHighLimit>19.4</RegHighLimit> <RegLowLimit>10.4</RegLowLimit> <RegCapOther>15.45</RegCapOther> <RegITOCap>10.45</RegITOCap> <RegServicePrice>9.45</RegServicePrice> <RHLCeil>21</RHLCeil> <RLLFloor>11</RLLFloor> <ServiceComment>Reduced Service factor</ServiceComment> <ServiceFactor>10</ServiceFactor> </RegOffer> <RegOffer day="2014-12-15" ID="5"> <ARR>45.5</ARR> <ARRCap>10</ARRCap> <Available>true</Available> <Capability>2.2</Capability> <DerateComment>Initial default derate factor</DerateComment> <DerateFactor>1</DerateFactor> <DispatchMethod>ENC</DispatchMethod> <ExcludedComment xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true" /> <Excluded>false</Excluded> <RegHighLimit>1.5</RegHighLimit> <RegLowLimit>-3</RegLowLimit> <RegCapOther>11.15</RegCapOther> <RegITOCap>7.15</RegITOCap> <RegServicePrice>3.1</RegServicePrice> <RHLCeil>15</RHLCeil> <RLLFloor>-5</RLLFloor> <ServiceComment>Initial default service factor</ServiceComment> <ServiceFactor>10</ServiceFactor> </RegOffer> </GetRegOfferResponse>

5.5.2 Regulation Hourly Updates XML Report

This report contains the information that is shown on the Regulation Hourly Updates display for every regulating unit or ATRR that belongs to you as Lead Participant, and for which hourly updates exist. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetRegHourlyResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <RegHourly day="2014-12-15" ID="444">
    <HourlyProfile>
      <HourlyValues time="2014-12-15T00:00:00-04:00">
        <ARR>21.5</ARR>
        <Available>true</Available>
        <Capability>37.5</Capability>
        <RegHighLimit>680</RegHighLimit>
        <RegLowLimit>400</RegLowLimit>
        <RegCapOther>70</RegCapOther>
        <RegITOCap>70</RegITOCap>
        <RegServPrice>70</RegServPrice>
      </HourlyValues>
      <HourlyValues time="2014-12-15T01:00:00-04:00">
        <ARR>21.5</ARR>
        <Available>true</Available>
        <Capability>37.5</Capability>
```

<RegHighLimit>680</RegHighLimit> <RegLowLimit>400</RegLowLimit> <RegCapOther>70</RegCapOther> <RegITOCap>70</RegITOCap> <RegServPrice>70</RegServPrice> </HourlyValues> <HourlyValues time="2014-12-15T02:00:00-04:00"> <Available>true</Available> <Capability>37.5</Capability> <RegHighLimit>680</RegHighLimit> <RegLowLimit>400</RegLowLimit> <RegCapOther>70</RegCapOther> <RegITOCap>70</RegITOCap> <RegServPrice>70</RegServPrice> </HourlyValues> <HourlyValues time="2014-12-15T23:00:00-04:00"> <Available>true</Available>

<Available>true</Available> <Capability>37.5</Capability> <RegHighLimit>680</RegHighLimit> <RegLowLimit>400</RegLowLimit> <RegCapOther>70</RegCapOther> <RegITOCap>70</RegITOCap> <RegServPrice>70</RegServPrice> </HourlyValues> </HourlyVrofile> <ShortName>UNIT</ShortName> </RegHourly> </GetRegHourlyResponse>
6. Demand Tab

To access the Demand Tab, you must be a registered user with a Demand Read/Write or Read-Only role. When you have this access, the Demand tab will be shown and selecting it will show all of its display sub tabs as shown below.

Public Genera	ation De	mand	ARD Reg	julation			
	Portfolio	Demand	Inc/Dec	Load Response	Demand By	Load Response	XML
	Manager	Bids	Bids	Offers	Portfolio	Results	Download

6.1 Demand Portfolio Manager Display

A demand portfolio is a collection of price locations (Pnodes) that are identified by each user. The user assigns a unique name to each portfolio when it is created. There is no limit to the number of demand portfolios a user can create. Once a portfolio is created the user assigns one or more price location. There is no limit to the number of price locations that can be "assigned" to a portfolio. A price location can be assigned to more than one portfolio.

Data is filtered by the demand portfolio on the following displays:

- Demand Bids
- Inc/Dec Bids
- Demand by Portfolio

Note: A Pnode that is not associated with a portfolio cannot be selected in these displays' tooltray filters.

See §3.1 Manage Portfolios for more information on the purpose of portfolios and details on using this display.

6.2 Demand Bids Display

This display is used to view and update fixed and price-sensitive demand bids. Demand bids are daily bids for energy to satisfy demand at Day-Ahead market prices for a given date. They specify a MW amount of energy to buy and the price per MWH the user is offering to pay for the energy. Demand bids can only be submitted at load zones (pricing location). A user can submit fixed demand bids and price-sensitive demand bids:

- A fixed demand bid does not have a price associated with it because the user is offering to unconditionally buy an amount of energy at a specified location irrespective of the Day-Ahead clearing price to satisfy his demand.
- A price-sensitive demand bid as its name implies has a price associated with it. The user is offering to buy an amount of energy at a specified location provided that the Day-Ahead clearing price at that location is below a user specified amount. The user can submit up to ten price-sensitive demand bids per hour for each pricing location.

ISO new england										emand <u>M</u> a	arket	Status: C						et Use 3(201506161306 Superuser		rface
												Partic	ipant: I	SO Ne	ew Engl	and Inc		Vser:	Superuser	
blic Generation Demand ARD Regulation																				
ivate Portfolio Demand Inc/Dec Load Resp. essages Manager Bids Bids Offers		Load Res Results		ML Jownload																
Demand Bids															_	_		Hour		_
lect Date and Node	Demand Bids for .Z.	CONNECT	TCUT (400)4) on 19-	Jun-20	15				_	_		_							ору
Date: 19-Jun-2015	Hour	# 1	4W #	MW	Price	#		Price	# MV	/ Price	#	MW	Price	#	MW	Price	Total		<u> </u>	
Portfolio: Mkt Adm All Pnodes	V V Copy	-	1	1		2			3		4			5				01		
Node: .Z.CONNECTICUT			6			7			8		9			10				02		
	01	F	1	62.8	999.00	2	62.9			2.7 999.		62.6		5	62.5	999.50		03		
	02	F	6	62.8	999.00	7	62.9	999.20	8 3 6	2.7 999.	9 30 4	62.6		10 5	62.5	999.50	313.5			
			8			7			8		9			10			313.5	04		
Сору То:	03	F	1	62.8	999.00	2	62.9		3 C 8	2.7 999.	30 4 9			5	62.5	999.50	313.5	05		
	04	F	1	62.8	999.00	2	62.9	999.20	з е	2.7 999.	30 4	62.6			62.5	999.50	313.5	06		
Сору	05	F	6	62.8	999.00	2	62.9	999.20	8 3 C	2.7 999.		62.6	999.40	10 5	62.5	999.50	313.5	07		
		F	8			7			8		9			10			313.5			
	06	F	6	62.8	999.00	2	62.9	999.20	3 C 8	2.7 999.	30 4 9	62.6		5 10	62.5	999.50	313.5	08		
	07	F	1	62.8	999.00	2	62.9	999.20	з е	2.7 999.	30 4	62.6			62.5	999.50		09		
	08	F	1	62.8	999.00		62.9	999.20	8 3 (2.7 999.	30 4	62.6		10 5	62.5	999.50	313.5	10		
			8			7			8		9			10			313.5	11		
	09	F	1	62.8	999.00	2	62.9	999.20	3 e 8	2.7 999.	30 4 9	62.6	999.40	10	62.5	999.50	313.5			
	10	F	1	62.8	999.00		62.9	999.20	з е	2.7 999.	30 4	62.6			62.5	999.50		12	-	
	11	F	6	62.8	999.00	7	62.9	999.20	8 3 6	2.7 999.	9 30 4	62.6		10 5	62.5	999.50	313.5	10	<u> </u>	
			6			7			8		9			10			313.5			
	12	F	1	62.8	999.00	2	62.9	999.20	3 e 8	2.7 999.	30 4 9	62.6		5 10	62.5	999.50	313.5			
	13	F	1	62.8	999.00		62.9	999.20		2.7 999.		62.6	999.40	5	62.5	999.50				
	14	F	6	62.8	999.00	7	62.9	999.20	8 3 6	2.7 999.	9 30 4			10 5	62.5	999.50	313.5			
			6			7			8		9			10			313.5			
	15	F	1	62.8	999.00	2	62.9	999.20	3 C	2.7 999.	30 4 9	62.6		5 10	62.5	999.50	313.5			
	16	F	1	62.8	999.00		62.9			2.7 999.		62.6	999.40	5	62.5	999.50				
	17	F	6	62.8	999.00	7	62.9	999.20	8 3 P	2.7 999.	9 30 4	62.6	999.40	10	62.5	999.50	313.5			
				02.0	555.00		52.5			2 000.		02.0	555.40		02.0	200.00				Refres

The format of the display used to submit demand bids is shown below:

The fields on this display are:

• **Hour** – Hour ending values are shown in the first column for each hour of the day. The row for hour ending 03 is not shown for the Spring Daylight Saving

Time transition day. Similarly, the value on the row of the repeating hour ending 02 of the Fall Daylight Saving Time transition day is marked with an asterisk (*).

- **F** Labels that identify the fields used to enter fixed demand bids.
- **#** This is the number of a price-sensitive demand bid for the hour.
- **MW** The MW amount of a fixed or price-sensitive demand bid.
- Price The price offered for a price-sensitive demand bid.
- Total The cumulative MW value of all demand bids for the hour.

The format of the display is slightly different for Participants who have registered subaccounts in the ISO-NE Customer and Asset Management System (CAMS). As shown below, there is an additional selection drop-down box labeled "SubAccount" in the **Filters** pane that is blank when the display is first called up. The user must select one of the subaccounts appearing on the drop-down list before the display data entry fields are shown. Only subaccounts active on the selected market day and inactive subaccounts for which there are demand bids on the selected market day will be listed. Inactive subaccounts will appear with an asterisk (*) appended to their name. See §1.8 for more information on subaccounts.

ISO new england		Demand Marke	: Status: CLOSED	eMarket User Interface Version: 4.0.3(201506161306) Certificate User: Superviser 18-Jun-2015 19:07:21
por a second			Participant: ACME Inc	Vser: Superuser Y
Public Generation Demand ARD Regulation				
Private Portfolio Demand Inc/Dec Load Response Manager Bids Bids Offers	Demand By Load Response XML Portfolio Results Download			
Demand Bids				
Select Date and Node 🛞 Dema	nand Bids (SubAccount Selection Required)			
Date: 19-Jun-2015	r # MW # MW Price	# MW Price # MW Price	# MW Price # MW I	Price Total Participant Total
Portfolio: Mkt_Adm_All_Pnodes				
Node: .Z.CONNECTICUT				
SubAccount:				
		15941 DOMR NRG Test5		
Messages				

An additional column of information is shown on the display:

• **Participant Total** – The cumulative MW value of all demand bids submitted by the Participant users across all subaccounts for the hour.

ISO new	england												Demai	nd Marke	et St	atus: OP	'EN(01)			Ce	eMa Version: • rtificate Use	4.0.3(201	50616130		rface
																Partie	cipant:	ACM	IE Inc				Y User:	Superuser	
olic Generation De	emand ARD Regulatio	1																					0.0.4		
		Respo			d Response		XML																		
	Bids Bids Offe	rs	Portfolio	Res	ults	_	Download																		
emand Bids																									
ect Date and Node		~	Demand Bids for .2	Z.CON	NECTICUT	r (40																			
Date: 19-Jun-2	015		Hour	"	MW	#	MW	Price	#	MW	Price	#	MW	Price	#	MW	Price	#	MW	Price	Total	Participa Tot			
Portfolio: Mkt_Adn	n_All_Pnodes	•	V V Cop	y		1			2			3			4			5							
Node: .Z.CONN	ECTICUT	•				6			7			8			9			10							
bAccount: Test5		•	01	F		1	12.0	888.00	2	121.0	888.10	3	12.2	888.20	4	12.3	888.30	5 10	12.4	888.40	169.9		69.9		
			02	F		1	12.0	888.00		121.0	888.10		12.2	888.20		12.3	888.30		12.4	888.40	105.5		05.5		
						6			7			8			9			10			169.9	1	69.9		
			03	F		1	12.0	888.00	2	121.0	888.10		12.2	888.20	4	12.3	888.30		12.4	888.40					
Сору То:		•	04	F		8	12.0	888.00		121.0	888.10	8	12.2	888.20		12.3	888.30	10 5	12.4	888.40	169.9	'	69.9		
	Сору					6			7			8			9			10			169.9	1	98.4		
			05	F		1	12.0	888.00		121.0	888.10		12.2	888.20		12.3	888.30		12.4	888.40					
			08			8	12.0	888.00	7	121.0	888.10	8	12.2	888.20	9 4	12.3	888.30	10 5	12.4	888.40	169.9	1	98.4		
						6	12.0	000.00	7	121.0	000.10	8	14.4	000.20	9	12.0	000.00	10	14.7	000.40	169.9	1	98.4		
			07	F		1	12.0	888.00		121.0	888.10		12.2	888.20		12.3	888.30		12.4	888.40					
			08			8	12.0	888.00	7	121.0	888.10	8	12.2	888.20	9 4	12.3	888.30	10 5	12.4	888.40	169.9	1	98.4		
			00			6	12.0	000.00	7	121.0	000.10	8	12.2	000.20	9	12.5	000.30	10	12.4	000.40	169,5	1	98.4		
			09	F		1	12.0	888.00	2	121.0	888.10	3	12.2	888.20	4	12.3	888.30	5	12.4	888.40					
			10			8	12.0	888.00	7	121.0	888.10	8	12.2	888.20	9 4	12.3	888.30	10 5	12.4	888.40	169.9	1	69.9		
			10			6	12.0	888.00	7	121.0	888.10	•	12.2	888.20	-	12.3	888.30	5	12.4	888.40	169.9	1	69.9		
			11	F		1			2	No	tice	th	at tk	o to)to	اد عا	ih a	ffc	ron	ъИ					
			12	F		8			7											-					
			12			6			7		son									_					
			13	F		1			2	de	man	d l	bids	for	the	ose l	hour	's d	on a						
				F		8			7	dif	fere	nt	sub		ur	nt									
			14	F		1 8			2	un	icic		5000		u										
			15	F		1			2			3			4			5							
						6			7			8			9			10						Update	Refre

6.2.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- MW values must be greater than zero; i.e. MW > 0.
- The Price of a price-sensitive demand bid must be less than or equal to the Energy Offer Cap price and greater than or equal to the Energy Offer Floor price; i.e.

Energy Offer Floor price \leq Price \leq Energy Offer Cap price²².

• Demand bids can only be submitted at Load Zone pricing nodes (Pnodes whose name begins with ".Z.").

Price-sensitive demand bids for a given hour can be entered in any order; each is treated independently when clearing the Day-Ahead market. Since each is treated independently, entering multiple duplicate bids for a given hour is equivalent to entering a single bid where the MW value is the sum of the MW

²² Energy Offer Floor price and Energy Offer Cap price are both defined in Market Rule 1, section I.2.2.

values of the duplicate bids; for example, three bids for hour-ending 02 of 2 MW @ \$10.00 is equivalent to one bid of 6 MW @ \$10.00.

6.2.2 View Demand Bids

- 1. In the Demand Navigation sub tabs, click the **Demand Bids** tab.
 - The Demand Bids display is shown in the display pane.

2. In the Filters pane:

- a. Select the desired date from the **Date** filter.
- b. Select the desired portfolio from the **Portfolio** filter.
- c. Select the desired load zone price location from the **Nodes** filter.
- d. Select the desired subaccount from the **SubAccount** filter if it is shown.
 - The data for the selected price location is shown.

6.2.3 Submit Fixed Demand Bids

- 1. View the demand bids for the location and date of interest (§6.2.2)
- 2. To submit various fixed demand bids for individual hours:
 - a. Click in the **MW** cell to the right of an "**F**" in the desired hour.
 - b. Enter the MW amount of the bid.
 - c. Repeat 2a-b for all hours as needed.
 - d. Click the **Update** button to save the data.
 - The totals column of each affected hour is updated to reflect the MW total of the bids; the Participant totals column if it is shown is also updated.
- 3. To submit the same demand bid for a range of hours:
 - a. Enter the **MW** amount of the bid in the first MW column on the first row.
 - b. Select the range of hours in the top row. Then select the "Copy" button. This will copy all of the values on the top row (including blanks) to the hourly rows for the range of hours selected.



- c. Click the **Update** button to save the data for the selected hours.
 - The totals column of each affected hour is updated to reflect the MW total of the bids; the Participant totals column if it is shown is also updated.

Note: You cannot submit fixed and price-sensitive demand bids independently using the first row of the grid. That is to say, if you enter only a fixed demand bid leaving the price-

sensitive demand bid cells empty, you are requesting eMarket to delete all the pricesensitive demand bids in the range of hours that you selected.

6.2.4 Submit Price-Sensitive Demand Bids

- 1. View the demand bids for the location and date of interest (§6.2.2)
- 2. To submit various price-sensitive demand bids for individual hours
 - a. Enter the **MW** amount and the **Price** of the bids in the cells provided for the hour of the day desired, up to ten per hour.
 - b. Repeat 2a for all hours as needed.
 - c. Click the **Update** button to save the data.
 - The totals column of each affected hour is updated to reflect the MW total of the bids; the Participant totals column if it is shown is also updated.
- 3. To submit the same demand bids for a range of hours:
 - a. Enter the **MW** amount and the **Price** of the bids in the cells provided on the first row.
 - b. Select the range of hours in the top row. Then select the "Copy" button. This will copy all of the values on the top row (including blanks) to the hourly rows for the range of hours selected.



- c. Click the **Update** button to save the data for the selected hours.
 - The totals column of each affected hour is updated to reflect the MW total of the bids; the Participant totals column if it is shown is also updated.

Note: You cannot submit fixed and price-sensitive demand bids independently using the first row of the grid. That is to say, if you enter only price-sensitive demand bids leaving the fixed demand bid cell empty, you are requesting eMarket to delete the fixed demand bids in the range of hours that you selected.

6.2.5 Deleting Ranges of Demand Bids

If you want to delete the fixed and price-sensitive demand bids at a location for a range of hours of a given date, you can use the top row of the grid to do this.

- 1. Leave the fixed and price-sensitive demand bid cells of the first row empty.
- Select the range of hours in the top row. Then select the "Copy" button. This will copy the blanks on the top row to the hourly rows for the range of hours selected.



3. Click the **Update** button. All of the existing bids shown on the display will be deleted.

Note: If you select an *inactive* subaccount, you can **ONLY** delete demand bids; you cannot modify the demand bids that exist on the selected day, nor can you submit new demand bids. The inactive subaccount will appear on the drop-down select list **only** if there are demand bids for the subaccount on the selected market day; if there are no demand bids for the subaccount, it won't even be listed and hence not be available for the submission of demand bids.

6.2.6 Duplicating one Day's Demand Bids

If you are interested in initializing your demand bids at a location with the bids that you submitted on another day, you can use the tool provided in the tooltray.

۵	Deman	d Bids		
Se	lect Date ar	nd Node		~
	Date:	20-Aug-2014		•
	Portfolio:	Mkt_Adm_All_Pnodes		~
	Node:	UN.JAY_T 46 KCW		~
				_
	Copy To:	21-Aug-2014		•
			Сору	

- 1. View the demand bids for the location and date that you want to copy to another date (§6.2.2)
- 2. In the "**Copy To**" field select the date on the drop-down calendar to which you want to copy your demand bids.
- 3. Select the **Copy** button in the tooltray to copy all of your demand bids, fixed and price-sensitive, for the selected subaccount when applicable. When the operation completes, you will notice that the "**Date**:" filter field is set to the date that you entered/selected.

Note: The request to copy your bid data will be rejected when the selected subaccount is terminated on the selected "**Copy To**" date. **Note**: You cannot use the **Copy** functionality to duplicate bids from one subaccount to another subaccount; rather, it can only be used to duplicate bids from one market day to another for a selected subaccount.

6.3 Increment Offers and Decrement Bids (Inc/Dec) Display

This display is used to view and enter incremental offers or decrement bids, also known as *virtual bids*, at a pricing location for each hour of a market day, Virtual bids can be made at all price node location (Pnode) for which ISO calculates an LMP. Virtual bids are supported in the Day-Ahead market only. Increment offers and decrement bids are used to indicate the prices at which you are willing to increase an injection (increment offer) or a withdrawal (decrement bids). An increment offer or decrement bid can be used to:

- Cover one side of a bilateral transaction
- Protect a day-ahead generation offer
- Cover congestion

The format of the display used to submit virtual bids is shown below:

																				eMarket		Inte	rface
ISO new england														Demand I	Market S			·		Certificate User: Superu	ser	100 C 100 C 100	2015 10:30
•																Pa	articipan	t: ACME	Inc		Y User:	Superuser	
	Regulation																						
Private Portfolio Demand Inc/De Messages Manager Bids Bids	C Load Response Offers	onse Demand By Portfolio	Load P Result	Respor Is		1L wnload																	
Inc/Dec Bids																							
Select Date and Node	<u>«</u>	Decrement Bids fo	or .Z.CO	NNECT	ICUT (4	004) on 2	0-Jur	-2015															
Date: 20-Jun-2015		Hour	+	*	MW	Price	*	MW	Price	*	MW	Price	*	MW	Price	*	MW	Price	Total				
Portfolio: Mkt_Adm_All_Pnodes	~	Cop		1			2			3			4			5							
Node: .Z.CONNECTICUT	~	01		1	180.0	1.15		112.0		3		110.60	4	5.6					308.8				
		02	_	1	180.0	1.15		112.0	11.50		11.2	110.60	4	5.6	178.96	5			308.8 180.0				
		03		1	180.0 180.0	1.15				3			4			5			180.0				
		05			180.0	1.15		112.0	11.50		11.2	110.60	4	5.6	178.96				308.8				
Сору То:		06			180.0	1.15		112.0		3		110.60	4	5.6		5			308.8				
		07		1	180.0	1.15		112.0		3		110.60	4	5.6		5			308.8				
Hour	Canu	08		1	180.0	1.15	2			3			4			5			180.0				
Copy	Сору	09		1	180.0	1.15	2			3			4			5			180.0				
<u> </u>		10	7 0	1	180.0	1.15	2	112.0	11.50	3	11.2	110.60	4	5.6	178.96	5			308.8				
01		11		1	180.0	1.15	2	112.0	11.50	3	11.2	110.60	4	5.6	178.96	5			308.8				
02		\mathbf{X}																				Update	Refrest
03		Increment Offers	for .Z.C	ONNEO	TICUT (4004) on	20-Ji	in-2015															
04		Heat	+	#	MW	Price	#	MW	Price	#	MW	Price	#	MW	Price	#	MW	Price	Total				
05	m k	V V Cop		1			2			3			4			6							
06		01		1	5.6	18.10				3			4			5			5.6 313.8				
07		02		1	5.6 5.6	18.10 18.10		11.2		3	112.0		4	185.0	226.78	5			313.8				
08		03	_	1	5.6	18.10		112.0 11.2			185.0 112.0		4	185.0	226.78				313.8				
09		05		1	5.6			11.2		3	112.0	113.00	4	105.0	220.70	5			16.8				
10		06			5.6	18.10		11.2		3	112.0	115.00	4	185.0	226.78		[313.8				
11		07		1	5.6			112.0		3			4			5			117.6				
12		08		1	5.6	18.10	2	11.2		3	112.0	115.00	4	185.0	226.78	5			313.8				
		09		1	5.6	18.10	2	11.2	36.25	3	112.0	115.00	4	185.0	226.78	5			313.8				
		10		1	5.6	18.10	2	11.2	36.25	3	112.0	115.00	4	185.0	226.78	5			313.8				
		11		1	5.6	18.10	2	11.2	36.25	3	112.0	115.00	4	185.0	226.78	5			313.8				
																						Update	Refresh
Messages																							

This display is divided into two identical sections – the upper is for decrement bids and the lower is for increment offers. Each has the same usage and fields:

- **Hour** Hour ending values are shown in the first column for each hour of the day. The row for hour ending 03 is not shown for the Spring Daylight Saving Time transition day. Similarly, the value on the row of the repeating hour ending 02 of the Fall Daylight Saving Time transition day is marked with an asterisk (*).
- **#** This is the number of a price-sensitive demand bid for the hour.
- **MW** The MW amount of the virtual bid.

- **Price** The price of the virtual bid. For an incremental offer, this is the price above which you are offering to pick up an obligation to sell energy; i.e. if the price at the Pnode you selected is equal to or greater than this offer price value, you are offering to sell the number of MW specified in your offer. Conversely for a decrement bid, this is the price below which you are willing to buy energy; i.e. if the price at the Pnode you selected is less than or equal to this bid price value, you are offering to buy the number of MW specified in your offer.
- **Total** The MW total of the bids for the hour.

Without exploding the rows, up to five virtual bids per hour can be entered on the display. However, each row can be exploded to provide the ability to enter up to fifty virtual bids per hour as shown below. You do this by clicking on the "+" icon at the beginning of the row; to collapse the row again, you click on the "-" icon at the beginning of the row.

Note: when the market is CLOSED, you will be **able to explode a row only if there are additional bids for the hour** that are not shown and the row will be exploded only enough to show those additional bids. Hence if you have five or less bids for an hour, you won't be able to explode the row to see more empty cells.

ISO new england												Dem	hand Mar		:: OPEN(01) Participant:	EMarket User Interface Versioni 2.0.10014072211190 Certificate Userni Usgenuer 23-Mug-2014 15122124 ISO New England End Versioni Usernier 30-Mug-2014 15122124
Public Generation Demand ARD Regulation																
Private Portfolio Demand Inc/Dec Load Response		Load Response	XML													
Messages Manager Bids Bids Offers	Portfolio	Results	Download													
Inc/Dec Bids																
Select Date and Node		s for UN.JAY_T 4														
Date: 25-Aug-2014	Hour	+ *	MW	Price #		Price	*	MW	Price M	100	Price		MOL	Price	Total	
Portfolio: Mkt_Adm_All_Phodes	01	Copy 🗉 1		2		5.01			4			5			38.0	
Node: UNJAY T 46 KCW	01	■ 1 ● 6	18.0	0.00 2		5.01	3		4			5 10			38.0	
Notes divident_1 for Kow		11		12			13		14			15				
		16		17			18		16			20				_
29-Apr-2012		21		23			23		24			26				
		26		27			28		25			30				
Copy To:		31		33			33		34			35				
Copy		36		33			38 43		36			40 45				
	/	46		4			48		4			50				
	02	0 1	18.0	0.00 2	18.0		з		4			5			36.0	
	03	0 1	18.0	0.00 2			з		4			5			18.0	-
																Update Refresh
	Increment Offe	ers for UNJAY_T	46 KCW on 2	25-Aug-201	•											
	Hour	+ N	MIV	Price #	MAN	Price		MW	Price #	1.00	Price		MN	Price	Total	<u>-</u>
	× ×	Copy 🔲 1		2			3		4			6				
	01	8 1		2			3		4			5				
		0		7			8		9			10				
		11		13			13		1.			15				_
E		16		15			18		16		_	20				
		21		23			23		24			25				
		26		27			28		29		_	30				
		31		30			33		34			35				
		36		33			38 43		30			40 45		\vdash		
				40			43		4							
							40									
	02	46		45			48		44			50 5				
	02			41			48 3		4			50 6				Update Refresh

The format of the display is slightly different for Participants who have registered subaccounts in the ISO-NE Customer and Asset Management System (CAMS). As shown below, there is an additional selection drop-down box labeled "SubAccount" in the **Filters** pane that is blank when the display is first called up. The user must select one of the subaccounts appearing on the drop-down list before the display data entry fields are shown. Only subaccounts active on the selected market day and inactive subaccounts for which there are demand bids on the selected market day will be listed. Inactive subaccounts will appear with an asterisk (*) appended to their name. See §1.8 for more information on subaccounts.

		eMarket User Interface
ISO new england	Demand Market Status: OPEN(01) Participant: ACME Inc	Certificate User: Superuser 19-Jun-2015 10:36:17
Private Concession Demaid MO Reputation Private Portfolio Demaid Mo//Dec Codd Reporter Demaid By Load Reporter Demaid By Select Date and Hode Image: Select Date and Hode	Participant ACHE Inc	1 Partiopant Total
Hessages		8

An additional column of information is shown on the display:

• **Participant Total** – The cumulative MW value of all virtual bids submitted by the Participant users across all subaccounts for the hour.

ISO new england												De	mand	larket Sta	atus: O	PEN(01)			Version	arket Use : 4.0.3(201506161306 ser: Superuser		
													manan				ACME		certificate o.		Superuser	
c Generation Demand ARD Re	gulation																					
ate Portfolio Demand Inc/Dec Bages Manager Bids Bids	Load Resp Offers	ponse Demand By Portfolio	Load Resp Results		ML Jownload																	
c/Dec Bids	Otters	Portrollo	Results		Jownload																	
ct Date and Node	«	Decrement Bids f	or .Z.CONNE	CTICUT (4004) on 2	0-Jun-	2015															
Date: 20-Jun-2015		Hour	+ #	MW	Price		MW	Price		MW	Price	*	MW	Price		MW	Price	Total	Participant Total			
Portfolio: Mkt Adm All Pnodes	*	~ ~ Co	py 🛛 1			2			3			4			5				Iotal			
Node: "Z.CONNECTICUT	~	01	0 1	145.6	7.60	2	15.7	26.40	3	12.3	45.30	4	11.1	48.60	5	8.0	51.20	197.8	198.9			
bAccount: Test5	~	02	□ 1	145.6	7.60	2	15.7	26.40	3	12.3	45.30	4	11.1	48.60	5	8.0	51.20	197.8	197.8			
		03	⊞ 1	145.6	7.60	2	15.7	26.40	3	12.3	45.30	4	11.1	48.60	5	8.0	51.20	197.8	197.8			
		04	1	145.6	7.60	2	15.7	26.40	3	12.3	45.30	4	11.1	48.60	5	8.0	51.20	197.8	364.9			
		05	1	145.6	7.60	2	15.7	26.40	3	12.3	45.30	4	11.1	48.60	5	8.0	51.20	197.8	364.9			
Copy To:		06	1	145.6	7.60	2	15.7	26.40	3	12.3	45.30	4	11.1	48.60	5	8.0	51.20	197.8	364.9			
	_	07	□ 1	145.6	7.60	2	15.7	26.40	3	12.3	45.30	4	11.1	48.60	5	8.0	51.20	197.8	364.9			
	Сору	08	E 1				15.7		3	12.3	45.30	4	11.1	48.60	5	8.0	51.20	197.8	364.9			
		09	1				15.7	26.40	3	12.3	45.30	4	11.1	48.60	5	8.0	51.20	197.8	364.9			
		10	1			2	15.7	26.40	3	12.3	45.30	4	11.1	48.60	5	8.0	51.20	197.8	364.9			
			m 1	146.6	7.60	²	15 7	06.40	3	10.0	45.20	4	44.4	10 60	5	0.0	17	107.8	364.0		Update	Re
		Increment Offers	for .Z.CON	ECTICUT	(4004) on	20-:	Not	ice t	hat	t th	e to	als	are	e diff	ere	ent	r					
		Hour	+ #	MW	Price	#	for	com	٥h		rs he	cai		ther	0 21	r۵	Price	Total	Participant			
		~ ~ Co	ov 🗆 1																Total			
		01	0 1	5.5	7.60	2	virt	ual k	bids	s to	r thc	se	hou	irs o	n a		99.50	254.1	254.1			
		02	0 1				diff	eren	it si	uba	ссоі	int.					99.50	254.1	254.1			
		03	. 1			2	5				2000						99.50	2511	516.6			
		04	□ 1	5.5	7.60	2	8.5	26.45	3	11.5	45.56	4	15.9	52.30	5	56.7	99.50	254.1	516.6			
		05	⊡ 1	5.5	7.60	2	8.5	26.45	3	11.5	45.56	4	15.9	52.30	5	56.7	99.50	254.1	516.6			
		06	1	5.5	7.60	2	8.5	26.45	3	11.5	45.56	4	15.9	52.30	5	56.7	99.50	254.1	516.6			
		07	1	5.5	7.60	2	8.5	26.45	3	11.5	45.56	4	15.9	52.30	5	56.7	99.50	254.1	516.6			
		08	1	5.5	7.60	2	8.5	26.45	3	11.5	45.56	4	15.9	52.30	5	56.7	99.50	254.1	516.6			
		09	⊡ 1	5.5	7.60	2	8.5	26.45	3	11.5	45.56	4	15.9	52.30	5	56.7	99.50	254.1	516.6			
		10	1	5.5	7.60	2	8.5	26.45	3	11.5	45.56	4	15.9	52.30	5	56.7	99.50	254.1	254.1			
					7.00	2	0 5	26.45	2	44 E	46 60		15.0	60.00		60 7	00.60	254.1	254.1			

6.3.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- MW values must be greater than zero; i.e. MW > 0.
- The Price of a bid must be less than or equal to the Energy Offer Cap price and greater than or equal to the Energy Floor price; i.e. Energy Offer Floor price ≤ Price ≤ Energy Offer Cap price²³.

Increment offers and decrement bids for a given hour can be entered in any order; each is treated independently when clearing the Day-Ahead market. Since each is treated independently, entering multiple duplicate offer/bids for a given hour is equivalent to entering a single bid where the MW value is the sum of the MW values of the duplicate offer/bids; for example, three decrements bids for hour-ending 05 of 2 MW @ \$10.00 is equivalent to one bid of 6 MW @ \$10.00.

²³ Energy Offer Floor price and Energy Offer Cap price are both defined in Market Rule 1, section I.2.2.

6.3.2 Submit/Modify Virtual Bids (Decrement Bids / Increment Offers)

Increment offers look like spot market sales or generation offers. Decrement bids look like spot market purchases or price-sensitive demand bids.

The method to enter/modify them using the Inc/Dec Bids display is the same; the only difference is that they are entered separately, decrement bids in the top portion of the display, increment offers in the low portion.

- 1. In the Demand Navigation sub tabs, click the Inc/Dec Bids tab.
 - The Inc/Dec Bids display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired price location from the **Node** filter.
 - d. Select the desired subaccount from the **SubAccount** filter if it is shown.
 - The data for the selected price location displays.
- 3. To submit virtual bids for individual hours
 - a. Enter the **MW** amount and the **Price** of the bids in the cells provided for the hour of the day desired. Explode the row for the hour to give you access to the 45 additional pairs of cells to enter bids. To delete a bid, you need to clear the MW and Price values of the bid in the cells.
 - b. Repeat 3a for all hours as needed.
 - c. Click the **Update** button to save the data. Remember that you have to submit Increments and Decrements separately and independently.
 - The totals column of each affected hour is updated to reflect the MW total of the bids; the Participant totals column if it is shown is also updated.
- 4. To submit the same virtual bid for a range of hours:
 - a. Enter the **MW** amount and the **Price** of the bids in the cells provided on the top row. Explode the row to give you access to the 45 additional pairs of cells to enter bids.
 - b. Select the range of hours in the top row. Then select the "Copy" button. This will copy all of the values on the top row (including blanks) to the hourly rows for the range of hours selected.



- c. Click the **Update** button to save the data for the selected hours.
 - The totals column of each affected hour is updated to reflect the MW total of the bids; the Participant totals column if it is shown is also updated.

6.3.3 Deleting Ranges of Virtual Bids

If you want to delete the incremental offers or decrements bids at a location for a range of hours of a given date, you can use the top row of the grids to do this.

- 1. Leave all the data entry cells of the top row empty.
- 2. Select the range of hours in the top row. Then select the "Copy" button. This will copy the blanks on the top row to the hourly rows for the range of hours selected.



3. Click the **Update** button. All of the existing offer/bids shown on the display will be deleted.

Note: If you select an *inactive* subaccount, you can **ONLY** delete virtual bids; you cannot modify the bids that exist on the selected day, nor can you submit new bids. The inactive subaccount will appear on the drop-down select list **only** if there are virtual bids for the subaccount on the selected market day; if there are no bids for the subaccount, it won't even be listed and hence not be available for the submission of virtual bids.

6.3.4 Duplicating one Day's Virtual Bids

If you are interested in initializing your virtual bids at a location with the bids that you submitted on another day, you can use the tool provided in the tooltray.

I	nc/Dec	Bids		
Se	elect Date ar	nd Node		«
	Date:	20-Aug-2014		
	Portfolio:	Mkt_Adm_All_Pnodes		~
	Node:	UN.JAY_T 46 KCW		~
	Copy To:	21-Aug-2014		•
			Copy	

- 1. View the virtual bids for the location and date that you want to copy to another date (§6.2.2)
- 2. In the "**Copy To**" filed select the date on the drop-down calendar to which you want to copy your virtual bids.
- Select the Copy button in the tooltray to copy all of your virtual bids, for the selected subaccount when applicable. When the operation completes, you will notice that the "Date:" filter field is set to the date that you entered/selected.

Note: The request to copy your bid data will be rejected when the selected subaccount is terminated on the selected "**Copy To**" date.

Note: You cannot use the **Copy** functionality to duplicate bids from one subaccount to another subaccount; rather, it can only be used to duplicate bids from one market day to another for a selected subaccount.

6.4 Load Response Offer Display

This display is used to view and update Load Response offers. These offers are used to indicate the Day-Ahead price at which you are willing to interrupt a specified amount of load in Real-Time, if cleared.

To access the Load Response Offer you must be a registered user with a Load Response Read/Write or Read-Only role. When you have this access, the Load Response Offers display sub-tab on the Demand display tab will be shown.

Public Generation	Demand	ARD R	tegulation DRR			
Private Portfolio		Inc/Dec	Load Response	Demand By	Load Response	XML
Messages Manager		Bids	Offers	Portfolio	Results	Download

Note: The Load Response roles will be retired/discontinued sometime after the implementation of the Price-Responsive Demand (PRD) project in June 2018.

A Load Response offer can be submitted for a single day or a range of days, as specified by the Start Day and Stop Day. The maximum duration of an offer is the end of the next month if the threshold price for the next month has been published; otherwise it is the end of the current month. In other words, a Load Response offer for the next month can only be posted if the threshold price for the next month has been published, which is normally done by the middle of the month.

The format of this display is shown below:

ISO	iew engla	ind							De	mand Market Status: C	PEN(01)	, ve		01407221119)	Interface 23-Aug-2014 15:34:31
										Part	icipant: ISO N	New England Inc		💙 User:	Superuser 💙
Public Generation	Demand	ARD	Regulation												
Private Portfolio Messages Manager	Demand Bids	Inc/Dec Bids	Load Resp Offers	nse Dema Portfo		Load Response Results	XML Download								
Load Respo	nse Off	ers													
Select Date and Load	Response			K Load Res	esponse Of	ffers for Mall of	America RTE	OR for Aug-2014							
Date:	Aug-2014			s	Start Day		op Day clusive)	MW	Price	Curtailment Initiation Price	Offer Cost	Minimum Interruption Duration	MIC	DRTP	
Load Response:		ica RTDR		08-	8-Aug-2014	1 20-/	Aug-2014	2.0	45.00	28.00	48.50	4	6.725	28.0	
				21-	I-Aug-2014	25-/	Aug-2014	1.0	40.00	28.00	47.00	4	6.725	28.0	
				26-	3-Aug-2014	1 05-9	Sep-2014	0.111	31.00	24.00	85.05	4	6.725	28.0	
									Note: Offers	will not apply to weekend d	lays or holidays.	New De	lete		Update Refresh

The fields on this display are:

- **Start Day** the first day that the offer is effective. The earliest day that a Load Response offer can be effective is the day following the current day
- **Stop Day** the last day (inclusive) that the offer is effective.
- **MW** the amount of load interruption offered. A single MW interruption amount applies to all program hours per day.

- **Price** the price per MWH at which the amount of load interruption is offered. A single offer price applies to all program hours per day.
- **Curtailment Initiation Price** the price per interruption at which the amount of interruption is offered. The Curtailment Initiation Price is paid each time a load curtailment is scheduled.
- Offer Cost The offer cost is computed as follows: offer cost = offer price + [curtailment initiation price / (minimum interruption duration x bid MW amount)]
- **Minimum Interruption Duration** the minimum amount of time that the resource must be scheduled to interrupt, if cleared.
- Maximum Interruptible Capacity (MIC) an estimate of the maximum hourly demand reduction amount that a Load Response resource can deliver.
- **Demand Reduction Threshold Price (DRTP)** the minimum offer price calculated pursuant to Market Rule 1, Appendix E. This value is calculated every month. The value shown in the column is the DRTP applicable to the month selected in the filter tooltray.

6.4.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- Start Day must be in the future, i.e. after the current day
- **Stop Day** must be later than or equal to the Start Day, i.e. ≥ Start Day
- Stop Day is subject to the limit of horizon months. The Stop Day date must fall in a month where the **Demand Response Threshold Price (DRTP)** is defined; in other words, Stop Day date ≤ date of the last day of a month where the DRTP is defined. Since the DRTP of a month will be defined and published around the 15th of the previous month, offers for the next month can be submitted only after the DRTP is published.
- The **MW** interruption amount must be greater than or equal to 0.1 MW and less than the minimum of the **Maximum Interruptible Capacity (MIC)** values of the Load Response resource for the dates spanned by the offer at the time the offer is submitted. Note that the monthly MIC of a resource can change at any time, even after an offer has been submitted and accepted. The market clearing software will use the lesser of the MIC effective for the market day and the MW amount of the offer.
- Offer price ≤ the maximum defined in Market Rule 1, Appendix E (e.g. \$1,000/MWH). Also, it must be greater than the maximum of the monthly Demand Response Threshold Price (DRTP) for the months spanned by the offer. The DRTP is calculated on a monthly basis as defined in Market Rule 1, Appendix E and posted to the ISO New England website by the 15th day of the preceding month in advance of the Demand Reduction Threshold Price's effective date. The DRTP applies to only the next month and does not change the DRTP for the current month.

- Offer Cost calculated value cannot exceed the maximum defined in Market Rule 1, Appendix E (e.g. \$1,000/MWH).
- **Curtailment Initiation Price** must be less than the maximum defined in Market Rule 1, Appendix E (e.g. \$1,000/MWH).
- **Minimum Interruption Duration** must be greater than or equal to 1 hour and less than or equal to 4 hours, and must be in whole hourly increments.

6.4.2 View a Load Response Offer

1. In the Demand Navigation sub tabs, click the **Load Response Offers** tab.

The Load Response Offers display is shown in the display pane.

- 2. In the **Filters pane**:
 - a. Select the desired month from the **Date** filter.
 - b. Select the desired resource from the Load Response filter.
 - If any offers exist for the selected resource for the selected month, the offer(s) will be shown in the display and available for update.

6.4.3 Create a new Load Response offer

- 1. View the offers for the Load Response resource (§6.4.2)
- 2. Click the **New** button; you can click more than once if you want to create multiple new offers.
- 3. Enter the following information in the data entry cells provided:
 - a. Start Day
 - b. Stop Day
 - c. MW
 - d. Price
 - e. Curtailment Initiation Price (optional it defaults to \$0.00)
 - f. Minimum Interruption Duration
- 4. Click the **Update** button to save the data.
- 5. Repeat steps 2 & 3 above to create additional offers.

6.4.4 Modify an existing Load Response offer

- 1. View the offers for the Load Response resource (§6.4.2)
- 2. Modify the data in the data entry cells for the offers shown as needed.
- 3. Click the **Update** button to save the data.

6.4.5 Delete a Load Response Offer

- 1. View the offers for the Load Response resource (§6.4.2)
- 2. Click a cell in the row that you want to delete and hit Enter. The cell will be highlighted in pink.
- 3. Click the **Delete** button.

6.4.6 Delete a Range of Load Response Offers

- 1. View the offers for the Load Response resource (§6.4.2)
- 2. Click a cell in each of the rows that you want to delete and hit Enter. Each cell will be highlighted in pink.
- 3. Click the **Delete** button. Each row with a cell highlighted is deleted.

6.5 Demand by Portfolio Display

You can view the net amount of your demand and virtual bids that cleared in the Day-Ahead Market at each of the Pnodes listed in your selected portfolio using the **Demand by Portfolio** display that is accessible on one of the Demand sub tabs. The information shown includes the name of the Pnode, the net MW amount of demand/virtual bids that cleared and the nodal price for each hour of the day. The 24-hour period is shown on two lines: the top line details the hours ending 01 through 12; the second line details hours ending 13 through 24. The last row of the display is labeled **Total Portfolio MW** and shows the total net MW amount of your bids for each hour that cleared in the Day-Ahead market for all the Pnodes listed. Information on the display is shown only after the Day-Ahead market has been cleared and the time for the opening of the Re-Offer Period of the selected day has passed.

														eMarket Us Version: 4.0.3(201506161)	
SO new england								De	mand Mark	et Status:	OPEN(01)			Certificate User: Superuser	19-Jun-201
									ReadOnl	y Pi	articipant:	ISO New	England In	c 🗸 Vs	sr: Superuser
blic Generation Demand ARD Regulation															
ivate Portfolio Demand Inc/Dec Load Res essages Manager Bids Bids Offers			XML Download												
ssages Manager Bids Bids Offers	POPTIOIIO	esuits I	Download												
•	By Portfolio for MBP o	n 20. Jun. 2015													
	Node	1 20-341-2013													
Date: 20-Jun-2015		01/13	02/14	03/15	04 / 16	05/17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24		
Portfolio: MBP															
	Z.CONNECTICUT	01/13	02/14	03/15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24		
		16.8@ \$18.00	16.8@ \$15.64	0@ \$13.00	0@ \$13.00	16.8@ \$13.00	16.8@ \$15.58	16.8@ \$15.45	0@ \$18.00	0@ \$17.99	16.8@ \$18.00	16.8@ \$17.97	11.2@ \$20.94		
		11.2@	11.2@	11.2@	11.2@	11.2@	11.2@	11.2@	16.8@	16.8@	16.8@	16.8@	16.8@		
		\$18.68	\$20.99	\$20.99	\$20.99	\$20.73	\$20.97	\$19.89	\$18.00	\$18.07	\$18.00	\$16.04	\$15.84		
	.Z.NEMASSBOST	01 / 13	02/14	03/15	04 / 16	05/17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24		
										46.101	10.100		10.101		
	.Z.RHODEISLAND	01/13	02/14	03/15	04 / 16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11 / 23	12/24		
	.Z.SEMASS	01/13	02/14	03/15	04/16	05 / 17	06/18	07/19	08/20	09/21	10/22	11 / 23	12/24		
	.2.0LIIIA33	17.1@	16.2@	15.8@	15.7@	16.1@	17.5@	20.8@	23.7@	25.7@	27@	28@	28.7@		
		\$18.00	\$15.64	\$13.00	\$13.00	\$13.00	\$15.58	\$15.45	\$18.00	\$17.99	\$18.00	\$69.83	\$16.68		
		28.9@	29@	28.7@	28.2@	27.7@	27.3@	26.7@	25.9@	26.3@	25.1@	21.4@	18.1@		
Portfolio totals		\$18.00	\$18.01	\$18.01	\$18.01	\$20.73	\$20.97	\$18.01	\$18.00	\$18.00	\$18.00	\$16.04	\$15.84	_	
shown on last	Total Portfolio MW	01 / 13	02/14	03/15	04 / 16	05/17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24		
		33.9	33.0	15.8	15.7	32.9	34.3	37.6	23.7	25.7	43.8	44.8	39.9		
rows of the display		40.1	40.2	39.9	39.4	38.9	38.5	37.9	42.7	43.1	41.9	38.2	34.9		

The format of this display is shown below:

The fields on this display are:

- **Node** the Pnode name of the price location.
- **MW** @ **Price** the net MW amount that cleared in the market and the LMP at which the bids cleared. Be aware that in the calculation of the net MW amount, the MW amount of increment offers that cleared are treated as negative quantities.

The format of the display is slightly different for Participants who register subaccounts in the ISO-NE Customer and Asset Management System (CAMS). As shown below, there is an additional selection drop-down box labeled "SubAccount" in the **Filters** pane that is blank when the display is first called up. The user must select one of the subaccounts appearing on the drop-down list before data is shown on the display. Only subaccounts active on the selected market day or inactive subaccounts for which there are demand bids on the selected market day will be listed. Inactive subaccounts will appear with an asterisk (*) appended to their name. See §1.8 for more information on subaccounts.

ISO new england		Demand Market Status: OPEN(0		eMarket User Interface Version: 4.0.3(201506161306) Certificate User: Superuser 19-Jun-2015 14:31:36
			ACME Inc	V User: Superuser
Public Generation Demand ARD Regulation				
Private Portfolio Demand Inc/Dec Load Response Demand By Messages Manager Bids Differs Portfolio	Load Response XML Download			
Demand By Portfolio				
Select Date and Portfolio 🦿 The day ahead	market results are not available for MBP on 20-Jun-2015			
Date: 20-Jun-2015	SubAccount: 15941 DOMR NRG Test5			
				Refresh
Messages				8

When a subaccount is selected, the net amount of the demand and virtual bids that were submitted in the selected subaccount and that cleared in the Day-Ahead Market at each of the Pnodes listed in the selected portfolio is shown on the display, along with a total for the portfolio.

ISO new england									De	mand Mark ReadOnl		OPEN(01) articipant:	ACME Inc		eMarko Version: 4.0.3(Sertificate User: Su	2015061613 peruser	•	06)	06) 19	06) 19-Ju	06)	06) 19-Jun-20	06) 19-Jun-2015	19-Jun-2015 1
vate Portfolio Demand Inc/Dec	ulation		Load Response	XML																				
essages Manager Bids Bids	Offers	Portfolio	Results	Download																				
lect Date and Portfolio	« E	By Portfolio for MB	3P on 20-Jun-201	5																				
Date: 20-Jun-2015	° _	Node																						
Portfolio: MBP	~	.H.INTERNAL_HUB	01/13	02/14	03/15	04 / 16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24										
ubAccount: Test5	~																							
		Z.CONNECTICUT	01/13	02/14	03/15	04 / 16	05/17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24										
			216.6@	216.6@	216.6@	216.6@	216.6@	216.6@	216.6@	216.6@	216.6@	216.6@	46.7@	46.7@										
			\$18.00 46.7@	\$15.64 46.7@	\$13.00 46.7@	\$13.00 46.7@	\$13.00 46.7@	\$15.58 46.7@	\$15.45 46.7@	\$18.00 46.7@	\$17.99 46.7@	\$18.00 46.7@	\$17.97 46.7@	\$20.94 46.7@										
			\$18.68	\$20.99	\$20.99	\$20.99	\$20.73	\$20.97	\$19.89	\$18.00	\$18.07	\$18.00	\$16.04	\$15.84										
		Z.NEMASSBOST	01/13	02 / 14	03/15	04 / 16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24										
		Z.RHODEISLAND	01/13	02/14	03/15	04/16	05/17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24										
		Z.SEMASS	01 / 13	02/14	03/15	04 / 16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24										
Deutfalie tetale																								
Portfolio totals		Total Portfolio MW	01/13	02/14	03/15	04/16	05/17	06/18	07 / 19	08/20	09/21	10/22	11/23	12/24	1									
shown on last			216.6	216.6	216.6	216.6	216.6	216.6	216.6	216.6	216.6	216.6	46.7	46.7										
rows of the display			46.7	46.7	46.7	46.7	46.7	46.7	46.7	46.7	46.7	46.7	46.7	46.7										
															_									R
ssages																								

6.5.1 View DA Cleared Bids at Pnodes in Portfolio

- 1. In the Demand Navigation sub tabs, click the **Demand by Portfolio** tab.
 - The Demand by Portfolio display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired subaccount from the **SubAccount** filter if it is shown.
 - Results are displayed for all Pnodes in the selected portfolio and for the selected date and when applicable, for the selected subaccount.

6.6 Load Response Results Display

You can view the results of the Load Response clearing which includes the cleared results for each hour for each of your Load Response resources that were offered into the Day-Ahead Market for the specified date. This data is available after the Day-Ahead market clears.

To access the Load Response Results you must be a registered user with a Load Response Read/Write or Read-Only role. When you have this access, the Load Response Results display sub-tab on the Demand display tab will be shown.

Public Ge	eneration	Demand	ARD	Regulation	DRR			
		Demand Bids	Inc/Dec Bids	Load Resp Offers		Demand By Portfolio	Load Response Results	XML Download

Note: The Load Response roles will be retired/discontinued sometime after the implementation of the Price-Responsive Demand (PRD) project in June 2018.

ISO new england									itatus: CLOSE					.1(201407	User Interfa 221119) 23-Aug-2014 11
the second s									Participant	ISO N	ew England I	nc		~	User: Superuser
blic Generation Demand ARD Regulation															
ivate Portfolio Demand Inc/Dec Load Response		Response	XML												
issages Manager Bids Bids Offers	Portfolio Result	ts	Download	d											
oad Response Results															
ct Date 《	Load Response Results	on 22-Aug	-2014												
	Load Response Name														
Date: 22-Aug-2014	Providence Place Mall	RTDR 0	1713	02 / 14	03 / 15	04716	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12 / 24	
	Providence Place Mall	RTDR								0@	0@	0@	0@	0@	
	Providence Place Mall	RTDR								\$18.92	\$29.41	\$24.61	\$25.97	\$26.60	
			0@	0@	0@	0@	0@	0@							
				\$31.50	\$31.73	\$32.04	\$32.04	\$33.71							
	Rhode Island Hospital-I Rhode Island Hospital-I		1713	02 / 14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22 0@	11/23 0@	12 / 24 0@	
	Rhode Island Hospital-I									\$18.92	\$29.41	\$24.61	\$25.97	\$26.60	
	Rifede Island Hospital-I		0@	0@	0@	0@	0@	0@		\$10.9Z	429.91	\$24.01	420.97	\$20.00	
				\$31.50	\$31.73	\$32.04	\$32.04	\$33.71							
	Verso Androscoggin G2	Inc 0	1713	02/14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12/24	
	Verso Androscoggin G2	Inc								0@	0@	0@	0@	0@	
	Verso Androscoggin G2	Inc								\$17.87	\$21.62	\$23.58	\$25.41	\$26.56	
			0@	0@	0@	0@	0@	0@							
		\$	27.69	\$30.75	\$30.81	\$31.04	\$31.04	\$30.46							
	Verso Androscoggin Loa		1 / 13	02 / 14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12/24	
	Verso Androscoggin Loa									0@	0@	0@	0@	0@	
	Verso Androscoggin Lo.									\$17.87	\$21.62	\$23.58	\$25.41	\$26.56	
			0@ 27.69	0@ \$30.75	0@ \$30.81	0@ \$31.04	0@ \$31.04	0@ \$30.46							
	Total MW			02/14	\$30.81 03 / 15	04/16	\$31.04 05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12/24	
										0	0	0	0	0	
			0	0	0	0	0	0							
															R

The format of this display is shown below:

The following fields are on this display:

- **MW** The MW amount of Load Response cleared at the location.
- **Price** The Day-Ahead Market LMP at the location.

6.6.1 View Day-Ahead Load Response Cleared Offers

- 1. In the Demand Navigation sub tabs, click Load Response Result.
 - The Load Response Result display is shown in the display pane.
- 2. Select the desired date from the **Date** filter located in the **Filters pane**.
- 3. Review the information on the display.

6.7 XML Download for Demand Display

The XML Download display is accessible through the Demand Navigation sub tab. This display enables you to download pre-defined Demand specific XML report files of data that you maintain in the ISO-NE Market System, including results from the cleared Day-Ahead market. It also provides you the means to upload data in XML formatted data files into the ISO-NE Market System to be used to clear the Day-Ahead and Real Time markets. The format of the XML files can be found in the **eMarket Data Exchange Specification**, which is found on the <u>ISO New England website</u> among the documents listed on the Participate > Support > Upload and Download File Format Protocols page.

ISO new england		Demand Market Status: CLOSED	Version: 4.0.3(2015 Certificate User: Superus	er 19-Jun-2015 15:27:03
Public Generation Demand ARD Regulation Private Portfolo Demand Inc/Dec Load Respute Manager Bids Offers Offers	nse Demand By Load Response XML Portfolio Results Download	Participant: ISO New Engla	nd Inc	Vuser: Superuser Y
XML Download Select Date Date: 20-Jun-2015	XML Download on 20-Jun-2015			
	Bids (Demand, Incs, Decs) Demand By Participant Detailed Demand by Participant Load Response Results	Upload XHL SOAP Requ	Browse	
	Load Response Offers			
Messages				

The format of the display is slightly different for Participants who register subaccounts in the ISO-NE Customer and Asset Management System (CAMS). As shown below, there is an additional selection drop-down box labeled "SubAccount" in the **Filters** pane that is blank when the display is first called up. The user can select one of the subaccounts appearing on the drop-down list before requesting a download of the reports. If a subaccount is not selected, the requested demand report will contain information for all of the Participant's subaccounts. Only subaccounts active on the selected market day and inactive subaccounts for which there are demand bids on the selected market day will be listed. Inactive subaccounts will appear with an asterisk (*) appended to their name. See §1.8 for more information on subaccounts.

ISO new england	Demand Market Status:	CLOSED ticipant: ACME Inc	eMarket User Interface Version: 4.0.3(201506161306) Certificate User: Superuser 19-Jun-2015 15:33:31 Voer: Superuser V
Public Generation Demand ARD Regulation Private Portfolio Demand Inc,Dec Load Resp Hersages Manager Bids XML Download Select Date (Date: 20-Jun-2015 SubAccount:		Upload XHL SOAP Request	: Browse
Messages			

The following XML reports of demand data are available for downloading:

- Bids (Demand, Increment, Decrement)
- Demand by Participant
- Detailed Demand by Participant
- Load Response Results
- Load Response Offers

Each returns an XML formatted report for the selected date and for all the Pnodes listed in all of the user's portfolios or for every Load Response resources that belongs to you as a Lead Participant and when applicable, for each of the user's subaccounts unless a specific one is selected. To access the Load Response reports you must be a registered user with a Load Response Read/Write or Read-Only role. When you have this access, the Load Response reports selections on the display will be shown.

Note: The Load Response roles will be retired/discontinued sometime after the implementation of the Price-Responsive Demand (PRD) project in June 2018.

See §3.2 for details on uploading and downloading files. The following sections provide samples of the downloaded reports.

6.7.1 Bids XML Report

This report contains the information that is shown on the Demand Bids and Inc/Dec Bids displays for all Pnodes listed in all of your portfolios, and when applicable, for each of your subaccounts or for the subaccount that you selected. The following is a sample of the report. If you are using subaccounts, each set of *DemandBid* information appears within *DemandBidBySubAccount* tags that identify the name of the subaccount as follows:

```
<PricePoint price="17.15" MW="3" />
           <PricePoint price="15.15" MW="5" />
       </HourlyBid>
      - <HourlyBid time="2014-08-24T01:00:00-04:00">
           <PricePoint price="222" MW="99" />
           <PricePoint price="17.15" MW="3" />
           <PricePoint price="15.15" MW="5" />
       </HourlyBid>
      - <HourlyBid time="2014-08-24T23:00:00-04:00">
           <PricePoint price="15.15" MW="2" />
           <PricePoint price="11.15" MW="2" />
       </HourlyBid>
    </HourlyProfile>
</DemandBid>
- <DemandBid bidType="Increment" day="2014-08-24" ID="4000">
    <NodeName>.H.INTERNAL_HUB</NodeName>
   - <HourlyProfile>
      - <HourlyBid time="2014-08-24T00:00:00-04:00">
           <PricePoint price="25.15" MW="3" />
           <PricePoint price="23.15" MW="2" />
       </HourlyBid>
      - <HourlyBid time="2014-08-24T01:00:00-04:00">
           <PricePoint price="25.15" MW="3" />
           <PricePoint price="22.15" MW="2" />
       </HourlyBid>
```

```
</DemandBid>
- <DemandBid bidType="Decrement" day="2014-08-24" ID="4004">
    <NodeName>.Z.CONNECTICUT</NodeName>
   - <HourlyProfile>
      - <HourlyBid time="2014-08-24T00:00:00-04:00">
           <PricePoint price="34.02" MW="30" />
           <PricePoint price="17.15" MW="3" />
           <PricePoint price="15.15" MW="5" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T23:00:00-04:00">
           <PricePoint price="34.02" MW="20" />
           <PricePoint price="15.15" MW="2" />
           <PricePoint price="11.15" MW="2" />
        </HourlyBid>
    </HourlyProfile>
 </DemandBid>
- <DemandBid bidType="Increment" day="2014-08-24" ID="4004">
    <NodeName>.Z.CONNECTICUT</NodeName>
   - <HourlyProfile>
      - <HourlyBid time="2014-08-24T00:00:00-04:00">
           <PricePoint price="25.15" MW="3" />
           <PricePoint price="23.15" MW="2" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T01:00:00-04:00">
           <PricePoint price="25.15" MW="3" />
           <PricePoint price="22.15" MW="2" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T23:00:00-04:00">
           <PricePoint price="25.15" MW="3" />
           <PricePoint price="22.15" MW="2" />
        </HourlyBid>
    </HourlyProfile>
</DemandBid>
- <DemandBid bidType="Fixed" day="2014-08-24" ID="4004">
    <NodeName>.Z.CONNECTICUT</NodeName>
      - <HourlyProfile>
           <HourlyBid time="2014-08-24T00:00:00-04:00" fixedMW="100" />
           <HourlyBid time="2014-08-24T01:00:00-04:00" fixedMW="101" />
           <HourlyBid time="2014-08-24T02:00:00-04:00" fixedMW="102" />
           <HourlyBid time="2014-08-24T22:00:00-04:00" fixedMW="122" />
           <HourlyBid time="2014-08-24T23:00:00-04:00" fixedMW="123" />
    </HourlyProfile>
 </DemandBid>
- <DemandBid bidType="PriceSensitive" day="2014-08-24" ID="4004">
    <NodeName>.Z.CONNECTICUT</NodeName>
   - <HourlyProfile>
      - <HourlyBid time="2014-08-24T00:00:00-04:00">
           <PricePoint price="400.13" MW="13.1" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T01:00:00-04:00">
           <PricePoint price="400.13" MW="11.9" />
        </HourlyBid>
```

```
- <HourlyBid time="2014-08-24T02:00:00-04:00">
           <PricePoint price="400.13" MW="11.4" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T22:00:00-04:00">
           <PricePoint price="400.13" MW="15.3" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T23:00:00-04:00">
           <PricePoint price="400.13" MW="14.2" />
        </HourlyBid>
    </HourlyProfile>
 </DemandBid>
- <DemandBid bidType="Decrement" day="2014-08-24" ID="4007">
    <NodeName>.Z.WCMASS</NodeName>
   - <HourlyProfile>
      - <HourlyBid time="2014-08-24T00:00:00-04:00">
           <PricePoint price="222" MW="44" />
           <PricePoint price="17.15" MW="3" />
           <PricePoint price="15.15" MW="5" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T23:00:00-04:00">
           <PricePoint price="222" MW="33" />
           <PricePoint price="15.15" MW="2" />
           <PricePoint price="11.15" MW="2" />
        </HourlyBid>
    </HourlyProfile>
 </DemandBid>
- <DemandBid bidType="Increment" day="2014-08-24" ID="4007">
    <NodeName>.Z.WCMASS</NodeName>
   - <HourlyProfile>
      - <HourlyBid time="2014-08-24T00:00:00-04:00">
           <PricePoint price="25.15" MW="3" />
           <PricePoint price="23.15" MW="2" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T01:00:00-04:00">
           <PricePoint price="25.15" MW="3" />
           <PricePoint price="22.15" MW="2" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T23:00:00-04:00">
           <PricePoint price="25.15" MW="3" />
           <PricePoint price="22.15" MW="2" />
        </HourlyBid>
    </HourlyProfile>
 </DemandBid>
- <DemandBid bidType="PriceSensitive" day="2014-08-24" ID="4007">
    <NodeName>.Z.WCMASS</NodeName>
   - <HourlyProfile>
      - <HourlyBid time="2014-08-24T00:00:00-04:00">
           <PricePoint price="400.13" MW="52.1" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T01:00:00-04:00">
           <PricePoint price="400.13" MW="46.7" />
```

```
</HourlyBid>
```

6.7.2 Demand by Participant XML Report

This report shows the net amount of your demand and virtual bids that cleared in the Day-Ahead Market at each of the distinct Pnodes that are listed in all of your Demand portfolios, and when applicable, for each of your subaccounts or for the subaccount that you selected, and the price at which they cleared. Be aware that in the calculation of the net MW amount cleared the MW amount of Increment offers that cleared are treated as negative quantities. The report contains information only after the time for the opening of the Re-Offer Period of the selected day has passed. The following is a sample of the report. If you are using subaccounts, each set of *DemandByParticipant* information appears within *DemandByParticipantBySubAccount* tags that identify the name of the subaccount as follows:

- <DemandByParticipantBySubAccount subAccount ="SubaccountName">

- <DemandByParticipant day="..." ID="...">

</DemandByParticipant>

</DemandByParticipantBySubAccount>

Sample Report:

<?xml version="1.0" encoding="UTF-8" standalone="yes" ?>

- <GetDemandByParticipantResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">

- <DemandByParticipant day="2014-08-24" ID="4000">
 - <NodeName>.H.INTERNAL_HUB</NodeName>

- <HourlyProfile>

```
<HourlyValues time="2014-08-24T00:00:00-04:00" MW="99" price="21.61" />
       <HourlyValues time="2014-08-24T01:00:00-04:00" MW="99" price="19.11" />
       <HourlyValues time="2014-08-24T02:00:00-04:00" MW="99" price="20.15" />
       <HourlyValues time="2014-08-24T03:00:00-04:00" MW="99" price="19.19" />
       <HourlyValues time="2014-08-24T04:00:00-04:00" MW="99" price="19.94" />
       <HourlyValues time="2014-08-24T05:00:00-04:00" MW="99" price="21.12" />
       <HourlyValues time="2014-08-24T06:00:00-04:00" MW="99" price="23.95" />
       <HourlyValues time="2014-08-24T07:00:00-04:00" MW="55" price="25.49" />
       <HourlyValues time="2014-08-24T08:00:00-04:00" MW="99" price="31.7" />
       <HourlyValues time="2014-08-24T09:00:00-04:00" MW="97" price="37.58" />
       <HourlyValues time="2014-08-24T10:00:00-04:00" MW="30" price="35.51" />
       <HourlyValues time="2014-08-24T11:00:00-04:00" MW="-8" price="34.09" />
       <HourlyValues time="2014-08-24T12:00:00-04:00" MW="-9" price="32.93" />
       <HourlyValues time="2014-08-24T13:00:00-04:00" MW="-103" price="26.14" />
       <HourlyValues time="2014-08-24T14:00:00-04:00" MW="-99" price="24.61" />
       <HourlyValues time="2014-08-24T15:00:00-04:00" MW="-99" price="24.11" />
       <HourlyValues time="2014-08-24T16:00:00-04:00" MW="-99" price="24.04" />
       <HourlyValues time="2014-08-24T17:00:00-04:00" MW="-99" price="25.83" />
       <HourlyValues time="2014-08-24T18:00:00-04:00" MW="-99" price="25.84" />
       <HourlyValues time="2014-08-24T19:00:00-04:00" MW="-99" price="31.99" />
       <HourlyValues time="2014-08-24T20:00:00-04:00" MW="99" price="36.85" />
       <HourlyValues time="2014-08-24T21:00:00-04:00" MW="94" price="33.68" />
       <HourlvValues time="2014-08-24T22:00:00-04:00" MW="-2" price="23.53" />
       <HourlyValues time="2014-08-24T23:00:00-04:00" MW="-2" price="23.67" />
    </HourlyProfile>
 </DemandByParticipant>
- <DemandByParticipant day="2014-08-24" ID="4004">
    <NodeName>.Z.CONNECTICUT</NodeName>
   - <HourlyProfile>
```

<HourlyValues time="2014-08-24T00:00:00-04:00" MW="43.1" price="21.86" />

<HourlyValues time="2014-08-24T01:00:00-04:00" MW="41.9" price="19.29" /> <HourlyValues time="2014-08-24T02:00:00-04:00" MW="41.4" price="20.34" /> <HourlyValues time="2014-08-24T03:00:00-04:00" MW="30.8" price="19.37" /> <HourlyValues time="2014-08-24T04:00:00-04:00" MW="10.7" price="20.09" /> <HourlyValues time="2014-08-24T05:00:00-04:00" MW="11" price="21.35" /> <HourlyValues time="2014-08-24T06:00:00-04:00" MW="11.5" price="23.86" /> <HourlyValues time="2014-08-24T07:00:00-04:00" MW="12.4" price="25.52" /> <HourlyValues time="2014-08-24T08:00:00-04:00" MW="13.3" price="31.86" /> <HourlyValues time="2014-08-24T09:00:00-04:00" MW="12.4" price="38.03" /> <HourlyValues time="2014-08-24T10:00:00-04:00" MW="12.1" price="35.93" /> <HourlyValues time="2014-08-24T11:00:00-04:00" MW="7.1" price="34.52" /> <HourlyValues time="2014-08-24T12:00:00-04:00" MW="35.6" price="33.24" /> <HourlyValues time="2014-08-24T13:00:00-04:00" MW="40.2" price="26.41" /> <HourlyValues time="2014-08-24T14:00:00-04:00" MW="13.7" price="24.83" /> <HourlyValues time="2014-08-24T15:00:00-04:00" MW="13.3" price="24.33" /> <HourlyValues time="2014-08-24T16:00:00-04:00" MW="13.2" price="24.15" /> <HourlyValues time="2014-08-24T17:00:00-04:00" MW="13.5" price="25.99" /> <HourlyValues time="2014-08-24T18:00:00-04:00" MW="14.2" price="25.98" /> <HourlyValues time="2014-08-24T19:00:00-04:00" MW="15.1" price="32.18" /> <HourlyValues time="2014-08-24T20:00:00-04:00" MW="16" price="37.09" /> <HourlyValues time="2014-08-24T21:00:00-04:00" MW="61.1" price="33.92" /> <HourlyValues time="2014-08-24T22:00:00-04:00" MW="33.3" price="23.5" /> <HourlyValues time="2014-08-24T23:00:00-04:00" MW="32.2" price="23.82" /> </HourlyProfile> </DemandByParticipant> - <DemandByParticipant day="2014-08-24" ID="4007"> <NodeName>.Z.WCMASS</NodeName> - <HourlyProfile> <HourlyValues time="2014-08-24T00:00:00-04:00" MW="96.1" price="21.75" /> <HourlyValues time="2014-08-24T01:00:00-04:00" MW="90.7" price="19.23" /> <HourlyValues time="2014-08-24T02:00:00-04:00" MW="88.8" price="20.3" /> <HourlyValues time="2014-08-24T03:00:00-04:00" MW="88" price="19.34" /> <HourlyValues time="2014-08-24T04:00:00-04:00" MW="89.3" price="20.08" /> <HourlyValues time="2014-08-24T05:00:00-04:00" MW="93.4" price="21.28" /> <HourlyValues time="2014-08-24T06:00:00-04:00" MW="75" price="24.07" /> <HourlyValues time="2014-08-24T07:00:00-04:00" MW="83.9" price="25.62" /> <HourlyValues time="2014-08-24T08:00:00-04:00" MW="93.3" price="31.88" /> <HourlyValues time="2014-08-24T09:00:00-04:00" MW="96.8" price="37.84" /> <HourlyValues time="2014-08-24T10:00:00-04:00" MW="98.3" price="35.77" /> <HourlyValues time="2014-08-24T11:00:00-04:00" MW="36.5" price="34.33" /> <HourlyValues time="2014-08-24T12:00:00-04:00" MW="35" price="33.14" /> <HourlyValues time="2014-08-24T13:00:00-04:00" MW="36.7" price="26.33" /> <HourlyValues time="2014-08-24T14:00:00-04:00" MW="38.6" price="24.78" /> <HourlyValues time="2014-08-24T15:00:00-04:00" MW="37.1" price="24.28" /> <HourlyValues time="2014-08-24T16:00:00-04:00" MW="39.4" price="24.18" /> <HourlyValues time="2014-08-24T17:00:00-04:00" MW="40.7" price="25.99" /> <HourlyValues time="2014-08-24T18:00:00-04:00" MW="40.3" price="25.99" /> <HourlyValues time="2014-08-24T19:00:00-04:00" MW="44.1" price="32.17" /> <HourlyValues time="2014-08-24T20:00:00-04:00" MW="115.4" price="37.03" /> <HourlyValues time="2014-08-24T21:00:00-04:00" MW="104.4" price="33.84" /> <HourlyValues time="2014-08-24T22:00:00-04:00" MW="98.3" price="23.64" /> <HourlyValues time="2014-08-24T23:00:00-04:00" MW="90.3" price="23.78" /> </HourlyProfile>

</DemandByParticipant>

</GetDemandByParticipantResponse>

6.7.3 Detailed Demand by Participant XML Report

This report lists each of your demand and virtual bids that cleared in the Day-Ahead Market at each of the distinct Pnodes that are listed in all of your Demand portfolios, and when applicable, for each of your subaccounts or for the subaccount that you selected. The report contains information only after the time for the opening of the Re-Offer Period of the selected day has passed.

It essentially is a breakdown of the bids that contributed to the net MW amount shown in the Demand by Participant report. If nothing cleared for a market hour, the Demand by Participant report will show a net MW amount of zero and the Detailed Demand by Participant report will list nothing. Note however that if the Demand by Participant report shows a net MW amount of zero, it does not mean that none of your bids cleared; you may have submitted demand or decrements bids that are countered by increment offers of total equal magnitude resulting in a net MW amount of zero since in the calculation of the net MW amount cleared the MW amount of Increment offers that cleared are treated as negative quantities. Note also that the sum of the MW amounts of the demand and decrement bids and increment offers listed for a given hour may not exactly match the net MW amount shown in the Demand by Participant report; this will happen when one of the bid/offers is marginal, thereby setting the price at the Pnode, which implies that only a portion of the MW amount bid/offered cleared.

The report lists the bids by type exactly as you've submitted them. In other words it does not report the MW amount of the bid that cleared, although it will be the same unless the bid is marginal, nor does it report the price at which the bid cleared but the bid price.

The following is a sample of the report. If you are using subaccounts, each set of *DetailedDemandByParticipant* information appears within *DetailedDemandByParticipantBySubAccount* tags that identify the name of the subaccount as follows:

```
- <DetailedDemandByParticipantBySubAccount subAccount ="SubaccountName">
- <DetailedDemandByParticipant bidType ="..." day="..." ID="...">
```

</DetailedDemandByParticipant> </DetailedDemandByParticipantBySubAccount>

Sample Report:

```
<PricePoint price="222" MW="0" />
        </HourlyBid>
    </HourlyProfile>
 </DetailedDemandByParticipant>
- <DetailedDemandByParticipant bidType="Increment" day="2014-08-24" ID="4000">
    <NodeName>.H.INTERNAL_HUB</NodeName>
   - <HourlyProfile>
      - <HourlyBid time="2014-08-24T21:00:00-04:00">
           <PricePoint price="25.15" MW="2" />
           <PricePoint price="30.15" MW="3" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T22:00:00-04:00">
           <PricePoint price="22.15" MW="2" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T23:00:00-04:00">
           <PricePoint price="22.15" MW="2" />
        </HourlyBid>
    </HourlyProfile>
 </DetailedDemandByParticipant>
- <DetailedDemandByParticipant bidType="Decrement" day="2014-08-24" ID="4004">
    <NodeName>.Z.CONNECTICUT</NodeName>
   - <HourlyProfile>
      - <HourlyBid time="2014-08-24T00:00:00-04:00">
           <PricePoint price="34.02" MW="30" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T01:00:00-04:00">
           <PricePoint price="34.02" MW="30" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T23:00:00-04:00">
           <PricePoint price="34.02" MW="20" />
        </HourlyBid>
    </HourlyProfile>
 </DetailedDemandByParticipant>
- <DetailedDemandByParticipant bidType="Increment" day="2014-08-24" ID="4004">
    <NodeName>.Z.CONNECTICUT</NodeName>
   - <HourlyProfile>
      - <HourlyBid time="2014-08-24T21:00:00-04:00">
           <PricePoint price="25.15" MW="2" />
           <PricePoint price="30.15" MW="3" />
        </HourlyBid>
       - <HourlyBid time="2014-08-24T22:00:00-04:00">
           <PricePoint price="22.15" MW="2" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T23:00:00-04:00">
           <PricePoint price="22.15" MW="2" />
        </HourlyBid>
    </HourlyProfile>
</DetailedDemandByParticipant>
- <DetailedDemandByParticipant bidType="PriceSensitive" day="2014-08-24" ID="4004">
    <NodeName>.Z.CONNECTICUT</NodeName>
   - <HourlyProfile>
      - <HourlyBid time="2014-08-24T00:00:00-04:00">
           <PricePoint price="400.13" MW="13.1" />
        </HourlyBid>
      - <HourlyBid time="2014-08-24T01:00:00-04:00">
           <PricePoint price="400.13" MW="11.9" />
```

```
</HourlyBid>
```

```
- <HourlyBid time="2014-08-24T23:00:00-04:00">
             <PricePoint price="400.13" MW="14.2" />
          </HourlyBid>
      </HourlyProfile>
   </DetailedDemandByParticipant>
  - <DetailedDemandByParticipant bidType="Decrement" day="2014-08-24" ID="4007">
      <NodeName>.Z.WCMASS</NodeName>
     - <HourlyProfile>
         - <HourlyBid time="2014-08-24T00:00:00-04:00">
             <PricePoint price="222" MW="44" />
          </HourlyBid>
         - <HourlyBid time="2014-08-24T01:00:00-04:00">
             <PricePoint price="222" MW="44" />
          </HourlyBid>
         - <HourlyBid time="2014-08-24T23:00:00-04:00">
             <PricePoint price="222" MW="33" />
          </HourlyBid>
      </HourlyProfile>
   </DetailedDemandByParticipant>
  - <DetailedDemandByParticipant bidType="Increment" day="2014-08-24" ID="4007">
      <NodeName>.Z.WCMASS</NodeName>
     - <HourlyProfile>
         - <HourlyBid time="2014-08-24T21:00:00-04:00">
             <PricePoint price="25.15" MW="2" />
             <PricePoint price="30.15" MW="3" />
          </HourlyBid>
         - <HourlyBid time="2014-08-24T22:00:00-04:00">
             <PricePoint price="22.15" MW="2" />
          </HourlyBid>
         - <HourlyBid time="2014-08-24T23:00:00-04:00">
             <PricePoint price="22.15" MW="2" />
          </HourlyBid>
      </HourlyProfile>
   </DetailedDemandByParticipant>
  - <DetailedDemandByParticipant bidType="PriceSensitive" day="2014-08-24" ID="4007">
      <NodeName>.Z.WCMASS</NodeName>
     - <HourlyProfile>
         - <HourlyBid time="2014-08-24T00:00:00-04:00">
             <PricePoint price="400.13" MW="52.1" />
          </HourlyBid>
         - <HourlyBid time="2014-08-24T01:00:00-04:00">
             <PricePoint price="400.13" MW="46.7" />
          </HourlyBid>
         - <HourlyBid time="2014-08-24T23:00:00-04:00">
             <PricePoint price="400.13" MW="59.3" />
          </HourlyBid>
      </HourlyProfile>
   </DetailedDemandByParticipant>
</GetDetailedDemandByParticipantResponse>
```

6.7.4 Load Response Offers XML Report

To obtain this report you must be a registered user with a Load Response Read/Write or Read-Only role.

Note: The Load Response roles will be retired/discontinued sometime after the implementation of the Price-Responsive Demand (PRD) project in June 2018.

This report contains the information that you entered and is on the Load Response Offers display for all unique Load Response resources listed in all of your portfolios. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes" ?>
- <GetLoadResponseOfferResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
   - <LoadResponseOffer name="Pratt & Whitney Middletown">
       <AssetID>31994</AssetID>
       <CurtailmentInitiationPrice>0</CurtailmentInitiationPrice>
       <MinimumInterruptionDuration>1</MinimumInterruptionDuration>
       <OfferMW>0.111</OfferMW>
       <OfferPrice>31</OfferPrice>
       <StartDay>2014-08-12</StartDay>
       <StopDay>2014-08-30</StopDay>
    </LoadResponseOffer>
   - <LoadResponseOffer name="Pratt & Whitney Middletown">
       <AssetID>31994</AssetID>
       <CurtailmentInitiationPrice>70</CurtailmentInitiationPrice>
       <MinimumInterruptionDuration>1</MinimumInterruptionDuration>
       <OfferMW>0.111</OfferMW>
       <OfferPrice>31</OfferPrice>
       <StartDay>2014-09-01</StartDay>
       <StopDay>2014-09-05</StopDay>
    </LoadResponseOffer>
   - <LoadResponseOffer name="Pratt & Whitney Middletown">
       <AssetID>31994</AssetID>
       <CurtailmentInitiationPrice>55</CurtailmentInitiationPrice>
       <MinimumInterruptionDuration>2</MinimumInterruptionDuration>
       <OfferMW>2</OfferMW>
       <OfferPrice>29</OfferPrice>
       <StartDay>2014-09-06</StartDay>
       <StopDay>2014-09-14</StopDay>
    </LoadResponseOffer>
   - <LoadResponseOffer name="Pratt & Whitney Middletown">
       <AssetID>31994</AssetID>
       <CurtailmentInitiationPrice>95</CurtailmentInitiationPrice>
       <MinimumInterruptionDuration>1</MinimumInterruptionDuration>
       <OfferMW>2</OfferMW>
       <OfferPrice>32</OfferPrice>
       <StartDay>2014-09-15</StartDay>
       <StopDay>2014-09-21</StopDay>
    </LoadResponseOffer>
   - <LoadResponseOffer name="Pratt & Whitney Middletown">
       <AssetID>31994</AssetID>
       <CurtailmentInitiationPrice>65</CurtailmentInitiationPrice>
       <MinimumInterruptionDuration>2</MinimumInterruptionDuration>
       <OfferMW>1.1</OfferMW>
       <OfferPrice>29</OfferPrice>
       <StartDay>2014-09-22</StartDay>
```

<StopDay>**2014-09-31**</StopDay> </LoadResponseOffer> </GetLoadResponseOfferResponse>
6.7.5 Load Response Results XML Report

To obtain this report you must be a registered user with a Load Response Read/Write or Read-Only role.

Note: The Load Response roles will be retired/discontinued sometime after the implementation of the Price-Responsive Demand (PRD) project in June 2018.

This report contains the same information that is shown on the Load Response Results display. This report shows the total MW amount of load reduction by resource and the price at which they cleared for each hour of the day. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes" ?>
- <GetLoadResponseResultResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
   - <LoadResponseResult day="2014-08-24" ID="37812">
       <Name>Madison Paper Industries</Name>
      - <HourlyProfile>
           <HourlyValues time="2014-08-24T07:00:00-04:00" LMP="24.98" MW="0" />
           <HourlyValues time="2014-08-24T08:00:00-04:00" LMP="31.34" MW="0" />
           <HourlyValues time="2014-08-24T09:00:00-04:00" LMP="36.43" MW="0" />
           <HourlyValues time="2014-08-24T10:00:00-04:00" LMP="34.26" MW="0" />
           <HourlyValues time="2014-08-24T11:00:00-04:00" LMP="33.05" MW="0" />
           <HourlyValues time="2014-08-24T12:00:00-04:00" LMP="32.05" MW="0" />
           <HourlyValues time="2014-08-24T13:00:00-04:00" LMP="25.61" MW="0" />
           <HourlyValues time="2014-08-24T14:00:00-04:00" LMP="24.05" MW="0" />
           <HourlyValues time="2014-08-24T15:00:00-04:00" LMP="23.57" MW="0" />
           <HourlyValues time="2014-08-24T16:00:00-04:00" LMP="23.52" MW="0" />
           <HourlyValues time="2014-08-24T17:00:00-04:00" LMP="25.3" MW="0" />
       </HourlyProfile>
    </LoadResponseResult>
   - <LoadResponseResult day="2014-08-24" ID="31994">
       <Name>Pratt & Whitney Middletown</Name>
      - <HourlyProfile>
           <HourlyValues time="2014-08-24T07:00:00-04:00" LMP="25.52" MW="0" />
           <HourlyValues time="2014-08-24T08:00:00-04:00" LMP="31.86" MW="0" />
           <HourlyValues time="2014-08-24T09:00:00-04:00" LMP="38.03" MW="0" />
           <HourlyValues time="2014-08-24T10:00:00-04:00" LMP="35.93" MW="0" />
           <HourlyValues time="2014-08-24T11:00:00-04:00" LMP="34.52" MW="0" />
           <HourlyValues time="2014-08-24T12:00:00-04:00" LMP="33.24" MW="0" />
           <HourlyValues time="2014-08-24T13:00:00-04:00" LMP="26.41" MW="0" />
           <HourlyValues time="2014-08-24T14:00:00-04:00" LMP="24.83" MW="0" />
           <HourlyValues time="2014-08-24T15:00:00-04:00" LMP="24.33" MW="0" />
           <HourlyValues time="2014-08-24T16:00:00-04:00" LMP="24.15" MW="0" />
           <HourlyValues time="2014-08-24T17:00:00-04:00" LMP="25.99" MW="0" />
       </HourlyProfile>
    </LoadResponseResult>
 </GetLoadResponseResultResponse>
```

7. Asset-Related Demand (ARD) Tab

To access the Generation Tab, you must be a registered user with an ARD Read/Write or Read-Only role. When you have this access, the ARD tab will be shown and selecting it will show all of its display sub tabs as shown below.

Public	Ge	eneration D	emand	ARD	Regulation	n				
Priva Mess								Schedule Offers Hourly Updates		

7.1 ARD Offer Concepts

An ARD asset is a physical load, either dispatchable or non-dispatchable, that has been discretely modeled within the ISO New England's Energy Management and Settlement systems that settles at a nodal location. Dispatchable ARD assets (aka DARDs) are loads that can be dispatched just like generators to increase or decrease consumption and get nodal pricing as opposed to load zone pricing as is the case for demand in general. Non-dispatchable ARD assets are special loads that satisfy all of the ISO-NE metering and are eligible for nodal pricing; unlike DARDs, they are not dispatchable and as such are simple price takers in the energy market. The following discussions are applicable to dispatchable ARDs.

When submitting data, it is important to understand the hierarchy of the ARD data and the implications of data overrides. Figure 9 below shows the hierarchy of the ARD displays used to submit data and where the same bid data appears in different displays, the data of which display has precedence over (overrides) the others. [**Note**: only the data elements subject to overrides are shown.]

The ARD Default Parameters display is at the root of the hierarchy. The ARD record itself is initially created by ISO New England with default bid data when the asset is instantiated in the Market system. Participants must update some data values, like the asset's default dispatch availability status and single value ramp rate, using the ARD Default Parameters display. For assets capable of providing off-line reserves, the default Claim10 and Claim30 values must be updated on this display.

To participate in the energy market, a Participant must define at least one schedule for the asset using the Schedule Manager display and all of the child data that goes with it. The schedule defines a container for which detail bid data must be submitted.

Managing ARD schedules is accomplished by performing the following general steps:

- Create or modify a schedule using the Schedule Manager display
- Create or modify schedule-specific default data, which include MW operating limits and market participation (day-ahead or real time) using the Schedule Detail Defaults display.

• Create or modify schedule-bid default data (MW and price) using the Schedule Offers Default display.



Figure 9 – ARD Display Hierarchy and Data Precedence

• Identify the schedule or schedules to be used in the clearing of the Day-Ahead and Real Time (Balancing) Markets using the Schedule Selection display. The other displays allow for overriding either schedule based default data or an asset's default dispatch availability data. Bid data in other displays higher in the hierarchy are overridden by the presence of data in these displays:

- The ARD Ramp Rate Default display can be used to define a default ramp rate curve for an asset that overrides the simple slope defined with the ARD Default Parameters display.
- The Minimum Consumption and Maximum Consumption limits on the Hourly Updates display override for specified hours of the day the corresponding limit values defined in the Schedule Detail Defaults display for all schedules.
- The dispatch availability status (Unavailable, Economic, Must Run) on the Hourly Updates display overrides for specified hours of the day the default dispatch availability status (Default Status) defined with the ARD Default Parameters display. For assets capable of providing off-line reserves, the display can be used to override for specific hours of the day the default Claim10 and Claim30 values defined on the ARD Default Parameters display.
- The incremental energy offers (Price/MW pairs) shown on the Schedule Offers Hourly Updates display override for specified hours of the day the default values defined in the Schedule Offers Default display for the corresponding schedule.

Note: For dispatchable ARDs that are part of a Continuous Storage Facility (CSF), refer to chapter 10 Continuous Storage Facilities (CSF) for additional considerations.

7.2 ARD Portfolio Manager Display

An ARD portfolio is a collection of ARD assets that is defined by each individual user. The user assigns a unique name to each portfolio when it is created. Once a portfolio is created, the user assigns it one or more ARD assets. An ARD asset can be assigned to more than one portfolio.

Data is filtered by ARD portfolios on the following displays:

- ARD Default Parameters
- ARD Multiple Ramp Rates Default
- ARD Ramp Rate Hourly Updates
- ARD Schedule Manager
- ARD Schedule Detail Defaults
- ARD Schedule Offers Default
- ARD Schedule Offers Hourly Updates
- ARD Schedule Selection
- ARD Hourly Updates
- ARD by Portfolio

Note: An asset that is not associated with a portfolio cannot be selected in these displays' tooltray filters.

See section 3.1, Manage Portfolios, for details on the purpose of portfolios and details using this display.

7.3 ARD Default Parameters Display

The ARD Default Parameters display allows you to view asset ownership information and to update an asset's default ramp rate, dispatch status and when applicable, its off-line reserve Claim Capability values. ARD bid data is defined on an asset basis, as well as on schedule basis. The ARD Default Parameters Display is for managing an ARD's more static bid data.

ISO New England requires that the Participant submit to the ISO, asset specific information about an ARD. This information is used to create the initial ARD asset record in the market system database and must be submitted in writing to ISO-NE and verified by ISO-NE before the Participant submits bids. (For more information on the data required, see the *ISO New England Manual for Market Operations Manual 11*, which is found on the *ISO New England website* among the documents listed on the Participate > Rules and Procedures > Manuals page.)

When the asset-specific information is received, ISO creates the necessary records in the market system database. Some of this information is shown on the ARD Default Parameters display, and users cannot modify the information from here.

ISO new england										eMarket Version: 2.0.1(20140	7221119)	
								Market Status: OPEN(01)		Certificate User: Superuse		Aug-2014 12:31
								Participant:	ISO New England Inc	*	User: Superu	ser
ublic Generation Demand ARD Regulation												
	o Rate Schedule Schedul Ily Updates Manager Detail De		Schedule Offers Hourly Updates	Schedule Selection	Hourly Updates	ARD By Portfolio	30ML Download					
ARD Daily Default Parameters												
•	ARD Details for ASSET NAME	on 24-Aug-2014										
Date: 24-Aug-2014												
Portfolio: AAA								Asset short Name:				
ARD: ASSET NAME	Asset Number:	888888888							AR.STATION_KV_NOD	ENAME		
ASSET NAME	Asset Longname:	ASSET NAME						DA Schedule Required:	No			
	Lead Participant(ID#):	88888888						Ramp Rate(MW/Min):	1.0			
	Lead Participant Name:	Participant Name						Ramp Rate Cap:	11			
	Dispatch Zone:	Nik-Stam						Claim 10 Capable:	Yes			
	Dispatchable:	Yes						Claim 10 (MW):	7.0			
								Claim 10 Cap:	7			
	Default Status:	Economic						Claim 30 Capable:	Yes			
		C Unavailable						Claim 30 (MW):	12.0			
		C Must Run						Claim 30 Cap:				
		- Hast Hall										
											Updat	e Refresh
issages												

The format of the ARD Default Parameters display is shown below:

The fields on this display are:

- **Asset Type** the physical type of the asset (e.g. Pump vs Non-Pump).
- Asset Number the ISO-NE generated unique identifier for the asset. This is the preferred reference number used in doing business with the Customer Service Department.
- **Node** the name of the pricing node, or bus, at which the ISO calculates dayahead and real-time LMPs and to which the asset is connected. ISO assigns

the pricing node for each ARD based on the asset's telemetry and on the ISO state estimator model.

- Asset Short Name the abbreviated name of the ARD asset
- Asset Long Name the full name or description of the Asset Related Demand. This label is shown on other ARD displays and in the ARD filter of all ARD displays.
- Lead Participant ID the ISO-NE assigned unique identifier of the Market Participant whose users are responsible for submitting the operating information of the ARD using the eMarket application.
- Lead Participant Name the full name of the Market Participant whose users are responsible for submitting the operating information of the ARD asset using the eMarket application.
- **Dispatch Zone** the electrical dispatch zone in which the ARD reside.
- **Dispatchable** indication as to whether the asset is a DARD and can respond to ISO New England dispatch signals or simply an ARD nodal price-taker.
- **Default Status (Economic/Unavailable/Must Run)** the default dispatch availability status of the ARD; applicable only to DARDs.
- **DA Schedule Required** indicates whether or not the ARD must submit a bid into the Day-Ahead market.
- Ramp Rate (MW/Min) the number of MW per minute with which the asset can respond to dispatch signals when it is on-line. This value is used in the calculation of reserves unless it is overridden by a default ramp rate curve on the Ramp Rate Default display or an override for an hour defined on the Ramp Rate Hourly Updates display.
- Ramp Rate Cap the asset's demonstrated ramp rate capability value that was established during the last audit conducted by ISO New England. The bid value will be capped (constrained) to this value if it is defined. If the demonstrated capability value is undefined, it will be shown as "UD".
- Claim 10 Capable indicates if the ARD asset is registered as capable of producing Ten Minute Non-Spinning Reserve. Only DARDs are Claim 10 eligible.
- Claim 10 (MW) the submitted bid for the number of consumption MW an asset can shed within 10 minutes. Used in the calculation of Ten Minute Non Spinning Reserves.
- **Claim 10 Cap** the asset's demonstrated Claim 10 Capability value that was established during the last audit conducted by ISO New England. The bid value will be capped (constrained) to this value if it is defined. If the demonstrated capability value is undefined or the asset is not Claim 10 Capable, it will be shown as "UD".
- **Claim 30 Capable** indicates if the ARD asset is registered as capable of producing Thirty Minute Reserve. Only DARDs are Claim 30 eligible.
- Claim 30 (MW) the submitted bid for the number of consumption MW an asset can shed within 30 minutes. Used in the calculation of Thirty Minute (non-spinning) Reserves.

 Claim 30 Cap – the asset's demonstrated Claim 30 Capability value that was established during the last audit conducted by ISO New England. The bid value will be capped (constrained) to this value if it is defined. If the demonstrated capability value is undefined or the asset is not Claim 30 Capable, it will be shown as "UD".

7.3.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- 0 ≤ Ramp Rate
- A warning will be issued if the Ramp Rate exceeds Ramp Rate Cap when the demonstrated capability is defined (not "UD")
- A warning will be returned if the asset is Claim10 capable and any of the following conditions exist
 - Claim 10 = 0 or NULL
 - Claim 10 > Claim 10 Cap when the demonstrated capability is defined (not "UD") meaning that the Claim 10 Cap value will be used instead of the offered value
- A warning will be returned if the asset is Claim 30 capable and any of the following conditions exist
 - Claim 30 = 0 or NULL
 - Claim 30 > Claim 30 Cap when the demonstrated capability is defined (not "UD") – meaning that the Claim 30 Cap value will be used instead of the offered value
 - Claim 30 < Claim 10 meaning that the Claim 10 value will be used instead of the offered Claim 30 value.

7.3.2 View Specific Asset Information:

- 1. In the ARD Navigation sub tabs, click the **ARD Default Parameters** tab.
 - The ARD Default Parameters display is shown in the display pane.
- 2. In the **Filters** pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired asset from the **ARD** filter.
 - The default data for the selected date and asset is shown in the ARD Default Parameters display.

7.3.3 Modify Asset Default Claim 10/30 and Ramp Rate

An asset's Claim 10/30 values and its default ramp rate are specified at the asset level. To define these values:

1. View the desired unit's information as described in §7.3.2

- 2. Enter all of the following information:
 - a. Claim 10 (MW) if Claim 10 Capable
 - b. Claim 30 (MW) if Claim 30 Capable
 - c. Ramp Rate (MW/Min)
- 3. Click the **Update** button to save the data.

7.3.4 Modify Asset Default Status

- 1. View the desired unit's information as described in §7.3.2
- 2. Select one of the following radio buttons:
 - a. Economic
 - b. Unavailable
 - c. Must Run
- 3. Click the **Update** button to save the selection.

7.4 ARD Multiple Ramp Rates Daily Default Display

This display provides for viewing and modifying an asset's default ramp rate curve. A ramp rate defines how quickly an asset can change its consumption in MW per min. Depending on the asset's physical characteristics this rate can vary as a function of the asset's MW consumption. When that is the case, the Participant can use this display to model the asset's ramp rate curve. Ramp rate curves consist of Energy Ramp Rate (MW/Minute) over a range of the asset's MW consumption. Up to 10 MW ranges can be defined. The default ramp rate curve shown on the display is effective for the date selected and for all future days until it is redefined or removed at a future date. Note that the effective date of the data may be a date prior to the date selected. See §1.5 Effective Dates of Data.

Note: The existence of a ramp rate curve on this display overrides the simple slope specified in the Ramp Rate field on the ARD Default Parameters display.

ISC	new england							Market :	Status: OPEN(01) Participant:	ISO New England Inc	Version: 2.0.1(201407221: Certificate User: Superuser	Cer Interface (119) 23-Aug-2014 12:44:26 r: Superuser
Public Gen	eration Demand ARD Regul	ation										
	ARD Default Portfolio Ramp Rate Parameters Manager Default	e Ramp Ra Hourly U		chedule etail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Hour Selection Upda	XML Download				
ARD MI	ultiple Ramp Rates Da	aily Defa	ault									
Select Date an	ad ARD	~	ARD Ramp Rates for A	SSET NAME or	24-Aug-2014							
Date:	24-Aug-2014		MW	Energy Rar								
Portfolio:		~	6		6.6							
ARD:	ASSET NAME	~	10		1.0							
										New	Delete	Update Refresh
Messages												8

The format of this display is shown below:

The fields on this display are:

- **MW** the MW consumption level up to which the ramp rate applies.
- Energy Ramp Rate the MW/Min the ARD can change consumption while its MW consumption is less than or equal to the MW value specified and greater than a lesser MW value specified on a separate row or zero if none. The ramp rate specified for the largest MW value will be used if the ARD's MW consumption is greater than this largest MW point defined for the curve.

7.4.1 Data Validation Rules

• Energy Ramp Rate \geq 0; MW > 0

- eMarket will give you a warning if any of the entered rates is greater than the Ramp Rate Cap shown on the ARD Default Parameters display.
- MW/Ramp Rate pairs can be entered in any order. eMarket will sort them and show them in increasing MW value order after the data is submitted and stored in the database.

7.4.2 View Ramp Rates

To view the ramp rate curve segments of an ARD:

- 1. In the ARD Navigation sub tabs, click the **Ramp Rate Default** tab.
 - The ARD Multiple Ramp Rates Daily Default display is shown in the display pane.

2. In the **Filters pane**:

- a. Select the desired date from the **Date** filter.
- b. Select the desired portfolio from the **Portfolio** filter.
- c. Select the desired asset from the **ARD** filter.
 - The display shows the selected asset's default ramp rate curve data for the selected date; if a ramp rate curve is not defined for the selected date, nothing is shown on the display.

7.4.3 Create Ramp Rates

To create segments of a ramp rate curve for an ARD:

- 1. View the asset's ramp rate curve segments (§7.4.2)
- 2. Click the **New** button; you can click more than once if you want to create multiple segments.
 - A new row for each click is added in the grid with a pink background to indicate it has not been saved.
- 3. Type in the desired **MW** and **Energy Ramp Rate** values in the appropriate cells.
- 4. Repeat steps 2-3 as desired.
- 5. Click the **Update** button to save the data.

7.4.4 Modify Ramp Rates

To update ramp rate curve segments:

- 1. View the asset's ramp rate curve segments (§7.4.2)
- 2. (optional) To add new segments to the curve, do as described in §7.4.3.
- 3. (optional) To delete existing segments in the curve:
 - a. Click a cell in the row to be deleted. The row will be highlighted.
 - b. Click the **Delete** button. The row is removed.

- 4. To change the parameters of a curve segment, enter the desired **MW** and **Energy Ramp Rate** values in the appropriate cells.
- 5. Repeat steps 2-4 as needed.
- 6. Click the **Update** button to save the data.

Note: to use the Ramp Rate defined on the ARD Default Parameters display, starting on the date selected, all Ramp Rate segments must be deleted from this display.

7.5 ARD Ramp Rate Hourly Updates Display

This display provides for viewing, entering and modifying hourly updates to an ARD's default ramp rate characteristics for a given market day. A ramp rate defines how quickly a unit can change its generation in MW per min. Depending on the ARD's physical characteristics this rate can vary as a function of the ARD's MW consumption, in other words the ARD's ramp rate characteristic is a curve defined as ramp rate for up to 10 ranges of the ARD's MW consumption. With this display, you can define an ARD's ramp rate characteristics on an hourly basis for a given market day to be either a single ramp rate or a ramp rate curve and thereby account for the ARD's physical configuration throughout the day.

Note: The existence of a ramp rate definition for an hour on this display overrides both the simple slope specified in the Ramp Rate field on the ARD Default Parameters display and the default ramp rate curve for the day, if one is defined, specified on the ARD Ramp Rate Default display.

The format of this display is shown below. There are two illustrations. The only difference is in the default values shown in the upper section when there is a default ramp rate curve defined for the ARD on the selected market day date and when there is none.

Generation Demand ARD Reput Generation Demand ARD Reput Berdidio Remo Reput Perdidio Remo Reduct	Ramp Rate Hourly Updates		Schedule Detail Default		chedule ffers Defau		dule Offer ly Updates				ARD By Portfolio	J04L Downl	oad					Market Status: OPEN(01) Participanti <u>ISO New England Inc</u>	eMarket User Inter Verson 2.0.2(0140722115) Certificate User Spanuer 23-Aug-201 Vurer Spanuer
Date and ARD		t Values for ASS	SET NAME o	in 24-A	Aug-2014														
Date: 24-Aug-2014			Daily RR		MN	RR	# 540	RR		MI	RR	8 h	6A/ I	R #	MO	RR			
ortfolio: AAA			1.0	8	5.0	6.6	2 1	10 1	8 3	16.0	1.0	4		10			1		
ARD: ASSET NAME																			
HID: HOSET NAME		de Hourly Value	es for ASSE	NAME	E on 24-A	un:2014													
	Hour				MIX		8 MN	RR		MI	RR		W F	R Ø	MN	RR			
	*	👻 Сору		1			2		3			4		5					
				6			7		8			9		10					
		01		1	10.0	10.0	2 1:	0 1	0 3			4		5					
		02		0	5.0	8.0		0 12	8 0 3	12.0	10.0		13.0	1.0 5				Shown only if the	ere is a default ramp
				6	0.0		7		8	12.0	10.0	9	10.0	10					
		03		1	10.0	10.0	2 1:	.0 1	0 3			4		6				rate curve define	ed for the selected
				6			7		8			9		10				market day date.	
		04		1	5.0	8.0	Z 10 Z	0 12	0 3	12.0	10.0	4	13.0	1.0 5				market uay uate.	•
		05		1	10.0	10.0		0 1	0 3			4		5					
				6			7		8			9		10					
		05		1	5.0	8.0		.0 12	0 3	12.0	10.0		13.0	1.0 5					
		07		6	10.0	10.0	7	0 1	8 0 3			9		10					
				6	10.0		7		8			9		10		1			
		08		1	5.0	8.0	2 10	.0 12	0 3	12.0	10.0	4	13.0	1.0 5					
				6			7		8			9		10					
		09		0	10.0	10.0	2 1:	0 1	0 3			4		5					
		10		1	5.0	8.0		0 12	0 3	12.0	10.0	-	13.0	1.0 5					
				6			7		8			9		10					
		11		1	10.0	10.0		.0 1	0 3			4		5					
		12	_	6	5.0	8.0	7	0 12	8	12.0	10.0	9	13.0	10					
				0	0.0		7	w 12	8	12.0	0.01		10.0	1.0 0					
		13		1	10.0	10.0		0 1	0 3			4		5					
				6			7		8			9		10					
							2 1/			12.0	10.0		10.0						

								_		_		eMarket User Interface Version: 2.0.1(201407221119)
ISO new england												et Status: OPEN(01) Certificate User: Superuser 23-Aug-2014 13:02 Participant: ISO New England Ind V User: Superuser
	p Rate rhy Updates Schedule Manager Detail D	lule Schedi Defaults Offers		dule Offers ly Updates		Hourly Updates	ARD By XI Portfolio D	n. xwnload				
lect Date and ARD Date: 24-Aug-2014	C Default Values for ASSET N Daily F		2014								1	
Portfolio: AAA	~											
	Override Hourly Values for											
		RR # MW		8 M// 2	RR #	MO	RR #	MA	RR # MW	RR	\mathbf{N}	
	V V Copy	6		7	3		4		10			
	01			2 13.0					6		<u>\</u> г	
	01	0		7	10 3		9		10		N	Notice that nothing is shown for a
	02		5.0 8.0	2 10.0		12.0	10.0 4	13.0	1.0 5			
		6		7	8		9		10		1	default ramp rate curve if one is not
	03		0.0 10.0				4		6			
	04	6		2 10.0	12.0 3	12.0	9	13.0	1.0 5			defined for the selected market day
	04	0		7	12.0 3	12.0	0.0 4	13.0	10 0			
	05	1 10	0.0 10.0	2 13.0	1.0 3		4		5		1	date.
		6		7	8		9		10			
	00	1 0	5.0 8.0	2 10.0	12.0 3	12.0	10.0 4	13.0	1.0 5		-	
	07		0.0 10.0				4		5			
		0		7	8		9		10			
	08	1 6	5.0 8.0	2 10.0		12.0	10.0 4	13.0	1.0 5			
	09	6		7	8		9		10			
	UV	1 10		2 13.0	1.0 3		4		5			
	10			2 10.0		12.0	10.0 4	13.0	1.0 5			
		0		7	8		9		10			
	11			2 13.0			4		5			
	12	8	5.0 8.0	2 10.0	12.0 3	12.0	9 10.0 4	13.0	10			
	14	6		7 10.0	12.0 3	12.0	10.0 4	13.0	1.0 0			
	13	1 10	0.0 10.0	2 13.0	1.0 3		4		6			
		0		7	8		9		10			
	14			2 100	100 2	10.0	10.0 A	10.0	10 4			Update Ref
												opeare
ages												

The Default Values section is provided as a reference to show what are the default daily values for the chosen day that are being overridden by the hourly update values entered in the section below. The fields in this section of the display are:

• **Daily RR** – the ARD's default energy ramp rate defined on the ARD Default Parameters display for the selected market day date.

 #/MW/RR – the ARD's default ramp rate curve segments or MW blocks; # is the segment number, MW is the ARD's MW consumption upper bound for the block and RR is the ramp rate value for the range of MW consumption from the MW value of the previous segment (0 if there is no previous segment) up to the MW value of the segment. The RR of the last segment is the value used for ARD MW consumptions values greater than the MW value of the last segment.

The Default Values section can be collapsed using the double up-arrow icon at the extreme right of the section title bar to show more rows in the hourly section as shown below. The collapsed section can be expanded again to see the default values using the double down-arrow icon at the extreme right of the collapsed section title bar.

ISO new england																			Market User Interfa m: 2.0.1(201407221119) User: Superuser 23-Aug-2014 1
																		Participant: ISO New England Inc	Superuser
Generation Demand ARD	Regulation																		
es Parameters Manager Defau		ste Jpdates		thedule stail Default	s Sel	hedule fers Defaul	Sch Hou	dule Offe rly Updat		Schedule Selection	Hourly Updates	ARD By Portfolio	De	ril. ownload					
D Ramp Rate Hourly U	pdates																		
Date and ARD	«	Default	alues for ASSE	T NAME (in 24-A	ug-2014													
Date: 24-Aug-2014		Override	Hourly Values	for ASSE	T NAME	on 24-A	ig-2014												
		Hour		RR		MN	RR	8 16	w 👘	RR #	MIC	RR		MN	RR	8 MI	w i		
tfolio: AAA	~ 7	~	Y Copy		1			2		3			4			5			
ARD: ASSET NAME	~				0			7		8			9			10			
	/		01		1	10.0	10.0	2	13.0	1.0 3			4			5			
					6			7		8			8			10			
01			02		1	6.0	8.0	2	10.0	12.0 3	12.0	10.0		13.0	1.0	6			
02			00		1	10.0	10.0		13.0	1.0 3	<u> </u>		9 4			10			
			60		0	10.0		7	13.0	10 3		1	9			10			
03			04		1	6.0	8.D	2	10.0	12.0 3		10.0	4	13.0	1.0				
04					8			7		8			9			10			
05			05		1	10.0	10.0		13.0	1.0 3			4			6			
06					6			7		8			9			10			
07			06		1 6	5.0	8.0	2	10.0	12.0 3	12.0	10.0	4	13.0	1.0	5			
08			07		1	10.D	10.D		13.0	1.0 3			4			6			
09					8	10.0		7	13.0	1.5 0	<u> </u>	1	8			10			
10			08		1	5.0	8.0	2	10.0	12.0 3	12.0	10.0	4	13.0	1.0	8			
10					6			7		8			9			10			
			09		1	10.0	10.0		13.0	1.0 3			4			5			
12					6			7		8			0			10			
10			10		1 6	6.D	8.D	2	10.0	12.0 3		10.0	4	13.0	1.0	5			
			11		1	10.0			13.0	1.0 3			4			6			
					0	10.0		7		10 0			9			10			
			12		1	5.0	8.D	2	10.0	12.0 3	12.0	10.0	4	13.0	1.0	5			
					8			7		8			9			10			
			13		1	10.0	10.0		13.0	1.0 3			4			6			
					8			7		8			8			10			
			14		1	5.0	8.0	2	10.0	12.0 3		10.0	4	13.0	1.0	5			
			15		1	10.0	10.0		13.0	1.0 3			4			6			
					8			7		12 8		1	9			10			
			16				0.0			120 2		10.0		12.0	1.0				
																			Update

The Override Hourly Values section shows each hour of the day and fields to enter overrides. When the fields are blank, it means that there is no override of the default values for that hour. Hence setting all data entry fields for an hour to blank (NULL) means that the applicable daily default value is to be used for the market hour; i.e. it is equivalent to removing the override for the market hour.

The description of the fields in the Override Hourly Values section of the display are the same as those in the Default Values section, except they define the unit's ramp rate characteristic for the hour-ending shown in the Hour column. *Either a single ramp rate override or a ramp rate curve override can be defined for an hour, not both.*

The top row of the grid provides you with the means to enter or clear overrides for a selected range of hours. You select the "from" and "to" hours using the drop-down list boxes under the Hour column, enter a single ramp rate or a ramp rate curve using the data entry fields on the right and then select the "Copy" button. As illustrated below, the values entered in the top row are copied to the rows for each hour in the range selected and the cells where values are being changed are highlighted. This process can be repeated to enter different ramp rate updates for other hour ranges before selecting the "Update" button to save the values in the database. If you do use the top row, you need to **remember** *that leaving all data entry field on this row blank means that you want to remove the overrides for all hours in the range selected*. For example, if you want to remove overrides for every hour of the day, simply select "01" as the "From" hour and "24" as the "To" hour in the Hour drop-down selections, select Copy to blank out all data entry fields below and then select Update. Since you did not enter anything in the data entry fields, all overrides are removed.

ic Generation Demand ARD Reg	ulation														Participant: ISO New England Inc. Y User: Superuser
te ARD Default Portfolio Ramp Ra			chedule etail Defaults	Schedu Offers D		dule Offe			Hourly	ARD By Portfolio	XML Download				
ges Parameters Manager Defauk	Hourly	updates Manager De	etali Deraulits	Offersi	erauit Hour	ly Update	s Select	ion	Updates	POPTIONO	Downios	2			
D Ramp Rate Hourly Up	dates														
Date and ARD	~	Default Values for ASSE													
Date: 24-Aug-2014		Override Hourly Values				V 144	BB		MW	RR	a 140	BB	a Mar	RR	
tfolio: AAA	*	Hour 07 ¥ 12 ¥ Copy.	RR 1.5	a Mar		2	C RR	3	MIV	нк	4	RR	5	RR	
ARD: ASSET NAME	~	07 12 00009		6							8		10		
		01		1 10.	0 10.0	2 1	3.0	0 3			4		6		
				0				8			9		10		
		02		1 6. 6		2 1	0.0 1:	:0 3 8		10.0	4 13	.0 1.1	5		
		03		1 10.			3.0	0 3			4	_	5	_	Hour RR
				6		7		8			9		10		
		04		1 5. 6	0.8.0		0.0 1:	.0 3			4 13	.0 1.1	5		
		05		1 10.			3.0	0 3			4		10	_	07 💙 12 🍸 Copy 1.5
				6		7		8			9		10		
		08		1 6.			0.0 1:	.0 3		10.0	4 13	.0 1.1			
		07		6		7		8 .0 3			9 4		10 5	_	
		07		6	0 10.0		3.0	9 8			9		10		Before selecting "Copy", example where
		08		1 5.			0.0 1:	0 3			4 13	.0 1.1			
		09		0		7		8			9		10		you want to set the unit's ramp rate
		09		1 10. 6	D 10.0		3.0	0 3 8			4	_	5	_	characteristics to a single ramp rate of 1.5
		10		1 6.	0.8.0	2 1	0.0 1:	.0 3	12.0	10.0	4 13	.0 1.1	5		
				6				8			9		10		for hours ending 07 through 12.
		11		1 10. 6		2 1	3.0	0 3			4	-	5	_	
		12		1 5.			0.0 1:	0 3	12.0		4 13	.0 1.1			
				•				8			0		10		
		13		1 10. 6	D 10.D		3.0	0 3			4		5		
		14		1 6.			0.0 1;	0 3	12.0	10.0	4 13	.0 1.1		_	
				6		7		8			9		10		
		16		1 10			3.0	0 3			4		6		
				0		7		8	10.0	10.0	9		10	_	

ISO new england																eMarket User Interfact Version: 2:0.1(201407221119) Ricate User: Superuser 23-Aug-2014 13 Vuser: Superuser Superuser
Generation Demand ARD Regulation	'n															
e ARD Defauk Portfolio Ramp Rate ges Parameters Manager Defauk	Ramp Rate Hourly Updates		nedule al Defaults	Schedule Offers Defai	schedul Hourly I		Schedule Selection	Hourly Updates	ARD By Portfolio	XML Download						
D Ramp Rate Hourly Update	es															
Date and ARD		alues for ASSET	F NAME on S	24-Aug-2014	F											
Date: 24-Aug-2014	Override	Hourly Values f	for ASSET N	AME on 24-/	vog-2014											
tfolio: AAA	₩ Hour		RR #	MI	RR N	MN	RR #	MI	RR	# MA	RR	# MN	RR			
	· ·	✓ Copy	1.5 1		2		3			4		5				
ARD: ASSET NAME			6		7		8			9		10				
		01	1	10.0	10.0 2	13.0	1.0 3			4		5				
		02	1		8.0 2	10.0	12.0 3	12.0		4 13.0	1.0					
			0		7		8			0		10				
		60	1		10.0 2	13.0	1.0 3			4		5				"
		P4	6		8.0 2	10.0	12.0 3	12.0	10.0	9 4 13.0	1.0	10	_		After selecting "Cop	by", the values
		D4		0.0	8.0 2	10.0	12.0 3	12.0	10.0	9		10			entered on the top	row are conied to
		05	1	10.0	10.0 2	13.0	1.0 3			4		5	1	/	entered on the top	Tow are copied to
			0		7		8			9		10			the rows for each c	f the selected
		08	1	5.0	8.0 2	10.0	12.0 3	12.0	10.0	4 13.0	1.0	5	¥ -			
		07	1.5 1		2		3			4	_	5	1	1	hours. The fields w	here values are
			6		7					9		10		1		
		08	1.6 1		2		3			4		5			changed are highlig	shted to show
		09	1.5 1		7		8			9		5	_	-	where data entries	will be made when
		100	1.5		7		8			0		10			where uata entries	will be made when
		10	1.5 1		2		3			4		5		1	the "Update" butto	n is selected
			0		7		8			9		10			and opdate butto	in is selected.
		11	1.6 1	1	2		3			4		5				
		12	1.5 1		2		3			4		8				
		_	0		7					0		10	1	1		
		10	1		10.0 2	13.0	1.0 3			4		5				
		14	0	5.0	8.0 2	10.0	12.0 3			9 4 13.0	1.0	10				
				0.0	8.0 2	10.0	12.0 3	12.0	10.0	9		10				
		16	1	10.0	10.0 2	13.0	1.0 3			4		6				
		_	6		7		8			9		10				
		18			00 2	10.0	12.0 2	10.0	10.0	4 49.0	1.0					Update Re
es																

IC Generation Demand ARD Regul Me ARD Default Portfolio Ramp Rate Parameters Manager Default	Ramp Rate Sched Hourly Updates Manag	e Schedule r Detail Defaults	Schedule Sche Offers Default Hou	edule Offers S rly Updates S	chedule Hourly election Updates	ARD By 10ML Portfolio Download			
RD Ramp Rate Hourly Upda t Date and ARD		or ASSET NAME on 24	Aug 2014						
	o de la companya de la	Values for ASSET NAME on 24							
Date: 24-Aug-2014	Hour	RR #	MW RR		RR # MW	RR # MW	RR # MA	W RR	
ARD: ASSET NAME	• 04 ¥ 06 ¥	Copy 1 0	5.0 9.0	2 10.0	7.0 3 12.0	2.0 4 14	1.0 5		
			90 00 00 00	7 100 2 100 7 - 2 100 2 100 2 100 2 100 2 100 2 100 2 100 2 100 2 100 2 - 2 - 2 - 2 - 2 - 2 - 2 - 3 - 4 - 5 - 6 - 7 - 9 - 9 - 9 - 10 - 10 -	10 2 120 3 120 </td <td>9 4 9</td> <td></td> <td></td> <td>Before selecting "Copy", example where you want to set the unit's ramp rate characteristics to a different ramp rate curve for hours ending 04 through 06.</td>	9 4 9			Before selecting "Copy", example where you want to set the unit's ramp rate characteristics to a different ramp rate curve for hours ending 04 through 06.
									eMarket User Interfi
Soonew england Grantikon Cemind AKD Regul		e Schedule 7 Detail Defaults	Schedule Offers Default Hou	edule Offers S	checkle Hourly election Updates	ARD By Partfele Downlead			
Sopraevengland Generation Demand ARD Regul ARD Ordrag Fortfold Reng Rate s Parameters Berlind Reng Rate	Ramp Rate Sched Hourly Updates Manag	le Schrebule 7 Detail Defaults	Schedule Sche Offers Defeut Hou	edule Offers S Irly Updates S	chedule Haurty election Updates	ARD By 25% Portfolio Downlow			eMarket User Interf Version: 20.1(201407221139) Narket Status: OPRII(1) Comficate User: Superiorer 23-Jug-2014
Greater Central AD Reput AD Octave Eventor Company Parameter Strategic Fordato D Ramp Rate Hourly Upda	Ramp Rate Sched Hourly Updates Manag ates	r Detail Defaults	-Aug-2014	rklute Offers S rly Updates S	chidule Hourly dection Updates	ARD By XML Portfolio Downlow			eMarket User Interf Version: 20.1(201407221139) Narket Status: OPRII(1) Comficate User: Superiorer 23-Jug-2014
Generation Demand ARD Regul term AD Default Perfolic Rump Rule Demand Rump Rule Demand Rump Rule DRamp Rate Hourly Upda Date: 24-Aug-2014	Ramp Rate Hourly Updates Sched Hourly Updates Manag Attes Override Hourl Hour	r Detail Defaults or ASSET NAME on 24 Values for ASSET NAI	-Aug-2014 ME on 24-Aug-2014 MW RR	9 1.667	RR # MAY	RR # MAN	RR # Md	w RR	eMarket User Interfi Version 2.0.1(20140722115) Narket Stetus: OPEN(Q1) Cenflicate Vier: Superiorer 23-kug-2014 Pertugent <u>50 New England Inc</u> Vier <u>Superior</u> <u>Viers</u> <u>Superior</u>
Software england arrenter Central ARD Regular Arrenter Central ARD Regular Arrenter Central Central December 2010	Ramp Rate Hourly Updates Sched Manag ates C Default Values Override Heurl Hour	r Detail Defaults or ASSET NAME on 24 Values for ASSET NAI	Amy 2014 Amy 2014 Mark Mark Mark Mark Amy 2014 Mark Mark Mark Mark Mark Mark Mark Mark Mark Mark Mark Mark Mark	Image: 1 Image: 1		RR J MM 2.0 4 144 4 4 13 4 4 13 4 4 13 2.0 4 14 0 2.0 4 0 9 14 0 9 14 0 9 14	RR J MM 1.0 6 10 6 10 10 1.0 6 10 2 1.0 6 3 1.0 6 3 1.0 6 3 1.0 6 3 1.0 6 5 10 10		eMarket User Interf Version: 20.1(201407221139) Narket Status: OPRII(1) Comficate User: Superiorer 23-Jug-2014

7.5.1 Data Validation Rules

- Ramp Rate \geq 0; MW > 0
- eMarket will give you a warning if any of the entered rates is greater than the Ramp Rate Cap shown on the Unit Default Parameters display.
- MW/Ramp Rate pairs can be entered in any order. eMarket will sort them and show them in increasing MW value order after the data is submitted and stored in the database.
- Either a single ramp rate or a ramp rate curve can be defined for an hour, not both.

7.5.2 Override Unit's Default Ramp Rate for a Day and Hour

As discussed in §7.3 ARD Default Parameters Display, you define on this display a unit's daily default ramp rate. Similarly, as discussed in §7.4 ARD Multiple Ramp Rates Daily Default Display, you can define on this display a daily default ramp rate curve for the unit that supersedes the daily default. Both the daily default single ramp rate and the daily default ramp rate curve for a selected market day are shown in the Default Values section of the display. If you want to profile the unit's ramp rate characteristics on an hourly basis,:

- 7. In the Generation Navigation sub tabs, click the **Ramp Rate Hourly Updates** tab.
 - The ARD Ramp Rate Hourly Updates display is shown in the display pane.
- 8. In the Filters pane:
 - a. Select the desired date from the Date filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired unit from the ARD filter.
 - The display reflects the hourly overrides for the selected unit.
- 9. To update the overrides for individual hours:
 - a. Enter the override value for a single ramp rate or for a ramp curve in the hours that you wish. Remember that if you are switching from a ramp rate curve override to a single ramp rate value, you need to blank each segment of the ramp rate curve in addition to entering the single ramp rate value.
 - b. Click the **Update** button to save the data.

10. To update the overrides for a range of hours

- a. Specify the range of hours by selecting "From" and "To" hour-ending values in the **Hour** drop-down list on the top row.
- b. Enter the override value for either a single ramp rate or a ramp rate curve in the top row.

- c. Select the **Copy** button to copy the entered values to the rows for each hour in the selected range.
- d. Repeat steps a through c to enter different values in other range of hours if necessary.
- e. Click the **Update** button to save the data for the selected hours.

7.6 ARD Schedule Manager Display

Schedule Manager enables users to view, create, and update schedules. The format of this display is shown below:

ISC	new england	ł									Market	Status: OPEN(01) Participant:	ISO New England Inc	Version: 2.0.1(2014072 Certificate User: Superuser	User Interface
		Y										Participante	150 New England Inc		osen operaser ·
Private /	ARD Default Portfolio Parameters Manager	RD Regular Ramp Rate Default	ion Ramp Rat Hourly U			Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Selection	Hourly Updates	ARD By Portfolio	XML Download				
ARD So	chedule Manag	ger													
Select ARD			~	Schedule Man	ager for ASSET NAM										
Portfolio:	000		~	Schedule Nan	e	Description					lule Type				_
	ASSET NAME		~	DFLPRICE		DFLPRICE ASSET Pri-				0 (Pric		 ✓ ✓ 			<u> </u>
				ASSET_12 DFLCOST		DFLCOST	0e			12 (P1 99 (Co	ice) ist)	× ×			
													New	Delete	Update Refresh
Messages															*

The fields on this display are:

- Schedule Name a short name for the asset schedule, defined by the Participant. This label appears in the Schedule filter on the ARD displays.
- **Description** a brief description of the asset's schedule, defined by the Participant. This label appears in the ARD by Portfolio display, which shows the DAM results by ARD portfolio.
- Schedule Type identifies the number and type of the schedule, as follows:
 - **0 94** price-based schedules
 - 95 99 cost-based schedules, reserved for use by ISO New England.

7.6.1 Create a New Schedule

Users can define more than one price-based schedule for an ARD asset. To assist in creating multiple schedules for an ARD asset, the Schedule Manager display allows you to assign a schedule name and a description to a specific ARD asset schedule. To create a new schedule:

- 1. In the ARD Navigation sub tabs, click the **Schedule Manager** tab.
 - The Schedule Manager display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired portfolio from the **Portfolio** filter.

- b. Select the desired asset from the **ARD** filter.
 - The existing schedules for the selected asset are listed.
- 3. Click the **New** button.
- 4. Type the following information in the new row:
 - a. Schedule Name
 - b. Description
- 5. Select an unused type of schedule from the **Schedule Type** drop-down menu.
- 6. Click the **Update** button to save the data.

7.6.2 Modify a Schedule's Name and/or Description

After a schedule is created and saved to the database, you cannot modify the Schedule Type field. Only the schedule's name and description can be changed.

To modify an existing schedule's parameters:

- 1. In the ARD Navigation sub tabs, click the **Schedule Manager** tab.
 - The Schedule Manager display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired portfolio from the **Portfolio** filter.
 - b. Select the desired asset from the **ARD** filter.
 - The existing schedules for the selected asset are listed.
- 3. Modify the either or both of the following in any of the listed schedules:
 - a. Schedule Name
 - b. Description
- 4. Click the **Update** button to save the data.

7.6.3 Delete a schedule

- 1. In the ARD Navigation sub tabs, click the **Schedule Manager** tab.
 - The Schedule Manager display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired portfolio from the **Portfolio** filter.
 - b. Select the desired asset from the **ARD** filter.
 - The existing schedules for the selected asset are listed.
- 3. Select a cell in the row of the schedule that you want to delete.
- 4. Click the **Delete** button. If the schedule is not flagged available in any of the market days for the period that market information is preserved in the market database, it is deleted; otherwise you will get an error message.

7.7 ARD Schedule Detail Defaults Display

The Schedule Detail Defaults display is used to view and manage an ARD asset's default market offer parameters. The format of this display is shown below:

ISO	new e	ngland										Market S		c		1(201407221119	r Interf	
													Participant:	ISO New England Inc		🗡 User:	Superuser	*
ublic Gene	eration Dem	and ARI	D Regulati	ion														
			Ramp Rate Default	Ramp Rate Hourly Up		Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Selection	Hourly Updates	ARD By Portfolio	XML Download						
	hedule D		efaults															
lect Date, Al	RD and Schedu	lle		~	Schedule Detail:	for ASSET NAME - D	FLPRICE (Price	based) on 24-Aug	2014									
Date:	24-Aug-2014			3			De	scription: DFLPRI	ICE					Max Daily Star	ts:			
Portfolio:	ААА			~				vailable: Yes					Max	Daily Consumption (MW	n):			_
ARD:	ASSET NAME			~				Market: 🗸						Min Downtime (h				
Schedule:	DFLPRICE			~				Market: 🗸						Min Runtime (h				
								_						Max Consumption (MV	v): 295.0			
														Min Consumption (MI				_
																	Update	Refresh
																	opoate	Reliesi

The fields on this display are:

- **Description** the schedule description as entered on the Schedule Manager display.
- Day Ahead Market this check box indicates whether the schedule is enabled for Day Ahead Market use.
- **Real Time Market** this check box indicates whether the schedule is enabled for Real Time Market use.
- Schedule Available indicates whether the schedule is to be used. This is managed on the Schedule Selection display.
- **Max Daily Starts** the maximum number of times that the ARD can be started in one operating day under **normal** operating conditions. A blank field means the value is unlimited. This parameter is primarily intended for dispatchable ARDs (pumps in a pump storage facility).
- **Max Daily Consumption** the maximum amount of energy that the ARD can consume in one operating day. A blank field means the value is unlimited. This parameter is primarily intended for dispatchable ARDs (pumps in a pump storage facility).
- **Min Downtime** the minimum number of hours that the ARD owner requires between the time the ARD is released for shutdown by the ISO and the time the ARD is scheduled to be released for dispatch to the ISO; non-enterable.
- **Min Runtime** the minimum number of hours that the ARD must remain online after it has been scheduled to reach its Minimum Consumption Limit before it can be released for shutdown from its Minimum Consumption Limit; non-enterable.

- Max Consumption (MW) the highest MW level at which the asset can be economically dispatched.
- Min Consumption (MW) the lowest MW level at which the asset can be economically dispatched. This value represents the minimum amount, in MW, available from a Dispatchable Asset Related Demand that is not available for economic dispatch and is based on its physical characteristics.

You cannot change the Min Downtime and Min Runtime values. They are for information purposes only. To have either value changed, you must submit a Heat ticket to the ISO New England.

7.7.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- Data is entered for each of the MW limit values, i.e. non-blank values appear in the data entry cells.
- Max Daily Starts and Max Daily Consumption values can be blank (unspecified)
- Max Consumption ≥ Min Consumption ≥ 0 for dispatchable ARDs Max Consumption = Min Consumption ≥ 0 for non-dispatchable ARDs The data submitted is rejected if any of these limit value validations fails.
- Max Daily Starts must be blank to specify an unlimited number of starts or a value greater than zero if you want to specify the maximum number of times the ARD can be started in a day. Max Daily Starts ≤ 24.
- Max Daily Consumption must be blank to specify an unlimited capacity or a value greater than zero if you want to define a limit on the ARD's consumption capability and be treated as a Limited Energy Generation (LEG) resource. If a non-blank value is entered, then Max Daily Consumption should be greater than or equal to Min Consumption x Min Runtime; a warning message will be reported if it is not.
- When an ARD self schedules for a total number of hours greater than its Min Runtime, the Max Daily Energy (if defined) must be greater than or equal to Min Consumption x the number of hours that the ARD self schedules or a warning will be returned.

7.7.2 View Schedule Detail Defaults

To view the details of an ARD asset schedule:

- 1. In the ARD Navigation sub tabs, click the **Schedule Detail Defaults** tab.
 - The ARD Schedule Detail Defaults display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired asset from the **ARD** filter.

- d. Select the desired schedule from the **Schedule** filter.
 - The display reflects the selected date's schedule detail.

7.7.3 Modify Schedule Constraint Data

Constraint data that can be modified includes maximum daily starts and maximum daily consumption. To modify schedule constraints:

- 1. View the unit's schedule details (§7.7.2)
- 2. Enter the desired changes for any or all of the following offer parameters:
 - a. Maximum Daily Starts
 - b. Maximum Daily Energy (MWh)
- 3. Click the **Update** button to save the data.

7.7.4 Modify Schedule Operating Limits

To modify an ARD's schedule operating limits

- 1. View the ARD's schedule detail (§7.7.2)
- 2. Enter the desired changes for any or all of the following operating limits:
 - a. Min Consumption (MW)
 - b. Max Consumption (MW)
- 3. Click the **Update** button to save the data.

7.7.5 Enable Schedule for a Market

Another schedule parameter is market participation. The Participant selects the schedules that are enabled for the Day-Ahead Market (DAM) and/or the Real-Time market (RTM). The following steps enable a schedule for the specified market(s):

- 1. View the ARD's schedule detail (§7.7.2)
- 2. To enable/disable a schedule for use in the DAM, click the **Day Ahead Market** checkbox.
- 3. To enable/disable a schedule for use in the RTM, click the **Real Time Market** checkbox.
- 4. Click the **Update** button to save the data.

Note: If the schedule is flagged as available, you will not be able to change the schedule's applicable market flags. That is because one price-based schedule must be flagged as available for each of the markets at all times. To change a schedule's market eligibility, you must first make one or more other schedules available (one for each market).

7.8 ARD Schedule Offers Daily Default Display

The purpose of this display is to view and update the ARDs daily default schedule bid segments and Use Offer Slope flag. Schedule bid segments specify a MW amount and the price per MWH for consumption from the previous segment's MW amount up to this segment's MW amount. In other words, schedule bids consist of MW/price segments. Up to 10 segments can be defined for each schedule.

Bids in MW are cumulative. For example,

- To bid the first block of 15 MW of consumption at \$30.00, you enter an offer segment of 15 MW @ \$30.00
- To bid the next block of 20 MW of consumption at \$10.00, you enter an offer segment of 35 MW @ \$10.00

And so on. You can enter the segment bids in any order of MW values; eMarket will organize the segments and display them in increasing MW value order. The price value of each segment must however be less than the price of the previous segment after the segments are sorted in increasing MW value order. In other words, the bid segments must be monotonically decreasing in MW and increasing in price value.

The slope from zero to the first MW point (which can be zero), is equal to zero; that is to say that the price/MWH is fixed and equal to the price of the first segment. The price along the segment between MW points is a calculated slope or block, depending on the status of the Use Offer Slope checkbox.

The last MW point should be greater than or equal to the Max Consumption Limit. If it is not, the curve is extended with a slope equal to zero.



This is all illustrated in Figure 10.

ARD assets are eligible to change segment bid Price/MW pairs after the close of the DA market during the daily Re-Offer period and during the RT Intraday Re-Offer period.

The format of the Schedule Offers Daily Default display is shown below:

ISÓ	new eng	land													Version: 2	orket Us	19)	
150	new eng	ianu										Market St	tatus: OPEN(01)		Certificate User			2014 13:15:03
													Participant:	ISO New England Ir	ic.	Y User	Superuser	×
ublic Gen	eration Demand	ARD	Regulati	on														
	RD Default Porti arameters Man		Ramp Rate Default	Ramp Rat Hourly U		Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates		Hourly Updates	ARD By Portfolio	XML Download						
RD Sc	hedule Off	ers D	aily De	fault														
lect Date, A	RD and Schedule			~	Schedule Offer for	ASSET NAME -	DFLPRICE (Price-b	ased) on 24-Aug-2	014									
Date:	24-Aug-2014					w	Price	Use Offer Slope	_									
Portfolio:				~		6.0	6.00	Use Offer Slope	e: 💌									
ARD:	ASSET NAME			~		10.0	4.75											
Schedule:	DFLPRICE			*														
														New	Delete		Update	Refresh
essages																		

The fields on this display are:

- **MW** the net MW consumption associated with this segment. The last MW point should be greater than or equal to the Max Consumption Limit.
- **Price** the \$/MWH price bid for this segment.
- Use Offer Slope this check box indicates whether to use the offer slope for the incremental energy bids within the market clearing and dispatch software as illustrated in Figure 10.

7.8.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- MW quantity must be strictly positive or zero; i.e. $MW \ge 0$
- Price must be less than or equal to the Energy Offer Cap price and greater than or equal to the Energy Offer Floor price; i.e. Energy Offer Floor price ≤ Price ≤ Energy Offer Cap price²⁴.
- Offers must be monotonically decreasing. The price value for each segment must be less than the one of the previous segment after the segments are ordered in increasing MW value²⁵.

²⁴ Energy Offer Floor price and Energy Offer Cap price are both defined in Market Rule 1, section I.2.2.

²⁵ The same price for successive segments is allowed; however it doesn't really make sense when considering Participant offers because the two segments are equivalent to a single segment with a MW value equal to that of the second segment. It is allowed only to permit mitigation reference price values for successive offer segments to be equal.

7.8.2 Modify Schedule Offers

- 1. In the ARD Navigation sub tabs, click the Schedule Offers Default tab.
 - The Schedule Offers Daily Default display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired asset from the ARD filter.
 - d. Select the desired schedule from the **Schedule** filter.
 - The display reflects the selected schedule's bid data for the selected date.
- 3. (optional) Click the **New** button to add a new segment. You can click the button more than once to add several rows.
 - A new segment row is added for each click of the button and the background of each is pink to show that it has not been saved.
- 4. (optional) To delete a segment:
 - a. Click in a cell in the row for the segment you wish to delete
 - The row is highlighted
 - b. Click the **Delete** button.
- 5. Type the following information for each segment as needed:
 - a. MW
 - b. Price
- 6. Repeat steps 3-5 as desired.
- 7. Click the **Update** button to save the data.

7.9 Schedule Offers Hourly Updates Display

This display provides for viewing, entering and modifying hourly updates to an ARD schedule's default offers for a given market day. Refer to §7.8 ARD Schedule Offers Daily Default Display for a description of these schedule offer parameters and how incremental energy offers must be structured. *This is the only ARD display enabled for submittal of hourly updates during the RT Intraday Re-Offer period*.

																										er Inter	face
IS	new	england																Market	Statu:	s: OPEN(Version: 2.0.1(2014 ficate User: Superus		(19) 23-Aug-201	4 13:16
	1																			Participai	nt: ISC) New England Inc			/ Use	superuser	
blic Ge	neration Der	mand AR	D Regulati	on																							
		Portfolio	Ramp Rate	Ramp Rate			Schedul Detail De		Schedule Offers Def		Schedule		Sch Sele		Hourly	ARD		ML .									
-		Manager	Default	Hourly Up		lanager	Detail D	shauks	Offers Def	ault	Hourly U	pdates	Sele	ction	Updates	Port	tholeo L	ownload									
RD S	chedule	Offers I	Hourly U	lpdate	s																						
ct Date,	ARD and Sched	lule		~	Default V	alues for A	SSET N	AME - DF	LPRICE (I	Price	based) on	24-Aug-2	014														
Date:	24-Aug-2014							Mil	Price	N	MØ	Price		MW	Price	N	MA	Price		MW	Price	Use Offer Slope					
Portfolio:	_			~			1	0.0	5.00	2	13.0	4.75	3			4			5								
	ASSET NAME			•																							
	DFLPRICE				Override	Hourly Val	ies for	ASSET N	AME - DFI	PRIC	E (Price-b	ased) on S	4-Au	g-2014													
icheadle:	DFLPRICE					Hour		M07	Price	#	MIN	Price		M00'	Price	#	MIN	Price	1	M00'	Price	Use Offer Slope					
					×	Y Cop				2			3			4			5					Legend			
							6			7			8			9			10								
						01	1	1.0			2.0	14.00	3	3.0	12.00		4.0	10.00		6.0	8.00	1 1000	-	No Override: False:			
						02	6	6.0			7.0	4.00	8	9.0 3.0	2.00		12.0	0.00		16.0	-10.00						
							6	6.0			7.0	4.00	8	9.0	2.00		12.0	0.00		16.0	-10.00			True:	V		
						83	1	1.0	16.00	2	2.0	15.00	3	5.0	12.00	4	10.0	5.00	6	15.0	0.00						
							6	20.0					8			9			10								
						04	1	1.0			2.0	15.00	3	6.0	12.00	4	10.0	5.00	5 10	15.0	0.00						
						05	1	1.0			2.0	14.00	3	3.0	12.00		4.0	10.00		5.0	8.00	✓					
							6	6.0	6.00	7	7.0	4.00	8	9.0	2.00	9	12.0	0.00	10	16.0	-10.00						
						06	1			2	2.0	14.00		3.0	12.00		4.0	10.00		5.0	8.00						
						07	6	6.0			7.0	4.00	8	9.0 6.0	2.00		12.0 10.0	0.00		16.0 15.0	- 10.00						
							6	20.0			1.0	10.00	8	0.0	12.00	9	10.0	0.00	10	10.0	0.00	V					
						88	1	1.0		2	2.0	15.00	3	5.0	12.00	4	10.0	5.00		15.0	0.00	V					
							6	20.0					8			9			10								
						09	1	1.0			2.0	14.00	3	0.C 9.0	12.00		4.0	10.00		6.0 16.0	8.00	a babad					
						10	1	1.0			2.0	4.00	3	3.0	12.00		4.0	10.00		5.0	8.00						
							6	6.0		7	7.0	4.00	8	9.0	2.00		12.0	0.00		16.0	-10.00						
						11	1	1.0			2.0	15.00		5.0	12.00		10.0	5.00		15.0	0.00						
						12	6	20.0				45.00	8		12.00	9 4	10.0		10								
						12	1	1.0			2.0	15.00	3	6.0	12.00	4	10.0	6.00	5 10	16.0	0.00						
						13	1	1.0			2.0	14.00	3	3.0	12.00		4.0	10.00		5.0	8.00	V					
							6	6.0	6.00	7	7.0	4.00	8	9.0	2.00	9	12.0	0.00	10	16.0	-10.00		-				
																										Update	Refres

The format of the display is shown below.

The Default Values section is provided as a reference to show what are the default offers for the chosen day that are being overridden by the hourly update values entered in the section below.

7.9.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

 At least one MW/Price pair is entered for an hour and the Use Offer Slope checkbox must be set to a state other than "No Override". When all data entry cells for an hour are set to blank and the Use Offer Slope is set to "No Override", it means that the override for the hour is to be removed/deleted.

- MW quantity must be strictly positive or zero; i.e. $MW \ge 0$
- Price must be less than or equal to the Energy Offer Cap price and greater than or equal to the Energy Offer Floor price; i.e. Energy Offer Floor price ≤ Price ≤ Energy Offer Cap price²⁶.
- Offers must be monotonically decreasing. The price value for each segment must be less than the one of the previous segment after the segments are ordered in increasing MW value²⁷.

7.9.2 Override Schedule Offers for One or More Hours of a Market Day

- 1. In the ARD Navigation sub tabs, click the **Schedule Offers Hourly Updates** tab.
 - The ARD Schedule Offers Hourly Updates display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired asset from the **ARD** filter.
 - d. Select the desired schedule from the **Schedule** filter.
 - The display reflects the hourly overrides for the selected ARD schedule.
- 3. To update the overrides for individual hours:
 - a. Enter override values and set the Use Offer Slope in the hours that you wish. Enter blanks and set the Use Offer Slope to "No Override" if you want to remove an override.
 - b. Click the **Update** button to save the data.
- 4. To update the overrides for a range of hours using the first row in the grid:
 - a. Enter override values and set the Use Offer Slope in the top row. Enter blanks and set the Use Offer Slope to "No Override" if you want to remove overrides.
 - b. Specify the range of hours to override by selecting the "From" (left dropdown selector) and "To" (right drop-down selector) hours in the top row
 - c. Click the **Copy** button; this will copy the data that you entered in the top row to each row for the hours in the range that you selected.
 - d. Repeat steps a through c for other range of hours as desired.
 - e. Click the **Update** button to save the data for the selected hours.

²⁶ Energy Offer Floor price and Energy Offer Cap price are both defined in Market Rule 1, section I.2.2.

²⁷ The same price for successive segments is allowed; however it doesn't really make sense when considering Participant offers because the two segments are equivalent to a single segment with a MW value equal to that of the second segment. It is allowed only to permit mitigation reference price values for successive offer segments to be equal.

7.10 ARD Schedule Selection Display

This display is used to make schedules available for a given day and market. The format of this display is shown below:

ISO new england							Ма	rket Status: OPEN(01				201412111610)	19-Dec-2014 18:3
								Participant	ISO New Engla	nd Inc		Y User: S	uperuser
olic Generation Demand AR	D Regulation												
vate ARD Default Portfolio ssages Parameters Manager	Ramp Rate Ramp Default Hour	Rate ly Updates	Schedule Manager	Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Selection	Hourly ARD By Updates Portfolio	XML Download				
DD Cala dula Cala du													
RD Schedule Selection													
ect Date and Portfolio			ules for Port	folio AAA on 21-		-							
Date: 21-Dec-2014		ARD	T NAME		Schedule ID 1422499	Type Cost	Soh	COST	Day Ahead X	Real Time X	Available		
Portfolio: AAA	~		TNAME		1422401	Price		PRICE	x	x			
			T NAME		1422412	Price		ET_12	x	x			

The fields on this display are:

- **ARD** the name of the ARD asset.
- Schedule ID the unique identifier of the schedule; it is the concatenation of the unique asset identifier and of the schedule type number.
- **Type** the type of schedule: Price or Cost.
- **Schedule** the name of the schedule as it was entered on the Schedule Manager display
- **Day Ahead** an X-mark indicates that the schedule is enabled to be used in the clearing of the Day Ahead market if the schedule is flagged as available.
- **Real Time** an X-mark indicates that the schedule is enabled to be used in the clearing of the Real Time market if the schedule is flagged as available.
- **Available** a checkmark indicates that the schedule is to be use in the clearing of the selected markets on the selected operating day.

7.10.1 Select Schedule for an Operating Day and Market

The ARD Schedule Selection operation is for modifying the availability of a schedule for a specific day.

Note: One and only one price-based schedule can be available for a market on any given day. Therefore to make a schedule that is enabled for both Day Ahead and Real Time markets unavailable, you must at the same time make another schedule that is enabled for both markets available or two other schedules, one enabled for the Day Ahead market and another enabled for the Real Time market, available.

To update data or enable a schedule:

- 1. In the ARD Navigation sub tabs, click the **Schedule Selection** tab.
 - The ARD Schedule Selection display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - The schedules for all assets in the chosen portfolio are displayed.
- 3. Toggle the **Available** checkboxes as desired.
- 4. Click the **Update** button to save your modifications.

Note: To change the market to which a schedule applies you must use the Schedule Detail Defaults display.

Note: You cannot change the availability of the cost-based schedules. The availability of these is managed by ISO New England.

7.11 ARD Hourly Updates Display

This display is used to view and modify the hourly overrides of an asset's MW limits, default dispatch availability status and when it is applicable, its off-line reserve capabilities for a specified market day. Changes made to the asset data using the ARD Hourly Updates display supersede all schedule limits (ARD Schedule Detail Defaults display) and default asset parameters (ARD Default Parameters Display).

																arket	User Inte 221119)	erface
IS	l new	england										Marke	t Status: OPEN(0)			ser: Superuser		-2014 13:13
													Participant	ISO New	England Inc	*	User: Superuser	
âc G	eneration De	emand AR	D Regulat	ion														
ate	ARD Default	Portfolio	Ramp Rate	Ramp Rate	Schedule	Schedule	Schedule	Schedule Offers	Schedule	Hourly	ARD By	XML						
	Parameters	Manager	Default	Hourly Up		Detail Defaults	Offers Default	Hourly Updates	Selection	Update:						_	_	
ם חם	lourly Up	dates																
		uales																_
ct Date	and ARD			~														
Date	: 24-Aug-2014	4			Default Values for	ASSET NAME on	24-Aug-2014											
Portfolio				~	Schedule	Min Consumpt		onsumption Type	Day A		Real Time X	Available		Claim10	Claim30	Default Status: (C Unavailable	
					ASSET_12 DFLCOST		0.0	13.0 Price Cost	X		x		Capable Default	Yes 7	Yes 12			
AKD	ASSET NAME			~	DFLPRICE		0.0	13.0 Price	×		x	V V	Cap	7			Economic	
																	🔿 Must Run	
					Override Hourly V	lues for ASSET	NAME on 24-A	ug-2014										
					Hour	Min C	onsumption	Max Consumption	Cla	im10	Claim:	0 Unavailat	le Economic	Mus	t Run			
					~ ~ c) 0		0			
					01		0.0	1:	1.0	6.0		11.0) ()		۲			
					02		0.0		1.0	7.0		12.0			۲			
					03		0.0	1:	:.0	0.8		13.0			۲			
					04		0.0		1.0	7.0		12.0			۲			
					05 06		0.0			0.0		11.0			•			
					07		0.0		0	7.0		12.0			() ()			
					08		0.0		1.0	7.0		12.0			Ő			
					09		0.0		:.0	6.0		11.0			ŏ			
					10		0.0		0	7.0		12.0) ()		0			
					11		0.0		.0	0.8		13.0			0			
					12		0.0		0.	7.0		12.0			0			
					13		0.0		1.0	6.0 7.0		11.0			0			
					15	_	0.0		.0	8.0		13.0			0			
					16		0.0		1.0	7.0		12.0			ŏ			
					17		0.0			6.0		11.0) ()		0			
					18		0.0		.0	7.0		12.0			0			
					19 20		0.0			8.0		13.0			0			
					20		0.0		0	7.0		12.0 ()			0			
					22		0.0		1.0	7.0		12.0			õ			
					23		0.0		.0	8.0		13.0) Ö					
					24		0.0	1:		7.0		12.0	0		۲			
																	Update	Refre

The format of this display is shown below:

The Default Values section is provided as a reference to show what are the limit values in the various schedules and the default dispatch availability status and off-line reserve claim capability for the chosen day that are being overridden by the hourly update values entered in the section below.

- **Schedule** The name of the schedule
- **Min Consumption** The Min Consumption limit specified in the schedule detail.
- **Max Consumption** The Max Consumption limit specified in the schedule detail.
- Claim 10 and Claim 30 Capable Indicate whether the ARD is registered as capable (Yes/No) of providing 10 and 30 minute non-spinning reserves from an offline state.

- Claim 10 and Claim 30 Default The daily default off-line 10 and 30 minute off-line reserve capability for the market day as submitted using the ARD Default Parameters display.
- Claim 10 and Claim 30 Cap The demonstrated claim capabilities that were established during the last audit conducted by ISO New England. If the demonstrated capability value is undefined, it will be shown as "UD".
- Default Status (Economic/Unavailable/Must Run) The dispatch availability status (Default Status) specified on the ARD Default Parameters display

The Default Values section can be collapsed using the double up-arrow icon at the extreme right of the section title bar to show more rows in the hourly section as shown below. The collapsed section can be expanded again to see the default values using the double down-arrow icon at the extreme right of the collapsed section title bar.

	ARD Default Portfolio	RD Regulati Ramp Rate	Ramp Rab			Schedule	Schedule Offers		Hourly	ARD By	XML			
-	Parameters Manager	Default	Hourly Up	odates Manager	Detail Defa	ults Offers Defau	t Hourly Updates	Selection	Updates	Portfolio	Download			🛎 📐
	ourly Updates													
ct Date a	nd ARD		*											
Date:	24-Aug-2014					1E on 24-Aug-201								
Portfolio:	AAA		~	Override Hourt		SET NAME on 24-	Max Consumption	Claim	-10	Claim30	Unavailable	Economic	Must Run	
ARD:	ASSET NAME		~		Сору	wan consumption	Max Consumption	Claim		Claim30	Unavailable	Economic	Must Run	
			1	01	C 0 0 9 7	0.0	12.		6.0	11.0		0	۲	
				02	_	0.0	12.		7.0	12.0		0		
				03		0.0	12.		8.0	13.0		ŏ	۲	
	01			04		0.0	13.	D	7.0	12.0		Ó	۲	
	02			05		0.0	12.	0	6.0	11.0		0	۲	
				06		0.0	13.		7.0	12.0		0	۲	
	03			07		0.0	12.		8.0	13.0		0	•	
	04			09		0.0	13.		6.0	12.0		0	0	
				10		0.0	13.		7.0	12.0		ŏ	ŏ	
	05			11		0.0	12.		8.0	13.0		ŏ	ŏ	
	06			12		0.0	13.	0	7.0	12.0		Õ	Õ	
				13		0.0	12.		6.0	11.0		$^{\circ}$	0	
	07			14		0.0	13.		7.0	12.0		0	0	
	08			15		0.0	12.		0.8	13.0		0	0	
				16		0.0	13.		7.0	12.0	~	00	0	
	09			18		0.0	12.		7.0	11.0		0	0	
	10			19		0.0	12.		8.0	12.0		ŏ	Ŏ	
				20		0.0	13.		7.0	12.0	۲	ŏ	ŏ	
	11			21		0.0	12.		0.0	11.0	۲	Ó	0	
	12			22		0.0	13.		7.0	12.0		0	0	
		-		23		0.0	12.		0.8	13.0		0	۲	
	12	-		24		0.0	13.		7.0	12.0	0	0	۲	

The Override Hourly Values section shows each hour of the day and fields to enter overrides. When the fields are blank and the radio buttons unchecked, it means that there is no override of the schedule or default values for that hour.

• **Hour** – Hour ending values are shown in the first column for each hour in the day. The row for hour ending 03 is not shown for the Spring Daylight Saving Time transition day. Similarly, the value on the row of the repeating hour ending 02 of the Fall Daylight Saving Time transition day is marked with an asterisk (*).

- **Min Consumption** Column to enter overrides of the Min Consumption limit for any hour of the day.
- **Max Consumption** Column to enter overrides of the Max Consumption limit for any hour of the day.
- **Claim 10** Column to enter overrides of the Claim 10 capability for any hour of the day if the ARD is Claim 10 capable. If the ARD is not Claim 10 capable, there will be no data entry cells in this column.
- **Claim 30** Column to enter overrides of the Claim 30 capability for any hour of the day if the ARD is Claim 30 capable. If the ARD is not Claim 30 capable, there will be no data entry cells in this column.

The following override the Default Status specified on the ARD Default Parameters display:

- **Unavailable** When checked, it indicates that the asset is unavailable for commitment and dispatch for that hour of the operating day.
- **Economic** When checked, it indicates that the asset is available for economic commitment and dispatch for that hour of the operating day.
- Must Run When checked, it indicates that the asset is self-scheduling for that hour of the operating day; that is, the asset must be committed at its Min Consumption limit and can be dispatched up to the Max Consumption limit based on economics.

If a value is entered for an hour in any of these columns, values must also be entered in every other column even if they are copies of the default values. Setting these limit values to blank (NULL) means that the corresponding values defined in the Schedule Detail data of the schedules that are flagged as available for the market day are to be used for that market hour; i.e. it is equivalent to removing the override of the values for that market hour. Setting the off-line reserve claim values to blank (NULL) means that the corresponding values defined in the ARD Default Parameters data for the market day are to be used for that market hour. Similarly clearing all of the Unavailable/Economic/Must Run flags means that the ARD's default status defined in the ARD Default Parameters data for the market day is to be used for that market hour.

The top row of the grid provides you with the means to enter or clear overrides for a range of hours:

If you do use the top row, you need to *remember that leaving all data entry fields on this row blank means that you want to remove the overrides for all hours in the range selected*. For example, if you want to remove all overrides for every hour of the day, select 01 for the "From" hour (left drop-down selector) and 24 for the "To" hour (right drop-down selector) in the Hour drop-down selection list, select the "Copy" button and then select "Update". Since you did not enter anything in the limit fields or select any of the dispatch availability override flags, all overrides are removed. *You cannot use the top row to selectively update the overrides in one column for a range of hours while retaining the override values in the other columns*.

7.11.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- Max Consumption ≥ Min Consumption ≥ 0 for dispatchable ARDs Max Consumption = Min Consumption ≥ 0 for non-dispatchable ARDs.
- A warning will be returned if the ARD is Claim 10 capable, a Claim 10 override is entered and any of the following conditions exist
 - Claim 10 = 0
 - Claim 10 > Claim 10 Cap meaning that the Claim 10 Cap value will be used instead of the offered value
- A warning will be returned if the ARD is Claim 30 capable, a Claim 30 override is entered and any of the following conditions exist
 - Claim 30 = 0
 - Claim 30 > Claim 30 Cap meaning that the Claim 30 Cap value will be used instead of the offered value
 - Claim 30 < Claim 10 meaning that the Claim 10 value will be used instead of the offered Claim 30 value.

7.11.2 Override MW Operating Limits for a Day and Hour

As discussed in §7.7 ARD Schedule Detail Defaults Display, you define on this display an ARD's MW operating limits under the selected schedule. These limit values are shown in the Defaults section for each schedule that you've created for the ARD. If you want to profile the ARD's MW operating limits on an hourly basis, irrespective of which schedule is designated as available for a given market, you override the schedule values using the ARD Hourly Updates display as follows:

- 1. In the ARD Navigation sub tabs, click the Hourly Updates tab.
 - The ARD Hourly Updates display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired asset from the **ARD** filter.
 - The display reflects the hourly overrides for the selected ARD.
- 3. To update the overrides for individual hours:
 - a. Enter override values for all of the limits in the hours that you wish. You must also enter ARD dispatch availability status overrides. If the ARD is Claim 10 or Claim 30 capable, you must also enter Claim 10 and Claim 30 override values. If you want to remove an override, you must enter blanks in every data entry cells and deselect all status overrides.
 - b. Click the **Update** button to save the data.

- 4. To update the overrides for a range of hours
 - a. Enter override values for all of the limits in the top row. You must also enter ARD dispatch availability status overrides. If the ARD is Claim 10 or Claim 30 capable, you must also enter Claim 10 and Claim 30 override values. If you want to remove overrides, leave all the data entry cells on the first row blank and the status overrides deselected.
 - b. In the Hour column, select a "From" hour (left drop-down selector) and a "To" hour (right drop-down selector) and then select the **Copy** button. This will copy the data entered in the top row to every row for the range of hours selected.
 - c. Repeat steps a and b if you want to enter/remove overrides for other ranges of hours.
 - d. Click the **Update** button to save the data for the selected hours.

7.11.3 Override the Status of an ARD Asset for a Day and Hour

As discussed in §7.3 ARD Default Parameters Display, you define on this display an ARD's default dispatch availability status (Default Status) for an entire market day. This status is shown in the Defaults section. If you want to profile the ARD's dispatch availability status on an hourly basis, you override this default status using the ARD Hourly Updates display.

- 1. In the ARD Navigation sub tabs, click the **Hourly Updates** tab.
 - The ARD Hourly Updates display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired asset from the **ARD** filter.
 - The display reflects the hourly overrides for the selected unit.
- 3. To update the overrides for individual hours:
 - a. Select the Unavailable/Economic/Must Run status for the hours that you want to override the default status. You must also enter override values for all the limits even if they are copies of the default values. If the ARD is Claim 10 or Claim 30 capable, you must also enter Claim 10 and Claim 30 override values. If you want to remove an override, deselect all the status overrides and enter blanks in all other data entry cells.
 - b. Click the **Update** button to save the data.
- 4. To update the overrides a range of hours
 - a. Select the status override value in the first row. You must also enter override values for all the limits even if they are copies of the default values. If the ARD is Claim 10 or Claim 30 capable, you must also enter Claim 10 and Claim 30 override values. If you want to remove overrides, leave all the data entry cells on the first row blank and the status overrides deselected.
- b. In the Hour column, select a "From" hour (left drop-down selector) and a "To" hour (right drop-down selector) and then select the **Copy** button. This will copy the data entered in the top row to every row for the range of hours selected.
- c. Repeat steps a and b if you want to enter/remove overrides for other ranges of hours.
- d. Click the **Update** button to save the data for the selected hours.

7.11.4 Self-Schedule an ARD Asset

If you want to self-schedule (must run) an ARD asset, you must submit a **Must Run** status and indicate whether the ARD is block loaded or if it can follow ISO dispatch instructions. To self-commit a ARD and allow the ISO to issue dispatch instructions, submit a must run status for the ARD, with a Min Consumption limit less than the Max Consumption limit. To submit a fixed MW schedule for the ARD with a must run status, set the Min Consumption limit equal to the Max Consumption limit.

Note: For Non-dispatchable ARDs the Minimum Consumption and Maximum Consumption values must be equal.

7.12 ARD by Portfolio Display

You can view the cleared results of the Day-Ahead Market for each ARD asset listed in a selected portfolio using the **ARD by Portfolio** display that is accessible on one of the ARD sub tabs. The day-ahead ARD information includes the name of the schedule used when committing the ARD. That will be the schedule that was designated to be available for the DA market on the selected day. The other information is the MW amount that the ARD is committed to consume along with the Locational Marginal Price (LMP) at the ARD Pnode for each hour of the market day. If the ARD was not committed, "(*No Schedule Committed*)" will be shown instead of the schedule name. Information for a 24-hour period is shown on two lines: the top line details the hours ending 01 through 12; the second line details hours ending 13 through 24. The last row of the display is labeled **Total Portfolio MW** and shows the total MW amount for each hour committed in the Day-Ahead market for all the ARDs in your portfolio. Information on the display is shown only after the Day-Ahead market has been cleared.

ISO new england										in: 2.0.1(2	014072211	19)	erface
la de la constante de la const				F	teadOnly P	articipant:	ISO New En	gland Inc			Y User	Superuser	*
Public Generation Demand ARD Regulation													
Private ARD Default Portfolio Ramp Rate Ramp Rat Messages Parameters Manager Default Hourly U		Schedule Offers Default	Schedule Offer Hourly Update		Hourly Updates	ARD By Portfolio	XML Download						
ARD By Portfolio													
Select Date and Portfolio	ARD By Portfolio for AAA on 24-	Aug-2014											
Date: 24-Aug-2014	Ard/Schedule												
	ASSET NAME	01 / 13	02 / 14 03			06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12/24	
Portfolio: AAA	(DFLPRICE)	0@	0@ 0		0@	0@	0@	0@	0@	0@	0@	0@	
Totals shown on		\$20.08 0@	\$18.18 \$18 0@ 0		\$19.12 0@	\$19.83 0@	\$22.25 0@	\$22.37 0@	\$22.51 0@	\$25.51 0@	\$28.33 0@	\$31.47 0@	
		\$32.23	\$36.89 \$30			\$40.54	\$35.98	\$34.03	\$31.91	\$23.94	\$22.44	\$21.31	
last rows of the	Total Portfolio MW	01/13	02/14 03			06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12/24	
display		0	0 0	0	0	0	0	0	0	0	0	0	
		0	0 0	0	0	0	0	0	0	0	0	0	
													Refresh
Messages													8

The format of this display is shown below:

The following ARD information is displayed for each ARD asset included in the selected ARD portfolio:

- **ARD/Schedule** the name of the ARD and of the schedule on which it was committed. If the ARD was not committed, "No Schedule Committed" is shown instead of a schedule name.
- **MW** @ **Price** the MW amount that ARD is committed to consume and the \$/MWH DA LMP at the ARD Pnode for each hour of the day.

7.12.1 View DA Commitment for each ARD in Portfolio

To view the commitment in the Day-Ahead market of the ARDs listed in your portfolios and the Day-Ahead LMPs at their Phodes:

1. In the ARD Navigation sub tabs, click the **ARD by Portfolio** tab.

• The ARD by Portfolio display is shown in the display pane.

2. In the Filters pane:

- a. Select the desired date from the **Date** filter.
- b. Select the desired portfolio from the **Portfolio** filter.
 - The display reflects the selected market day's day-ahead commitment and LMPs for each of the ARDs listed in the selected portfolio. If the Day-Ahead market has not yet been cleared for the selected day, the following text is shown in the display title bar:

"The day ahead market results are not available ...".

7.13 XML Download for ARD Display

The XML Download display is accessible through the ARD Navigation sub tab. This display enables you to download pre-defined ARD specific XML report files of data that you maintain in the ISO-NE Market System, including results from the cleared Day-Ahead market. It also provides you the means to upload data in XML formatted data files into the ISO-NE Market System to be used to clear the Day-Ahead and Real Time markets. The format of the XML files can be found in the **eMarket Data Exchange Specification**, which is found on the <u>ISO New</u> <u>England website</u> among the documents listed on the Participate > Support > Upload and Download File Format Protocols page.

ISO new england								Market Status: OPEN(01) Participant:	GDF Suez Energy Market	eMarket Use Version: 2.0.1(201407221119 Certificate User: Superuser Ing NA VUser:) 23-Aug-2014 12:21:10
Public Generation Demand ARD Regulation Private ARD Default Portfolio Ramp Rate Ramp R			edule Schedule Offers	Schedule	Hourly	ARD By	XML				
Messages Parameters Manager Default Hourly XML Download	Updates Manager	Detail Defaults Offe	rs Default Hourly Updates	Selection	Updates	Portfolio	Download				
Select Date	XML Download or	24-Aug-2014									
Date: 24-Aug-2014	Schedules Schedule Do Schedule Of Schedule Of Schedule Se Asset-Relate Asset-Relate	ad Demand Ram stails fors fers Hourly section ad Demand Hour	iy Updates p Rate Hourly Update	95				Uplead XML SOAP Request	Browse		
Messages											۲

The following XML reports of ARD data are available for downloading:

- Asset-Related Demand
- Asset-Related Demand Ramp Rates
- Asset-Related Demand Ramp Rate Hourly Updates
- ARD Schedules
- ARD Schedule Details
- ARD Schedule Offers
- ARD Schedule Offers Hourly (Updates)
- ARD Schedule Selection
- Asset-Related Demand Hourly Updates
- Asset-Related Demand Result

All but the Asset-Related Demand Result report returns an XML formatted report for the selected date and for every ARD that belongs to you as Lead Participant. Asset-Related Demand Result report returns information for all the unique ARDs listed in all of the user's portfolios.

See §3.2 for details on uploading and downloading files. The following sections provide samples of the downloaded reports.

7.13.1 Asset-Related Demand XML Report

This report contains the information that is shown on the ARD Default Parameters display for every ARD that belongs to you as Lead Participant. The following is a sample of the report.

<?xml version="1.0" encoding="UTF-8" standalone="yes"?> <GetARDResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages"> <ARD day="2014-08-24" ID="12345"> <ARDLongName>MY LOAD</ARDLongName> <ARDShortName>MYLD1</ARDShortName> <AssetType>Pump</AssetType> <AssetNumber>12345</AssetNumber> <Claim10Settings claim10Capable="true"> <Claim10>0</Claim10> <Claim10Cap xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> </Claim10Settings> <Claim30Settings claim30Capable="true"> <Claim30>0</Claim30> <Claim30Cap xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> </Claim30Settings> <DayAheadRequired>false</DayAheadRequired> <DefaultStatus>Economic</DefaultStatus> <Dispatchable>true</Dispatchable> <DispatchZone>West MA</DispatchZone> <LeadParticipantID>88888</LeadParticipantID> <LeadParticipantName>IAM PARTICIPANT</LeadParticipantName> <NodeID>99999</NodeID> <NodeName>ANODE</NodeName> <RampRateCap>11</RampRateCap> <RampRateDefault>9999</RampRateDefault> </ARD> <ARD day="2014-08-24" ID="55555"> <ARDLongName>MY LOAD 2</ARDLongName> <ARDShortName>MYLD2</ARDShortName> <AssetType>Pump</AssetType> <AssetNumber>555555</AssetNumber> <Claim10Settings claim10Capable="true"> <Claim10>0</Claim10> <Claim10Cap xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> </Claim10Settings> <Claim30Settings claim30Capable="true"> <Claim30>0</Claim30> <Claim30Cap xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> </Claim30Settings> <DayAheadRequired>false</DayAheadRequired> <DefaultStatus>Economic</DefaultStatus> <Dispatchable>true</Dispatchable> <DispatchZone>West MA</DispatchZone> <LeadParticipantID>99999</LeadParticipantID> <LeadParticipantName> IAM PARTICIPANT 2</LeadParticipantName> <NodeID>55555</NodeID> <NodeName>ANODE2</NodeName> <RampRateCap xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <RampRateDefault>9999</RampRateDefault> </ARD>

</GetARDResponse>

7.13.2 Asset-Related Demand Ramp Rates XML Report

This report contains the information that is shown on the ARD Ramp Rates display for every ARD that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetARDRampRateResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <ARDRampRate day="2014-08-24" ID="55555">
   <RampRateCap>11</RampRateCap>
   <RampRateCurve>
     <RampRatePoint rate="1" MW="10"/>
   </RampRateCurve>
   <ShortName>ARDUX</ShortName>
  </ARDRampRate>
  <ARDRampRate day="2014-08-24" ID="666666">
   <RampRateCurve>
     <RampRatePoint rate="30" MW="25"/>
   </RampRateCurve>
   <ShortName>ARDUY</ShortName>
  </ARDRampRate>
</GetARDRampRateResponse>
```

7.13.3 Asset-Related Demand Ramp Rate Hourly Updates XML Report

This report contains the information that is shown on the ARD Ramp Rate Hourly Updates display for every ARD that belongs to you as Lead Participant, and for which hourly updates exist. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetARDRampRateHourlyResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <ARDRampRateHourly day="2014-08-24" ID="55555">
    <HourlvProfile>
      <HourlyValues time="2014-08-24T00:00:00-04:00">
        <RampRateCurve>
         <RampRatePoint rate="10" MW="10"/>
         <RampRatePoint rate="1" MW="13"/>
       </RampRateCurve>
      </HourlyValues>
      <HourlyValues time="2014-08-24T01:00:00-04:00">
        <RampRateCurve>
         <RampRatePoint rate="8" MW="5"/>
         <RampRatePoint rate="12" MW="10"/>
         <RampRatePoint rate="10" MW="12"/>
         <RampRatePoint rate="1" MW="13"/>
        </RampRateCurve>
      </HourlyValues>
      <HourlyValues time="2014-08-24T22:00:00-04:00">
       <RampRateCurve>
         <RampRatePoint rate="10" MW="10"/>
         <RampRatePoint rate="1" MW="13"/>
        </RampRateCurve>
      </HourlyValues>
      <HourlyValues time="2014-08-24T23:00:00-04:00">
       <RampRateCurve>
         <RampRatePoint rate="8" MW="5"/>
```

```
<RampRatePoint rate="12" MW="10"/>
         <RampRatePoint rate="10" MW="12"/>
         <RampRatePoint rate="1" MW="13"/>
       </RampRateCurve>
     </HourlyValues>
   </HourlyProfile>
   <ShortName>ARPXP</ShortName>
 </ARDRampRateHourly>
 <ARDRampRateHourly day="2014-08-24" ID="44444">
   <HourlyProfile>
     <HourlyValues time="2014-08-24T00:00:00-04:00">
       <RampRateCurve>
         <RampRatePoint rate="27" MW="240"/>
         <RampRatePoint rate="29" MW="245"/>
         <RampRatePoint rate="30" MW="250"/>
       </RampRateCurve>
     </HourlyValues>
     <HourlyValues time="2014-08-24T01:00:00-04:00">
       <RampRateCurve>
         <RampRatePoint rate="27" MW="240"/>
         <RampRatePoint rate="29" MW="245"/>
         <RampRatePoint rate="30" MW="250"/>
       </RampRateCurve>
     </HourlyValues>
     <HourlyValues time="2014-08-24T22:00:00-04:00">
       <RampRateCurve>
         <RampRatePoint rate="27" MW="240"/>
         <RampRatePoint rate="28" MW="245"/>
         <RampRatePoint rate="29" MW="246"/>
         <RampRatePoint rate="29.5" MW="247"/>
         <RampRatePoint rate="30" MW="248"/>
         <RampRatePoint rate="31" MW="249"/>
         <RampRatePoint rate="30" MW="250"/>
       </RampRateCurve>
     </HourlyValues>
     <HourlyValues time="2014-08-24T23:00:00-04:00">
       <RampRateCurve>
         <RampRatePoint rate="27" MW="240"/>
         <RampRatePoint rate="28" MW="245"/>
         <RampRatePoint rate="29" MW="246"/>
         <RampRatePoint rate="29.5" MW="247"/>
         <RampRatePoint rate="30" MW="248"/>
         <RampRatePoint rate="31" MW="249"/>
         <RampRatePoint rate="30" MW="250"/>
       </RampRateCurve>
     </HourlyValues>
   </HourlyProfile>
   <ShortName>ARPVP</ShortName>
 </ARDRampRateHourly>
</GetARDRampRateHourlyResponse>
```

7.13.4 Asset-Related Demand Schedules XML Report

This report contains the information that is shown on the ARD Schedule Manager display for all schedules of every ARD that belongs to you as Lead Participant. The following is a sample of the report.

<?xml version="1.0" encoding="UTF-8" standalone="yes"?> <GetARDScheduleResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages"> <ARDSchedule schedID="5555500"> <ARDID>55555</ARDID> <ARDShortName>ARDXP</ARDShortName> <SchedLongName>DFLPRICE</SchedLongName> <SchedShortName>DFLPRICE</SchedShortName> <SchedType>Price</SchedType> <SchedTypeID>0</SchedTypeID> </ARDSchedule> <ARDSchedule schedID="5555599"> <ARDID>55555</ARDID> <ARDShortName>ARDXP</ARDShortName> <SchedLongName>DFLCOST</SchedLongName> <SchedShortName>DFLCOST</SchedShortName> <SchedType>Cost</SchedType> <SchedTypeID>99</SchedTypeID> </ARDSchedule> <ARDSchedule schedID="6666600"> <ARDID>66666</ARDID> <ARDShortName>ARDVP</ARDShortName> <SchedLongName>DFLPRICE</SchedLongName> <SchedShortName>DFLPRICE</SchedShortName> <SchedType>Price</SchedType> <SchedTypeID>0</SchedTypeID> </ARDSchedule> <ARDSchedule schedID="66666612"> <ardid>66666</ardid> <ARDShortName>ARDVP</ARDShortName> <SchedLongName>ARDVP Price</SchedLongName> <SchedShortName>ARDVP1</SchedShortName> <SchedType>Price</SchedType> <SchedTypeID>12</SchedTypeID> </ARDSchedule> <ARDSchedule schedID="66666699"> <ARDID>66666</ARDID> <ARDShortName>ARDVP</ARDShortName> <SchedLongName>DFLCOST</SchedLongName> <SchedShortName>DFLCOST</SchedShortName> <SchedType>Cost</SchedType> <SchedTypeID>99</SchedTypeID> </ARDSchedule> </GetARDScheduleResponse>

7.13.5 Asset-Related Demand Schedule Details XML Report

This report contains the information that is shown on the ARD Schedule Detail display for all schedules of every ARD that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetARDScheduleDetailResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <ARDScheduleDetail day="2014-08-24" ID="5555500">
   <MarketType>Both</MarketType>
   <MaxConsumption>295</MaxConsumption>
   <MinConsumption>240</MinConsumption>
   <MaxDailyConsumption>5760</MaxDailyConsumption>
   <MaxDailyStarts>4</MaxDailyStarts>
   <MinDownTime>1.00</MinDownTime>
   <MinRunTime>1.00</MinRunTime>
   <ScheduleAvailable>true</ScheduleAvailable>
   <ShortName>DFLPRICE</ShortName>
  </ARDScheduleDetail>
  <ARDScheduleDetail day="2014-08-24" ID="5555599">
   <MarketType>Both</MarketType>
   <MaxConsumption xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
   <MinConsumption xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
   <MaxDailyConsumption>5760</MaxDailyConsumption>
   <MaxDailyStarts>4</MaxDailyStarts>
   <MinDownTime>1.00</MinDownTime>
   <MinRunTime>1.00</MinRunTime>
   <ScheduleAvailable>true</ScheduleAvailable>
   <ShortName>DFLCOST</ShortName>
  </ARDScheduleDetail>
  <ARDScheduleDetail day="2014-08-24" ID="66666600">
   <MarketType>Both</MarketType>
   <MaxConsumption>295</MaxConsumption>
   <MinConsumption>240</MinConsumption>
   <MaxDailyConsumption xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
   <MaxDailyStarts xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
   <MinDownTime>1.00</MinDownTime>
   <MinRunTime>1.00</MinRunTime>
   <ScheduleAvailable>true</ScheduleAvailable>
   <ShortName>DFLPRICE</ShortName>
  </ARDScheduleDetail>
  <ARDScheduleDetail day="2014-08-24" ID="66666699">
   <MarketType>Both</MarketType>
   <MaxConsumption>0</MaxConsumption>
   <MinConsumption>0</MinConsumption>
   <MaxDailyConsumption xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
   <MaxDailyStarts xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
   <MinDownTime>1.00</MinDownTime>
   <MinRunTime>1.00</MinRunTime>
   <ScheduleAvailable>true</ScheduleAvailable>
   <ShortName>DFLCOST</ShortName>
  </ARDScheduleDetail>
  <ARDScheduleDetail day="2014-08-24" ID="88888800">
   <MarketType>Both</MarketType>
   <MaxConsumption xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
   <MinConsumption xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
   <MaxDailyConsumption>5760</MaxDailyConsumption>
   <MaxDailyStarts>4</MaxDailyStarts>
```

<MinDownTime>1.00</MinDownTime> <MinRunTime>1.00</MinRunTime> <ScheduleAvailable>false</ScheduleAvailable> <ShortName>DFLPRICE</ShortName> </ARDScheduleDetail> <ARDScheduleDetail day="2014-08-24" ID="8888812"> <MarketType>Both</MarketType> <MaxConsumption>13</MaxConsumption> <MinConsumption>0</MinConsumption> <MaxDailyConsumption>5760</MaxDailyConsumption> <MaxDailyStarts>4</MaxDailyStarts> <MinDownTime>1.00</MinDownTime> <MinRunTime>1.00</MinRunTime> <ScheduleAvailable>true</ScheduleAvailable> <ShortName>RR-P_12</ShortName> </ARDScheduleDetail> <ARDScheduleDetail day="2014-08-24" ID="88888899"> <MarketType>Both</MarketType> <MaxConsumption xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MinConsumption xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MaxDailyConsumption>5760</MaxDailyConsumption> <MaxDailyStarts>4</MaxDailyStarts> <MinDownTime>1.00</MinDownTime> <MinRunTime>1.00</MinRunTime> <ScheduleAvailable>true</ScheduleAvailable> <ShortName>DFLCOST</ShortName> </ARDScheduleDetail> </GetARDScheduleDetailResponse>

7.13.6 Asset-Related Demand Schedule Offers XML Report

This report contains the information that is shown on the ARD Schedule Offers display for all schedules of every ARD that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetARDScheduleBidResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <ARDScheduleBid day="2014-08-24" ID="3333300">
    <PriceCurve>
      <PricePoint price="20" MW="200"/>
    </PriceCurve>
    <UseOfferSlope>false</UseOfferSlope>
    <ShortName>DFLPRICE</ShortName>
  </ARDScheduleBid>
  <ARDScheduleBid day="2014-08-24" ID="3333399">
    <PriceCurve>
      <PricePoint price="21.5" MW="295"/>
    </PriceCurve>
    <UseOfferSlope>true</UseOfferSlope>
    <ShortName>DFLCOST</ShortName>
  </ARDScheduleBid>
  <ARDScheduleBid day="2014-08-24" ID="4444400">
    <PriceCurve>
      <PricePoint price="23" MW="295"/>
    </PriceCurve>
    <UseOfferSlope>false</UseOfferSlope>
    <ShortName>DFLPRICE</ShortName>
  </ARDScheduleBid>
  <ARDScheduleBid day="2014-08-24" ID="4444499">
    <PriceCurve>
      <PricePoint price="15" MW="295"/>
    </PriceCurve>
    <UseOfferSlope>false</UseOfferSlope>
    <ShortName>DFLCOST</ShortName>
  </ARDScheduleBid>
  <ARDScheduleBid day="2014-08-24" ID="55555512">
    <PriceCurve>
      <PricePoint price="6" MW="6"/>
      <PricePoint price="5.99" MW="13"/>
    </PriceCurve>
    <UseOfferSlope>false</UseOfferSlope>
    <ShortName>RR-P_12</ShortName>
  </ARDScheduleBid>
  <ARDScheduleBid day="2014-08-24" ID="5555599">
    <PriceCurve>
      <PricePoint price="61" MW="6"/>
      <PricePoint price="59" MW="13"/>
    </PriceCurve>
    <UseOfferSlope>false</UseOfferSlope>
    <ShortName>DFLCOST</ShortName>
  </ARDScheduleBid>
</GetARDScheduleBidResponse>
```

7.13.7 Asset-Related Demand Schedule Offers Hourly Updates XML Report

This report contains the information that is shown on the ARD Schedule Offers Hourly Updates display for every ARD that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetARDScheduleBidHourlyResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <ARDScheduleBidHourly day="2014-08-24" ID="5555500">
    <HourlyProfile>
      <HourlyValues time="2014-08-24T00:00:00-04:00">
        <PriceCurve>
          <PricePoint price="45.45" MW="240"/>
          <PricePoint price="40.4" MW="255"/>
          <PricePoint price="35.35" MW="270"/>
          <PricePoint price="30.3" MW="280"/>
          <PricePoint price="25" MW="295"/>
        </PriceCurve>
        <UseOfferSlope>false</UseOfferSlope>
      </HourlyValues>
      <HourlyValues time="2014-08-24T01:00:00-04:00">
        <PriceCurve>
          <PricePoint price="50.45" MW="241"/>
          <PricePoint price="45.4" MW="256"/>
          <PricePoint price="40.35" MW="271"/>
          <PricePoint price="35.3" MW="281"/>
          <PricePoint price="30" MW="296"/>
        </PriceCurve>
        <UseOfferSlope>false</UseOfferSlope>
      </HourlyValues>
      <HourlyValues time="2014-08-24T22:00:00-04:00">
        <PriceCurve>
          <PricePoint price="45.45" MW="240"/>
          <PricePoint price="40.4" MW="255"/>
          <PricePoint price="35.35" MW="270"/>
          <PricePoint price="30.3" MW="280"/>
          <PricePoint price="25" MW="295"/>
        </PriceCurve>
        <UseOfferSlope>false</UseOfferSlope>
      </HourlyValues>
      <HourlyValues time="2014-08-24T23:00:00-04:00">
        <PriceCurve>
          <PricePoint price="50.45" MW="241"/>
          <PricePoint price="45.4" MW="256"/>
          <PricePoint price="40.35" MW="271"/>
          <PricePoint price="35.3" MW="281"/>
          <PricePoint price="30" MW="296"/>
        </PriceCurve>
        <UseOfferSlope>false</UseOfferSlope>
      </HourlyValues>
    </HourlyProfile>
    <ShortName>DFLPRICE</ShortName>
  </ARDScheduleBidHourly>
  <ARDScheduleBidHourly day="2014-08-24" ID="66666612">
    <HourlyProfile>
```

```
<HourlyValues time="2014-08-24T00:00:00-04:00">
        <PriceCurve>
          <PricePoint price="16" MW="1"/>
          <PricePoint price="14" MW="2"/>
          <PricePoint price="12" MW="3"/>
          <PricePoint price="10" MW="4"/>
          <PricePoint price="8" MW="5"/>
          <PricePoint price="6" MW="6"/>
          <PricePoint price="4" MW="7"/>
          <PricePoint price="2" MW="9"/>
          <PricePoint price="0" MW="12"/>
          <PricePoint price="-10" MW="16"/>
        </PriceCurve>
        <UseOfferSlope>true</UseOfferSlope>
      </HourlyValues>
      <HourlyValues time="2014-08-24T01:00:00-04:00">
        <PriceCurve>
          <PricePoint price="16" MW="1"/>
          <PricePoint price="14" MW="2"/>
          <PricePoint price="12" MW="3"/>
          <PricePoint price="10" MW="4"/>
          <PricePoint price="8" MW="5"/>
          <PricePoint price="6" MW="6"/>
          <PricePoint price="4" MW="7"/>
          <PricePoint price="2" MW="9"/>
          <PricePoint price="0" MW="12"/>
          <PricePoint price="-10" MW="16"/>
        </PriceCurve>
        <UseOfferSlope>true</UseOfferSlope>
      </HourlyValues>
      <HourlyValues time="2014-08-24T23:00:00-04:00">
        <PriceCurve>
          <PricePoint price="16" MW="1"/>
          <PricePoint price="15" MW="2"/>
          <PricePoint price="12" MW="5"/>
          <PricePoint price="5" MW="10"/>
          <PricePoint price="0" MW="15"/>
          <PricePoint price="-15" MW="20"/>
        </PriceCurve>
        <UseOfferSlope>true</UseOfferSlope>
      </HourlyValues>
    </HourlyProfile>
    <ShortName>DFLPRICE</ShortName>
  </ARDScheduleBidHourly>
</GetARDScheduleBidHourlyResponse>
```

7.13.8 Asset-Related Demand Schedule Selection XML Report

This report contains the information that is shown on the ARD Schedule Selection display for all schedules of every ARD that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetARDScheduleSelectionResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <ARDScheduleSelection day="2014-08-24">
    <ARDSchedule ID="2222200">
      <Available>true</Available>
      <ShortName>DFLPRICE</ShortName>
    </ARDSchedule>
    <ARDSchedule ID="2222299">
      <Available>true</Available>
      <ShortName>DFLCOST</ShortName>
    </ARDSchedule>
    <ARDSchedule ID="3333300">
      <Available>true</Available>
      <ShortName>DFLPRICE</ShortName>
    </ARDSchedule>
    <ARDSchedule ID="3333399">
      <Available>true</Available>
      <ShortName>DFLCOST</ShortName>
    </ARDSchedule>
  </ARDScheduleSelection>
</GetARDScheduleSelectionResponse>
```

7.13.9 Asset-Related Demand Hourly Updates XML Report

This report contains the information that is shown on the ARD Hourly Updates display for every ARD that belongs to you as Lead Participant, and for which hourly updates exist. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetARDHourlyResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <ARDHourly day="2014-08-24" ID="33333">
    <HourlyProfile>
      <HourlyValues time="2014-08-24T00:00:00-04:00">
       <MaxConsumption>0</MaxConsumption>
       <MinConsumption>0</MinConsumption>
       <ResourceStatus>Economic</ResourceStatus>
        <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
       <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
      </HourlvValues>
      <HourlyValues time="2014-08-24T01:00:00-04:00">
        <MaxConsumption>0</MaxConsumption>
       <MinConsumption>0</MinConsumption>
        <ResourceStatus>Economic</ResourceStatus>
       <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
        <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
      </HourlyValues>
      <HourlyValues time="2014-08-24T22:00:00-04:00">
       <MaxConsumption>0</MaxConsumption>
        <MinConsumption>0</MinConsumption>
       <ResourceStatus>Economic</ResourceStatus>
```

```
<Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
      <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
    </HourlyValues>
    <HourlyValues time="2014-08-24T23:00:00-04:00">
      <MaxConsumption>0</MaxConsumption>
      <MinConsumption>0</MinConsumption>
      <ResourceStatus>Economic</ResourceStatus>
      <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
     <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
    </HourlyValues>
  </HourlyProfile>
  <ShortName>ARDXP</ShortName>
</ARDHourly>
<ARDHourly day="2014-08-24" ID="44444">
  <HourlyProfile>
    <HourlyValues time="2014-08-24T00:00:00-04:00">
     <MaxConsumption>0</MaxConsumption>
      <MinConsumption>0</MinConsumption>
      <ResourceStatus>Economic</ResourceStatus>
     <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
      <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
    </HourlyValues>
    <HourlyValues time="2014-08-24T01:00:00-04:00">
     <MaxConsumption>288</MaxConsumption>
     <MinConsumption>280</MinConsumption>
      <ResourceStatus>Economic</ResourceStatus>
      <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
     <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
    </HourlyValues>
    <HourlyValues time="2014-08-24T23:00:00-04:00">
     <MaxConsumption>0</MaxConsumption>
     <MinConsumption>0</MinConsumption>
      <ResourceStatus>Economic</ResourceStatus>
      <Claim10 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
      <Claim30 xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
    </HourlyValues>
  </HourlyProfile>
  <ShortName>ARDVP</ShortName>
</ARDHourly>
```

</GetARDHourlyResponse>

7.13.10 Asset-Related Demand Results XML Report

This report contains the same information that is shown on the ARD by Portfolio display. The report contains information only after the Day-Ahead market has been cleared and the time for the opening of the Re-Offer Period of the selected day has passed. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetARDByParticipantResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <ARDBvParticipant day="2014-08-24" ID="44444">
    <ARDShortName>ARDXP</ARDShortName>
    <ARDScheduleID>4444400</ARDScheduleID>
    <ARDScheduleShortName>DFLPRICE</ARDScheduleShortName>
    <HourlyProfile>
      <HourlyValues time="2014-08-24T00:00:00-04:00" MW="0" price="36.2"/>
      <HourlvValues time="2014-08-24T01:00:00-04:00" MW="0" price="34.71"/>
      <HourlyValues time="2014-08-24T02:00:00-04:00" MW="0" price="33.81"/>
      <HourlyValues time="2014-08-24T03:00:00-04:00" MW="0" price="32.68"/>
      <HourlyValues time="2014-08-24T04:00:00-04:00" MW="0" price="33.83"/>
      <HourlyValues time="2014-08-24T05:00:00-04:00" MW="0" price="34.61"/>
      <HourlyValues time="2014-08-24T06:00:00-04:00" MW="0" price="38.54"/>
      <HourlyValues time="2014-08-24T07:00:00-04:00" MW="0" price="41.6"/>
      <HourlyValues time="2014-08-24T08:00:00-04:00" MW="0" price="41.61"/>
      <HourlyValues time="2014-08-24T09:00:00-04:00" MW="0" price="43.22"/>
      <HourlyValues time="2014-08-24T10:00:00-04:00" MW="0" price="51"/>
      <HourlyValues time="2014-08-24T11:00:00-04:00" MW="0" price="49.75"/>
      <HourlyValues time="2014-08-24T12:00:00-04:00" MW="0" price="52.61"/>
      <HourlyValues time="2014-08-24T13:00:00-04:00" MW="0" price="55.27"/>
      <HourlyValues time="2014-08-24T14:00:00-04:00" MW="0" price="57.03"/>
      <HourlyValues time="2014-08-24T15:00:00-04:00" MW="0" price="57.14"/>
      <HourlyValues time="2014-08-24T16:00:00-04:00" MW="0" price="57.18"/>
      <HourlyValues time="2014-08-24T17:00:00-04:00" MW="0" price="55.1"/>
      <HourlyValues time="2014-08-24T18:00:00-04:00" MW="0" price="52.06"/>
      <HourlyValues time="2014-08-24T19:00:00-04:00" MW="0" price="48.8"/>
      <HourlyValues time="2014-08-24T20:00:00-04:00" MW="0" price="51.63"/>
      <HourlyValues time="2014-08-24T21:00:00-04:00" MW="0" price="42.16"/>
      <HourlyValues time="2014-08-24T22:00:00-04:00" MW="0" price="40.7"/>
      <HourlyValues time="2014-08-24T23:00:00-04:00" MW="0" price="40.36"/>
    </HourlyProfile>
  </ARDByParticipant>
  <ARDBvParticipant day="2014-08-24" ID="55555">
    <ARDShortName>ARDVP</ARDShortName>
    <ARDScheduleID>5555500</ARDScheduleID>
    <ARDScheduleShortName>DFLPRICE</ARDScheduleShortName>
    <HourlvProfile>
      <HourlyValues time="2014-08-24T00:00:00-04:00" MW="0" price="36.2"/>
      <HourlyValues time="2014-08-24T01:00:00-04:00" MW="288" price="34.71"/>
      <HourlyValues time="2014-08-24T02:00:00-04:00" MW="288" price="33.81"/>
      <HourlyValues time="2014-08-24T03:00:00-04:00" MW="288" price="32.68"/>
      <HourlyValues time="2014-08-24T04:00:00-04:00" MW="286" price="33.83"/>
      <HourlyValues time="2014-08-24T05:00:00-04:00" MW="286" price="34.61"/>
      <HourlyValues time="2014-08-24T06:00:00-04:00" MW="0" price="38.54"/>
      <HourlvValues time="2014-08-24T07:00:00-04:00" MW="0" price="41.6"/>
      <HourlyValues time="2014-08-24T08:00:00-04:00" MW="0" price="41.61"/>
      <HourlyValues time="2014-08-24T09:00:00-04:00" MW="0" price="43.22"/>
      <HourlyValues time="2014-08-24T10:00:00-04:00" MW="0" price="51"/>
      <HourlyValues time="2014-08-24T11:00:00-04:00" MW="0" price="49.75"/>
      <HourlyValues time="2014-08-24T12:00:00-04:00" MW="0" price="52.61"/>
```

<HourlyValues time="2014-08-24T13:00:00-04:00" MW="0" price="55.27"/>
<HourlyValues time="2014-08-24T14:00:00-04:00" MW="0" price="57.03"/>
<HourlyValues time="2014-08-24T15:00:00-04:00" MW="0" price="57.14"/>
<HourlyValues time="2014-08-24T16:00:00-04:00" MW="0" price="57.18"/>
<HourlyValues time="2014-08-24T17:00:00-04:00" MW="0" price="55.1"/>
<HourlyValues time="2014-08-24T18:00:00-04:00" MW="0" price="55.1"/>
<HourlyValues time="2014-08-24T19:00:00-04:00" MW="0" price="52.06"/>
<HourlyValues time="2014-08-24T19:00:00-04:00" MW="0" price="51.63"/>
<HourlyValues time="2014-08-24T20:00:00-04:00" MW="0" price="51.63"/>
<HourlyValues time="2014-08-24T21:00:00-04:00" MW="0" price="40.8"/>
<HourlyValues time="2014-08-24T22:00:00-04:00" MW="0" price="40.7"/>
<HourlyValues time="2014-08-24T23:00:00-04:00" MW="0" price="40.36"/>
</HourlyValues time="2014-08-24T23:00:00-04:00" MW="0" price="40.36"/>
</HOURlyProfile>

</GetARDByParticipantResponse>

8. Public Tab

The Public Tab selection provides all market system users' access to displays showing general information from the cleared Day-Ahead Market for a selected market day. Selecting the Public tab will show all of the display sub tabs shown below.



The information that is shown on these public displays is available only after the Day-Ahead Market for the selected day is cleared. On each of the public displays, information for a 24-hour period is shown on two lines as shown below: the top line shows data for hours ending 01 through 12; the second line shows data for hours ending 13 through 24.

Day-Ahead Demand	01/13	02/14	03/15	04 / 16	05/17	06/18	07 / 19	08 / 20	09/21	10/22	11 / 23	12/24
	10347	9869	10078	9963	10085	10145	10408	11277	12153	12535	12635	12403
	12248	12082	11772	11622	11651	11831	12034	12335	12734	12129	11309	10331

8.1 System Summary Display

The System Summary display shows the hourly values of the following:

- Day-Ahead Demand the total demand (net of all the demand bids) cleared in the Day-Ahead Market
- Day-Ahead Load Forecast the ISO forecasted system load.
- Day-Ahead Operating Reserve Objectives the system operating reserve requirements cleared in the Day-Ahead Market

IS	ne	ew engla	and									larket Statu				Vers	Mark ion: 2.0.3(e User: Sup	201408181	.502)	terface
											F	leadOnly F	articipant:	Constellat	ion NewEner	gy_Inc.		💙 Us	er: Superuse	r I
ublic	Generation	Demand	ARD Re	gulation																
ublic Iessages	System Summary	Prices	Binding Constraints	Reactive Interface Lim	Net Tie Schedules	XML Download														
Syste	m Sumi	mary																		
lect Date	,			~	System Summa	ry for 24-Au	g-2014													
Deb					Name															
Dau	e: 24-Aug-2	2014			Day-Ahead Der	nand		01/13	02/14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24	
								11517	11380	11203	11056	11190	11812	12358	13455	14564	15579	16211	16598	
								16832	17055	17153	17210	17198	17038	16738	16543	16460	15535	13785	12245	
					Day-Ahead Loa	d Forecast		01/13	02 / 14	03 / 15	04 / 16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24	
								11600	11020	10720	10600	10550	11290	12660	14070	14900	15490	15930	16270	
								16480	16660	16710	16730	16820	16850	16570	16220	16230	15310	13810	12400	
					Day-Ahead Op	erating Reserve	e Objective	01/13	02/14	03 / 15	04 / 16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24	
								2404	2324	2337	2337	2337	2404	2665	2657	2715	2715	2715	2715	
								2715	2715	2715	2715	2715	2715	2715	2715	2715	2715	2452	2324	
																				Refresh
essages																				

8.2 Prices Display

This display shows the Day-Ahead Locational Marginal Prices (LMPs) at every pricing location (Pnode) that were calculated by the clearing of the Day-Ahead Market for every hour of the market day.

																	orket U:	ser Inte	erfac
ISC	new engla	and											_			ficate User	: Superuser	-	2014 16:
X	P	W. W.									ReadOnly	Participant:	ISO Nev	# England I	nc		V Us	er: Superuser	
blic Ge	neration Demand	ARD Regu	lation																
	System Prices Summary	Binding Constraints	Reactive Interface Limi	Net Tie Schedules	XML Download														
ay Ah	ead Prices																		
t Date a	nd Portfolio		~	Day Ahead Pri															
Date:	24-Aug-2014		1	.H.INTERNAL_H	UB	01 / 13 20.04	02 / 14 18.21	03 / 15 18.21	04 / 16 17.04	05 / 17	06 / 18 19.86	07 / 19 22.46	08 / 20 22.57	09 / 21 22.62	10 / 22 25.59	11/23 28.44	12 / 24 31.49	-	
				.I.HQHIGATE120	12	32.14 01 / 13	36.71 02 / 14	39.18 03 / 15	42.01 04 / 16	41.50 05 / 17	40.38 06 / 18	35.87 07 / 19	33.92 08 / 20	31.87 09 / 21	23.98 10 / 22	22.58 11/23	21.43 12 / 24	_	
or (1 0110)	ALL PNODES		•			18.45 29.30	17.08 33.51	18.56 35.48	17.35 34.29	19.57 38.08	18.40 37.22	20.56 33.30	20.67 31.51	20.76 29.79	23.59 22.19	26.61 20.60	28.76 19.65	_	
				.I.HQ_P1_P234	5	29.30 01 / 13 19.76	02/14	03 / 15 17.93	04/16	05 / 17	06/18	07 / 19 22.25	08/20	29.79 09 / 21 22.25	22.19	20.00	12 / 24 30.96		
						31.57	18.00 36.07	38.45	16.78 40.52	40.82	19.59 39.75	35.29	22.25 33.34	31.35	23.61	22.25	21.12		
				.I.NRTHPORT13	85	01 / 13 20.00	02 / 14 18.15	03 / 15 18.33	04 / 16 17.15	05 / 17	19.78	07 / 19 22.19	08 / 20 22.30	09 / 21 22.41	10 / 22 25.36	11 / 23 28.15	12 / 24 31.23	-	
				I.ROSETON 34	51	32.04 01 / 13	36.67 02 / 14	39.20 03 / 15	41.76 04 / 16	41.62	40.49 06 / 18	35.89 07 / 19	33.96 08 / 20	31.84 00 / 21	23.85 10 / 22	22.35 11/23	21.26		
						19.55 31.49	18.20 36.05	18.48 38.40	17.29 40.25	19.39 40.82	19.68 39.76	21.89 35.32	21.93 33.34	21.99 31.34	24.96 23.46	28.06 21.85	30.75 20.85	_	
				.I.SALBRYNB34	51	01/13	02/14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24		
						18.06 26.73	16.43 26.65	16.37 26.55	15.37 27.20	17.31 27.15	17.82 21.13	20.56 25.43	19.99 24.23	19.94 26.86	22.45 20.98	25.03 19.89	26.80 19.59		
				.I.SHOREHAM13	899	01/13 20.08	02 / 14 18,59	03 / 15 18.69	04 / 16 17,12	05 / 17	06 / 18 19.37	07 / 19 22.22	08/20 21.87	09 / 21 22.47	10 / 22 24,90	11/23 28.28	12 / 24 31,39		
				ZICONNECTIO	7	31.38 01 / 13	38.77 02 / 14	39.24 03 / 15	42.02	41.59 05 / 17	40.49	35.89 07 / 19	33.93 08 / 20	32.51 09 / 21	23.89 10 / 22	22.42 11/23	21.28		
				.z.connectice		20.00	18.12	18.22	17.06	19.02	19.75	22.16	22.29	22.41	25.38	28.19	31.29	_	
				.Z.MAINE		32.03 01 / 13	36.65 02 / 14	39.09 03 / 15	41.68 04 / 16	41.45 05 / 17		35.78 07 / 19	33.83 08 / 20	31.74 09/21	23.83 10 / 22	22.37 11 / 23	21.24 12 / 24		
						19.12 30.41	17.38 34.09	17.29 35.98	16.20 36.73	18.17 38.12	18.97 36.40	21.68 33.33	21.54 31.53	21.57 30.25	24.36 22.81	27.16 21.44	29.80 20.58		
				Z.NEMASSBOS	T	01 / 13 20.12	02 / 14 18.28	03 / 15 18.23	04 / 16 17.06	05 / 17		07 / 19 22.69	08 / 20 22.81	09 / 21 22.85	10 / 22 25.85	11 / 23 28.79	12 / 24 41.52		
				7. 11710110100	0.00	46.71	49.46	64.99	265.09	51.73	44.16	36.25	34.23	32.20	24.21	22.77	21.58		
				.Z.NEWHAMPSH	IRE	01 / 13 19.96	02 / 14 18.15	03 / 15 18.11	04 / 16 16.94	05 / 17 18.94	19.82	07 / 19 22.56	08 / 20 22.55	09 / 21 22.58	10 / 22 25.58	11 / 23 28.58	12 / 24 31.48		
				Z.RHODEISLAN	D	32.14 01 / 13	36.64 02 / 14	39.10 03 / 15	41.79 04 / 16	41.48 05 / 17	40.37 06 / 18	35.94 07 / 19	33.97 08 / 20	32.01 09 / 21	23.98 10 / 22	22.44 11/23	21.27 12 / 24	_	
						20.14 31.67	18.26 36.23	18.25 38.63	17.07 41.00	19.04 41.00	19.93 39.92	22.57 35.41	23.11 33.44	22.64 31.40	25.37 23.77	28.14 22.75	31.05 21.56	_	
				Z.SEMASS		01/13	02/14	03 / 15	04/16	05 / 17	06/18	07 / 19	08/20	09/21	10/22	11/23	12/24		
						20.01 32.16	18.17 36.75	18.15 39.24	16.99 42.26	18.93 41.59	19.82 40.49	22.48 35.90	22.71 33.92	22.69 31.88	25.62 24.04	28.49 22.68	31.55 21.45		
				.Z.VERMONT		01 / 13 19.82	02 / 14 18.13	03 / 15 18.50	04 / 16 17.30	05 / 17	06 / 18 19.72	07 / 19 22.20	08/20 22.28	09/21 22.35	10/22 25.36	11/23 28.39	12 / 24 31.06		
				.Z.WCMASS		31.70 01 / 13	36.18 02 / 14	38.54 03 / 15	40.58	41.02	40.00	35.69 07 / 19	33.78 08 / 20	31.86	23.81 10/22	22.21	21.05	_	
						20.11	18.28	18.31	17.14	19.12	19.95	22.52	22.62	22.66	25.64	28.52	31.79		
				AR.BEARSWMP	13.8BSW1P	32.54 01 / 13	37.09 02 / 14	39.85 03 / 15	46.90 04 / 16	41.84 05 / 17		35.98 07 / 19	34.03 08 / 20	31.97 09 / 21	24.04 10 / 22	22.61 11 / 23	21.48 12 / 24		
						20.09 32.17	18.64 36.74	19.26 39.23	18.05 41.76	20.08 41.50	20.37 40.39	22.37 35.88	22.43 33.97	22.55 31.86	25.55 23.92	28.31 22.42	31.42 21.35		
				AR.BEARSWMP	13.8BSW2P	01 / 13	02 / 14 18.64	03 / 15	04 / 16 18.05	05 / 17		07 / 19 22.37	08/20	09/21	10 / 22 25.55	11 / 23 28.31	12 / 24		
						20.09	26.74	20.22	/14.76	44.60		22.37	22.43	22.00	20.00	28.31	31.42		
																			Ref
ages																			

8.3 Binding Constraints Display

This display shows all the transmission constraints that were applied and binding in the clearing of the Day-Ahead Market.

									Mark				c	Version	2.0.1(201 ser: Super	140722111 [.] Jser	23-Aug-201	
· Y									Rea	dOnly Parti	cipant: 1	SO New Engl	and Inc			Y User:	Superuser	
blic Generation Dem	and ARD Reg	ulation																
blic System Pric ssages Summary	es Binding Constraints	Reactive Interface Lin	Net Tie Schedules	XML Download														
inding Constrai	nts																	
ect Date		~	Binding Constra	aints on 24-Au	g-2014													
			Day Ahead Lim			14												
Date: 24-Aug-2014		•	Contingency															
			Interface		01/13	02 / 14	03 / 15	04 / 16	05 / 17	06 / 18	07 / 19	08 / 20	09/21	10 / 22	11/23	12/24		
																×		
					×	х	х	х	×	×								
			Day Ahead Lim	it for Node_No	rthport_Exp	port on 24-/	Aug-2014											
			Contingency															
			Generic Constra	aint	01/13	02 / 14	03 / 15	04 / 16	05 / 17	06 / 18	07 / 19	08 / 20	09/21	10 / 22	11 / 23	12/24		
									×									
			Day Ahead Lim	it for Node_Sh	oreham_Ex	port on 24-/	Aug-2014		~									
			Contingency															
			Generic Constra	aint	01/13	02 / 14	03 / 15	04 / 16	05 / 17	06 / 18	07 / 19	08 / 20	09/21	10 / 22	11/23	12/24		
					×	×	×	×	×	х	×	×	×	×	×	×		
					х	х	х	×	×	х	×	х	х	х	х	х		
			Day Ahead Lim	it for ORR-SO	on 24-Aug-	2014												
			Contingency															
			Interface		01/13	02 / 14	03 / 15	04 / 16	05 / 17	06 / 18	07 / 19	08 / 20	09/21	10 / 22	11/23	12/24		
					х	X	X	X	X	х	×	X	X			×		
							~	~	~	~	~	~	~					
			Day Ahead Lim	it for RI-IMP o	on 24-Aug-S	2014												
			Contingency Interface		01/13	02 / 14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12/24		
			Internative		01/13	02714	00710	04710	05717	00710	07710	X	09721	10722	11723	12724		
												~						
																		Refr

8.4 Reactive Interface Limits Display

This display shows the MW flow on each of the system interfaces along with the MW flow limit on the interface for each hour of the market day in the Day-Ahead Market clearing solution.

												2.0.1(2014		r Interfac
SO new england						Marke				Cer	tificate Use	er: Superus	er	23-Aug-2014 16
						Read	Only Particip	oant: ISC	New Englar	nd Inc			💙 User:	Superuser
olic Generation Demand ARD Regul	ition													
	Reactive Net Tie Interface Limits Schedu													
eactive Interface Limits														
ect Date	Reactive Int	erface Limits on 24-Aug-201	14											
	BH+KRE In	erface on 24-Aug-2014												
Date: 24-Aug-2014	Type													
	Operating L	imit 01 / 13	02 / 14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24	i i
		99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	
		99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	
	Actual Flow	01 / 13	02 / 14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24	
		326	314	319	323	323	333	406	386	538	618	607	629	
		627	607	610	634	607	547	621	617	610	573	407	409	
	BH-EXP Inte	erface on 24-Aug-2014												
	Туре													
	Operating L	imit 01/13	02 / 14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24	
		99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	
		99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	
	Actual Flow		02 / 14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24	
		219	207	210	209	205	211	284	263	416	495	485	502	
		515	509	509	522	526	458	493	496	530	475	260	261	
	BRAY-2 Int	erface on 24-Aug-2014												
	Туре													
	Operating L		02 / 14	03 / 15	04716	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24	
		99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	
		99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	99999	
	Actual Flow	01/13	02 / 14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12/24	
		298 485	298 465	298 463	298 461	298 463	298 465	298 467	385 472	383 474	422	450 385	459 298	
		465	400	403	401	403	405	407	4/2	4/4	388	380	298	

8.5 Net Tie Schedules Display

This display shows the scheduled MW flow on each of the ISO New England interfaces with its neighboring control areas as calculated in the clearing of the Day-Ahead Market.

ISO new england												Version: :	2.0.1(2014	07221119)	
non ongiana								t Status: C	_			tificate Use		-	23-Aug-2014 16:2
							Read	Only Partici	pant: ISC) New Englar	nd Inc			Vser:	Superuser
lic Generation Demand ARD	Regulation														
lic System Prices Bindin sages Summary Const		Net Tie XML Schedules Download													
sages Summary Const	aints Interface Limit	s schedules Download													
et Tie Schedules															
ct Date	«	Net Tie Schedules on 24-A													
ct Date	<u>>></u>	Area	ug-2014												
Date: 24-Aug-2014		.I.HQHIGATE120.2	01 / 13	02/14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12/24	
			193	175	00710	0	0	191	193	218	218	218	218	218	
			218	218	218	218	218	218	218	218	218	218	218	193	
		.I.HQ_P1_P2346.5	01/13	02 / 14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12/24	
			1296	1172	1236	1236	1236	1296	1504	1417.1	1537.7	1545	1545	1545	
			1545	1545	1545	1545	1545	1545	1545	1545	1545	1545	1270.9	896	
		.I.NRTHPORT138.5	01 / 13	02 / 14	03 / 15	04716	05 / 17	06 / 18	07 / 19	08/20	09/21	10/22	11/23	12 / 24	
			0	0	0	0	0	0	-50	-75	-50	-25	0	-25	
			-75	-80	-155	-127	-200	-175	-165	-155	-150	-75	-75	-75	
		J.ROSETON 345 1	01 / 13	02 / 14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12 / 24	
			143	293	243	236	253	194	119	202	152	205.7	302	268.7	
		J.SALBRYNB345 1	284 01 / 13	213 02 / 14	246 03/15	296 04 / 16	296 05 / 17	289 06 / 18	383 07 / 19	333 08 / 20	284 09 / 21	233 10 / 22	229 11/23	91 12 / 24	
			440	440	439	414	365	510	362	634	589	565	567	571	
			573	593	590	566	593	671	579	583	590	591	591	321	
		.I.SHOREHAM138 99	01 / 13	02/14	03 / 16	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12/24	
			-330	-330	-330	-330	-330	-330	-330	-330	-330	-330	-330	-330	
			-330	-330	-330	-330	-330	-330	-330	-330	-330	-330	-330	-330	
		Total MW	01 / 13	02 / 14	03 / 15	04/16	05 / 17	06 / 18	07 / 19	08/20	09/21	10 / 22	11/23	12/24	
			1742	1750	1588	1556	1524	1861	1798	2066.1	2116.7	2178.7	2302	2247.7	
			2215	2159	2114	2168	2122	2218	2230	2194	2157	2182	1903.9	1096	
															Refr

8.6 Public XML Download Display

The XML Download Display is used to view the following public XML reports of data from the clearing of the Day-Ahead Market:

- Day-Ahead System Summary
- Day-Ahead Prices
- Day-Ahead Binding Constraints
- Day-Ahead Reactive Limits
- Day-Ahead Net Tie Schedules

The reports are available for download after the Day-Ahead Market results are posted and the Re-Offer Period for the Real-Time Market opens. The format of the XML files can be found in the *eMarket Data Exchange Specification*, which is found on the *ISO New England website* among the documents listed on the Participate > Support > Upload and Download File Format Protocols page.

ISO new england	Market St	atus: CLOSED	eMarket User Interface Version: 2.0.1(201407221119) Certificate User: Superuser 23-Aug-2014 16:26:40
		Participant: ISO New England Inc	Vser: Superuser V
Public Generation Demand ARD Regulation			
Public System Prices Binding Reactive Messages Summary Constraints Interface Limit	s Net Tie XML Schedules Download		
XML Download	J		
Select Date «	XML Download on 24-Aug-2014		
Date: 24.Aug-2014	Day-Ahead System Summary Day-Ahead Prices Day-Ahead Binding Constraints Day-Ahead Reactive Limits Day-Ahead Net Tie Schedules	Upload XML SDAP Request	Browse
Messages			

See <u>Section</u> 4 for details on uploading and downloading files. The following sections provide samples of the downloaded reports.

8.6.1 DA System Summary XML Report

This report contains the information that you see on the System Summary display. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes" ?>

_ <GetSystemSummaryResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">

_ <SystemSummary day="2014-08-24">

_ <AreaSummary name="NEPEX">

_ <HourlyValues time="2014-08-24T00:00:00-04:00">

< DemandBidMW>10347</DemandBidMW>

< DemandForecastMW>10600</DemandForecastMW>

< Reserve>0</Reserve>

</HourlyValues>

_ <HourlyValues time="2014-08-24T01:00:00-04:00">

< DemandForecastMW>10600</DemandForecastMW>

< Reserve>0</Reserve>

</HourlyValues>

_ <HourlyValues time="2014-08-24T01:00:00-04:00">

< DemandBidMW>9869</DemandBidMW>

< DemandForecastMW>10170</DemandForecastMW>

< Reserve>0</Reserve>
```



8.6.2 DA Prices XML Report

This is a large report. It contains the information that you see on the Prices display, specifically the hourly DA nodal LMPs at every public pricing node (Pnode) in the data model. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes" ?>
- <GetPricesResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
   - <Prices day="2014-08-24">
      - <NodePrices ID="4000" name=".H.INTERNAL_HUB">
           <HourlyPrice time="2014-08-24T00:00:00-04:00" price="21.61" />
           <HourlyPrice time="2014-08-24T01:00:00-04:00" price="19.11" />
           <HourlyPrice time="2014-08-24T23:00:00-04:00" price="23.67" />
       </NodePrices>
       - <NodePrices ID="4001" name=".Z.MAINE">
           <HourlyPrice time="2014-08-24T00:00:00-04:00" price="21.02" />
           <HourlyPrice time="2014-08-24T01:00:00-04:00" price="18.67" />
           <HourlyPrice time="2014-08-24T23:00:00-04:00" price="23.17" />
        </NodePrices>
      - <NodePrices ID="39004" name="LD.VT YK 115">
           <HourlyPrice time="2014-08-24T00:00:00-04:00" price="21.46" />
           <HourlyPrice time="2014-08-24T22:00:00-04:00" price="23.08" />
           <HourlyPrice time="2014-08-24T23:00:00-04:00" price="23.23" />
       </NodePrices>
    </Prices>
 </GetPricesResponse>
```

8.6.3 DA Binding Constraints XML Report

This report contains the information that you see on the Binding Constraints display. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes" ?>
_<GetBindingLimitsResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
   - <BindingLimits day="2014-08-24">
      - <BindingLimit>
          <BranchName>CONVERTRK42-3 A LN</BranchName>
          <ContingencyName>Actual</ContingencyName>
         - <HoursImpacted>
              <Hour>2014-08-24T00:00:00-04:00</Hour>
              <Hour>2014-08-24T03:00:00-04:00</Hour>
              <Hour>2014-08-24T05:00:00-04:00</Hour>
              <Hour>2014-08-24T06:00:00-04:00</Hour>
              <Hour>2014-08-24T23:00:00-04:00</Hour>
          </HoursImpacted>
       </BindingLimit>
      - <BindingLimit>
          <BranchName>CONVERTR_K42-3_A_LN</BranchName>
          <ContingencyName>Actual</ContingencyName>
         - <HoursImpacted>
              <Hour>2014-08-24T01:00:00-04:00</Hour>
              <Hour>2014-08-24T02:00:00-04:00</Hour>
          </HoursImpacted>
       </BindingLimit>
    </BindingLimits>
</GetBindingLimitsResponse>
```

8.6.4 DA Reactive Limits XML Report

This report contains the information that you see on the Reactive Interface Limits display. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes" ?>
- <GetInterfaceLimitsResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
   - <InterfaceLimits day="2014-08-24">
       - <InterfaceLimit>
           <Name>BSTN</Name>
          - <HourlyProfile>
               <HourlyValues time="2014-08-24T00:00:00-04:00" flow="1509" limit="99999" />
               <HourlyValues time="2014-08-24T01:00:00-04:00" flow="1571" limit="99999" />
               <HourlyValues time="2014-08-24T02:00:00-04:00" flow="1520" limit="99999" />
               <HourlyValues time="2014-08-24T22:00:00-04:00" flow="1528" limit="99999" />
               <HourlyValues time="2014-08-24T23:00:00-04:00" flow="1341" limit="99999" />
           </HourlyProfile>
        </InterfaceLimit>
       - <InterfaceLimit>
           <Name>MENH</Name>
          - <HourlvProfile>
               <HourlyValues time="2014-08-24T00:00:00-04:00" flow="13" limit="99999" />
               <HourlyValues time="2014-08-24T01:00:00-04:00" flow="-41" limit="99999" />
               <HourlyValues time="2014-08-24T22:00:00-04:00" flow="-96" limit="99999" />
               <HourlyValues time="2014-08-24T23:00:00-04:00" flow="-26" limit="999999" />
```

```
</HourlyProfile>
      </InterfaceLimit>
     - <InterfaceLimit>
          <Name>MYS8-9</Name>
         - <HourlyProfile>
              <HourlyValues time="2014-08-24T00:00:00-04:00" flow="1297" limit="10000" />
              <HourlyValues time="2014-08-24T01:00:00-04:00" flow="986" limit="10000" />
              <HourlyValues time="2014-08-24T22:00:00-04:00" flow="1774" limit="10000" />
              <HourlyValues time="2014-08-24T23:00:00-04:00" flow="1780" limit="10000" />
          </HourlyProfile>
      </InterfaceLimit>
     - <InterfaceLimit>
          <Name>NE_E-W</Name>
         - <HourlyProfile>
              <HourlyValues time="2014-08-24T00:00:00-04:00" flow="757" limit="99999" />
              <HourlyValues time="2014-08-24T01:00:00-04:00" flow="557" limit="99999" />
              <HourlyValues time="2014-08-24T22:00:00-04:00" flow="-136" limit="999999" />
              <HourlyValues time="2014-08-24T23:00:00-04:00" flow="125" limit="99999" />
          </HourlyProfile>
      </InterfaceLimit>
     - <InterfaceLimit>
          <Name>NHME</Name>
         - <HourlyProfile>
              <HourlyValues time="2014-08-24T00:00:00-04:00" flow="-13" limit="99999" />
              <HourlyValues time="2014-08-24T01:00:00-04:00" flow="41" limit="999999" />
              <HourlyValues time="2014-08-24T22:00:00-04:00" flow="96" limit="99999" />
              <HourlyValues time="2014-08-24T23:00:00-04:00" flow="26" limit="99999" />
          </HourlyProfile>
      </InterfaceLimit>
   </InterfaceLimits>
</GetInterfaceLimitsResponse>
```

8.6.5 DA Net Tie Schedules XML Report

This report contains the information that you see on the Net Tie Schedules display. The following is a sample of the report.

```
<HourlyValue time="2014-08-24T22:00:00-04:00" MW="0" />
          <HourlyValue time="2014-08-24T23:00:00-04:00" MW="0" />
      </NetTieSchedule>
     - <NetTieSchedule area=".I.NRTHPORT138 5">
          <HourlyValue time="2014-08-24T00:00:00-04:00" MW="-100" />
          <HourlyValue time="2014-08-24T01:00:00-04:00" MW="-100" />
          <HourlyValue time="2014-08-24T22:00:00-04:00" MW="-185" />
          <HourlyValue time="2014-08-24T23:00:00-04:00" MW="-100" />
      </NetTieSchedule>
     - <NetTieSchedule area=".I.ROSETON 345 1">
          <HourlyValue time="2014-08-24T00:00:00-04:00" MW="-23" />
          <HourlyValue time="2014-08-24T01:00:00-04:00" MW="44" />
          <HourlyValue time="2014-08-24T22:00:00-04:00" MW="278" />
          <HourlyValue time="2014-08-24T23:00:00-04:00" MW="181" />
      </NetTieSchedule>
     - <NetTieSchedule area=".I.SALBRYNB345 1">
          <HourlyValue time="2014-08-24T00:00:00-04:00" MW="293" />
          <HourlyValue time="2014-08-24T01:00:00-04:00" MW="202" />
          <HourlyValue time="2014-08-24T22:00:00-04:00" MW="293" />
          <HourlyValue time="2014-08-24T23:00:00-04:00" MW="293" />
      </NetTieSchedule>
     - <NetTieSchedule area=".I.SHOREHAM138 99">
          <HourlyValue time="2014-08-24T00:00:00-04:00" MW="-330" />
          <HourlyValue time="2014-08-24T01:00:00-04:00" MW="-330" />
          <HourlyValue time="2014-08-24T22:00:00-04:00" MW="-330" />
          <HourlyValue time="2014-08-24T23:00:00-04:00" MW="-330" />
      </NetTieSchedule>
   </NetTieSchedules>
</GetNetTieSchedulesResponse>
```

9. Demand Response Resource (DRR) Tab

To access the DRR Tab, you must be a registered user with a DRR Read/Write or Read-Only role. When you have this access, you will be able to select the DRR tab and all its sub tabs.

 Public
 Demand
 ARD
 Regulation
 DRR

 Private
 DRR Default
 Portfolio
 Schedule
 Schedule
 Schedule
 Schedule
 Manager

 Private
 DRR Default
 Portfolio
 Schedule
 Schedule
 Schedule
 Schedule
 Manager
 Portfolio
 Download

9.1 DRR Offer Concepts

When submitting data, it is important to understand the hierarchy of the DRR offer data and the implications of data overrides. Figure 11 below shows the hierarchy of the DRR displays used to submit data, where the same offer data parameter appears in different displays, and the data from which display has precedence over (overrides) the others for the same parameter. [**Note**: only the data elements subject to overrides are shown.]

The DRR Default Parameters display is at the root of the hierarchy. The DRR record itself is initially defined by ISO New England with default offer data when the asset is instantiated in the Market system. Participants must update some data values, like the DRR's default dispatch availability status and ramp rate using the display. For DRRs capable of providing off-line reserves, the default Claim10 and Claim30 values must be updated on this display.

To participate in the energy market, a Participant must define at least one schedule for the DRR using the Schedule Manager display and all of the related data. The schedule defines a container for which detail offer data must be submitted.

Managing DRR schedules is accomplished by performing the following general steps:

- Create or modify a schedule using the Schedule Manager display
- Create or modify schedule-specific default data, which include MW operating limits, Startup time, Notification time, Min Reduction Time, Min Time Between Reductions, and market participation (day-ahead or real time) using the Schedule Detail Defaults display.
- Create or modify schedule-offer default data (Interruption Cost and Incremental Energy Offers [Price/MW pairs] using the Schedule Offers Default display.
- Identify the schedule or schedules to be used in the clearing of the Day-Ahead and Real Time (Balancing) Markets using the Schedule Selection display.





The other displays allow for overriding schedule based default data or a DRR's default dispatch availability status data. Offer data in other displays which are higher in the hierarchy are overridden by the presence of data in the hourly override displays:

- The limits on the DRR Hourly Updates display (Min Reduction and Max Reduction) override for specified hours of the day the corresponding limit values defined in the Schedule Detail Defaults display for all schedules.
- The dispatch availability status (Available, Unavailable) on the Hourly Updates display overrides for specified hours of the day the default dispatch availability status (Default Status) defined on the DRR Default Parameters display. For DRRs capable of providing off-line reserves, the display can be used to override for specific hours of the day the default Claim10 and Claim30 values defined on the DRR Default Parameters display.
- The Interruption cost and the incremental energy offers (Price/MW pairs) shown on the Schedule Offers Hourly Updates display override for specified hours of the day the default values defined in the Schedule Offers Default display for the corresponding schedule.
- The Startup and Notification times shown on the Schedule Times Hourly Updates display override for specified hours of the day the default values defined in the Schedule Detail Defaults display for the corresponding schedule.

9.2 DRR Portfolio Manager Display

A DRR portfolio is a collection of DRRs that is defined by each individual DRR user. The user assigns a unique name to each portfolio when it is created. Once a portfolio is created, the user assigns it one or more DRRs. A DRR can be assigned to more than one portfolio.

Data is filtered by DRR portfolios on the following displays:

- DRR Default Parameters
- Schedule Manager
- Schedule Detail Defaults
- Schedule Offers Default
- Schedule Offers Hourly Updates
- Schedule Times Hourly Updates
- Schedule Selection
- DRR Hourly Updates
- DRR by Portfolio

Note: A DRR that is not associated with a portfolio cannot be selected in the tooltray filters of these displays.

See §3.1 Manage Portfolios for more information on the purpose of portfolios and details on using this display.

9.3 DRR Default Parameters Display

The DRR Default Parameters display allows you to view DRR ownership information and to update a DRR's default ramp rate, dispatch status, and its offline reserve claim capability values. Offer information is defined on a DRR basis, as well as on schedule basis.

ISO New England requires that the Participant submit to the ISO, DRR specific information about a DRR. This information is used to create the initial DRR record in the market system database and must be submitted in writing to ISO-NE and verified by ISO-NE before the Participant submits offers. (For more information on the data required, see *ISO New England Manual for Market Operations Manual 11*, which is found on the *ISO New England website* among the documents listed on the Participate > Rules and Procedures > Manuals page.

When the DRR-specific information is received, the ISO creates the necessary records in the market system database. Some of this information is shown on the DRR Default Parameters display, and users cannot modify this information using the display. As shown below, only the fields with white background in the DRR Details section are modifiable.

		DR	R Market Status:	OPEN(01)				Market User Version: 5.0.0(201709181808	
ISO new england							ficate User:		11-Dec-2017 14:20:50
÷			Pa	rticipant:	ISO New En	gland Inc		Vser: Superuser	*
Public Generation Demand ARD Regulation I	ORR								
		Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Updates	DRR By Portfolio	XML Download		
DRR Daily Default Parameters									
Select Date and DRR	DRR Details for DRR 99999	(99999) on 12-	Dec-2017						
Date: 12-Dec-2017									
Portfolio: DRR_A	DR Resource ID:					Lead Pa	articipant ID:	99999	
DRR: Long Name	DRR Short Name:	Short Name				Lead Partic	ipant Name:	Participant Name	
Didit Long Name	DRR Long Name:	Long Name				DA Schedu	le Required:	No	
	Location:	Location							
	Ramp Rate (MW/Min):	1.0				De	fault Status:	 Available 	
	Ramp Rate Cap:	UD						🔘 Unavailable	
	Claim 10 Capable:	Yes							
	Claim 10 (MW):	1.0							
	Claim 10 Cap:	UD							
	Claim 30 Capable:	Yes							
	Claim 30 (MW):	2.0							
	Claim 30 Cap:	UD							
									Update Refresh
Messages									۲

The format of the DRR Default Parameters display is shown below:

The fields on this display are:

- **DR Resource ID** the ISO-NE generated unique identifier for the market asset. This is the preferred reference number used in doing business with the Customer Service Department.
- **DRR Short Name** the abbreviated name of the DRR.
- **DRR Long Name** the full name or description of the DRR. This label is shown on other DRR displays and in the DRR filter of all DRR displays.
- **Location** the pnode associated with the DRR.
- Ramp Rate (MW/Min) the number of MW per minute with which the DRR can respond to dispatch signals when it is on-line. This value is used in the calculation of spinning reserves unless it is overridden by an override for an hour defined on the DRR Hourly Updates display.
- Ramp Rate Cap the DRR's demonstrated ramp rate Capability value that was established during the last audit conducted by ISO New England. The offered value will be capped (constrained) to this value if it is defined. If the demonstrated capability value is undefined, it will be shown as "UD".
- **Claim 10 Capable** indicates if the DRR is registered as capable of producing Ten Minute Non-Spinning Reserve from an offline state.
- Claim 10 (MW) the number of MW with which a resource can respond within 10 minutes from an offline state. Used in the calculation of Ten Minute Non Spinning Reserves.
- **Claim 10 Cap** the DRR's demonstrated Claim 10 Capability value that was established during the last audit conducted by ISO New England. The offered value will be capped (constrained) to this value if it is defined. If the demonstrated capability value is undefined or the DRR is not Claim 10 Capable, it will be shown as "UD".
- **Claim 30 Capable** indicates if the DRR is registered as capable of producing Thirty Minute Reserve from an offline state
- Claim 30 (MW) the number of MW with which a resource can respond within 30 minutes from an offline state. Used in the calculation of Thirty Minute (non-spinning) Reserves.
- Claim 30 Cap the DRR's demonstrated Claim 30 Capability value that was established during the last audit conducted by ISO New England. The offered value will be capped (constrained) to this value if it is defined. If the demonstrated capability value is undefined or the DRR is not Claim 30 Capable, it will be shown as "UD".
- Lead Participant ID the ISO-NE assigned unique identifier of the Market Participant whose users are responsible for submitting the operating information of the DRR using the eMarket application.

- Lead Participant Name the full name of the Market Participant whose users are responsible for submitting the operating information of the DRR using the eMarket application.
- **DA Schedule Required** indicates if the DRR is classified as a capacity resource and therefore required to submit offers in the DA Market
- **Default Status (Available/Unavailable)** the default dispatch availability status of the DRR0 ≤ Ramp Rate
- A warning will be issued if the Ramp Rate exceeds Ramp Rate Cap when the demonstrated capability is defined (not "UD")
- A warning will be returned if the DRR is Claim 10 capable and any of the following conditions exist
 - Claim 10 = 0 or NULL
 - Claim 10 > Claim 10 Cap when the demonstrated capability is defined (not "UD") – meaning that the Claim 10 Cap value will be used instead of the offered value
- A warning will be returned if the DRR is Claim 30 capable and any of the following conditions exist
 - Claim 30 = 0 or NULL
 - Claim 30 > Claim 30 Cap when the demonstrated capability is defined (not "UD") – meaning that the Claim 30 Cap value will be used instead of the offered value
 - Claim 30 < Claim 10 meaning that the Claim 10 value will be used instead of the offered Claim 30 value.

9.3.1 View Specific DRR Information

- 1. In the DRR Navigation sub tabs, click the **DRR Default Parameters** tab.
 - The DRR Default Parameters display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired DRR from the **DRR** filter.
 - The default DRR data for the selected date and selected DRR is shown in the DRR Default Parameters display.

9.3.2 Modify DRR Default Claim 10/30 and Ramp Rate

A DRR's Claim 10/30 values and its default ramp rate data are specified at the DRR level. To define this data:

- 1. View the desired DRR's information as described in §9.3.1.
- 2. Enter all of the following DRR-specific information:

- a. Ramp Rate (MW/Min)
- b. Claim 10 (MW) if Claim 10 Capable
- c. Claim 30 (MW) if Claim 30 Capable

9.3.3 Modify DRR Default Status

- 1. View the desired DRR's information as described in §9.3.1.
- 2. Select one of the following radio buttons:
 - a. Available
 - b. Unavailable
- 3. Click the **Update** button to save the selection.

9.4 DRR Schedule Manager Display

Schedule Manager enables users to view, create, and update schedules. The format of this display is shown below.

ISC	new e	ngland						ĥ	DRR Market Statu	IS: CLOSED		Ce		Version: 5.0	et Use		rface
									P	articipant:	ISO New E			and the second second	Iser: Superus	er	~
Public Ge	eneration Der	mand ARI	D Regulatio	n DI	RR												
Private Messages	DRR Default Parameters	Portfolio Manager	Schedule Manager	Sched Detail	lule Defaults	Schedule Offers Default	Schedule Off Hourly Updat		Schedule Times Hourly Updates	Schedule Selection	Hourly Updates	DRR By Portfolio					
Sched	ule Man	ager															
Select DRR				~	Schedule	e Manager for DR	RR 60810 (60	810)									
Portfolio:	DRR A			~	Schedule	Name	D	escrij	ption			S	chedule Type				
	Long Name			~	DFLPRIC	E	D	FLPF	RICE			1	(Price)	~			
DRR.	Long reame				PRICE2		P	RICE	2			2	(Price)	~			
												New	r De	lete		Update	Refresh
Messages																	۵

The fields on this display are:

- **Schedule Name** a short name for the schedule, defined by the Participant. This label appears in the Schedule filter on the DRR displays.
- **Description** a brief description of the schedule, defined by the Participant. This label appears in the DRR by Portfolio display, which shows the DAM results for DRRs by generation portfolio.
- Schedule Type identifies the number and type of the schedule:
 - 0 94 price-based schedules

9.4.1 Create a New Schedule

Users can define more than one price-based schedule for a DRR. To assist in creating multiple schedules for DRR, the Schedule Manager display allows you

to provide a schedule name and description for each DRR schedule. To create a new schedule:

- 1. In the DRR Navigation sub tabs, click the **Schedule Manager** tab.
 - The Schedule Manager display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired portfolio from the **Portfolio** filter.
 - b. Select the desired DRR from the **DRR** filter.
 - Existing schedules for the selected DRR are listed in the Schedule Manager pane.
- 3. Click the **New** button.
- 4. Type the following information in the new row:
 - a. Schedule Name
 - b. Description
- 5. From the **Schedule Type** drop-down menu, select a schedule type that has not been used in other schedules for this DRR.
- 6. Click the **Update** button to save the data.

9.4.2 Modify a Schedule's Name and/or Description

After a schedule is created and saved to the database, you cannot modify the Schedule Type field. Only the schedule's name and description can be changed.

To modify an existing schedule's names:

- 1. In the DRR Navigation sub tabs, click the **Schedule Manager** tab.
 - The Schedule Manager display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired portfolio from the **Portfolio** filter.
 - b. Select the desired DRR from the **DRR** filter.
 - The existing schedules for the selected DRR are listed.
- 3. Modify either or both of the following in any of the listed schedules:
 - a. Schedule Name
 - b. Description
- 4. Click the **Update** button to save the data.

9.4.3 Delete a Schedule

- 1. In the DRR Navigation sub tabs, click the **Schedule Manager** tab.
 - The Schedule Manager display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired portfolio from the **Portfolio** filter.
- b. Select the desired DRR from the **DRR** filter.
 - The existing schedules for the selected DRR are listed.
- 3. Select a cell in the row of the schedule that you want to delete.
- 4. Click the **Delete** button. If the schedule is not flagged available in any of the market days for the period that market information is preserved in the market database, it is deleted; otherwise you will get an error message.

9.5 DRR Schedule Detail Defaults Display

The Schedule Detail Defaults display is used to view and manage a DRR's default market offer parameters. The format of the display is shown below.

ISO new england		DRR Market Status: OPEN(01)	Ver	arket User Interface rsion: 5.0.0(201709181808) e User: Superuser 11-Dec-2017 15:31:30
		Participant:	Participant Name	Vser: Superuser V
Public Generation Demand ARD Regulation D	RR			
Private DRR Default Portfolio Manager Schedule Detail	I Defaults Schedule Schedule Offers Offers Default Hourly Updates	Schedule Times Schedule Hourly Hourly Updates Selection Update		
Schedule Detail Defaults				
Select Date, DRR and Schedule	Schedule Details for DRR 60810 (60810) - 0	DFLPRICE (6081001 Price-based) on 1	12-Dec-2017	
Date: 12-Dec-2017 3 Portfolio: DRR_A DRR: Long Name Schedule: DFLPRICE	Schedule Available: Yes Day Ahead Market: Real Time Market: Fast Start (10 min): Yes Fast Start (30 min): Yes	Max Daily Star Max Daily Energy (for D Min Time Between Redu Min Reductior DR Resource Notification DR Resource Startup	A in MWh): uctions (hr): 1.0 n Time (hr): 1.0 n Time (hr): 0.01	Min Reduction: 1.0 Max Reduction: 40.0
Messages				Update Refresh

The fields on this display are:

- Schedule Available indicates whether the schedule is to be used. This is managed on the Schedule Selection display.
- **Day Ahead Market** this check box indicates whether the schedule is enabled for Day Ahead Market use.
- **Real Time Market** this check box indicates whether the schedule is enabled for Real Time Market use.
- Fast Start (10 min) if the DRR is Claim 10 capable²⁸, this indicates whether the entered parameters qualify the DRR for 10-min fast startup operation and therefore make it eligible for off-line non-spinning reserves.
- Fast Start (30 min) if the DRR is Claim 30 capable²⁸, this indicates whether the entered parameters qualify the DRR for 30-min fast startup operation and therefore make it eligible for off-line non-spinning reserves.
- Max Daily Starts the maximum number of times that the DRR can be started in one operating day under **normal** operating conditions. This field is only used for the Day Ahead Market. A blank field means the value is unlimited.
- **Max Daily Energy** the maximum amount of energy that the DRR can reduce in one operating day. A blank field means the value is unlimited.

²⁸ The indication of whether the DRR is Claim 10 or Claim 30 Capable can be seen on the DRR Default Parameters display (§9.3)

- Min Time Between Reductions the minimum time (in hours) that the DRR owner requires between the time the DRR is released for shutdown by the ISO and the time the DRR is scheduled to be released for dispatch to the ISO.
- Min Reduction Time the minimum time (in hours) that the DRR must remain online after it has been scheduled to reach its Minimum Reduction Limit before it can be released for shutdown from its Minimum Reduction Limit.
- **DR Resource Notification Time** the notification time (in hours) needed for the DRR to begin reducing; the time is measured from the time the startup dispatch instruction is received from the ISO.
- **DR Resource Startup Time** the time (in hours) for the DRR to startup after the notification time elapses.
- **Min Reduction** the lowest MW level at which the DRR can be economically dispatched.
- **Max Reduction** the highest MW level at which the DRR can be economically dispatched.

9.5.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- Data is entered for each of the following values, i.e. non-blank values appear in the data entry cells:
 - Min Time Between Reductions and Min Reduction Time
 - Startup and Notification times
 - Min Reduction and Max Reduction

Only Max Daily Starts and Max Daily Energy can be blank (unspecified).

- Min Reduction > 0 if Max Reduction > 0
- Min Reduction ≤ Max Reduction
- Startup Time ≥ 0
- Notification Time ≥ 0
- Min Time Between Reductions ≥ 0
- Min Time Between Reduction ≤ number of consecutive unavailable hours in Hourly Updates if hourly updates exist for the schedule market day
- $24 \ge Min Reduction Time \ge 0$
- Max Daily Energy must be blank to specify an unlimited capacity or a value greater than zero if you want to the Day Ahead Market to limit the energy allocated. If a non-blank value is entered, then Max Daily Energy should be greater than or equal to Min Reduction x Min Reduction Time; a warning message will be reported if it is not.

- For Claim 10 capable DRRs, a warning will be returned when the sum of notification and startup time exceeds the Claim 10 criteria, i.e. ≥ 10min because time values make the DRR ineligible for off-line reserve capability.
- For Claim 30 capable DRRs, a warning will be returned when the sum of notification and startup time exceeds the Claim 30 criteria, i.e. ≥ 30min because time values make the DRR ineligible for off-line reserve capability.
- For Claim 10 capable DRRs and Claim 30 capable DRRs, a warning will be returned when Min Reduction Time > 1 or Min Time Between Reductions > 1 because time values greater than 1 hour make the DRR ineligible for off-line reserve capability.

9.5.2 View Schedule Detail Defaults

To view the default values of a DRR schedule:

- 1. In the DRR Navigation sub tabs, click the **Schedule Detail Defaults** tab.
 - The Schedule Detail Defaults display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired DRR from the **DRR** filter.
 - d. Select the desired schedule from the **Schedule** filter.
 - The display shows the selected schedule's default detail information.

9.5.3 Modify Schedule Constraint Data

Constraint data includes Minimum Time Between Reductions, Minimum Reduction Time, Maximum Daily Starts and Maximum Daily Energy. To modify schedule constraints:

- 1. View the DRR's schedule details (§9.5.2).
- 2. Enter the desired changes for any or all of the following offer parameters:
 - a. Minimum Time Between Reduction (hours)
 - b. Minimum Reduction Time (hours)
 - c. Maximum Daily Starts
 - d. Maximum Daily Energy (MWh)
- 3. Click the **Update** button to save the data.

9.5.4 Modify Schedule Operating Limits

- 1. View the DRR's schedule details (§9.5.2).
- 2. Enter the desired changes for any or all of the following operating limits:
 - a. Min Reduction (MW)
 - b. Max Reduction (MW)

3. Click the **Update** button to save the data.

9.6 DRR Schedule Offers Daily Default Display

The purpose of this display is to provide viewing and updating capabilities of the DRR daily default interruption cost and incremental energy offers. Incremental energy offers consist of price/MW pairs, also referred to as schedule offer segments. Schedule offer segments specify a MW amount and the price per MWh for reduction from the previous segment's MW amount up to this segment's MW amount. Up to 10 segments can be defined for each schedule.

Offers in MW are cumulative. For example,

- To offer the first block of 15 MW of reduction at \$10.00, you enter an offer segment of 15 MW @ \$10.00
- To offer the next block of 20 MW of reduction at \$30.00, you enter an offer segment of 35 MW @ \$30.00

And so on. You can enter the segment offers in any order of MW values; eMarket will organize the segments and display them in increasing MW value order. The price value of each segment must however be greater than the price of the previous segment after the segments are sorted in increasing MW value order. In other words, the offer segments must be monotonically increasing in MW and price value.

The slope from zero to the first MW point (which can be zero) is equal to zero; that is to say that the price/MWh is fixed and equal to the price of the first segment.

The last MW point should be greater than or equal to the Maximum Reduction Limit. If it is not, the offer curve is extended using the last segment.

This was illustrated in Figure 2. Only block interpolation of offer values applies to DRRs as, contrary to offers for Units, Use Offer Slope cannot be set to True for DRRs.

DRRs are eligible to change segment offer Price/MW pairs and Interruption Cost offers after the close of the DA market during the daily Re-Offer period.

The format of the Schedule Offers Default display is shown below.

ISC	new england					DRR Marke	Status: CL	OSED		Certi		/ersion: 5.0.0(201709181808)	Interface
							Particip	oant: IS	50 New En				superuser	
ublic Gen	neration Demand ARD	Regulation	ORR											
	DRR Default Parameters Manager	Schedule Sched Manager Detail	lule Defaults	Schedule Offers Default	Schedule Hourly Up			thedule	Hourly Updates	DRR By Portfolio	XML Download			
Schedu	ule Offers Daily	Default												
elect Date, I	DRR and Schedule	«	Schedu	le Offer for DRR 6	0810 (608	10) - DFLPRICE (081001 Pri	ce-based	d) on 12-D	ec-2017				
Date:	12-Dec-2017	•												
Portfolio:	DRR_A	~			DRTP (\$):	16.00								
DRR:	Long Name	~		Interruption										
Schedule:	DFLPRICE	~			MW	Price								
					1.0	17.00								
					10.0	21.00								
					40.0	34.00								
														Refrest
														Reffest

The updateable fields on this display are:

- Interruption Cost the interruption cost of the DRR.
- **MW** the MW DRR output associated with this segment. The last MW point should be greater than or equal to the Maximum Reduction MW.
- **Price** the \$/MWh price offered for this segment.

9.6.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- A non-blank value is entered for the Interruption Cost.
- Interruption Cost ≥ 0
- At least one non blank Price/MW pair value is entered
- MW quantity must be strictly positive or zero; i.e. $MW \ge 0$
- Price of Incremental Energy Offers must be less than or equal to the Energy Offer Cap price and greater than or equal to the Demand Reduction Threshold Price (DRTP); i.e.
 DRTP ≤ Price ≤ Energy Offer Cap price²⁹.

²⁹ Energy Offer Cap price is defined in Market Rule 1, section I.2.2. DRTP is defined in Market Rule 1, appendix E, section 3.1.

• Offers must be monotonically increasing. The price value for each segment must be greater than the one of the previous segment after the segments are ordered by increasing MW value³⁰.

Note: If for a DRR there are no price/MW pair values specified in the Schedule Offers Default display, and no hourly overrides of price/MW have been specified in Schedule Offers Hourly Updates, then the DRTP value will be used in the clearing process as the default price for this DRR.

9.6.2 Modify Schedule Offers

- 1. In the DRR Navigation sub tabs, click the **Schedule Offers Default** tab.
 - The Schedule Offers Daily Default display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired DRR from the **DRR** filter.
 - d. Select the desired schedule from the **Schedule** filter.
 - The display reflects the selected schedule's offer data for the selected date.
- 3. Enter or modify the values on the display. To delete an incremental energy offer segment, simply blank each value on the row where the segment is shown.
- 4. Click the **Update** button to save the data.

³⁰ The same price for successive segments is allowed; however it doesn't really make sense when considering Participant offers because the two segments are equivalent to a single segment with a MW value equal to that of the second segment.

9.7 DRR Schedule Offers Hourly Updates Display

This display provides capability for viewing, entering and modifying hourly updates to a DRR schedule's default offers for interruption cost and incremental energy price/MW pairs for a future market day. Schedule offers cannot be modified for the current market day. Refer to §9.5.3 Schedule Offers Daily Default Display for a description of these schedule offer parameters and how incremental energy offers must be structured. There are multiple tabs on this display:

- The "DEF" tab shows the default offers for the selected market day. These were entered and are shown on the Schedule Offers Default display. They cannot be changed using the Schedule Offers Hourly Updates display.
- There is one tab for each hour of the market day. Each tab is labeled with the hour-ending value of the market hour. There is an extra tab labeled "02*" for the extra hour on the Fall Daylight Saving Time transition day. On the Spring Daylight Saving Time transition day, the tab labeled "03" will not be shown.

ISO	new england							D	RR Marke	t Status: Of	PEN(01)		(Ve	lark rsion: 5. e User:	0.0(20	17091			erf		
										Parti	cipant:	Participant	Name				¥ Us	er: S	uperus	er		~
Public Genera	tion Demand ARD	Regulation	DRR																			
		hedule Scheo nager Detai	lule Defaults	Schedule Offers De		Schedu Hourly	e Offers Jpdates	Schedul Hourly	e Times Jpdates	Schedule Selection	Hourly Updates	DRR By Portfolia		L wnload								
Schedule	e Offers Hourly	Updates																				
Select Date, DR	R and Schedule	«	Schedu	le Offer fo	r DRR	10 (10) - [OFLPRICE (100	Price-base	d) on 12-	Dec-2017										
Date:	12-Dec-2017		DEF	01 02	03	04	05 06	07	08 09	10 11	12	13 14	15	16 1	7 18	19	20	21	22	23	24	
Portfolio:	DRR_A	~																				
DRR:	Long Name	~				DRTP (): 16.0	0														
Schedule:	DFLPRICE	*		Int	erruptio	on Cost (\$): 0.00															
						MW		Price														
						0.0		16.0	D													~
						10.0		21.0	D													
Copy From:		~				40.0		400.0	D													
Copy To:	✓ Through:	*																				
		Сору																				
																						\sim
																			Update		Refres	
Messages																						

The format of the Schedule Offers Hourly Updates display is shown below.

The horizontal bar above the HE label indicates that hourly updates have been submitted and saved for the hour (HE 01 - 06 and 11 - 24 in the example above).

The Copy From/Copy To/Through selections allow you to copy hourly offers from Default or from a selected hour to another hour or to a range of hours. It is designed to minimize the amount of data entry that you are required to perform, especially when there are blocks of hours where the override values are to be the same.

For instance, to copy hourly offers from hour 06 to hours 09 - 10, the following data would be selected in the drop-down lists:

ISO new england				DRR Mark	et Status: OPI	EN(01)		Ve	rsion: 5.0	et Us 0.0(2017091: Superuser			erfac	
					Partic	ipant: P	articipant N	ame		✓ Us	er:	Superuser		~
Public Generation Demand ARD Regu	ation D	DRR												
Private DRR Default Portfolio Schedule Messages Parameters Manager Manager		ule Schedule Defaults Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Updates	DRR By Portfolio	XML Download						
Schedule Offers Hourly Up	dates													
Select Date, DRR and Schedule	~	Schedule Offer for DRR	10 (10) - D	FLPRICE (100	1 Price-based) on 12-D	ec-2017							
Date: 12-Dec-2017		DEF 01 02 03	04 05 06	07 08 09	10 11	12	13 14	15 16	17 18	19 20	21	22	23 24	
Portfolio: DRR_A	~													
DRR: Long Name	~		DRTP (\$): 16.00)										
Schedule: DFLPRICE	~	Interrupt	tion Cost (\$): 0.00											
			MW	Price										
			10.0	19.00										~
			25.0	27.00										
Copy From: 06	~		40.0	45.00										
Copy To: 09 Y Through: 10	~													
	Copy													
														~
								Dele	te			Update	Refr	resh
Messages														۲

When the **Copy** button is clicked, the data are copied to the selected target hours and the displays shows updates for these hours as unsaved data, i.e. their corresponding tabs displayed in red:

ISO	new england						DRR	Marke	t Status: C			(v		5.0.0		81808	11-De	c-2017 1	16:51:	
ħ.,									Par	icipant:	Participant	Name				v ∪	ser: S	uperuser	_		~
Public Genera	ation Demand ARD	Regulation D	RR																		
		nager Schedu		chedule)ffers Default	Schedule Hourly Up		Schedule 1 Hourly Up		Schedule Selection	Hourly Updates	DRR By Portfolio		L wnload								
Schedule	e Offers Hourly	Updates																			
Select Date, DR	R and Schedule	**	Schedule (Offer for DRR	10 (10) - DF	FLPRICE (1001	Price-bas	ed) on 12-0	ec-2017										
Date:	12-Dec-2017		DEF 01	ī <u>02</u> <u>03</u>	04 03	5 06	07 08	09	10 1	1 12	13 14	15	16	17 1	8	19 20	21	22	23 24		
Portfolio:	DRR_A	~																			
DRR:	Long Name	~			DRTP (\$)	: 16.00															
Schedule:	DFLPRICE	~		Interrupti	on Cost (\$)																
					MW		Price														
					10.0		19.00														
					25.0		27.00														^
Copy From:	06	~			40.0		45.00														
Copy To:	✓ Through:	~																			
		Сору																			
																					~
												(Dele	te				Update	R	efresh	
Messages													C an					- pouro		- needl	

At this point, you can click the **Update** button to save the changes, or you can navigate to the modified tabs and make additional changes before clicking **Update**.

When entering new overrides, you can minimize the amount of data entry by first copying the schedule's default values to the first hour of the range that you want to override and then edit the values for that hour.

ISO	new england					DRR Market S	tatus: CLOSED Ver	rsion: 5.0.0(201709181812)/ Ser	larket User ver: QAEMKT2/ Database: ite User: Superuser	
1	±••*						Participant:	Verso Energy Services LL	.C 💙 User	Superuser 👻
Public	Generation	Demand	ARD	Regulation	DRR					
Private Messages	DRR Default Parameters	Portfolio Manager	Schedule Manager	Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Hourly Selection Updates	DRR By Portfolio	XML Download
Schedu	le Offers Hourl	y Updates						u U		
Select Date, I	RR and Schedule	«	Schedule Offer for Verso	Androscoggin G1 I	nc DG (9338393) - DFL	PRICE (933839301 P	rice-based) on 10-(Oct-2017		
Date:	12-Dec-2017		DEF 01 02 03	04 05 06	07 08 09 10	11 12 13	14 15 16	17 18 19 20 21 22	23 24	
Portfolio:	DRR_A	*								
DRR:	Long Name	~		DRTP (\$): 35.00						
Schedule:	DFLPRICE	*	Interruptio	n Cost (\$): 222.00	1					
				MW	Price					
				999.0	25.00					
	n: Default	*								
Copy To	o: Through	•								
		Сору								
										Refresh

Furthermore, if there are updates for an hour (whether saved in the database or entered on the screen) that can be changed for the selected market day, a **Delete** button is provided on the display. Selecting the button sets all the data entry fields to blank without saving the entered data. You must click the **Update** button to actually delete the hourly updates and save the change. That way, you can delete the updates to multiple hours with one **Update** selection. For instance, the deletion of hourly updates for hours 11 - 13 can be achieved by clicking the **Delete** button on each corresponding tab:

ISO	new england				DRR Mark	et Status: O	PEN(01)			Version:	ket U 5.0.0(20170 : Superuser			face
1						Part	icipant: P	articipant N	ame		~	User: Su	iperuser	*
Public Generat	tion Demand ARD Regula	tion DF	RR											
	R Default Portfolio Schedule ameters Manager Manager	Schedul Detail D		Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Updates	DRR By Portfolio	XML Download					
		otoo												
	e Offers Hourly Upd													
Select Date, DRF	R and Schedule	~	Schedule Offer for DR			1 Price-base								_
Date:	12-Dec-2017		DEF 01 02	03 04 05 06	07 08 09	10 11	i <u>12</u>	13 14	15 16	17 1	8 19 2	0 21	22 23	24
Portfolio:	DRR_A	¥												
DRR:	Long Name	v		DRTP (\$):										
Schedule:	-	v	Intern	ption Cost (\$):										
			Interro											
				MW	Price									
														^
Copy From:	Defects	~												
Copy To:	✓ Through:	~												
	Co	ру												
														~
									De	elete			pdate	Refresh
Messages														

At this point, the changes are made only in the display. Use the **Update** button to save changes.

After the changes have been saved using the **Update** button, tabs for hours 11 - 13 show that there are no hourly updates for these hours:

ISO	new england						DRI	R Marke	t Status: Of	PEN(01)			v	lark ersion: 5. te User: 1	0.0(20	170918			erf		
									Parti	cipant:	Participan	t Name				Y Use	r: S	uperus	r		~
Public Genera	ation Demand ARD Regulat	ion D	RR																		
	R Default Portfolio Schedule rameters Manager Manager	Sched	ule Defaults	Schedule Offers Default		dule Offers ly Updates			Schedule Selection	Hourly Updates	DRR B Portfol		4L ownload								
Sabadula	e Offers Hourly Upd	otoo			_																
Select Date, DR	R and Schedule	~		e Offer for DR			DFLPRICE (Price-base	- XC XC											
Date:	12-Dec-2017	•	DEF	01 02 0	03 04	05 06	5 07 08	09	10 11	12	13 14	15	16	17 18	19	20	21	22	23	24	
Portfolio:	DRR_A	~																			
DRR:	Long Name	~			DRTF	(\$): 16.0	0														
Schedule:	DFLPRICE	~		Interru	ption Cost		-														
					MW		Price														
																					^
Copy From:	Default	~				1															
Copy To:	✓ Through:	~																			
	Co	ov																			
																					~
																		Update		Refre	sh
Messages																					8

The **Copy** functionality may also be used to remove hourly updates for a range of hours. In this case, the "Copy From" hour will be an hour with no hourly updates, and "Copy To" hours will be the hours for which hourly updates need to be removed.

9.7.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks for each market hour where data has been entered to ensure that the entered data is valid:

- At least one non-blank Price/MW pair value is entered except when you are deleting all overrides for the selected market hour.
- Interruption Cost ≥ 0
- MW quantity must be strictly positive or zero; i.e. $MW \ge 0$
- Price of Incremental Energy Offers must be less than or equal to the Energy Offer Cap price and greater than or equal to the Demand Reduction Threshold Price (DRTP); i.e.
 DRTP ≤ Price ≤ Energy Offer Cap price³¹.Incremental Energy Offers must be monotonically increasing. The price value for each segment must be greater than or equal to the one of the previous segment after the segments are ordered by increasing MW value³².
- The market hour for which updates are being submitted must be a future day hour. In other words, the selected market day date must be tomorrow or a later date.

Note: If for a DRR there are no price/MW pair values specified in the Schedule Offers Default display, and no hourly overrides of price/MW have been specified in Schedule Offers Hourly Updates, then the DRTP value will be used in the clearing process as the default price for this DRR.

9.7.2 Override Schedule Offers for One or More Hours of a Market Day

- 1. In the DRR Navigation sub tabs, click the **Schedule Offers Hourly Updates** tab.
 - The "DEF" Schedule Offers Hourly Updates display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired DRR from the **DRR** filter.
 - d. Select the desired schedule from the **Schedule** filter.

³¹ Energy Offer Cap price is defined in Market Rule 1, section I.2.2. DRTP is defined in Market Rule 1, appendix E, section 3.1.

³² The same price for successive segments is allowed; however it doesn't really make sense when considering Participant offers because the two segments are equivalent to a single segment with a MW value equal to that of the second segment.

- The "DEF" display tab reflects the selected schedule's default offer data for the selected date.
- 3. To update the overrides of individual hours:
 - a. Select the display tab of the hour for which you want to enter new hourly overrides of the schedule offers or modify any existing hourly overrides.
 - b. Enter the override value for each of the schedule offers or modify the existing values.
 - c. Repeat steps a and b for each hour that you want to enter new overrides or modify the existing ones.
 - d. Click the **Update** button to save the data.
- 4. To update the overrides for a range of hours use the "Copy From" tool in the tooltray.
 - a. First, create the offer hourly updates for one hour.
 - b. Then copy this hour's values to the various ranges (one or more) of hours where you want this set of overrides to apply. Optionally, instead of creating data for the "Copy From" hour, you can select "Default" to copy daily offers to the selected ranges of hours.
 - c. You can repeat this process (steps a and b) for other ranges of hours where you want to enter different sets of overrides.
 - d. If necessary, modify the offer hourly updates for selected hours.
 - e. When you have finished entering/modifying the hourly overrides to the schedule offer, click the **Update** button to save the data.

9.7.3 Remove/Delete Hourly Updates to Schedule Offers

- 1. In the DRR Navigation sub tabs, click the **Schedule Offers Hourly Updates** tab.
 - The "DEF" Schedule Offers Hourly Updates display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired DRR from the **DRR** filter.
 - d. Select the desired schedule from the **Schedule** filter.
 - The "DEF" display tab reflects the selected schedule's default offer data for the selected date.
 - A horizontal bar appears above the hour-ending value of the display tabs where there are overrides/updates in place.

- 3. To remove/delete the overrides of individual hours:
 - a. Select the display tab of the hour for which you want to remove the hourly overrides. If you are allowed to remove the overrides for the selected hour, a **Delete** button will be shown in the lower display toolbar.
 - b. Click on the **Delete** button. All of the data entry fields on the display are set to blank.
 - c. Repeat steps a and b for each hour that you want to remove the overrides.
 - d. Click the **Update** button to save the changes.
- 4. To remove/delete the overrides for a range of hours use the "Copy From" tool in the tooltray.
 - a. Pick the display tab of the first hour in a range of hours where you want to delete the overrides; click the **Delete** button to set all data entry fields to blank. You don't need to save ("Update") this data entry.
 - b. Copy the selected hour's values to the various ranges (one or more) of hours where you want to remove the overrides. *Note*: if there is an hour with no overrides in the selected market day, you can skip step a and copy this hour to the various ranges (one or more) of hours where you want to remove the overrides.
 - c. Repeat this process (steps a and b) for other ranges of hours where you want to also remove the overrides.
 - d. When you have finished, click the **Update** button to save the changes.

9.8 DRR Schedule Times Hourly Updates Display

This display provides capability for viewing, entering, and modifying hourly updates to a DRR schedule default Notification and Startup times. Refer to §9.5 Schedule Detail Defaults Display for a description of the schedule default daily parameters and the impact of these parameters on the DRR's eligibility for offline reserve designations.

In the upper section, the display shows the default daily values:

- Notification and Startup Times
- The DRR's default eligibility to be treated as a 10-minute and 30-minute Fast Start resource; i.e. its eligibility to receive 10-minute and 30-minute off-line (Non-Spin) reserve designations
- The DRR's default Minimum Reduction Time (MRT) and Minimum Time Between Reductions (MTBR) that were entered on the Schedule Detail Defaults display

In the lower section, called Override Hourly Values, the display shows:

- Hourly override values for Notification and Startup Times
- The DRR's hourly eligibility to be treated as a 10-minute and 30-minute Fast Start resource; i.e. its eligibility to receive 10-minute and 30-minute off-line (Non-Spin) reserve designations

To be treated as a 10-minute resource, the DRR must be designated as Claim10 capable. Similarly, to be treated as a 30-minute resource, the DRR must be designated as Claim 30 capable. These two capability flags are asset registration parameters and they are shown on the DRR Default Parameters display (§9.3). In addition, each eligibility is a function of the DRR's Notification Time + Startup Time along with its Minimum Reduction Time and Minimum Time Between Reductions. That is why the information is shown both at the schedule default level and for each hour of the operating day where the hourly time values override the default values and can be different for each hour of the operating day. Thus a DRR can be eligible based on the schedule's default offer parameters but not eligible for a particular hour of the operating day because of the hourly overrides of either Notification or Startup Times.

The format of the display is shown below.

ISC	new e	ngland						DRR Market	Status: OPEN(01)				1(201711162319	Interfa)) 21-Dec-2017	
									Participant:	Participant N	lame		✓ User:	Superuser	
ublic Ge	eneration De	mand ARI	D Regulat	tion D	RR										
vate issages	DRR Default Parameters	Portfolio Manager	Schedule Manager	Sched Detail	ule Defaults	Schedule Offers Default	Schedule Offers Hourly Updates			Hourly Updates	DRR By Portfolio	XML Download			
ched	ule Start	up and	Notific	atior	n Tim	e Hourly	Updates								
	, DRR and Sch			~											
Date	22-Dec-2017				Default	Values for DRI	R 10 (10)	- DFLPRICE (1001 Price-based	i) on 22-Dec	-2017				
				-			Notification	Startup	Fast Start (10 min)	Fast Start (3	30 min)	MRT	MTBR		
	DRR_A						0.4	0.01	No		Yes	0.1	1		
DRR:	Long Name			*											
Schedule:	DFLPRICE			~	0		- (000 10	((0010) DEL				2017			
					Overnic	le Hourly Value	Notification	(60810) - DFLI Startup	Fast Start (10 min)	ce-based) or Fast Start (3		2017			
					-	✓ Сору	recation	oranop	Past Start (10 mm)	Past Start (50 mm)				
						01 Copy	0.5	0.5	No		No				
						02	0.5	0.5	No		No				
						03	0.1	0.3	No		Yes				
						04	0.1	0.05	Yes		Yes				
						05									
						08									
						07									
														Update F	Refres

The Default Values section is provided as a reference to show the default schedule offers for the chosen day which are being overridden by the hourly update values entered in the lower section. Only the Notification Time and Startup Time in the Hourly Values section are enterable. The other information shown in this section is derived from the entered values as explained above, and is for reference only.

9.8.1 Data Validation Rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

- Data is entered for each of the parameter values for an hour, i.e. non-blank values appear in all the data entry cells, if any one of the cells contains a nonblank value. In other words, when you want to override one time value of an hour, you must enter values for all the other time parameters even if they are the same as the default values. When all data entry cells for an hour are set to blank, it means that the override for the hour is to be removed/deleted.
- Startup Time ≥ 0
- Notification Time ≥ 0

9.8.2 Override/Remove Startup and Notification Times for One or More Hours of a Market Day

1. In the DRR Navigation sub tabs, click the **Schedule Times Hourly Updates** tab.

- The Schedule Startup and Notification Time Hourly Updates display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the Date filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired DRR from the **DRR** filter.
 - d. Select the desired schedule from the **Schedule** filter.
 - The display reflects the hourly overrides for the selected DRR schedule.
- 3. To update the overrides for individual hours:
 - a. Enter override values for each time parameter in the hours that you wish. Enter blanks if you want to remove an override. Remember that you must enter values for all time parameters.
 - b. Click the **Update** button to save the data.
- 4. To update the overrides for a range of hours using the first row in the grid:
 - a. Enter override values for each time parameter in the top row. Enter blanks in each field of the top row if you want to remove overrides.
 - b. Specify the range of hours to override by selecting the "From" (left dropdown selector) and "To" (right drop-down selector) hours in the top row



- c. Click the **Copy** button; this will copy the data that you entered in the top row to each row for the hours in the range that you selected.
- d. Repeat steps a through c for other range of hours as desired.
- e. Click the **Update** button to save the changes.

9.9 DRR Schedule Selection Display

This display is used to make schedules available for a given day and market. The format of the display is shown below.

Demand ARD Regulation DRR vivate PRR Default Portfolio Schedule Schedule Offers Default Hourly Updates Schedule Hourly Updates Variantes Hourly Updates Participant Participant DRR By XML	and the second	21-Dec-2017 15 Superuser
tvate DRR Default Portfolio Schedule Schedule Schedule Schedule Offers Schedule Times Schedule Hourly DRR By XML		
Schedule Selection		
lect Date and Portfolio CRR_B on 22-Dec-2017		
Date: 22-Dec-2017 OR Schedule ID Type Schedule Day Ahr	ead Real Time	Available
DRR 10 1001 Price DFLPRICE X	х	
Portfolio: DRR_B V DRR 10 1002 Price PRICE2 X		
DRR 10 1004 Price PRICE4	×	
DRR 11 1101 Price DFLPRICE X	х	
DRR 22 2201 Price DFLPRICE X	x	

The fields on this display are:

- **DRR** the name of the DRR. All DRRs listed in the selected portfolio along with each of their schedules are shown on this display.
- **Schedule ID** the unique identifier of the schedule; it is the concatenation of the unique DRR identifier and the schedule type number.
- **Type** the type of schedule: Price
- Schedule the name of the schedule as it was entered on Schedule Manager display
- **Day Ahead** an X-mark indicates that the schedule is enabled to be used in the clearing of the Day Ahead market if the schedule is flagged as available.
- **Real Time** an X-mark indicates that the schedule is enabled to be used in the clearing of the Real Time market if the schedule is flagged as available.
- **Available** a checkmark indicates that the schedule is to be use in the clearing of the selected market on the selected operating day.

9.9.1 Select Schedule for an Operating Day and Market

The DRR Schedule Selection operation is for modifying the availability of a schedule for a specific day.

Note: One and only one price-based schedule that is enabled for the Real-Time market must be available for any given market day.

To change the availability of schedules:

- 1. In the DRR Navigation sub tabs, click the **Schedule Selection** tab.
 - The Schedule Selection display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - The schedules for all DRRs in the chosen portfolio are displayed.
- 3. Toggle the **Available** checkboxes as desired.
- 4. Click the **Update** button to save your modifications.

Note: To change the market to which a schedule applies, you must use the Schedule Detail Defaults display.

9.10 DRR Hourly Updates Display

This display is used to view and modify the hourly overrides of a DRR's MW limits, default dispatch availability status and its off-line reserve capabilities for a specified market day. Changes made to the DRR data using the DRR Hourly Updates display supersede all schedule limits (Schedule Detail Defaults display) and default DRR parameters (DRR Default Parameters Display). When the market is open for the next day, all the rows are updateable. During the Intraday Re-Offer, only the current and future hours are available for updates. *This is the only DRR display enabled for submittal of hourly updates during the RT Intraday Re-Offer period*.

ISO new england						D	RR Market Sta	tus: OPEN(01)		Ň	Marke Version: 5.0.1(cate User: Supe	2017111623	r Interfa ³¹⁹⁾ 22-Dec-2017	
								Participant:	Participant l	Name		Vser:	Superuser	
olic Generation Demand AR	D Regulation	DRR												
rate DRR Default Portfolio ssages Parameters Manager	Schedule Sche Manager Deta				edule Times Sche urly Updates Sele	ction Hourly		XML Download						
RR Hourly Updates														
ect Date and DRR	~													
Date: 23-Dec-2017	3	Default Val	lues for DRR 10	(10) on 23-l	Dec-2017									
Portfolio: DRR B	~	Schedule	Min Red	Max Red Ty			Available		Claim10	Claim30	Ramp Rate	Default	Status: 🔿 Unavailab	
and the second	*	PRICE4	0.1	20.0 Pri		×	V	Capable	Yes	Yes 27				
DRR: DRR 10	*	PRICE2	0.1	45.0 Pri	ce X		V	Default Cap	11	45	5		 Available 	
		Override Ho	ourly Values for DR	R 10 (10) on 23-Dec-2017									
		Override Ho	ourly Values for DR) on 23-Dec-2017 Claim10	Claim30	Ramp Rate	Available						
		Hour				Claim30	Ramp Rate	Available						
		Hour	Min Re ❤ Copy		Claim10	Claim30 38.0	Ramp Rate							
		Hour	Min Re Copy	d Max Red	Claim10									
		Hour V 01	Copy	d Max Red	Claim10 20.0 20.0	38.0	5.0							
		Hour V 01 02	V Copy	d Max Red 1.0 38.0 1.0 38.0	Claim10 20.0 20.0 20.0	38.0 38.0	5.0							
		Hour V 01 02 03	Min Re	Max Red 1.0 38.0 1.0 38.0 1.0 38.0	Claim10 20.0 20.0 20.0 20.0 20.0	38.0 38.0 38.0	5.0 5.0 5.0							
		Hour 01 02 03 04	V Copy	Max Red 1.0 38.0 1.0 38.0 1.0 38.0 1.0 38.0	Claim10 20.0 20.0 20.0 20.0 20.0 20.0 20.0	38.0 38.0 38.0 38.0	5.0 6.0 5.0 5.0							
		Hour 01 02 03 04 05	V Copy	Max Red 1.0 38.0 1.0 38.0 1.0 38.0 1.0 38.0 1.0 38.0 1.0 38.0	Claim10 20.0 20.0 20.0 20.0 20.0 20.0 20.0 20	38.0 38.0 38.0 38.0 38.0	5.0 5.0 5.0 5.0 5.0							
		Hour 01 02 03 04 05 06	V Copy	Max Red 1.0 38.0 1.0 38.0 1.0 38.0 1.0 38.0 1.0 38.0 1.0 38.0 1.0 38.0	Claim10 20.0 20.0 20.0 20.0 20.0 20.0 20.0 20	28.0 38.0 38.0 38.0 38.0 38.0 38.0	5.0 5.0 5.0 5.0 5.0 5.0							
		Hour 01 02 03 04 05 08 07	V Copy.	Max Red 1.0 38.0 1.0 38.0 1.0 38.0 1.0 38.0 1.0 38.0 1.0 38.0 5.0 45.0	Claim10 20.0 20.0 20.0 20.0 20.0 20.0 20.0 11.0 11	38.0 38.0 38.0 38.0 38.0 38.0 38.0 38.0	5.0 5.0 5.0 5.0 5.0 5.0 6.0							

The format of this display is shown below.

The Default Values section is provided as a reference to show the limit values in the available schedules and the default parameters for the selected market day. These values may be overridden in the Hourly Overrides section. The Default Values section contains the following information:

- Schedule The name of the schedule
- Min Red The Minimum Reduction limit specified in the schedule detail
- Max Red The Maximum Reduction limit specified in the schedule detail
- **Type** The schedule type: Price
- Day Ahead/Real Time The market type
- Claim 10 and Claim 30 Capable Indicate whether the DRR is registered as capable (Yes/No) of providing 10 and 30 minute non-spinning reserves from an offline state.

- Claim 10 and Claim 30 Default The daily default off-line 10 and 30 minute off-line reserve capability for the market day as submitted using the DRR Default Parameters display.
- Claim 10 and Claim 30 Cap The demonstrated claim capabilities that were established during the last audit conducted by ISO New England.
- **Default Status (Available/Unavailable)** The dispatch availability status (Default Status) specified on the DRR Default Parameters display.

The Default Values section may be collapsed using the double up-arrow icon at the extreme right of the section title bar in order to show more rows in the hourly section. The collapsed section can be expanded again to see the default values using the double down-arrow icon at the extreme right of the collapsed section title bar.

ISO new england							D	RR Market Stat	tus: OPEN(01)	eMarket User Interfa Version: 5.0.1(201711162319) Certificate User: Superuser 22-Dec-2017	
									Participant:	Participant Name 💙 User: Superuser	
lic Generation Demand ARD	Regulation	DRR									
	chedule Scher lanager Detai	tule I Defaults	Schedule Offers Default	Schedule Hourly Up			thedule Hourly		XML Download		
RR Hourly Updates											
ect Date and DRR	44										
Date: 23-Dec-2017	173	Default	Values for DR	R 10(10) on 23-D	ec-2017					7
	3	Override	Hourly Value	s for DRR	10 (10)	on 23-Dec-201	7				
Portfolio: DRR_B		Hour		Min Red	Max Red	Claim10	Claim30	Ramp Rate	Available		
DRR: DRR10	*	-	- Copy						1		
		01		11.0	38.0	20.0	38.0	5.0	13		
		02		11.0	38.0	20.0	38.0	5.0			
		03		11.0	38.0	20.0	38.0	5.0			
		04		11.0	38.0	20.0	38.0	5.0			
		05		11.0	38.0	20.0	38.0	5.0			
		08		11.0	38.0	20.0	38.0	5.0			
		07		5.0	45.0	11.0	45.0	6.0			
		08		5.0	45.0	11.0	45.0	6.0			
		09		5.0	45.0	11.0	45.0	6.0			
		10		5.0	45.0	11.0	45.0	6.0			
		11		5.0	45.0	11.0	45.0	6.0			
		12		5.0	45.0	11.0	45.0	6.0			
		13		5.0	45.0	11.0	45.0	6.0			
		14									
										Update R	efre

The Override Hourly Values section shows each hour of the day and fields to enter overrides. When the fields are blank and the check box is not specified, it means that there is no override of the schedule or default values for that hour.

- **Hour** Hour ending values are shown in the first column for each hour in the day. The row for hour ending 03 is not shown for the Spring Daylight Saving Time transition day. Similarly, the value on the row of the repeating hour ending 02 of the Fall Daylight Saving Time transition day is marked with an asterisk (*).
- **Min Red** Column to enter overrides of the Minimum Reduction limit for any hour of the day.
- Max Red Column to enter overrides of the Maximum Reduction limit for any hour of the day.

- **Claim 10** Column to enter overrides of the Claim 10 capability for any hour of the day if the DRR is Claim 10 capable. If the DRR is not Claim 10 capable, there will be no data entry cells in this column.
- **Claim 30** Column to enter overrides of the Claim 30 capability for any hour of the day if the DRR is Claim 30 capable. If the DRR is not Claim 30 capable, there will be no data entry cells in this column.
- Ramp Rate Rate at which the DRR can move (MW/minute).

The following override values of the Default Status are specified on the Hourly Updates display:

• **Available** – When checked, it indicates that the DRR is available for economic commitment and dispatch for that hour of the operating day. When blank, it indicates that the DRR is not available. If empty (green background inside the checkbox), it means that no status override exists for that hour.

If a value is entered for an hour in any of the columns, values must also be entered in every other column even if they are copies of the default values. Setting the entire row values to blank (NULL) means that the corresponding values defined in the Schedule Detail data of the schedules that are flagged as available for the market day are to be used for that market hour; i.e. it is equivalent to removing the override of the values for that market hour. Setting the off-line reserve claim values to blank (NULL) means that the corresponding values defined in the DRR Default Parameters data for the market day are to be used for that market hour. Similarly, clearing of the Available flag means that the DRR's default status defined in the DRR Default Parameters data for the market day is to be used for that market hour.

The top row of the grid provides you with the means to enter or clear overrides for a range of hours. If you do use the top row, you need to **remember that leaving all data entry fields on this row blank means that you want to remove the overrides for all hours in the range selected**. For example, if you want to remove all overrides for every hour of the day, select 01 for the "From" hour (left drop-down selector) and 24 for the "To" hour (right drop-down selector) in the Hour drop-down selection lists, select the "Copy" button and then click **Update**. Since you did not enter anything in the limit fields or select any of the dispatch availability override flags, all overrides are removed. **You cannot use the top row to selectively update the overrides in one column for a range of hours while retaining the override values in the other columns**.

9.10.1 Validation rules

When you click the **Update** button, eMarket makes the following checks to ensure that the entered data is valid:

• Min Reduction ≤ Max Reduction. The data submitted is rejected if this rule is violated.

- When a DRR is declared unavailable for an hour (through the combination of the DRR's default status for the day and the hourly overrides of the status), it should not be declared available until its Min Time Between Reductions is elapsed. Therefore if the DRR is set to Unavailable for any hour, it *should* be set for a consecutive number of hours sufficient to cover its Min Time Between Reductions. A warning will be returned if the rule is violated.
- A warning will be returned if the DRR is Claim 10 capable, a Claim 10 override is entered and any of the following conditions exist
 - Claim 10 = 0
 - Claim 10 > Claim 10 Cap meaning that the Claim 10 Cap value will be used instead of the offered value
- A warning will be returned if the DRR is Claim 30 capable, a Claim 30 override is entered and any of the following conditions exist
 - Claim 30 = 0
 - Claim 30 > Claim 30 Cap meaning that the Claim 30 Cap value will be used instead of the offered value
 - Claim 30 < Claim 10 meaning that the Claim 10 value will be used instead of the offered Claim 30 value.

9.10.2 Override MW Operating Limits for a Day and Hour

As discussed in §9.5 DRR Schedule Detail Defaults Display, you define on this display a DRR's MW operating limits under the selected schedule. These limit values are shown in the Defaults section for each schedule that you've created for the DRR. If you want to profile the DRR's MW operating limits on an hourly basis, irrespective of which schedule is designated as available for a given market, you override the schedule values using the DRR Hourly Updates display as follows:

- 1. In the DRR Navigation sub tabs, click the **Hourly Updates** tab.
 - The DRR Hourly Updates display is shown in the display pane.
- 2. In the **Filters pane**:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired DRR from the **DRR** filter.
 - The display reflects the hourly overrides for the selected DRR.
- 3. To update the overrides for individual hours:
 - a. Enter override values for all of the limits in the hours that you wish. You must also enter DRR dispatch availability status overrides. If the DRR is Claim 10 or Claim 30 capable, you must also enter Claim 10 and Claim 30 override values. If you want to remove an override, you must enter blanks in every data entry cells and deselect all status overrides.

- b. Click the **Update** button to save the data.
- 4. To update the overrides for a range of hours
 - a. Enter override values for all of the limits in the top row. You must also enter DRR dispatch availability status overrides. If the DRR is Claim 10 or Claim 30 capable, you must also enter Claim 10 and Claim 30 override values. If you want to remove overrides, leave all the data entry cells on the first row blank and the status overrides deselected.
 - b. In the Hour column, select a "From" hour (left drop-down selector) and a "To" hour (right drop-down selector) and then select the **Copy** button. This will copy the data entered in the top row to every row for the range of hours selected.
 - c. Repeat steps a and b if you want to enter/remove overrides for other ranges of hours.
 - d. Click the **Update** button to save the data for the selected hours.

9.10.3 Override the Status of a DRR for a Day and Hour

As discussed in §9.3 DRR Default Parameters Display, you define on this display a DRR's default dispatch availability status (Default Status) for an entire market day. This status is shown in the Defaults section. If you want to profile the DRR's dispatch availability status on an hourly basis, you override this default status using the DRR Hourly Updates display.

- 1. In the DRR Navigation sub tabs, click the Hourly Updates tab.
 - The DRR Hourly Updates display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - c. Select the desired DRR from the **DRR** filter.
 - The display reflects the hourly overrides for the selected DRR.
- 3. To update the overrides for individual hours:
 - a. Select the Available status checkbox for the hours that you want to override the default status to available. You must also enter override values for all the limits even if they are copies of the default values. If the DRR is Claim 10 or Claim 30 capable, you must also enter Claim 10 and Claim 30 override values. If you want to remove an override, set the Available checkbox to its default value (not checked with green background) and enter blanks in all other data entry cells.
 - b. Click the **Update** button to save the data.
- 4. To update the overrides for a range of hours
 - a. Select the status override value in the first row. You must also enter override values for all the limits even if they are copies of the default values. If the DRR is Claim 10 or Claim 30 capable, you must also enter

Claim 10 and Claim 30 override values. If you want to remove overrides, leave all the data entry cells on the first row blank and the status overrides deselected.

- b. In the Hour column, select a "From" hour (left drop-down selector) and a "To" hour (right drop-down selector) and then select the **Copy** button. This will copy the data entered in the top row to every row for the range of hours selected.
- c. Repeat steps a and b if you want to enter/remove overrides for other ranges of hours.
- d. Click the **Update** button to save the data for the selected hours.

9.11 DRR by Portfolio Display

You can view the cleared results of the Day-Ahead Market for each DRR listed in a selected portfolio using the **DRR by Portfolio** display. The schedule name in this display is the name of the schedule used when committing the DRR, i.e. the schedule that was designated to be available for the Day-Ahead market on the selected day. The other information is the MW amount that the DRR is committed to provide, along with the Locational Marginal Price (LMP) at the DRR's pnode for each hour of the market day.

If a DRR was not committed for any hour of the day, its name will not be listed in the DRR By Portfolio display. The display will be empty if none of the DRRs in the selected portfolio has been committed for any hour of the selected day.

For committed DRRs, information for a 24-hour period is shown on two lines: the top line details the hours ending 01 through 12; the second line details hours ending 13 through 24. The last row of the display is labeled **Total Portfolio MW** and shows the total MW amount for each hour committed in the Day-Ahead market by all the DRRs in your portfolio. Information on the display is shown only after the Day-Ahead market has been cleared.

ISO new england		DRR Market Status: CLOSED						eMarket User Interface Version: 5.0.1(201711162319) Certificate User: Superver 26-Dec-2017 12:06:									
									ReadOnly	Participant:	Partic	ipant Nar	ne		Y User	Superuser	
Public Generation Demand ARD	Regulation	ORR															
	Schedule Sched Manager Detail	lule Defaults	Schedule Offers Default	Schedule Offers Hourly Updates		dule Times ly Updates	Schedule Selection	Hourly Updates	DRR By Portfolio	XML Download							
ORR By Portfolio																	
elect Date and Portfolio	«	DRR By	Portfolio for DF	RR_B on 19-Dec	-2017												
Date: 19-Dec-2017		Name															
Portfolio: DRR_B	×	DRR		01 / 13	02/14	03/15	04/16	05/17	06 / 18		8/20	09/21	10/22	11/23	12/24		
		(DFLPF	RICE)								87.5@	37.5@	37.5@				
											58.44	\$61.67	\$64.09				
					37.5@	25@	37.5@				7.5@	37.5@	37.5@				
		DRR	12		\$51.58 02/14	\$49.64 03 / 15	\$54.16	05/17	06/18		59.68 8 / 20	\$67.71 09/21	\$57.73	44.(22	12/24		
		(DFLPF	1000	01713	02714	03715	a , 7 16	05717	06716		30@	09721	30@	11/23	30@		
		OFCFF	(ICE)								58.52		\$64.33		\$56.12		
									30@		30@		304.33		350.12		
									\$95.71		63.14						
		Total P	ortfolio MW	01 / 13	02/14	03/15	04/16	05/17	06/18		8/20	09/21	10/22	11/23	12/24		
							in relation of the				67.5	37.5	67.5		30		
					37.5	25	37.5		30		67.5	37.5	37.5				
																	Refres
essages																	

The format of this display is shown below.

The following generation information is displayed for each DRR included in the selected generation portfolio:

- **DRR/Schedule** the name of the DRR and of the schedule on which it was committed.
- **MW** @ **Price** the MW amount that DRR is committed to provide and the \$/MWh DA LMP at the resource Pnode for each hour of the day.
- **Total Portfolio MW** the total amount of MWh committed for all the DRRs of the selected portfolio for each hour of the day.

9.11.1 View DA Commitment for each DRR in Portfolio

To view the commitment in the Day-Ahead market of the DRRs listed in your portfolios and the Day-Ahead LMPs at their resource Pnodes:

- 1. In the DRR Navigation sub tabs, click the **DRR by Portfolio** tab
 - The DRR by Portfolio display is shown in the display pane.
- 2. In the Filters pane:
 - a. Select the desired date from the **Date** filter.
 - b. Select the desired portfolio from the **Portfolio** filter.
 - The display reflects the selected market day's day-ahead commitment and LMPs for each of the DRRs listed in the selected portfolio. If the Day-Ahead market has not yet been cleared for the selected day, or none of the DRRs in the selected portfolio has been committed, the following text is shown in the display title bar: "The day ahead market results are not available ..."

9.12 DRR XML Download Display

The XML Download display is accessible through the DRR tab. This display enables you to download pre-defined DRR specific XML report files of data that you maintain in the ISO-NE Market System, including results from the cleared Day-Ahead market. It also provides the means to upload data in XML formatted data files into the ISO-NE Market System to be used to clear the Day-Ahead and Real Time markets. The format of the XML files can be found in the **eMarket Data Exchange Specification**, which is found on the <u>ISO New England</u> <u>website</u> among the documents listed on the Participate > Support > Upload and Download File Format Protocols page.

ISO new england										DRI	R Market St	atus: CLOSED	eMarket User Interface Version: 5.0.1(201711162319) Certificate User: Superuser 26-Dec-201712:49:1					
												Participant:	Participant Name	✓ User:	Superuser	*		
Public Generation Demand ARD Regulation DRR																		
Private DRR Defau Messages Parameter			Schedule Manager	Schedu Detail	le Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Updates	DRR By Portfolio	XML Download						
XML Download																		
Select Date				<	XML Do	ownload on 12-D	Dec-2017											
Date: 12-Dec-2	017				Scher Scher DRR Scher Scher	- dules dule Details dule Offers dule Selectio Hourly Upda dule Offers I	on ates Hourly Update Hourly Update					Upload XML S	OAP Request Browse					
Messages																*		

The format of the XML Download display is shown below.

The following XML reports are available for downloading:

- DRRs
- DRR Schedules
- DRR Schedule Details
- DRR Schedule Offers
- DRR Schedule Selection
- DRR Hourly Updates
- Schedule Offers Hourly Updates
- Schedule Times Hourly Updates
- Reduction by DRR

All but the Reduction by DRR returns an XML formatted report for the selected date and every DRR that belongs to you as Lead Participant. The Redution by DRR report returns information for all the unique DRRs listed in all of the user's portfolios.

See §3.2 for details on uploading and downloading files. The following sections provide samples of the downloaded reports.

9.12.1 DRRs XML Report

This report contains the information that is shown on the DRR Default Parameters display for every DRR that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetDRRResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <DRR day="2017-12-27" ID=" ... ">
    <CapacityResource>false</CapacityResource>
    <Claim10Settings claim10Capable="true">
      <Claim10>18</Claim10>
      <Claim10Cap xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
    </Claim10Settings>
    <Claim30Settings claim30Capable="true">
      <Claim30>20</Claim30>
      <Claim30Cap xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
    </Claim30Settings>
    <DefaultStatus>Available</DefaultStatus>
    <LeadParticipantID> ... </LeadParticipantID>
    <LeadParticipantName> . . . </LeadParticipantName>
    <Location> . . . </Location>
    <RampRateCap xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
    <RampRateDefault>10</RampRateDefault>
    <DRRLongName> . . . </DRRLongName>
    <DRRNumber> ... </DRRNumber>
    <DRRShortName> ... </DRRShortName>
  </DRR>
  <DRR day="2017-12-27" ID=" ... ">
    <CapacityResource>false</CapacityResource>
    <Claim10Settings claim10Capable="true">
      <Claim10>11</Claim10>
      <Claim10Cap xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
    </Claim10Settings>
    <Claim30Settings claim30Capable="true">
      <Claim30>22</Claim30>
      <Claim30Cap xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
    </Claim30Settings>
    <DefaultStatus>Available</DefaultStatus>
    <LeadParticipantID> ... </LeadParticipantID>
    <LeadParticipantName> . . . </LeadParticipantName>
    <Location> . . . </Location>
    <RampRateCap xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/>
    <RampRateDefault>10</RampRateDefault>
    <DRRLongName> . . . </DRRLongName>
    <DRRNumber> ... </DRRNumber>
    <DRRShortName> . . . </DRRShortName>
  </DRR>
</GetDRRResponse>
```

9.12.2 DRR Schedules XML Report

This report contains the information that is shown on the DRR Schedule Manager display for all schedules of every DRR that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetDRRScheduleResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <DRRSchedule schedID=" ... 0701">
    <DRRID> ... 07</DRRID>
    <DRRShortName> ... </DRRShortName>
    <SchedLongName>DFLPRICE</SchedLongName>
    <SchedShortName>DFLPRICE</SchedShortName>
    <SchedType>Price</SchedType>
    <SchedTypeID>1</SchedTypeID>
  </DRRSchedule>
  <DRRSchedule schedID=" ... 0702">
    <DRRID> . . . 07</DRRID>
    <DRRShortName> . . . </DRRShortName>
    <SchedLongName>ALTPRICE 2</SchedLongName>
    <SchedShortName>ALTPRICE</SchedShortName>
    <SchedType>Price</SchedType>
    <SchedTypeID>2</SchedTypeID>
  </DRRSchedule>
  <DRRSchedule schedID=" ... 0801">
    <DRRID>...08</DRRID>
    <DRRShortName> . . . </DRRShortName>
    <SchedLongName>DFLPRICE</SchedLongName>
    <SchedShortName>DFLPRICE</SchedShortName>
    <SchedType>Price</SchedType>
    <SchedTypeID>1</SchedTypeID>
  </DRRSchedule>
</GetDRRScheduleResponse>
```

9.12.3 DRR Schedule Details XML Report

This report contains the information that is shown on the DRR Schedule Detail display for all schedules of every unique DRR that belongs to you as Lead Participant. The following is a sample of the report.

<?xml version="1.0" encoding="UTF-8" standalone="yes"?> <GetDRRScheduleDetailResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages"> <DRRScheduleDetail day="2017-12-27" ID=">...0701"> <NotifyTime>0.01</NotifyTime> <StartupTime>0.2</StartupTime> <MaxReduction>44</MaxReduction> <MinReduction>1</MinReduction> <MarketType>Both</MarketType> <MaxDailyEnergy xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MaxDailyStarts xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MinTimeBetweenReductions>1</MinTimeBetweenReductions> <MinReductionTime>0.01</MinReductionTime> <ScheduleAvailable>true</ScheduleAvailable> <ShortName>DFLPRICE</ShortName> </DRRScheduleDetail> <DRRScheduleDetail day="2017-12-27" ID="> ... 0702"> <NotifyTime>1</NotifyTime> <StartupTime>0.1</StartupTime> <MaxReduction>55</MaxReduction> <MinReduction>10</MinReduction> <MarketType>Both</MarketType> <MaxDailyEnergy xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MaxDailyStarts xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MinTimeBetweenReductions>0.5</MinTimeBetweenReductions> <MinReductionTime>1</MinReductionTime> <ScheduleAvailable>false</ScheduleAvailable> <ShortName>ALTPRICE</ShortName> </DRRScheduleDetail> <DRRScheduleDetail day="2017-12-27" ID="> . . . 0801"> <NotifyTime>0.1</NotifyTime> <StartupTime>0.01</StartupTime> <MaxReduction>60</MaxReduction> <MinReduction>5</MinReduction> <MarketType>Both</MarketType> <MaxDailyEnergy xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MaxDailyStarts xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:nil="true"/> <MinTimeBetweenReductions>2</MinTimeBetweenReductions> <MinReductionTime>1</MinReductionTime> <ScheduleAvailable>true</ScheduleAvailable> <ShortName>DFLPRICE</ShortName> </DRRScheduleDetail> </GetDRRScheduleDetailResponse>

9.12.4 DRR Schedule Offers XML Report

This report contains the information that is shown on the DRR Schedule Offers display for all schedules of every DRR that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetDRRScheduleBidResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <DRRScheduleBid day="2017-12-27" ID=" ... 0701">
    <PriceCurve>
      <PricePoint price="19.5" MW="44"/>
    </PriceCurve>
    <ShortName>DFLPRICE</ShortName>
    <DRTP>16</DRTP>
    <InterruptionCost>600</InterruptionCost>
  </DRRScheduleBid>
  <DRRScheduleBid day="2017-12-27" ID=" ... 0702">
    <PriceCurve>
      <PricePoint price="19" MW="10"/>
      <PricePoint price="25" MW="55"/>
    </PriceCurve>
    <ShortName>ALTPRICE</ShortName>
    <DRTP>16</DRTP>
    <InterruptionCost>640</InterruptionCost>
  </DRRScheduleBid>
  <DRRScheduleBid day="2017-12-27" ID=" ... 0801">
    <ShortName>DFLPRICE</ShortName>
    <DRTP>16</DRTP>
    <InterruptionCost>0</InterruptionCost>
  </DRRScheduleBid>
</GetDRRScheduleBidResponse>
```

9.12.5 DRR Schedule Selection XML Report

This report contains the information that is shown on the DRR Schedule Selection display for all schedules of every DRR that belongs to you as Lead Participant. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="ves"?>
<GetDRRScheduleSelectionResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <DRRScheduleSelection day="2017-12-27">
    <DRRSchedule ID="...0701">
      <Available>true</Available>
      <ShortName>DFLPRICE</ShortName>
    </DRRSchedule>
    <DRRSchedule ID="...0702">
      <Available>false</Available>
      <ShortName>ALTPRICE</ShortName>
    </DRRSchedule>
    <DRRSchedule ID=" ... 0801">
      <Available>true</Available>
      <ShortName>DFLPRICE</ShortName>
    </DRRSchedule>
  </DRRScheduleSelection>
</GetDRRScheduleSelectionResponse>
```

9.12.6 DRR Hourly Updates XML Report

This report contains the information that is shown on the DRR Hourly Updates display for every DRR that belongs to you as Lead Participant, and for which hourly updates exist. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetDRRHourlyResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <DRRHourly day="2017-12-27" ID=" ... 08">
    <HourlyProfile>
      <HourlyValues time="2017-12-27T00:00:00-05:00">
       <MinReduction>1</MinReduction>
       <MaxReduction>45</MaxReduction>
       <RampRate>5</RampRate>
        <ResourceStatus>Available</ResourceStatus>
        <Claim10>9</Claim10>
        <Claim30>18</Claim30>
      </HourlyValues>
      <HourlyValues time="2017-12-27T01:00:00-05:00">
       <MinReduction>1</MinReduction>
       <MaxReduction>45</MaxReduction>
        <RampRate>5</RampRate>
       <ResourceStatus>Available</ResourceStatus>
       <Claim10>9</Claim10>
       <Claim30>18</Claim30>
      </HourlyValues>
      <HourlyValues time="2017-12-27T02:00:00-05:00">
       <MinReduction>5</MinReduction>
        <MaxReduction>60</MaxReduction>
        <RampRate>10</RampRate>
       <ResourceStatus>Unavailable</ResourceStatus>
       <Claim10>11</Claim10>
        <Claim30>22</Claim30>
      </HourlyValues>
      <HourlyValues time="2017-12-27T03:00:00-05:00">
       <MinReduction>5</MinReduction>
        <MaxReduction>60</MaxReduction>
       <RampRate>10</RampRate>
        <ResourceStatus>Unavailable</ResourceStatus>
        <Claim10>11</Claim10>
       <Claim30>22</Claim30>
      </HourlyValues>
    </HourlyProfile>
    <ShortName> . . . </ShortName>
  </DRRHourly>
</GetDRRHourlyResponse>
```

9.12.7 Schedule Offers Hourly Updates XML Report

This report contains the information that is shown on the DRR Schedule Offers Hourly Updates display for every unique DRR that belongs to you as Lead Participant, and for which Schedule Offer Hourly Updates exist. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetDRRScheduleBidHourlyResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <DRRScheduleBidHourly day="2017-12-27" ID=" ... 01">
    <HourlyProfile>
      <HourlyValues time="2017-12-27T02:00:00-05:00">
        <PriceCurve>
          <PricePoint price="31" MW="30"/>
          <PricePoint price="56" MW="60"/>
        </PriceCurve>
        <InterruptionCost>450</InterruptionCost>
      </HourlyValues>
      <HourlyValues time="2017-12-27T03:00:00-05:00">
        <PriceCurve>
          <PricePoint price="30" MW="15"/>
          <PricePoint price="40" MW="30"/>
          <PricePoint price="56" MW="60"/>
        </PriceCurve>
        <InterruptionCost>200</InterruptionCost>
      </HourlyValues>
      <HourlyValues time="2017-12-27T04:00:00-05:00">
        <PriceCurve>
          <PricePoint price="56" MW="60"/>
        </PriceCurve>
        <InterruptionCost>370</InterruptionCost>
      </HourlyValues>
    </HourlyProfile>
    <ShortName>DFLPRICE</ShortName>
  </DRRScheduleBidHourly>
</GetDRRScheduleBidHourlyResponse>
```

9.12.8 Schedule Times Hourly Updates XML Report

This report contains the information that is shown on the DRR Schedule Times Hourly Updates display for every DRR that belongs to you as Lead Participant, and for which hourly updates exist. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetDRRScheduleTimesHourlyResponse xmIns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <DRRScheduleTimesHourly day="2017-12-27" ID=" ... 02">
    <HourlyProfile>
      <HourlyValues time="2017-12-27T00:00:00-05:00">
        <StartupTime>0.01</StartupTime>
        <NotifyTime>1</NotifyTime>
      </HourlyValues>
    </HourlyProfile>
    <ShortName>ALTPRICE</ShortName>
  </DRRScheduleTimesHourly>
  <DRRScheduleTimesHourly day="2017-12-27" ID=" ... 01">
    <HourlyProfile>
      <HourlyValues time="2017-12-27T00:00:00-05:00">
        <StartupTime>0.02</StartupTime>
        <NotifyTime>0.2</NotifyTime>
      </HourlyValues>
      <HourlyValues time="2017-12-27T01:00:00-05:00">
        <StartupTime>0.02</StartupTime>
        <NotifyTime>0.2</NotifyTime>
      </HourlyValues>
      <HourlyValues time="2017-12-27T02:00:00-05:00">
        <StartupTime>0.01</StartupTime>
        <NotifyTime>0.1</NotifyTime>
      </HourlyValues>
      <HourlyValues time="2017-12-27T03:00:00-05:00">
        <StartupTime>0.1</StartupTime>
        <NotifyTime>0.1</NotifyTime>
      </HourlyValues>
    </HourlyProfile>
    <ShortName>DFLPRICE</ShortName>
  </DRRScheduleTimesHourly>
</GetDRRScheduleTimesHourlyResponse>
```

9.12.9 Reduction by DRR XML Report

This report contains the same information that is shown on the DRR by Portfolio display. All the DRRs that belong to any of your portfolios, and which have been cleared for any hour, are listed. The report contains information only after the Day-Ahead market has been cleared. The following is a sample of the report.

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<GetDRRByParticipantResponse xmlns="http://www.markets.iso-ne.com/MUI/eMkt/Messages">
  <DRRByParticipant day="2017-12-12" ID=" ... 10">
    <DRRShortName> ... </DRRShortName>
    <DRRScheduleID>...1001</DRRScheduleID>
    <DRRScheduleShortName>DFLPRICE</DRRScheduleShortName>
    <HourlyProfile>
      <HourlyValues time="2017-12-12T09:00:00-05:00" MW="25" price="33.02"/>
      <HourlvValues time="2017-12-12T10:00:00-05:00" MW="10" price="38.15"/>
      <HourlyValues time="2017-12-12T13:00:00-05:00" MW="27" price="38.13"/>
      <HourlyValues time="2017-12-12T14:00:00-05:00" MW="27" price="38.26"/>
      <HourlyValues time="2017-12-12T17:00:00-05:00" MW="35" price="56.2"/>
      <HourlyValues time="2017-12-12T18:00:00-05:00" MW="35" price="46.59"/>
      <HourlyValues time="2017-12-12T21:00:00-05:00" MW="35" price="35.98"/>
      <HourlyValues time="2017-12-12T22:00:00-05:00" MW="35" price="36.52"/>
    </HourlyProfile>
  </DRRByParticipant>
  <DRRByParticipant day="2017-12-12" ID=" ... 11">
    <DRRShortName> . . . </DRRShortName>
    <DRRScheduleID> . . . 1101
    <DRRScheduleShortName>DFLPRICE</DRRScheduleShortName>
    <HourlvProfile>
      <HourlyValues time="2017-12-12T08:00:00-05:00" MW="15" price="29.48"/>
      <HourlyValues time="2017-12-12T09:00:00-05:00" MW="15" price="32.94"/>
      <HourlyValues time="2017-12-12T13:00:00-05:00" MW="20" price="37.57"/>
      <HourlyValues time="2017-12-12T14:00:00-05:00" MW="20" price="37.67"/>
      <HourlyValues time="2017-12-12T15:00:00-05:00" MW="20" price="38.47"/>
      <HourlyValues time="2017-12-12T19:00:00-05:00" MW="35" price="41.18"/>
      <HourlyValues time="2017-12-12T20:00:00-05:00" MW="35" price="38.46"/>
      <HourlyValues time="2017-12-12T21:00:00-05:00" MW="35" price="35.55"/>
    </HourlyProfile>
  </DRRByParticipant>
  <DRRByParticipant day="2017-12-12" ID=" ... 13">
    <DRRShortName> ... </DRRShortName>
    <DRRScheduleID> ... 1301/DRRScheduleID>
    <DRRScheduleShortName>DFLPRICE</DRRScheduleShortName>
    <HourlyProfile>
      <HourlyValues time="2017-12-12T09:00:00-05:00" MW="20" price="33.61"/>
      <HourlyValues time="2017-12-12T10:00:00-05:00" MW="20" price="38.76"/>
      <HourlyValues time="2017-12-12T11:00:00-05:00" MW="20" price="33"/>
      <HourlyValues time="2017-12-12T12:00:00-05:00" MW="20" price="39.47"/>
      <HourlyValues time="2017-12-12T17:00:00-05:00" MW="25" price="56.84"/>
      <HourlyValues time="2017-12-12T18:00:00-05:00" MW="25" price="47.16"/>
      <HourlyValues time="2017-12-12T19:00:00-05:00" MW="25" price="42.2"/>
      <HourlyValues time="2017-12-12T20:00:00-05:00" MW="25" price="39.34"/>
    </HourlvProfile>
  </DRRByParticipant>
</GetDRRByParticipantResponse>
```
10. Continuous Storage Facilities (CSF)

10.1 CSF Concepts

Continuous Storage Facilities (CSF) are capable of receiving electricity from the grid and storing the energy for later injection of electricity back to the grid. CSFs are enabled to participate in the energy and regulation markets. In eMarket application, these facilities are referred to as Energy Storage Devices (ESD).

A CSF is a combination of three asset types:



All three components of a CSF are registered in Customer and Asset Management System (CAMS) as parts of a single facility, but with separate asset IDs and names.

All three components of a CSF may participate simultaneously in the applicable market. Each of the components manages only its own responsibilities.

- The generator manages energy and reserves for MW at and above zero
- The DARD manages energy and reserves for MW below zero (consumption)
- The ATRR manages the regulation, positive or negative

CSFs operate in a continuous range of generation and consumption. They can switch automatically between generation and consumption. CSFs can switch between maximum generation and maximum consumption in 10 minutes or less.

The regulation can be provided while generating or consuming energy.

CSF Generator and CSF DARD participate in the Day Ahead market by bidding in eMarket. CSF Generators are required to offer in eMarket if they have capacity supply obligation through Forward Capacity Market.

Both CSF Generator and CSF DARD are considered to be always online in Day Ahead unless Participant declares them unavailable. Clearing in Day Ahead is based on offers and economics independently for CSF Generator and CSF DARD. Participant is responsible for developing bids and offers so that a CSF does not get a Day Ahead obligation to charge and discharge at the same time. In Real Time market, CSF assets are dispatched based on charge status.

- When fully discharged (i.e., available energy = 0 MWh) only DARD is available
- When fully charged (i.e., available storage = 0 MWh) only the generator asset is available
- When partially charged or discharged, dispatch is based on offer parameters, availability status, and telemetered dispatch limits

CSFs can provide 10-minute spinning reserves.

10.2 Managing CSFs in eMarket

CSF Generators, DARDs, and ATRRs are managed using the same eMarket displays as non-CSF assets. However, some specific validation rules apply to CSF assets.

Each component of a CSF is managed separately.

- Generation function group to support CSF Generator offers
- ARD function group to support CSF DARD offers
- Regulation function group to support CSF ATRR regulation offers

	SO new	v engla	and			Market Read(: Status: OPEN(01 Dnly Participant:		Version	rket Us 5.0.12(2018080 er: Superuser Use	12148) 28	terfa -Dec-2018	
Public	Generation	Demar	d ARD	Regulation	DRR								
	vate Unit D ssages Param		Portfolio Manager	Ramp Rate Default	Ramp Rate Hourly Updates	Schedule Manager	Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Updates	Fuel Pr Adjustr

There is no cross-validation between offers of the three assets of a CSF.

For each asset type, the corresponding Portfolio Manager functionality is used to create a portfolio. There is no difference in Portfolio Manager functionality between creating portfolios for CSF and non-CSF assets. For more information, refer to section 3.1 Manage Portfolios.

10.3 CSF Generator

CSF Generator's offers and limits are managed using sub-tabs in the Generation group function.

10.3.1 Unit Daily Default Parameters

The Unit Daily Default Parameters display allows the Participant to view the CSF Generator's default parameters, as well as update the default status and ramp rate.

	_														
Public	Ge	eneration	Demand	ARD	Regulation	DRR									
	ivate issages	Unit D Param		Portfolio Manager	Ramp Rate Default	Ramp Rate Hourly Updates	Schedule Manager	Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Updates	Fuel Price Adjustments	€→ E
Un	it Da	aily De	efault	Paran	neters										
Select	Date	and Unit			«	Unit Details fo	MyFacility_	_GEN (12345) on 2	9-Dec-2018						
	Date:	29-Dec-20	018				Asset ID:	12345		L	ead Participant ID:				^
Po	tfolio:	My_CS	F_GEN		*	Unit	Short Name:	ESD1		Lead	Participant Name:				
	Unit:	MyFaci	lity_GEI	N	*	Unit	Long Name:	MyFacility G	GEN	DA	Schedule Required:	No			
						_		Energy Storage		_	qulation Resource:				
						L -		Linergy storage			,				
						Ramp Rat	e (MW/Min):	100.0			Default Status:	O Economi	c		
						Ran	np Rate Cap:	UD			[🔿 Unavaila	ble		
						Claim	10 Capable:	No				Must Rur	,		
						Cla	im 10 (MW):			-	L				
							laim 10 Cap:	UD		-					
						Claim	30 Capable:	No		_					
							im 30 (MW):			-					
							laim 30 Cap:			-					
							ann ao cap.	00							~

The format of the Unit Daily Default Parameter display is shown below:

Fields on this display are the same for CSF and non-CSF generators. For CSF generators, the Type of Unit is "Energy Storage Device", which is a synonym of Continuous Storage Device. Regulation Resource is always displayed as "No". Claim 10 and Claim 30 fields are not updateable because CSF Generators are not Claim10/Claim30 capable.

For more information on this display, refer to section 4.3 Unit Default Parameters Display.

10.3.1.1 Modify CSF Generator Default Status

To specify the CSF Generator's status:

- 1. Select Unit Default Parameters in Generation function group.
- 2. Select the desired Portfolio.
- 3. Select the desired Unit.
- 4. In Default Status, select one of the following radio buttons
 - a. Must Run
 - b. Unavailable
- 5. Click the **Update** button to save the selection.

10.3.1.2 Modify CSF Generator Ramp Rate

To specify the CSF Generator's ramp rate:

- 1. Select Unit Default Parameters in Generation function group.
- 2. Select the desired Portfolio.
- 3. Select the desired Unit.

- 4. In Ramp Rate (MW/min), enter the ramp rate value. Note that the ramp rate must be high enough to allow the CSF to switch between maximum generation and maximum consumption in 10 minutes or less.
- 5. Click the **Update** button to save the changes.

10.3.2 Schedule Detail Defaults

The Schedule Detail Defaults display allows the Participant to view the CSF Generator's schedule default parameters, as well as update the default operating limits, and maximum daily energy.

The format of the Schedule Detail Defaults display is shown below:

Public G	eneration	Demand /	ARD Regulat	ion	DRR											
Private Messages	Unit Default Parameters	Portfolio Manager	Ramp Rate Default		p Rate 1y Updates	Schedule Manager	Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Updates	Fuel Price Adjustments	Generation By Portfolio		beok
Sched	ule Det	ail Defa	ults													
Select Date	, Unit and Sc	hedule		~	Schedule	Details for	MyFacility_GEN (123	45) – DFLPRICE (12	34501 Price-based) o	on 31-Dec-2018						
Date:	31-Dec-201	18		9		Schedule #	Available: Yes		Max	Daily Starts:	_		Eme	rgency Min:	0.0	^
Portfolio:	My_CSF_	GEN		~			i Market: 🔽		-	ergy (MWh): 120.	0			pnomic Min:		
Unit	MyFacilit	y_GEN		~			e Market: 🗸							nomic Max:		
Schedule	DFLPRICE			~		Redi Time	e markec 🔽		1000	wntime (hr): 0.0	_					
									10000	tuntime (hr): 0.0			RT High Opera	ting Limit *:	30.0	
						Fast Start	(10 min): No		Hot to Co	Id Time (hr): 0.0						
						Fast Start ((30 min): No		Hot to Inte	er Time (hr): 0.0			Condens	e Available:		
									Hot Notificatio	on Time (hr): 0.0		(Condense Start	up Cost (\$):	0.00	
									Inter Notificatio	on Time (hr): 0.0			Condense Hou	rly Cost (\$):	0.00	
									Cold Notificatio	on Time (hr): 0.0			Cond	ense Power:	0.0	
									Hot Startu	up Time (hr): 0.0		Conde	nse Notificatio	n Time (hr):	0.0	
									Inter Startu	p Time (hr): 0.0						
									Cold Start	up Time (hr): 0.0						~

For CSF generators, the following rules are enforced:

- Emergency Min and Economic Min must be set to zero.
- Intertemporal parameters (runtime, downtime, notification and startup times) must be set to zero.

For more information on this display, refer to section 4.7 Schedule Detail Defaults Display.

10.3.2.1 Modify Schedule Constraint Data

Constraint data include minimum downtime, minimum runtime, maximum daily starts and maximum daily energy. To specify the CSF Generator's schedule constraint data:

- 1. Select Schedule Detail Defaults in Generation function group.
- 2. Select the desired Portfolio.
- 3. Select the desired Unit.
- 4. Enter changes in any of the following fields:

- a. Min Downtime must be set to zero
- b. Min Runtime must be set to zero
- c. Maximum Daily Starts
- d. Maximum Daily Energy (MWh) if left blank it means that the value is unlimited.
- 5. Click the **Update** button to save changes.

10.3.2.2 Modify Schedule Operating Limits

To specify the CSF Generator's schedule operating limits:

- 1. Select Schedule Detail Defaults in Generation function group.
- 2. Select the desired Portfolio.
- 3. Select the desired Unit.
- 4. Enter changes in any of the following fields:
 - a. Emergency Min must be set to zero
 - b. Economic Min must be set to zero
 - c. Economic Max
 - d. RT High Operating Limit
- 5. Click the **Update** button to save changes.

10.3.3 Hourly Updates

The Hourly Updates display allows the Participant to view the CSF Generator's schedule default parameters, as well as update the hourly values for operating limits and generator's status.

The format of the Hourly Updates display is shown below:

Public G	eneration Den	mand A	ARD Regulat	ion	DRR															
Private Messages		Portfolio Manager	Ramp Rate Default		Rate y Updates	Schedule Manager	Schedule Detail Defa	ults Offer		Schedul Hourly U	e Offers Jpdates		le Times Updates	Schedule Selection	Hourly Updates	Fuel P Adjust		Seneration by Portfolio		load
Unit H	ourly Upd	ates																		
Select Date	and Unit			~																
Date:	25-Jan-2019			•	Default Va	lues for N	lyFacility_G	EN (12345)	on 25-Jan-2	019										8
	My_CSF_GE	N			Schedule	E	imerg Min	Econ Min	Econ Max			ype	Day Ahea						im10	Claim30
					DFLPRICE		0.0	0.0	30.		40.0 P		×	x			Capa		No	· · · ·
Unit:	MyFacility_0	GEN		~	ESD9_M		0.0	0.0	99999.		999999.9 C		×	х	v		Defa			
					ESD9_R1		0.0	0.0	99999.	9 9	99999.9 C	ost	×	x			0	ap		· ·
					<															>
					Override H	lourly Valu	es for MyFa	acility_GEN	(12345) on	25-Jan-	2019									
					Hour		Emerg Min	Econ Mi	n Econ M	fax	RTHOL		Claim10	Claim	30 Unavai	lable	Economic	٨	/lust Run	
					~	✓ Copy										0	0		$^{\circ}$	
					01		0	.0	0.0	25.0	30.0	2				0	0		۲	
					02		C	.0	0.0	25.0	25.0	0				0	0		۲	^
					03		0	.0	0.0	20.0	25.0	0				0	0		۲	
					04		C	.0	0.0	30.0	30.0	0				۲	0		0	
					05		0	.0	0.0	30.0	30.0	0				0	0		۲	

For CSF generators, the following rules are enforced:

- Emergency Min and Economic Min must be set to zero.
- Status must be either Must Run or Unavailable.

For more information on this display, refer to section 4.12 Unit Hourly Updates Display.

10.3.3.1 Enter Hourly Updates

To specify the CSF Generator's hourly updates of default schedule and default status:

- 1. Select Hourly Updates in Generation function group.
- 2. Select the desired Portfolio.
- 3. Select the desired Unit.
- 4. For any Hour of the market day, enter changes in any of the following fields:
 - a. Emergency Min must be set to zero
 - b. Economic Min must be set to zero
 - c. Select either Unavailable or Must Run radio button.
- 5. Click the **Update** button to save changes.

10.3.4 Schedule Offers Default

The Schedule Offers Default display allows the Participant to specify the CSF Generator's daily incremental energy offers.

The format of Schedule Offers Default display is shown below:

Public Ge	neration Demand A	ARD Regulatio	n	DRR										
	Unit Default Parameters Portfolio Manager	Ramp Rate Default		p Rate ly Updates	Schedule Manager	Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Schedule Selection	Hourly Updates	Fuel Price Adjustments	Generation By Portfolio	XML Download
Schedu	ule Offers Daily	y Default												
Select Date,	Unit and Schedule		46	Schedule	Offer for N	lyFacility_GEN (12	345) – DFLPRICE (1234	1501 Price-based) or	29-Dec-2018					
Date:	29-Dec-2018		3		Fue	Selection: MW	н							
Portfolio:	My_CSF_GEN		~											
Unit:	MyFacility_GEN		*		Hot Startu	p Cost (\$):	0.00							
Schedule:	DFLPRICE		*	Intermed	diate Startu	p Cost (\$):	0.00							
					Cold Startu	p Cost (\$):	0.00							
					No Loa	d Cost (\$):	0.00							
					Incre	mental Energy	Offers							
					Use C	Offer Slope: 🗌								
						MW	Price							
						10.0	60.00							~
						999.0	260.00							
														~

For CSF generators, the following rule is enforced:

• Startup and No Load Costs must be set to zero.

For more information on this display, refer to section 4.8 Schedule Offers Daily Default Display.

10.3.4.1 Update Schedule Offers Default

To specify the CSF Generator's daily offer:

- 1. Select Schedule Offers Default in Generation function group.
- 2. Select the desired Portfolio.
- 3. Select the desired Unit.
- 4. Enter changes in any of the following fields:
 - a. Startup Cost (Hot, Intermediate, Cold) must be set to zero
 - b. No Load Cost must be set to zero
 - c. Use Offer Slope checked on unchecked
 - d. MW/Price pairs for incremental energy offer
- 5. Click the **Update** button to save changes.

10.3.5 Schedule Offers Hourly Updates

The Schedule Offers Hourly Updates display allows the Participant to specify the CSF Generator's daily incremental energy offers for particular hours of the market day.

The format of Schedule Offers Hourly Updates display is shown below:

Public Gene	ration Demand	ARD Regulati	on DRR						
	nit Default Portfolio Manager		Ramp Rate Hourly Upd		Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates	Schedule Times Hourly Updates	Sched Select
Schedule	Offers Hou	rly Update	es						
Select Date, Unit	t and Schedule		Scheel	dule Offer for My	acility_GEN (12	345) – DFLPRICE (1234501 Price-base	d) on 29-Dec-2018	3
Date:	25-Jan-2019	(3 DEF	01 02 03	04 05	06 07 08	09 10 11	12 13 14	15
Portfolio:	My_CSF_GEN		~	Fuel S	Selection: MWH				
Unit:	MyFacility_GEN		~						
Schedule:	DFLPRICE			Hot Startup	Cost (\$):	0.00			
			Int	ermediate Startup (Cost (\$):	0.00			
				Cold Startup		0.00			
				No Load	Cost (\$):	0.00			
Copy From: Copy To:	Default			Increm	ental Energy O	ffers			
				Use Off	er Slope: 📃				
		Сору			MW	Price			
					10.0	115.00			~
					15.0	120.00			
					30.0	130.00			~

For CSF generators, the following rule is enforced:

• Startup and No Load Costs must be set to zero for every hour for which the update is made.

For more information on this display, refer to section 4.9 Schedule Offers Hourly Updates Display.

10.3.5.1 Update Schedule Offers Default

To specify the CSF Generator's daily offer:

- 1. Select Schedule Offers Default in Generation function group.
- 2. Select the desired Portfolio.
- 3. Select the desired Unit.
- 4. Enter changes in any of the following fields:
 - a. Startup Cost (Hot, Intermediate, Cold) must be set to zero
 - b. No Load Cost must be set to zero
 - c. Use Offer Slope checked on unchecked
 - d. MW/Price pairs for incremental energy offer
- 5. Click the **Update** button to save changes.

10.4 CSF DARD

CSF DARD's offers and limits are managed using sub-tabs in the ARD group function.

10.4.1 ARD Daily Default Parameters

The ARD Daily Default Parameters display allows the Participant to view the CSF DARD's default parameters, as well as update the default status and ramp rate.

The format of the ARD Daily Default Parameter display is shown below:

Public Ger	neration Dem	and ARD	Regulation	DRR										
	ARD Default Parameters	Portfolio Manager	Ramp Rate Default	Ramp Rate Hourly Updates	Schedule Manager	Schedule Detail Defaults	Schedule Offers Default	Schedule Offers Hourly Updates		Hourly Updates	ARD By Portfolio	XML Download		
ARD D	aily Defa	ult Par	ameters											
Select Date	and ARD			ARD Detail	s for MyFacil	ity_ARD (12345)	on 29-Dec-2018							
Date:	29-Dec-2018		8		Asset Typ	e: ESD			Ass	et short Narr	e: CSF	ARD		^
Portfolio:	My_CSF_AR	D	~		Asset Numb	er: 12345				Noc	le: 10000			
ARD:	MyFacility_A	ARD	~		sset Longnam	e: MyFacility	ARD		DA Sch	edule Require				
				Lead P	articipant(ID#	·):	- www.www.		Ramp	Rate(MW/Mir	n): 100.0)	1	
				Lead Pa	articipant Nam	ne: (((((((((((F	amp Rate Ca	ap: UD		1	
					Dispatch Zor	e: Boston			Cla	im 10 Capab	le: No			
					Dispatchab	le: Yes				Claim 10 (MV	v):			
										Claim 10 Ca	ap: UD			
					Default State	us: 🔿 Economic			Cla	im 30 Capab	le: No			
						🔿 Unavailab	le			Claim 30 (MV	v):			
						Must Run				Claim 30 Ca	ap: UD			~

For CSF DARDs, Asset Type is "ESD" (Energy Storage Device, synonym of Continuous Storage Facility) and the following rules are enforced:

- Default Status either Must Run or Unavailable
- Claim 10, Claim 30 not updateable. CSF DARDs are not Claim10/Claim30 capable.

For more information on this display, refer to section 7.3 ARD Default Parameters Display.

10.4.1.1 Modify CSF DARD Default Status

To specify the CSF DARD's status:

- 1. Select ARD Default Parameters in ARD function group.
- 2. Select the desired Portfolio.
- 3. Select the desired ARD.
- 4. In Default Status, select one of the following radio buttons
 - a. Must Run
 - b. Unavailable
- 5. Click the **Update** button to save the selection.

10.4.1.2 Modify CSF DARD Ramp Rate

To specify the CSF DARD's ramp rate:

- 1. Select ARD Default Parameters in ARD function group.
- 2. Select the desired Portfolio.
- 3. Select the desired ARD.
- 4. In Ramp Rate (MW/min), enter the ramp rate value. Note that the ramp rate must be high enough to allow the CSF to switch between maximum generation and maximum consumption in 10 minutes or less.
- 5. Click the **Update** button to save the changes.

10.4.2 Schedule Detail Defaults

The Schedule Detail Defaults display allows the Participant to view the CSF DARD's schedule default parameters, as well as update the default operating limits, and maximum daily consumption.

The format of the Schedule Detail Defaults display is shown below:

Public Generation Demand ARD Regulation	DRR			
		chedule Schedule Offers Default	Schedule Offers Hourly Updates Selection Update	
ARD Schedule Detail Defaults				
Select Date, ARD and Schedule	Schedule Details for MyFa	acility_ARD (12345) - DFLPRICE	(1234501 Price-based) on 29-Dec-2010	3
Date: 29-Dec-2018	Schedu Day Ah	Description: DFLPRICE fule Available: Yes thead Market: 🗹 Time Market: 🗹	Max	Max Daily Starts: ionsumption (MWh): 80.0 Min Downtime (hr): 0.0 Min Runtime (hr): 0.0 Consumption (MW): 30.0 Consumption (MW): 0.0

For CSF DARDs, the following rules are enforced:

- Min Consumption must be set to zero.
- Min Downtime and Min Runtime are not updateable.

For more information on this display, refer to section 7.7.2 View Schedule Detail Defaults.

10.4.2.1 Modify Schedule Default Data

To specify the CSF DARD's schedule constraint data and operating limits:

- 1. Select Schedule Detail Defaults in ARD function group.
- 2. Select the desired Portfolio.
- 3. Select the desired DARD.
- 4. Enter changes in any of the following fields:
 - a. Max Daily Consumption (MW/h)
 - b. Max Consumption (MW)
 - c. Min Consumption (MW) must be set to zero
- 5. Click the **Update** button to save changes.

10.4.3 Hourly Updates

The Hourly Updates display allows the Participant to view the CSF DARD's schedule default parameters, as well as update the hourly values of status and operating limits.

The format of the Hourly Updates display is shown below:

Public Generation Demand ARD Reg	ulation DRR											
Private ARD Default Portfolio Ramp F Messages Parameters Manager Default	Rate Ramp Rate Hourly Upd		Schedule Detail Defaults	Schedule Offers Default	Schedule Offe Hourly Update		Hourly		ML ownload			
	nouny opo	nariagei	Detail Derauts	Offers Delaur	Hourry opdate	is j selection	opuate		OWINODU			
ARD Hourly Updates												
elect Date and ARD												
Date: 25-Jan-2019		ault Values for N										
Portfolio: My_CSF_ARD	Sche	edule M .COST	fin Consumption 0.0	Max Consumptio	n Type 0.0 Cost	Day Ahead X	Real Time X	Available	Capable	Claim10 No	Claim30 No	Default Status: 🔘 Unavailable
ARD: MyFacility ARD		PRICE	0.0		15.0 Price	×	×	 ✓ 	Default	NO	IND	
ing denity_Aitb									Cap			
												Must Run
	Ove	rride Hourly Value	es for MyFacili	ty_ARD (123	45) on 25-Jar	-2019						
	Hour	r [Min Consumpt	ion Max 0	Consumption	Claim10	Claim	30 Unavailable	Economic	Must	Run	
		Y Y Copy						0	0		0	
	08							0	0		0	
	09							0	0		0	
	10							0	0		0	
	11							0	0		0	
	12							0	0		0	
	14							ŏ	ŏ		ŏ	
	15							ŏ	ŏ		ŏ	
	16			0.0	20.0			۲	0		0	
	17			0.0	20.0			۲	0		0	
	18			0.0	20.0			0	0		•	
	20			0.0	20.0			0	0		•	
	21			0.0	35.0			ŏ	ŏ			
	22			0.0	30.0			ŏ	Ő			
	23			0.0	20.0			۲	0		0	
	24			0.0	20.0			۲	0		0	
												Refresi

For CSF DARDs the following rules are enforced:

• Min Consumption must be set to zero.

• Status must be either Must Run or Unavailable.

For more information on this display, refer to section 7.11 ARD Hourly Updates Display.

10.4.3.1 Enter Hourly Updates

To specify the CSF DARD's hourly updates of default status and operating limits:

- 1. Select Hourly Updates in ARD function group.
- 2. Select the desired Portfolio.
- 3. Select the desired DARD.
- 4. For any Hour of the market day, enter changes in any of the following fields:
 - a. Min Consumption must be set to zero
 - b. Max Consumption must be set to zero
 - c. Select either Unavailable or Must Run radio button.
- 5. Click the **Update** button to save changes.

10.5 CSF ATRR

CSF ATRR's parameters are managed using sub-tabs in the Regulation group function.

10.5.1.1 Validation rules for CSF ATRR regulation limits

CSF ATRRs must operate symmetrically on positive and negative side, or be biased slightly to charging. The mid-point limits for each ATRR are set by the system. If the regulation range entered in this display violates the mid-point limits, an error message is displayed and the change is not accepted. This rule applies to both daily and hourly regulation limits.

Example:

Mid-point limits set by the system:

- Max mid-point = 0
- Min mid-point = -0.15
 - If the Participant sets Reg High to 10, and Reg Low to -8, the calculated mid-point is 1. The ratio of mid-point to Reg High is 0.1, which exceeds the limit of 0. An error message is displayed.
 - If the Participant sets Reg High to 10, and Reg Low to -18, the calculated mid-point is -3. The ratio of mid-point to 16 is -0.19, which exceeds the limit of -0.15. An error message is displayed.

10.5.2 Regulation Offers Daily Default

The ATRR Daily Default Parameters display allows the Participant to update the asset status and regulation limits.

The format of the Regulation Offers Daily Default display is shown below:

Public Generation Demand ARD	Regulation	DRR													
Private Portfolio Messages Manager Offers Default	Regulation Hourly Updates	XML Download													
Regulation Offers Daily	Default														
Select Date and Portfolio	~	Regulation Offe	rs for Portfolio	My_CSF on	26-Jan-2019	•									
Date: 26-Jan-2019		Resource Short Name	Reg Capability	Reg Cap ITO Cost	Reg Cap Offer Oth	Reg Serv Price	ARR	ARR Cap	Dispatch Method	Reg High	RHL Ceil	Reg Low	Floor	Unavail	Derate Factor
Portfolio: My_CSF_ATRR	~	CSF_ATRR	10	0.00	0.00	1.00	75.0	999.0	ENC	10.0	999.0	-10.0	-999.0		1.0

For details on validation of CSF ATRR regulation range, refer to section 10.5.1.1 Validation rules for CSF ATRR regulation limits.

For more information on this display, refer to section 5.3 Regulation Offers Default Display.

10.5.2.1 Modify CSF ATRR Default Status

To specify the CSF ATRR's status:

- 1. Select Regulation Offers Default in Regulation function group.
- 2. Select the desired Portfolio.
- 3. Select the desired ATRR.
- 4. Check the Unavail checkbox to make the asset unavailable, or uncheck the checkbox to make the asset available.
- 5. Click the **Update** button to save the change.

10.5.2.2 Modify CSF ATRR Regulation Range

To specify the CSF ATRR's regulation low and high limits:

- 1. Select Regulation Offers Default in Regulation function group.
- 2. Select the desired Portfolio.
- 3. Select the desired ATRR.
- 4. Enter values in Reg Low and Reg High.
- 5. Click the **Update** button to save the change.

10.5.3 Regulation Hourly Updates

The Regulation Hourly Updates display allows the Participant to update the asset status and regulation limits for any particular hour of the market day.

The format of the Regulation Hourly Updates display is shown below:

7 V	Y Y															
Public Ger	neration Demand ARD	Regulation	DRR													
Private Messages	Portfolio Regulation Manager Offers Default	Regulation Hourly Updates	XML Download													
Regula	ation Hourly Upd	ates														
Select Date	and Resource	«	Regulation Daily Def	ault Values for	MyFacili	ty_ATRR (1	2345) on 2	26-Jan-201	9							*
Date:	26-Jan-2019		Resource Short Name	Reg Capability	Reg Low	Reg High	ARR	Unavail	Reg Cap ITO Cost	Reg Cap Offer Oth	Reg Serv Price	RLL Floor	RHL Ceil	Active ARR Cap	Dispatch Method	
Portfolio:	My_CSF_ATRR	*	CSF_ATRR	10	-10.0	10.0	75.0		0.00	0.00	1.00	-999.0	999.0	999.0	ENC	
Resource:	MyFacility ATRR	~	Regulation Hourly U	odates for My	Facility_A	TRR (1234	5) on 26-Ja	in-2019								
			Hour	Reg Capability	Reg Low	Reg High	ARR	Unavail	Reg Cap ITO Cost	Reg Cap Offer Oth	Reg Serv Price		Legend			
			Y Y Copy										No Overr	de: 📧		
			09	10.0								~	Available			
			10	9.5	-10.0	9.0	30.0	v	0.00	0.00	0.00	~	Unavailab	le: 🔽		
			11	9.0	-10.0	8.0	30.0	V	0.00	0.00	0.00					
			12	10.0												

For details on validation of CSF ATRR regulation range, refer to section 10.5.1.1 Validation rules for CSF ATRR regulation limits.

For more information on this display, refer to section 5.4 Regulation Hourly Updates Display.

10.5.3.1 Modify CSF ATRR Hourly Status

To specify the CSF ATRR's status for a particular hour:

- 1. Select Regulation Hourly Updates in Regulation function group.
- 2. Select the desired Portfolio.
- 3. Select the desired ATRR.
- 4. For a selected hour, check the Unavail checkbox to make the asset unavailable, or uncheck the checkbox to make the asset available.
- 5. Click the **Update** button to save the change.

10.5.3.2 Modify CSF ATRR Hourly Regulation Range

To specify the CSF ATRR's regulation low and high limits for a particular hour:

- 1. Select Regulation Hourly Updates in Regulation function group.
- 2. Select the desired Portfolio.
- 3. Select the desired ATRR.
- 4. For a selected hour, enter values in Reg Low and Reg High.
- 5. Click the **Update** button to save the change.

FUEL INDEX NAME	FUEL INDEX DESCRIPTION	Third Party Vendor
Coal_HS	Central Appalachia 2.0 12500 CSX quarter	Argus
Coal_LS	Central Appalachia 1.2 12500 CSX quarter	Argus
No6_030	Fuel oil No 6 0.3% low-pour NYH	Argus
No6_050	Fuel oil No 6 0.5% NYH	Argus/Internal Market Monitor
No6_070	Fuel oil No 6 0.7% NYH	Argus/Internal Market Monitor
No6_100	Fuel oil No 6 1% NYH	Argus
No6_300	Fuel oil No 6 3% NYH	Argus
LS_Jet_kero	Kerosene NYH barge fob prompt	Argus
Jet_fuel	Jet fuel NYH barge fob prompt	Argus
No2	Heating oil NYH barge fob prompt	Argus
ULSDiesel	Diesel ULSD NYH barge fob prompt	Argus
ULSKerosene	Kerosene ULSK NYH barge fob	Argus
Henry	Hub: Henry	Intercontinental Exchange
NY_natgas	Hub: Transco-Z6 (NY)	Intercontinental Exchange
CT_natgas	Hub: Iroquois-Z2	Intercontinental Exchange
MA_natgas1	Hub: Algonquin Citygates	Intercontinental Exchange
MA_natgas2	Hub: TGP-Z6 200L	Intercontinental Exchange
MA_natgas3	Hub: Dracut	Intercontinental Exchange
MA_natgas4	Hub: Algonquin Citygates (non-G)	Intercontinental Exchange
MA_natgas5	Hub: TGP-Z6 200L North	Intercontinental Exchange
MA_natgas6	Hub: TGP-Z6 200L South	Intercontinental Exchange
avg_MA_natgas	Average Massachusetts Index	Intercontinental Exchange/Internal Market Monitor
Tetco_M3	Hub: TETCO-M3	Intercontinental Exchange
Transco_Z6	Hub: Transco-Z6 (non-NY)	Intercontinental Exchange
Portland_gas	Hub: PNGTS	Intercontinental Exchange
Marcellus	TGP-Z4 Marcellus	Intercontinental Exchange
Propane	Propane Gas	n/a
Refuse	Refuse	n/a
Nuclear	Nuclear	n/a
Wood	Wood	n/a
Wind	Wind	n/a
Bio_Mass	Bio Mass	n/a
Solar	Solar	n/a
Hydro_Pondage	Hydro_Pondage	n/a
Hydro_Pumped Storage	Hydro_Pumped Storage	n/a
Hydro_Run_of_River	Hydro_Run_of_River	n/a
Tire_Derived_Fuel	Tire_Derived_Fuel	n/a
Black_Liquor	Black_Liquor	n/a
Other_Biomass_Liquid	Other_Biomass_Liquids	n/a
Landfill_Gas	Landfill_Gas	n/a
Other_Biomass_Solids	Other_Biomass_Solids	n/a

Appendix A. List of Fuel Index Names used by ISO-NE Market Monitoring